

University of Southampton Research Repository

Copyright © and Moral Rights for this thesis and, where applicable, any accompanying data are retained by the author and/or other copyright owners. A copy can be downloaded for personal non-commercial research or study, without prior permission or charge. This thesis and the accompanying data cannot be reproduced or quoted extensively from without first obtaining permission in writing from the copyright holder/s. The content of the thesis and accompanying research data (where applicable) must not be changed in any way or sold commercially in any format or medium without the formal permission of the copyright holder/s.

When referring to this thesis and any accompanying data, full bibliographic details must be given,

Alagil J (2022) " A qualitative study to explore the decision making and clinical reasoning processes among experienced Pelvic health physiotherapists in the United Kingdom and the Kingdom of Saudi Arabia while managing patients with urinary incontinence", University of Southampton, School of Health Sciences, PhD Thesis.

University of Southampton

Faculty of Environmental and Life Sciences

School of Health Sciences

A qualitative study to explore the decision making and clinical reasoning processes among experienced Pelvic health physiotherapists in the United Kingdom and the Kingdom of Saudi Arabia while managing patients with urinary incontinence

By Jawahr Alagil

Thesis for the degree of Doctor of Philosophy

September 2022

University of Southampton Abstract

Faculty of Environmental and Life Sciences/ School of Health Sciences

Thesis for the degree of Doctor of Philosophy

A qualitative study to explore the decision making and clinical reasoning processes among experienced pelvic health physiotherapists in the United Kingdom and the Kingdom of Saudi Arabia while managing patients with urinary incontinence. By Jawahr Alagil

Urinary Incontinence can significantly affect (and be affected by) different aspects of a patient's life and can be a chronic and recurrent disease that may require selfmanagement; this may lead to unique clinical reasoning by pelvic health physiotherapists. Clinical reasoning is complex in nature and can be represented by different models that may overlap with each other in terms of their terminology and meaning. The existing bio-psychosocial model is limited and has weak implementation in the current literature. Also, the clinical reasoning of physiotherapists working within different countries and organisational cultures are likely to differ. However, there is a lack of research regarding the influence of different cultures and other factors, such as patients' social and organisational factors, on physiotherapists' clinical reasoning processes and the factors that might affect their decision making. This study aims to explore the clinical reasoning process and decision making of physiotherapists in the United Kingdom (UK) and the Kingdom of Saudi Arabia (KSA) while assessing and treating patients with urinary incontinence. Using qualitative research that involved focus groups and interviews with experienced physiotherapists in the UK and KSA. A framework approach to analysis was used to explore and interpret findings. A total of 28 participants were recruited from both countries to undertake semi-structured interviews to understand their thinking processes while managing patients with urinary incontinence. Another 20 participants were recruited to discuss the factors influencing physiotherapists' decision making across three focus groups in both countries.

The results showed four main themes: 1) contextual factors (culture, resources, and healthcare system), 2) Physiotherapists' factors (multifaceted knowledge and experience, interpersonal characteristics and dealing with complex emotional and ethical issues), 3) patients' factors (patients' characteristics and expectations) and 4) making sense of physiotherapists' clinical reasoning and decision making. The existing models of reasoning are insufficient on their own because they do not include individual and organisational cultures. Sense making theory was used to interpret the

participants' thinking process because it helps in understanding ambiguity and uncertainties; in addition, it considers organisational cultures. Comparing two different countries increase understanding of the influence of individual and organisational cultures on decision making. Introducing sense making theory, institutional logics and cultural humility assist in understanding physiotherapists' clinical reasoning and decision making in the UK and KSA while managing patients with urinary incontinence. The findings of this study can be relevant to physiotherapists' clinical reasoning and decision making in other chronic conditions. Further studies are needed to explore the effect of adding cultural humility to cultural competency in physiotherapists' management plans of ethnic minority patients within the UK.

Tabl	le of C	ontentsi
Tabl	le of Ta	ablesix
Tabl	le of Fi	guresxi
Rese	earch T	Thesis: Declaration of Authorshipxv
Ack	nowled	lgements xvii
Abb	reviati	ons xix
Cha	pter 1	Introduction and Background1
1.1	Intro	oduction and Background1
1.2	Clin	ical Reasoning4
	1.2.1	Definition4
	1.2.2	A personal perspective on Clinical Reasoning in Pelvic Health Physiotherapy 5
1.3	The	sis structure
Cha	ntor ?	Literature review Q
Cha	pter 2	
2.1	Mec	chanisms of continence and incontinence9
2.2	Pelv	ric floor muscles
2.3	Evic	lence supporting physiotherapy management for Urinary Incontinence
	2.3.1	Pelvic floor muscles training13
	2.3.2	Non-pelvic floor muscles training supporting / alternative options for treating
		urinary incontinence
	2.3.3	Social and psychological support19
	2.3.4	Lifestyle modifications
	2.3.5	Urinary incontinence and patients' adherence21
	2.3.6	Section summary23
2.4	Orga	anisational culture
2.5	Pelv	ic health physiotherapists' education in the UK and KSA27
	2.5.1	Physiotherapy education and the regulatory bodies in KSA and the UK27
	2.5.2	Pelvic health physiotherapists' competency frameworks27
2.6	Vary	ving degrees of difficulties related to digital palpation29
2.7	Неа	Ith care systems in the UK

2.8	Health care systems in KSA	. 31
2.9	Understanding clinical reasoning	. 32
2.10	Scoping review and narrative synthesis of Clinical Reasoning in Physiotherapy	. 33
2.11	Clinical Reasoning in physiotherapy	. 35
2	2.11.1 Hypothetico-deductive clinical reasoning and pattern recognition	. 35
2	2.11.2 Narrative reasoning	. 37
2	2.11.3 Ethical and moral reasoning	. 38
2	11.4 Literature for an integrated approach: Bio-psychosocial clinical reasoning	
	model	. 39
2	2.11.5 Dialectical reasoning	.41
2	2.11.6 Limitations of the bio-psychosocial model in clinical practice	. 42
2.12	Summary of clinical reasoning literature	. 42
2.13	Theories in clinical reasoning and decision making	. 43
2	2.13.1 Classic decision making theory	. 43
2	2.13.2 Dual Process Theory	. 44
2.14	Perspective Theory	. 45
2	2.14.1 Naturalistic Decision making	. 45
2	14.2 Institutional Theory and Institutional logics	. 46
2	2.14.3 Sense making processes	. 48
	2.14.3.1 Theoretical background	. 49
	2.14.3.2 Dimensions of sense making	. 50
	2.14.3.3 An overview of the existing literature in sense making among different	nt
	health practitioners	. 51
2.15	Factors that might influence clinical reasoning	. 51
2	2.15.1 Physiotherapist factors	. 52
2	15.2 Patients' factors	. 53
2	.15.3 Contextual Factors	. 53
	2.15.3.1 Cultural humility'	. 56
	2.15.3.2 Studies in culture as a factor that influences health practitioners	
	clinical reasoning and decision making	. 58
2.16	Summary and Study rationale	. 61

2.17 Research	question	62
2.18 Study Aim	۱	62
2.19 Objectives	S	63
Chapter 3 Meth	hodology	65
3.1 Research	Paradigm	65
3.2 Methodol	ogy and Research Design	67
3.3 Methods.		67
3.4 Other Me	thods Considered	68
3.5 Interview	method	68
3.5.1 What	t are semi-structured interviews?	68
3.5.2 Key f	eatures of and rationale for the use of interviews	69
3.5.3 Limit	ations of semi-structured interviews	70
3.6 Focus Gro	up Method	70
3.6.1 What	t are focus groups?	71
3.6.2 Key f	eatures of and rationale for the use of focus groups	71
3.6.3 Grou	p homogeneity	72
3.6.4 The r	ole of the focus group facilitator and the observer	72
3.6.5 Limit	ation of focus groups	73
3.7 Procedure	2	74
3.7.1 Samp	oling	74
3.7.2 The p	participants	74
3.7.3 In the	e semi-structured interviews	74
3.7.4 In the	e focus groups	75
3.7.4.1	Inclusion criteria	75
3.7.4.2	Exclusion criteria	76
3.8 Settings		76
3.9 Ethical ap	proval	77
3.10 Recruitme	ent	78
3.10.1 UK		78
3.10.2 KSA		78

3	3.10.3 Recruitment and data collection processes	79
3.11	Pilot interviews and focus groups	
3.12	The interview and the focus group topic guide	82
3.13	Conducting the interviews and the focus groups	
3	3.13.1 During the interview	
3	3.13.2 During the focus groups	
3.14	Different Phases of Data Collection	
3.15	NVivo	90
3.16	Framework analysis	91
3.17	Data Analysis	92
3.18	Analysis of Focus Groups	106
3	3.18.1 Two main ways to analyse focus group data	107
3.19	Methods to ensure rigour	108
3	3.19.1 Credibility	108
3	3.19.2 Confirmability	110
3	3.19.3 Transferability	111
3.20	Ethical Considerations	112
3.21	Data management and confidentiality	112
3.22	Reflexivity	113
3	3.22.1 Engaging in critical self-reflection	114
3	3.22.2 Positioning the researcher in the study	115
3	3.22.3 Reflection in action	116
3.23	Summary of the research paradigm, methods of data collection and analy	sis 117
Chapt	ter 4 Factors that influence Pelvic Health Physiotherapists' Clinical Reasc	oning and
	Decision Making	119
4.1	Introduction	119
4.2	Themes and Categories	120
4.3	Interview Participants	122
4.4	Focus Group Participants	126
4.5	Theme 1 - Contextual / Organisational factors	127

4.5.1 Cultural Context	
4.5.1.1 Impact of indivi	dual culture on clinical reasoning/decision making128
4.5.1.2 An Impact of He	terogeneous culture in the UK131
4.5.1.3 Homogenous cu	lture in KSA132
4.5.1.4 Impact of social	and religious factors on decision making133
4.5.2 Healthcare system/ Or	ganisational culture133
4.5.3 Resources	
4.5.3.1 Influence of the	availability of assessment and treatment options on
decision making	
4.5.3.2 Differences betw	ween the UK and KSA in terms of resources138
4.6 Theme 2 – Pelvic health phy	vsiotherapist's factors139
4.6.1 Physiotherapists' mult	faceted knowledge, training, and experiences140
4.6.1.1 Differences in t	he decision making and clinical reasoning
capabilities of J	participants in relation to introductory and
advanced post-	graduate training and experiences143
4.6.1.2 Influence of pos	tgraduate training on decision making144
4.6.1.3 Impact of reflec	tion on decision making and clinical reasoning144
4.6.2 Physiotherapists' Emot	ional Intelligence
4.6.2.1 Pelvic health ph	ysiotherapists' personal characteristics and life
experiences	
4.6.2.2 Pelvic health ph	ysiotherapists' interaction skills147
4.6.2.3 Communication	between physiotherapists and patients147
4.6.2.4 Complexity of p	elvic floor muscles and lifestyle modification147
4.6.2.5 Variations in ma	naging ethical or sensitive issues between the UK and
KSA	
4.6.2.6 Impacts of patie	nt disclosure153
4.7 Theme 3: Patients' factors .	
4.7.1 Patients' demographic	s and socio-economic features157
4.7.2 Acceptance of digital p	alpation157
4.7.3 Associated comorbidit	es158
4.7.4 Uniqueness of urinary	incontinence159

4.8	Sun	nmary of the findings of the factors that could influence physiotherapists	.) >
	dec	cision making and clinical reasoning in the UK and KSA	160
Chap	ter 5	Theme 4: Making sense of physiotherapist's Decision making and Clin	ical
		Reasoning	163
5.1	Int	troduction	163
5.2	Wh	nat is sense making?	163
5.3	Exa	amples of how the seven properties of sense making constructs are used	by
	phy	ysiotherapists, including quotes/ descriptions	164
	5.3.1	Plausibility rather than accuracy	164
	5.3.2	Social context	165
	5.3.3	Extracted cues	166
	5.3.4	Ongoing	167
	5.3.5	Enactment of environmental context	168
	5.3.6	Retrospective	169
	5.3.7	Identity	170
5.4	Vig	nettes to demonstrate physiotherapists clinical reasoning using sense ma	aking
	pro	ocesses	171
	5.4.1	Vignette 1: Sense making to overcome uncertainty and ambiguity - Soc	ial,
		retrospective and plausibility rather than accuracy	172
	5.4.2	Sense making process	174
	[5.4.2.1 Social context	174
	5	5.4.2.2 Plausibility rather than accuracy (Figure 5-1)	175
	5.4.3	Vignette 2: Properties of sense making: Extracted cues, 'social' and 'gro	ounded
		in identity'	177
	5.4.4	Sense making process	179
	Ę	5.4.4.1 Extracted cues (Figure 5-1)	179
	5	5.4.4.2 Social (Figure 5-1)	180
	Ę	5.4.4.3 Grounded in identity	181
	5.4.5	Vignette 3: Ongoing extracted cues, enactment of sensible environme	ent and
		sense making process to overcome uncertainty and ambiguity	183

	5	.4.5.1	Sense making process	
	5	.4.5.2	'Ongoing' and based on extracted cues	
	5	.4.5.3	Enacting sensitive environment	
Chapt	ter 6	Discu	ssion	189
6.1	Wha	at this i	research has identified	
6.2	Insu	fficien	cy of clinical reasoning models	
6	5.2.1	Rewo	rking of the physiotherapists' thinking processes: using sense ma	aking
c		Troati	y	192
C	5.2.2	omoti	ing from the neck up father than the waist down. How the work	
		a hum	ons and psychological state affect their biadder, and on a wider	10/
		a nun		194
6.3	Insti	tution	al theory: how the differences between institutions influence de	cision
	mak	ing		
6	5.3.1	Facto	rs affecting pelvic health physiotherapists' decision making and o	clinical
		reaso	ning	
6	5.3.1	Institu	utional logics shaped KSA physiotherapists' decision making	
6.4	Cult	ural hu	imility theory: how to incorporate patients' cultural needs into the	he
	man	ageme	ent plan	200
e	5.4.1	Cultu	ral competency in physiotherapy and this study	201
6	5.4.2	Partic	ipants' consideration of culture in the management plan using c	ultural
		humil	ity rather than competency	202
6.5	Refl	ection	and reflexivity	204
6.6	Sum	imary o	of the discussion	206
6.7	Limi	tations	5	207
Chapt	ter 7	Concl	usion	209
7.1	Con	clusion	and implications	209
7.2	Reco	ommer	ndations	212
Арреі	ndix A	Scopi	ng review of the decision making and clinical reasoning literatu	ire in
		physi	otherapy	215
Арреі	ndix B	Chart	ing the data 2021 and 2016	225

Appendix C Ethical approval in the UK 255
Appendix D Ethical approval in KSA 256
Appendix E Summary of the UK and KSA matrices 257
Appendix F Invitation email 282
Appendix G Invitation Letter 283
Appendix H Poster 286
Appendix I Participants' Information Sheet 287
Appendix J Twitter Message 291
Appendix K Interview Consent form 292
Appendix L Focus Group Consent Form 294
Appendix MThe UK Demographics Form 296
Appendix N KSA Demographics Form 299
Appendix O Interview Guide 302
Appendix P Focus Group Guide
Appendix Q Thank you e-mail
Glossary 308
List of References

Table of Tables

Table 2-1: Weick (1995) describes seven properties that are part of the sense making process	es49
Table 2-2: shows the	ne differences between cultural competency and cultural humility adapted	from
Т	ervalon and Murray-Garcia (1998) and Fisher-Borne et al. (2015	. 57
Table 3-1: the data	analysis diary	. 98
Table 4-1: Themes	and categories of the factors and processes that influence pelvic health	
pl	hysiotherapist decision making and clinical reasoning	121
Table 4-2: Demogra	aphics of UK interview participants	123
Table 4-3: Demogra	aphic detail of KSA interview participants	124
Table 4-4: Demogra	aphic details of the two UK focus groups Table 4-5: Demographic details	of the
fc	ocus group participants in the KSA.4F	126
Table 4-6: Themes	include contextual factors, with a category on individual culture, and diffuse ubcategories derived from the data and reported by physiotherapists' particular provides the statement of the stat	erent cipants
		129
Table 4-7: Themes	include contextual factors, with a category on individual culture, and diffe	erent
sı	ubcategories derived from the data and reported by physiotherapists'	
pa	articipants6	134
Table 4-8: Theme is	includes physiotherapists' factors, with a category on physiotherapist's	
m	nultifaceted knowledge, training and experiences	141
Table 4-9: Theme is	nvolving pelvic health physiotherapists' factors/ Emotional intelligence a	nd
de	escription of subcategories8	149
Table 7-1: Developi	ing search terms using PEO Format, which is Population, exposure and	
01	utcomes	216
Table 7-2: the object	ctives, related key words, and Boolean operators	217
Table 7-3: Setting t	the inclusion and exclusion criteria	219
Table 7-4: Database	e selection criteria	221
Table 7-5: Grey Lit	terature	222

Table of Tables

Table 7-6: A detailed data extraction sheet of the literature review conducted in 2016 and 2021.225

Table 7-7: illustrates charting of the data from the UK and KSA interviews and focus groups 257

Table of Figures

Figure 2-1 Illustrate the connection of the endopelvic fascia and vaginal wall 10
Figure 2-2 The levator ani muscle is seen from above, looking over the sacral promontory 12
Figure 2-3: The selection processes of different studies in 2016 and 2021 in clinical reasoning in pelvic health physiotherapy and other physiotherapy area using the Preferred Reporting Items for Systematic Review Chart (Moher <i>et al.</i> , 2009)
Figure 3-1: Different Phases of Data Collection in the UK and the KSA
Figure 3-2: Framework analysis phases outlining the processes of data analysis
Figure 3-3: A diagram used to identify one of the participant's clinical reasoning models. The main researcher raised the questions, and the answers were given by the participant.94
Figure 3-4 An extraction from NVivo to show the indexing process
Figure 3-5: Framework matrices in NVivo that summarise each participant's quotes related to each category and subcategory
Figure 3-6: A schematic diagram illustrates the process of data analysis in interpreting and abstracting the findings
Figure 4-1: Illustrates that contextual factors are a strong recurrent theme with pelvic health physiotherapist decision making and clinical reasoning
Figure 4-2: Contextual factors impacting clinical reasoning/decision making 127
Figure 4-3: Factors influencing physiotherapists' decision making and clinical reasoning. PG: postgraduate, DM: decision making, CR: clinical reasoning
Figure 4-4: the elements that demonstrate physiotherapists' emotional intelligence and influence their ethical and sensitive decision making and clinical reasoning
Figure 4-5: Patients' factors may include patients' demographics, associated morbidities, uniqueness of UI and socio-economic features
Figure 5-1: this is a pictorial image of what is going on: Sense making process using different proprieties of sense making that include extracted cues, and identity 175

Table of Figures

Research Thesis: Declaration of Authorship

Jawahr Alagil

A qualitative study to explore the decision making and clinical reasoning processes among experienced pelvic health physiotherapists in the United Kingdom and the Kingdom of Saudi Arabia while managing patients with urinary incontinence

I declare that this thesis and the work presented in it are my own and have been generated by me as the result of my original research.

I confirm that:

- This work was done wholly or mainly while in candidature for a research degree at this University;
- Where any part of this thesis has previously been submitted for a degree or any other qualification at this University or any other institution, this has been clearly stated;
- Where I have consulted the published work of others, this is always clearly attributed;
- Where I have quoted from the work of others, the source is always given. With the exception of such quotations, this thesis is entirely my own work;
- I have acknowledged all main sources of help;
- Where the thesis is based on work done by myself jointly with others, I have made clear exactly what was done by others and what I have contributed myself;
- None of this work has been published before submission

Signature: Jawahr Alagil

Date: 20.09.2022

Acknowledgements

Firstly, all praise and thanks are for Allah the Most Merciful, the Compassionate.

I would like to thank my supervisors, Sara Demain, Ellen Kitson-Reynolds and Cathy Murphy. I would like to extend special thanks for their continued support, critique, and advice to improve my thesis throughout my study. In addition, for supporting me scientifically, mentally, and emotionally, especially during the COVID-19 pandemic, which has been a difficult and testing time for everyone.

Many thanks also to Gill Brook, Teresa Cook and Katie Mann from POGP for helping me with the data collection in the UK. In addition, Hana Alsobayel, Rafeef Aljuraifany, Hind Alhugail, Alanoud Aldawish and Hala Aldossary from Saudi physiotherapy women's health association for helping me with the data collection in KSA. Without your help, the data collection process would not have been as successful. Much appreciation to all my wonderful participants who gave such valuable and insightful information that contributed greatly to my study.

Thanks, must also go to my father, who always believed in me and encouraged me to follow my dreams. To my sisters and brother, who provided so much support and assisted me so much in the early years of my study, taking care of my children and looking after me too.

I can't say enough words to express thanks to my husband, who has tolerated my mood swings throughout my studies, who has always been my rock and who believed in me and always assisted me in re-building my confidence when I felt lost. You are a wonderful husband and incredible father, and you mean the world to me.

Special thanks to my children, Suliman, who gave me cuddles and kisses when I felt lonely and to Lulu, who reassured me that I'll still be a mother even after getting a degree. To my friend who was wise and gave me the advice to overcome difficult and tough days, my daughter Norah. I love you all with all my heart and I hope for bright and successful futures for you all.

To all my other family and friends and everyone that I have had the pleasure of meeting during this time – thank you for all your help and support, there are too many names to mention, but you know who you are. In particular, Zinab Jafar who provided me with academic feedback and emotional support, especially in my early years. Also, very special thanks to Amy Watson – you have been an incredible support to me and an invaluable source of care and kindness to my children, taking care of them and opening your door and your heart to them. We all appreciate you and everything that you have done for us.

Finally, much appreciation to the Saudi Cultural Bureau for sponsoring my studies. I hope that my work will contribute to the development and enhancement of KSA health sciences in the future.

This thesis is dedicated to my beautiful mother **Munirah Alyahya**, who passed away in 2013 after suffering from illness – My inspiration and my love, a wonderful mother who I will always miss and who made me the person I am today. May she always rest in peace.

Abbreviations

- **BF: Biofeedback**
- CR: Clinical reasoning
- DM: Decision Making
- EBM: Evidence-Based Medicine
- EBP: Evidence-Based Practice
- FGM: Female Genital Mutilation.
- FGs: Focus groups
- HD: Hypothetico-Deductive
- HRA: Health Research Authority
- IOPTPWH: International Organization of Physical Therapists in Pelvic and Women's Health
- ICS: International Continence Society
- IUGA: International Uro-gynecological Association
- MOH: Ministry of Health
- MUI: Mixed urinary incontinence
- NHS: National Health Service
- NICE: National Institute for Health and Care Excellence
- **OPTION: Observing Patient Involvement Instrument**
- **OC: Organisational Culture**
- PFM: Pelvic floor muscle
- PFMT: pelvic floor muscle training
- PGT: Post-graduate training
- PHE: Public Health England
- PHPTs: Pelvic Health Physiotherapists
- POGP: Pelvic Obstetric and Gynaecological Physiotherapy
- POP: Pelvic Organ Prolapse
- PRR: Progression review report

Abbreviations

- PTNS: Posterior Tibial Nerve Stimulation
- KSA: Kingdom of Saudi Arabia
- SDM: Shared Decision Making
- SUI: Stress Urinary Incontinence
- UI: Urinary incontinence
- UUI: Urgency Urinary incontinence
- WPT: World Physiotherapy
- SPTWHA: Saudi Physiotherapy Women's Health Association
- WHO: World Health Organisation
- WHPTs: Women's Health Physiotherapists
- UK: United Kingdom
- USA: United State of America

Chapter 1 Introduction and Background

1.1 Introduction and Background

This study explored the clinical reasoning (CR) and decision making (DM) process of physiotherapists assessing and treating patients with Urinary Incontinence (UI). In simple terms, clinical reasoning is the process by which health professionals 'collect and evaluate data and make judgments about the diagnosis and management of patient problems' (Higgs and Jones, 2008p.4).

The problem of Urinary Incontinence

Urinary incontinence is defined by the International Continence Society and the International Uro-gynecological Association (IUG) 'as any involuntary loss of urine' (Haylen *et al.*, 2010 p 7). The true prevalence of UI is difficult to estimate and is influenced by the sample population studied and the definition of incontinence. It is agreed that UI is more frequent in women than men (Hunskaar *et al.*, 2000), which is often related to the effects of childbirth, obesity, increasing age and menopause (MacArthur, Lewis and Knox, 1991; Hannestad *et al.*, 2000). The prevalence of UI, using the most inclusive definitions of UI ('ever' 'any' or 'at least once in the past 12 months') ranges from 25 to 45% (Botlero *et al.*, 2008).

Urinary incontinence is common; Coyne *et al.* (2009) conducted a multinational crosssectional study in the United State of America (USA), Sweden, and the United Kingdom (UK), comprising 15,861 female participants. Stress Urinary Incontinence (SUI) was reported in over a third of all women surveyed (31.8%), while nearly a quarter of all women surveyed reported Urge Urinary Incontinence (UUI) (24.4%). Similar reports have also been recorded in the KSA, where SUI was found to be the most common type of UI, with prevalence estimated at 50% of women. However, the sample was taken from women visiting the family doctors in five different hospitals, so these estimates may not be fully representative of the general population (Altaweel and Alharbi, 2012). Nevertheless, it is clear that UI is a highly prevalent problem globally, especially in women, which can have a major impact on the patients' quality of life. This is discussed further in the following section.

Chapter 1

Experiencing UI has been shown to have numerous psychosocial impacts. Lee (2005) argues that the voluntary control of the bladder is essential for the sense of normality, selfesteem, and independence. Hence, a loss of bladder control resulting in urine loss can have a profound impact on social activities of women. Although it is not considered a lifethreatening condition, UI can have a significant psychosocial impact on a woman's life, and can substantially interfere with a patient's general health, wellbeing, and quality of life. Women with UI report higher levels of anxiety and depression than the general population (Perry, McGrother and Turner, 2006; Melville *et al.*, 2009). Senra and Pereira (2015) conducted a study involving 80 women with UI in Portugal that compared the link between quality of life and urine loss severity (mild, moderate, and severe). They highlighted the negative effect of UI on women's sexual satisfaction.

Urinary incontinence may have a greater effect on the quality of life of Muslim women due to the disruption to religious observance (Chaliha and Stanton, 1999; Siddiqui *et al.*, 2014). Muslims pray five times a day at different intervals, and Muslim women are required to do ablution (Wudu¹). The actions of prayers can lead to leakage in UI women because it requires them to stand, bend and sit (Chapra, Khan and Al Shaikh-Ali, 2008). Qualitative studies conducted in mixed Western populations (White, Black, Hispanic, Asian and Arab) suggested that Muslim women and other ethnic minorities only seek help after having severe urinary leakage (Siddiqui *et al.*, 2014). The culture of secrecy, sense of shame and self-blame for UI limit help-seeking for UI in women from ethnic minorities (Siddiqui *et al.*, 2014; Toye and Barker, 2020). Some Muslim women may be hesitant to undertake Uro-gynecological tests because there are customs that demand that women should expose certain body parts to their husbands only (Khattab, 2001).

Due to the effects of UI, women (of any race or religion) may often feel unable to leave their home. This could be due to feelings of fear, embarrassment and shame of leaking urine in a public place, mainly because of anxiety regarding odour, feeling wet and not being able to locate a bathroom in time (Huang *et al.*, 2006). Consequently, women with UI will often change their daily routines, isolate themselves from social activities and

¹ "Wudu: is a form of worship to Allah that aims to purify the body and mind by washing water to some predetermined body parts, which are carried out before the prayer." Reza, I.F. (2020) 'Wudu as Islamic Psychotherapy to Improve Sleep Quality in Young Women', *Journal An-Nafs: Kajian Penelitian Psikologi*, 5(1), pp. 64-75.

avoid social gatherings. This in turn can affect their emotional, working lives, wellbeing, restricting physical activities such as, walking, running and dancing (Bø, Talseth and Vinsnes, 2000; Coyne *et al.*, 2003; Sinclair and Ramsay, 2011). It is clear from the existing literature that UI can have a significantly negative impact on the quality of life of the patient (McClurg *et al.*, 2017). It can be assumed that the lifestyles women have will affect how UI is experienced. It is therfore likely that women in different countries and cultures experience and perceive incontinence differently. It is also possible therefore that pelvic health physiotherapists decision making and clinical reasoning may vary in different cultures.

There are a number of interventions to reduce the symptoms of UI in women; for instance, pharmacological and surgical interventions (Virkud, 2011; Nambiar *et al.*, 2018). However, for a number of women, the first line of treatment is physiotherapy intervention (Smith *et al.*, 2013; guideline, 2019).

Pelvic Floor Muscle Training (PFMT) with or without the use of adjunctive therapy such as vaginal cones, biofeedback or electrostimulation aims to strengthen the pelvic floor muscles to improve bladder control (Dumoulin, Hay-Smith and Mac Habee-Seguin, 2014). Physiotherapy interventions can also include a combination of lifestyle modification and education to improve management of UI. These may include bladder training to increase the capacity of the bladder by consciously increasing the time between voiding by contracting the PFM to inhibit the detrusor muscles' over-activity (Ostaszkiewicz, Chestney and Roe, 2004; Dumoulin and Hay-Smith, 2010). It also includes advice on lifestyle changes, for instance, through the management of fluid intake, caffeine and levels of physical activity (Imamura *et al.*, 2015). PFMT can also be carried out with electrical or magnetic stimulation as a supplemental treatment, depending on the severity and type of UI (Bo *et al.*, 2014).

Since UI is more frequent in women than in men (Buckley and Lapitan, 2010), particularly following childbirth; the number of women seeking physiotherapy treatment for UI is high. The International Organization of Physical Therapists in Pelvic and Women's Health (IOPTPWH), an official subgroup of the World physiotherapy, defined women's health physiotherapists as those who provide specialist skills and knowledge related to the prevention and management of many conditions unique to women's health. These conditions include, but are not limited to, incontinence, pregnancy, pelvic pain,

Chapter 1

osteoporosis, bone health, breast health, cardiovascular health, and other musculoskeletal conditions (Porter, 2008).

Pelvic health physiotherapists (PHPTs) provide management to a range of conditions that can cause pelvic problems, such as incontinence of the bladder and bowel, pelvic pain, pelvic organ prolapse (POP) and sexual dysfunction (Bo et al., 2014; Gynaecological and Physiotherapy, 2019). Pelvic and women's health physiotherapy differs from other physiotherapy specialities, as PHPTs deal with sensitive and intimate body parts, meaning these physiotherapists require a specific knowledge base and post-qualification, specialised clinical training (Bo et al., 2014; Frawley, Neumann and Delany, 2019). Physiotherapists often work autonomously and treat complex cases drawing upon a range of models that inform their decisions. The researcher's own practical experience as a pelvic health physiotherapist means there is an awareness of how physiotherapists have to make decisions that account for a complex range of physical, psychological, and socially sensitive issues (Coyne et al., 2003). It is not always clear how to best manage different UI conditions and take account of this complexity. PHPTs assess (including identifying symptoms, mechanisms, and impact on women's life) and carry out treatments using the most appropriate regime. To do this, PHPTs need to clinically reason about likely causation and impact and make decisions about what management to apply. This study therefore seeks to explore and understand the clinical reasoning undertaken by pelvic health physiotherapists, while acknowledging the complexity and uncertainty experienced by physiotherapists during assessments. It also aims to see how physiotherapists consider the impact of UI on women's bodies and their lives. The next section will explore the meaning of clinical reasoning.

1.2 Clinical Reasoning

1.2.1 Definition

Different terms have been used interchangeably within the literature; decision making, clinical judgement, problem solving and clinical reasoning. There are controversies in the literature regarding the concept of clinical reasoning. In nursing, it has been differentiated from decision making (Murphy, 2004), while in physiotherapy clinical reasoning and decision making are used interchangeably as suggested by Oostendorp, Elvers and Trijffel van (2020) clinical reasoning is a process by which physiotherapists work together with

the patient and significant others; make inductive and deductive decisions, step-by-step on the ultimate diagnostics, management options, and outcome measurements related to patients (Hendriks *et al.*, 2000; Oostendorp, Elvers and Trijffel van, 2020). In this study, clinical reasoning focuses on the thinking processes that a physiotherapist uses to make a decision or judgement and solve problems (Hendriks *et al.*, 2000; Kautz *et al.*, 2005), clinical reasoning is the pioneer to decision making and action.

Clinical decision making (DM) has often been defined as the process of choosing between alternatives or options (Thompson and Stapley, 2011). However, Tiffen, Corbridge and Slimmer (2014) developed a definition of clinical decision making by reviewing the literature and asking a panel of nurse practitioners to provide feedback on the new definition. Clinical decision making is defined as a continuous, contextual, and evolving process, where data are gathered from different resources, interpreted, and evaluated in order to select an evidence-based choice of action (Tiffen, Corbridge and Slimmer, 2014).

1.2.2 A personal perspective on Clinical Reasoning in Pelvic Health Physiotherapy

The researcher's own experience as a physiotherapist, working in the Kingdom of Saudi Arabia (KSA) showed her that even after explaining the benefits of pelvic floor muscles (PFMs) many patients found it difficult to contract these muscles and isolate them from another adjacent muscles. Many of her patients were shy during the consent process, required before assessing pelvic floor muscles, and often did not accept digital palpation (vaginal examination). The researcher understood from her own experiences that these women were desperate to get help; she did not want to discharge them with a home exercise programme. She believed they would not adhere to the exercises without a follow-up appointment. In fact, she felt caught between her eagerness to follow research evidence which supports using digital palpation as a way to demonstrate how to contract pelvic floor muscles correctly, and satisfying and respecting her patients' preferences and choices. The latter was contradicting the current available research evidence: isolating pelvic floor muscles from other surrounding muscles was challenging without digital palpation or using other biofeedback tools. Even though the physiotherapy department where the researcher worked prioritised following high quality research evidence over patients' preferences, it was challenging to ignore individual patients' preferences and to follow the department's recommendations solely. However, some of her colleagues were

Chapter 1

supportive of the researcher's approach and followed their patients' preferences also; others did not.

To summarise my professional past, I worked as a pelvic health physiotherapist for about five years at KSA. In 2008, I was offered a position at King Saud University in Riyadh, KSA, to work as a lecturer in general physiotherapy and pelvic health physiotherapy. During my work as a clinician and lecturer, I was influenced by the positivist paradigm, which, with its search for unique answers to determinate questions, dominated my professional education. In the years that followed, I was excited about the need for high-quality research evidence, and I shared this with my students. As a physiotherapist, my main concern was to develop my knowledge and skills in women's health and pelvic rehabilitation in the form of cause-effect relationships. I used clinical reasoning in my daily practice; I saw it as a purely hypothetico-deductive (HD) process that helped me during this period, and I taught it to my students.

I thought following the HD process was sufficient for clinicians to make a competent diagnosis. I assumed that clinical reasoning should be the domain of clinicians and that patients should not have any involvement. However, I became aware that my perspective of clinical reasoning was too narrow, and I began to feel uncomfortable with some of my beliefs, so I decided to reconsider my experiences as a lecturer and a physiotherapist. My assumptions seemed limited, and I needed to understand the complexity of physiotherapists' clinical reasoning to grow fully and develop as a practitioner. Therefore, I began a personal journey to increase my knowledge in this area. The purpose of clearing a researcher's concepts and thoughts about a topic, in addition to their personal biases, is to bring awareness and disclose what is assumed about a topic. By carrying the researcher's beliefs to the consciousness level, the researcher is in an excellent position to approach the topic fairly and openly (Speziale and Carpenter, 2007).

Therefore, in 2013, I relocated to the UK to obtain my master's degree in physiotherapy at the University of Brighton. During my studies, I became even more aware of the importance of clinical reasoning in physiotherapy; this became the focus of my master's thesis, which I completed in 2015. This subsequently led me to my PhD at Southampton University. My research idea, which reflected my previous experiences, was to identify the gaps in clinical reasoning in the Kingdom of Saudi Arabia and the UK. I was interested in exploring clinical reasoning processes whilst managing patients with UI among experienced pelvic health physiotherapists in the UK and the KSA. In addition, I started to reflect on the distinctive characteristics of the Saudi healthcare system and the status of the physiotherapy profession in Saudi Arabia compared with those of the UK, and I developed a keen interest in progressing the clinical reasoning knowledge of pelvic health physiotherapists. See section 3.22 for more details about my reflection and reflexivity.

I realised that there are limited studies on the clinical reasoning processes of pelvic health physiotherapists for UI that consider cultural perspectives. However, given its importance, exploring the impact of culture is necessary. Therefore, this study was mainly conducted to understand how pelvic health physiotherapists clinically reason while considering the factors that impact UI management. It also sought to explore similarities and differences in the UK and KSA reasoning and decision making.

Chapter 1

1.3 Thesis structure

This thesis consists of seven chapters; this first chapter introduces the topic and presents a brief background of the research field, including definitions and presenting a personal perspective on clinical reasoning in pelvic health physiotherapy. The subsequent chapter presents a detailed literature review, including the mechanism of incontinence and pelvic floor muscle training and alternative treatment options for UI. Clinical reasoning methods in physiotherapy and a summary of the literature are also presented. A rationale for the present study and the research questions and aims are also included. Chapter three outlines the methodology and data analysis methods that are utilised in this study, which include semi-structured interviews and focus groups. Chapter four presents the findings regarding the factors that were found to influence pelvic health physiotherapists' clinical reasoning and decision making, while chapter five provides an overview of the sense making process of pelvic health physiotherapists' decision making and clinical reasoning. Chapter six discusses the findings with regard to the literature, while chapter seven presents the conclusions that have been drawn from the findings and their implications and recommendations for future research.
Chapter 2 Literature review

This chapter presents a contemporary understanding of what is already established within existing physiotherapy clinical reasoning literature and presents an exploration of the knowledge gaps. The literature review chapter has been divided into four sections: the first section presents an overview of the field and defines relevant terms; it also presents an introduction of the health systems in the UK and the KSA. This is followed by the second section, which presents the process and the results of a scoping review and narrative synthesis of clinical reasoning and decision making within physiotherapy practice in general; it also considers the limitations of the existing models of clinical reasoning. The third section highlights the need for introducing different decision theories to overcome the limitations of the clinical reasoning models. The fourth section identifies the factors that influence pelvic health physiotherapists' decision making and clinical reasoning, in addition to outlining the frequently used decision making and clinical reasoning models of health practitioners in the KSA and, finally, the study rationale.

An aim of this study is to understand pelvic health physiotherapists' decision making and clinical reasoning when assessing and treating patients with UI. It is therefore, of interest to outline the mechanisms associated with continence/incontinence and mention the anatomy of the pelvic floor muscles in more detail and to highlight the physical, psychological, and social impacts of the condition. This will highlight the complexities of mechanisms, symptoms, and treatment options available, which physiotherapists are required to take into account of in their reasoning.

2.1 Mechanisms of continence and incontinence

There are different types of UI. The most common types are Stress Urinary Incontinence (SUI), Urgency Urinary Incontinence (UUI), and Mixed Urinary Incontinence (MUI). The definition of each type1² are in the footnote (Berghmans, Seleme and Bernards, 2020; Frawley *et al.*, 2021).

² Urgency: is the complaint of a sudden compelling desire to pass urine, which is difficult to defer.

The primary anatomical basis of continence is the urethra, whose wall contains small muscles that are able to generate intra-urethral pressure (Laycock and Haslam, 2013). The continuity of the endo-pelvic fascia, vaginal wall, arcus tendinous fascia pelvis, and levator ani muscle forms the hammock-like structural layer against which the urethra is compressed. This is facilitated by an intact nervous system control. The support of the urethra, as illustrated in Figure 2-1, is determined by the state of relaxation or contraction of the levator ani muscle (pelvic floor muscle) (DeLancey, 1994).

Figure 2-1 Illustrate the connection of the endopelvic fascia and vaginal wall The endo-pelvic fascia and vaginal wall that lie under urethra to arcus tendinous fascia

pelvis; its connection to levator ani muscle forms the hammock-like structural layer.



Source: (DeLancey, 1994) (Permission to use image granted by J.O. DeLancey)

Urgency Urinary Incontinence: is the complaint of involuntary loss of urine associated with urgency.

Mixed Urinary Incontinence: is the complaint of involuntary leakage associated with urgency and exertion, effort, sneezing, or coughing.

Stress Urinary Incontinence: is the complaint of involuntary leakage on effort, exertion, sneezing, or coughing.

Voluntary contraction of the levator ani muscle causes bladder neck elevation, while voluntary relaxation causes it to be lowered at the onset of micturition. With increased intra-abdominal pressure (for example, during coughing, sneezing, and carrying heavy weight) the urethra is compressed between abdominal pressure and pelvic fascia, which leads to higher pressure in the urethra compared to that in the abdomen and the bladder. This pressure differential maintains continence and avoids urine leakage (DeLancey, 1994; Haylen *et al.*, 2010; Frawley *et al.*, 2021). The role of the pelvic floor in giving support to the bladder and urethra, and in allowing normal abdominal pressure transmission to the proximal urethra is also considered necessary in the maintenance of continence. If, however, the layer under the urethra becomes weak and unstable (for example, due to frequent spontaneous vaginal delivery, nerve damage, post-menopause or obesity), the opposing force that causes urethral closure is lost, and the occlusive action is diminished, which leads to incontinence (DeLancey, 1994; Rehman *et al.*, 2017). Urethral support is not the only factor involved in the incontinence mechanism; competence of the bladder neck and urethral function also plays a vital role (DeLancey, 1994).

2.2 Pelvic floor muscles

The main muscular supports for the pelvic organs (including the uterus, vagina, urethra, and rectum) are provided by the levator ani muscles. The levator ani, which is of crucial importance for the pelvic floor re-education (Laycock and Haslam, 2013), can be divided into different parts: pubovisceral muscle, pubovaginalis, iliococcygeus, puborectalis and pubococcygeus **Error! Reference source not found.** (DeLancey, 1994; Kearney, S awhney and DeLancey, 2004; Laycock and Haslam, 2013). The levator ani muscle has a constant tone and relaxes during urine voiding and defecation. It has the ability to fast contract with an immediate increase in intra-abdominal pressure, for instance, during sneezing and coughing (DeLancey et al., 2007).

Figure 2-2 The levator ani muscle is seen from above, looking over the sacral promontory.

Showing the pubovaginal muscle (PVM). The urethra, vagina, and rectum have been transacted just above the pelvic floor. PAM = puboanal muscle; ATLA = arcus tendinous levator ani; and ICM = iliococcygeal muscle (The internal obturator muscles have been removed to clarify the levator muscle origins. The vagina and rectum have been transacted just above the pelvic floor).



Source: (DeLancey et al., 2003) (Permission to use image granted by J.O. DeLancey)

The levator ani muscles are composed of striated muscles. It is known that two-thirds of the fibres are type I (slow-twitch) muscle fibres responsible for the resting tone of the levator ani muscle (Gilpin *et al.*, 1989). One-third of the fibres are type II (fast-twitch) fibres, which are thought to act when a quick, powerful contraction is needed during coughing and sneezing (Gosling *et al.*, 1981). These might contribute to the management of UI by using constant pelvic floor muscle contraction that stimulates slow-twitch muscle fibres and using 'the knack'; this is the fast contraction of the pelvic floor muscles before coughing and sneezing, which would stimulate the fast-twitch fibres (Bø, Talseth and Holme, 1999; Miller *et al.*, 2008).

2.3 Evidence supporting physiotherapy management for Urinary Incontinence

There are several options for pelvic health physiotherapists to use in terms of UI management; this section presents some of the most frequently used options.

2.3.1 Pelvic floor muscles training

Pelvic floor muscle training (PFMT) is the most widely used option and is supported by scholars Smith et al. (2013); McClurg et al. (2017); Dumoulin, Cacciari and Hay-Smith (2018). An international standardisation committee defined pelvic floor muscle training as an exercise to improve pelvic floor muscle strength, endurance, power, relaxation or a combination of these parameters (Frawley et al., 2021). A Cochrane review by Dumoulin, Cacciari and Hay-Smith (2018) compared pelvic floor muscle treatment for managing female stress, urgency and mixed UI with no treatment or control treatments. The review included 31 trials, with one study of women with MUI and only one study with UUI alone, with no data on cure or improvement or the number of episodes of UI of these participants. Dumoulin, Cacciari and Hay-Smith (2018) found that pelvic floor muscle training can cure or improve symptoms of stress UI and the other types of UI. Women undertaking pelvic floor muscle training lost smaller amounts of urine on the clinic-based pad tests. They reported emptying their bladder less often during the day and having better sexual outcomes. However, some trials found no evidence of an improvement in general health status quality of life measures, probably because these types of measures are less sensitive to changes in continence management or because of a lack of evidence (Beuttenmüller et al., 2010; Carneiro et al., 2010; Pereira, Correia and Driusso, 2011). Furthermore, the review, conducted using a robust methodology, reported difficulties in drawing any conclusions about the most effective type of exercise or setting for exercise provision because of variations in the regime type and incomplete description of the pelvic floor muscles training programmes.

There was also a lack of follow-up beyond the end of treatment in most trials, meaning that the long-term outcomes of using pelvic floor muscles training remain uncertain. There were potential biases in the clinical trials; for instance, some of the studies lacked genuine randomisation, limited blinding of the outcome assessment and significant differences in the participants' age. The review's findings suggest pelvic floor muscle training should be included in first-line conservative management programmes for women with UI. Dumoulin,

Cacciari and Hay-Smith (2018) indicated that further attention needs to be paid to socioeconomics, as it has not been addressed enough until now; other studies are required in order to understand the significance of the follow-up beyond the end of supervised treatment because the purpose of therapy is longterm continence, it would be convenient too if there were measured at least one year after the end of treatment.

Moreover, Bascur-Castillo *et al.* (2022) analysed the systematic evidence of 32 reviews on the beneficial effects of pelvic floor muscle training, biofeedback, electrical stimulation, and lifestyle changes such as weight loss and vaginal cones. The result of the systematic review showed that pelvic floor muscles training is the most successful treatment for UI, with 70.6% of the studies reporting an improvement using this strategy. Of all the 32 reviews that Bascur-Castillo (2022) included, only three systematic reviews and one meta-analysis were of high quality; they analysed the effect of pelvic floor muscle training and non-surgical weight loss on UI. Therefore, there is scientific evidence to confirm the impact of these conservative management in treating UI. Pelvic floor muscle training has been considered the first-line treatment for UI and has several forms, however. However, there is not enough evidence to suggest which program of pelvic floor muscles treatment is better than others (Dumoulin, Cacciari and Hay-Smith, 2018; guideline, 2019; Okeahialam *et al.*, 2022).

At present, there are some inconsistencies within the literature regarding the effective exercise parameters, such as differences in the type (near maximal, submaximal, 'the Knack', with or against gravity), the number (per set, per day or per week), or the duration of contractions. However, it is recommended that the basic physiological principles to improve muscle strength and endurance should be adhered to, which may include: overloading muscles by performing more work than usual to the point of muscles fatigue; in addition, a specificity that may consist of training the muscles with exercise or physical activity that is as close as possible to the functional movement required (Garber et al., 2011). Maintenance is essential in pelvic floor muscle training because the benefits of the exercise will reverse if they are not undertaken on a regular basis (Garber et al., 2011). Smith et al. (2013) suggest pelvic floor muscle training of at least three months' duration and pelvic floor muscle training programmes are most effective when containing at least 8-12 slow to moderate velocity, moderate to maximal intensity contractions performed 2-4 times per day, 2-3 days per week with progression over 16 weeks (Kraemer et al., 2002). Often pelvic floor muscle training can be performed by an individual as self-management, but the literature suggests that it is most effective when carried out under the supervision of a physiotherapist (Smith *et al.*, 2013; Fitz *et al.*, 2020). Also, physiotherapy treatment can be provided individually or during group exercises. The effect of pelvic floor muscle training is better if it is delivered regularly in supervised training (Hay-Smith *et al.*, 2012). Supervised training is defined as a pelvic floor muscle training program taught and monitored by a health professional field (Bo *et al.*, 2017). A Cochrane review by Hay-Smith *et al.* (2012) found that pelvic floor muscle training with regular weekly supervision was better than pelvic floor muscle training with regular weekly supervision was better than pelvic floor muscle training with little or no supervision; however, it was not clear if the supervision was best provided individually or in a group. There are several limitations of the review: because there was a possibility of a relationship between attention and reporting of more improvement in participants who were not blind to treatment allocation, some trials did not describe their intervention in detail. This Cochrane review found that the evidence was insufficient to make strong recommendations about the best approach to pelvic floor muscle training.

Pelvic floor muscle training can also be undertaken through facilitating co-contraction of another muscle group, such as the transversus abdominus; however, adding other muscles to pelvic floor muscle contraction did not improve the outcome of UI. A systematic review conducted by Bø et al. (2009) found two randomised control trials by Dumoulin et al.(2004) that shown there was no additional effect of adding transversus abdominus training to pelvic floor muscle training in the treatment of patients with urinary incontinence. Small co-contractions of the inner abdominal muscles have always been encouraging in rehabilitation programs by health practitioners, but exercising accessory muscles such as abdominal muscles in an attempt to activate the pelvic floor sufficiently to elicit a training effect is not recommended (Kruger et al., 2019). Also, in a systematic review conducted by Bø and Herbert (2013) to assess the existing evidence of RCTs to treat mixed and stress UI using alternative interventions to pelvic floor muscle training such as Pilates exercises, abdominal conditioning, and the Paula method, which means contracting ring muscles of the mouth, eyes, or nose will result in co-contraction of pelvic floor muscles. The result of the systematic review showed that there is no substantial evidence that alternative exercise interventions can decrease urinary leakage in women with stress urinary incontinence; this is supported by other work by Bø et al. (2009); Hay-Smith et al. (2012); Dumoulin, Cacciari and Hay-Smith (2018) who reported that there is no substantial evidence that using additional therapeutic exercises to pelvic floor muscle training will lead to more activation of the pelvic floor muscles and decrease UI.

Research has shown that pelvic floor muscle training is the most effective option for UI management compared to other alternatives, such as electrical stimulation and vaginal cones in the treatment of stress urinary incontinence (Bø, Talseth and Holme, 1999). In international consultation on incontinence (ICI) regarding the National Institute for Health and Care Excellence (NICE) guidelines, in two editions of the IC1, both Okeahialam et al. (2022) and Abrams et al. (2018) recommended pelvic floor muscle training as the first line of treatment for women with stress or mixed UI. In addition, a high-quality randomised controlled trial conducted by Bø, Talseth and Holme (1999) compared pelvic floor muscle training, electrical stimulation and vaginal cones in the treatment of SUI in 107 women with SUI; women were divided into four groups: pelvic floor muscle exercise group (n=25), electrical stimulation group (n=25), vaginal cones (n=27) and a control group with no treatment (n=30). Using a pad test with standardised urinary volume and self-reported severity, the amount of urine leakage was tracked over a period of six months. Significantly reduced urine leakage was noted in the pelvic floor muscles exercise group in comparison to electrostimulation (difference of 22.8g, p=0.02) and vaginal cones group (difference of -15.5g, p<0.01).

Moreover, it was reported that the group with the highest compliance rate was the pelvic floor muscle exercise group, which might be the reason why this group had the best outcome. Moreover, pelvic floor muscle training was found to be superior to other interventions in the treatment of SUI, with pelvic floor muscle training exercise recommended as the gold standard by a Cochrane systematic review (Dumoulin, Hay-Smith and Mac Habee-Seguin, 2014). Dumoulin, Cacciari and Hay-Smith (2018) also compared pelvic floor muscle training with no treatment or placebo in women with SUI, MUI, and UUI and concurred that pelvic floor muscle training should be the first line of treatment for all types of UI.

There are some other options available for the management of UI within the literature; for example, electrical stimulation, vaginal cones and the use of different muscles as facilitators; however, these alternatives do not have a strong evidence base and, as such, are not highly supported by researchers (Dumoulin, Hay-Smith and Mac Habee-Seguin, 2014). As a result, it has been noted within the literature that pelvic floor muscle training often report that they have insufficient management options to offer to their patients (Slade *et al.*, 2020). Nevertheless, the following section highlights the common alternatives to pelvic floor muscle training.

2.3.2 Non-pelvic floor muscles training supporting / alternative options for treating urinary incontinence

The effectiveness of physiotherapy alone in improving SUI and MUI is established within the literature (Bo *et al.*, 2014; Dumoulin, Hay-Smith and Mac Habee-Seguin, 2014; McClurg *et al.*, 2017), while Urge Urinary Incontinence (UUI) may be caused by detrusor muscle contraction, which can be inhibited by a pelvic floor muscles contraction induced by electrical stimulation (McClurg *et al.*, 2017). Bo *et al.* (2014) suggest that when physiotherapy intervention fails, UUI may require additional pharmaceutical therapy to decrease urgency and bladder hyperactivity. Pharmacological therapy interventions can include botulinum toxin injected into the bladder for a persistent overactive bladder (Duthie *et al.*, 2011), anticholinergic drugs taken to relax the bladder (Marcelissen *et al.*, 2018) or oestrogen injected locally to the vagina, to decrease frequency and urgency (Araklitis and Cardozo, 2017). Anticholinergic drugs significantly correlate with increased risk of cognitive impairment and dementia, especially in frail and elderly patients. The provider needs to prescribe the lowest effective doses or alternative medications for patients at risk (Cancelli *et al.*, 2009).

Surgery has also been suggested as an alternative to improve UI, which generally aims to lift and support the ureterovesical junction. It has been suggested that the decision to manage UI using such an invasive surgical procedure depends on several factors, including the patient's physical features and the surgeons' preferences. Surgical interventions most frequently include mid-urethral sling procedures and sacral neuromodulation (SNM). The latter involves the placement of an electrode into the third sacral (S3) foramen, which is electrically stimulating the nerve root and suppressing the reflexes responsible for involuntary detrusor muscle contractions. The latter treatment only commences after failed physiotherapy intervention and where Botox injection proved to be an ineffective treatment for UUI. Furthermore, there are debates about suspending mesh mid-urethral slings from regular use in the UK, Australia and other countries due to the long-term morbidity; it is only considered for use as a last resort (Kmietowicz, 2018).

Different management strategies are available for physiotherapists to treat UI, such as acupuncture, functional magnetic chair, and posterior tibial nerve stimulation.

Acupuncture may include dry needling to increase or decrease muscle tone of the pelvic floor muscles. While electro-acupuncture (EA) uses needles placed in certain spots, a small electrode is attached to the needles. Costa et al. (2018) reviewed different systematic reviews of non-surgical treatment options available, ranging from pharmacological approaches to pelvic exercises. In addition, they assessed the benefits and side effects of acupuncture among adults with UI compared to pharmaceutical treatment; the result of the review showed 20 systematic reviews on pelvic floor muscle training programmes, vaginal cones, and acupuncture. Only one RCT was found, it compared acupuncture versus midodrine with 60 women, and the study result showed no difference between the groups. The RCT is of low methodological quality and poor reporting, and no conclusion could be reached regarding the efficacy and safety of acupuncture among patients with urinary incontinence (Wang et al., 2013). In another systematic review by Zhao et al., (2018) with a meta-analysis of ten RCTs including 794 patients, compared EA to sham acupuncture, or EA plus tolterodine vs tolterodine alone. The studies were of low quality and the authors reported that acupuncture might effectively reduce overactive bladder symptoms compared to sham treatment (Zhao et al., 2018). However, acupuncture is comparatively well tolerated with few adverse reactions and is safe. There is a need for high-quality studies and larger sample sizes in the future to understand the effect of acupuncture on UI.

Functional magnetic chair treatment involves using a unique chair with a therapeutic head placed on the seats; as the magnetic field can penetrate through clothes, stimulation can be carried out in clothing, which increases the comfort and privacy of the treatment (Quek, 2005). The magnetic field stimulates the pelvic floor muscle and the pelvic organs; it also inhibits the reflex mechanism of emptying the bladder to increase bladder capacity. In a recent systematic review to understand the beneficial effect of different physiotherapy modalities in UI (Peng *et al.*, 2019), it was found that the functional magnetic chair is effective in the treatment of UUI and overactive bladder, while the results were inconclusive regarding the management of SUI (Mazur-Bialy *et al.*, 2020). Booth *et al.* (2021) used transcutaneous posterior tibial nerve stimulation to treat care home residents with UI. A total of 408 participants were randomised to receive transcutaneous tibial nerve stimulation or sham stimulation over a six-week period. The trial showed that transcutaneous posterior tibial nerve stimulation is ineffective in reducing UI. The study has some limitations because the participants had cognitive impairment, which increased the challenge of completing 24-hour pad collections for care home staff, resulting in

missing data. In addition, the participants were complaining of UI, and it is unclear which type of UI was included in the study because the literature suggested that transcutaneous posterior tibial nerve stimulation is a practical, minimally invasive option for the management of the UUI (Vandoninck *et al.*, 2004).

Feedback is sensory information available as the result of an activity of an intrinsic source from the individual or an extrinsic source from the physiotherapists (Bo et al., 2017). Biofeedback uses specific equipment, such as vaginal cones, a perineometer, pelvic floor electromyography or an educator, to visualise the pelvic floor muscle contraction to demonstrate to the patient whether they are contracting the correct muscles. Regarding the effectiveness of biofeedback, a Cochrane review by Herderschee et al. (2013) found that women who received biofeedback combined with pelvic floor muscle training were significantly more likely to report improved UI compared to those who received pelvic floor muscle training alone. However, it could also be argued that women who received biofeedback had more contact with health professionals than those who did not, which may explain this result. Another study by Smith et al. (2013) also suggested that electrical stimulation and biofeedback might be considered in women who cannot actively contract the pelvic floor muscle to aid motivation and adherence to therapy. Based on current evidence, it is unclear if the apparent benefit of biofeedback can be related to the biofeedback or some other variable, such as more health professionals in contact with those women receiving biofeedback (Hagen et al., 2020). A common issue in previous trials was unclear statements related to the purpose of biofeedback, or to describe the intervention protocol (Herderschee et al., 2013). Thus, it was unclear if the way the biofeedback was used could theoretically or in practice improve the effectiveness of the pelvic floor muscle training (Hagen et al., 2020). Despite this, National Institute for Health and Care Excellence guidelines discourage the use of perineometer or pelvic floor electromyography as biofeedback in a routine part of pelvic floor muscle training (Smith et al., 2013; guideline, 2019;2021). Instead, pharmaceutical and surgical interventions are used as a last resort, when physiotherapy interventions do not work (Duthie *et al.*, 2011; Kim, Kim and Kim, 2020; Robinson and Cardozo, 2021).

2.3.3 Social and psychological support

UI is a chronic problem that can lead to stigma, social withdrawal and resulting psychological problems, such as anxiety and depression. Moreover, anxiety may also

contribute directly to overactive bladder symptoms and therefore, has to be considered in the management process (Lai *et al.*, 2016). Psychological disorders can also affect patient motivation, which is why they need to be considered, especially in elderly patients (Tarcan *et al.*, 2020). When managing chronic diseases there are different barriers that may necessitate social and emotional support (Bayliss *et al.*, 2003); hence, the treatments of chronic conditions have increasingly emphasised self-management of the condition via physical as well as psychological approaches (Keles *et al.*, 2007). Although it is important to acknowledge patients' social and psychological factors, there are inconclusive results regarding the relationship between patients' psychological issues and UI (Warner and Greenwell, 2018).

Nevertheless, there is no doubt that UI impacts womens' daily living; it has been shown to interfere with the physical, psychological and social activities of women, reducing general health and wellbeing, as well as their quality of life (Smith *et al.*, 2013; Warner and Greenwell, 2018). Hence, it can be suggested that due to the physiological, sociological and psychological complexity of UI, it could be advantagous for physiotherapists to incorporate social and psychological factors into decision making and clinical reasoning when determining the type and causes of incontinence and deciding the most appropriate treatment approaches.

2.3.4 Lifestyle modifications

The ICS and NICE guidelines have highlighted the importance of using lifestyle modifications for women as a potential UI management option for urgency or mixed UI issues Smith *et al.* (2013); Abrams *et al.* (2018); guideline (2019;(2021); Okeahialam *et al.* (2022). This includes bladder training that lasts for a minimum of six weeks and incorporates advice on lifestyle changes, such as reducing caffeine intake and achieving weight reduction. Both motivation and adherence are central factors in the therapeutic success of life-style modifications when managing UI, alongside other patients' factors such as co-morbidities and mental health problems (Smith, Higgs and Ellis, 2008). Although lifestyle modifications can be useful and helpful, they are not sufficiently effective in managing UI, when used as a stand-alone option, but rather are most effective when used in conjunction with another therapy (Abrams *et al.*, 2018). Moreover, there is evidence to suggest that adherence to physiotherapy management is often low among patients with a chronic disease that requires self-management (Peek *et al.*, 2016). The

determining factors of therapeutic adherence to pelvic floor muscles training have been analysed and classified based on patient personal parameters or factors dependent on the physiotherapeutic performance (Dumoulin *et al.*, 2015). There are potential differences in adherence in patients in the UK and KSA. Patients' adherence to physiotherapy management will be explored further in the next section, in addition to 2.6.

2.3.5 Urinary incontinence and patients' adherence

High levels of patient adherence to management programmes have been linked to an increase in positive patient outcomes, such as a decrease in urinary leakage and improved function (Karnad and McLean, 2011). A scoping review was conducted by McClurg et al. (2015) on adherence promotion theories in pelvic floor muscle training. Twelve promising theoretical models were identified in an attempt to increase patients' adherence to pelvic floor muscle training, including such models as a social cognitive theory where self-efficacy is considered the core to the social cognitive theory. It refers to patients' beliefs in their ability to control or perform specific behaviours such as pelvic floor contractions. The importance of selfefficacy was highlighted by women doing pelvic floor muscles training for UI and was a common predictor of the intention to adhere to the pelvic floor muscles training (Messer et al., 2007). Self-efficacy is considered to be one of the main determinants of adherence within the field of pelvic floor muscle training and behavioural change process, which is seen to encourage adherence. In addition to the self-regulatory model that views individuals as active problem solvers and their behaviour to health as an attempt to close the gap between current health status and future goals (Alewijnse et al., 2001). The World Health Organisation suggests that a health behavioural approach directed toward adherence in longterm conditions, such as UI, needs to be a process of motivation to do pelvic floor muscles training that involves family influence on individual motivation; however, it is still underdeveloped and requires further research (Organization, 2003). Motivational interviews are another theory suggested by McClurg et al. (2015) as outperforming traditional advice given by health professionals, and Miller and Rollnick (2012) suggested that clinical training is required for the effective implementation of motivational interviewing. Furthermore, selfcare behaviour and its potential for the use of information for self-management; the behavioural change technique, embedding pelvic floor muscle training as part of a patient's routine; a simple explanation of the exercise; and patient education about the importance of pelvic floor muscle training (Griffthis, 1993; Fisher et al., 2002; Abraham and Michie, 2008; May and Finch, 2009). The models that were identified in the scoping review showed there

is a need for future studies to explore how these strategies may improve pelvic floor muscle training adherence in different populations.

There are also different patients' factors that affect the outcome of physiotherapy treatment for pelvic floor muscle training. Reed *et al.* (2020) conducted a prospective observational study on the effect of a patient's health values on adherence to pelvic floor muscle trainings. The values were assessed using the Personal Values Questionnaire (PVQ). The health value section asks nine questions, each rated on a five-point scale. There are also three subscales (Intrinsically-held Value Choice, Aversively-controlled Value Choice, and Value-related Behaviour): 'Intrinsic Choice' represents the individual's commitment to the value; 'Aversive Choice' reflects the degree of external regulation of the value; and 'Behaviour' measures the degree to which the value is acted upon. The PVQ assessed the internal and external values towards change and how change was dealt with. The key findings were that the strength of the patient's health-related values predicted their ability to fully adhere to the treatment entirely. Levin and colleagues reported acceptable test-pre-test reliability within three weeks. Furthermore, there is no validation study published for the PVQ (Blackledge, Ciarrochi and Bailey, 2006; Levin *et al.*, 2014).

Many studies implement adherence aid to help with exercising, such as using a mobile device application (app), offering a physiotherapy consultation at the time of the initial urogynecologist's evaluation, providing supplementary printed information, and asking patients to repeat the prescribed strategy (Araujo, Marques and Juliato, 2020; Brown *et al.*, 2020; Peek *et al.*, 2020). Nonetheless, there is insufficient data to examine their use within clinical practice and to evaluate other strategies that promote self-management and adherence (Peek *et al.*, 2016).

Peek *et al.* (2020) argue that labelling patients as adherent or non-adherent puts all of the responsibility on the patient and does not acknowledge the physiotherapists' role in changing patients' adherence. Therefore, physiotherapists need to consider different factors, such as simplifying patients' education and exercise instructions rather than labelling patients as non-adherent to improve patients' outcomes.

2.3.6 Section summary

There are a number of management options available to treat patients with urinary incontinence, although the most commonly used option is pelvic floor muscles training, which is currently recommended as the gold standard and first-line treatment option by ICS and NICE guidelines. Regarding other options, there are inconsistencies within the literature regarding their effectiveness.

It could be argued that the complexity of UI presentation and the need to consider the impact of UI not only on women's bodies but also on their lives, is apparent and considered necessary. It can therefore be assumed that the lifestyles of these women will affect how they experience UI. This means that women in different countries and cultures are likely to experience incontinence differently. It is also possible that physiotherapists' decision making and clinical reasoning may vary in different cultures. This study is directed towards exploring physiotherapists' decision making and clinical reasoning in two differing cultures, the UK and KSA, concentrating on the experiences of womenwomen's experiences. The next section will explore an overview of KSA and UK culture. Culture in KSA and the UK

There are several ways to define culture. Minkov (2012) discussed cultural complexity, explaining that culture can be studied at different levels and not merely at the national level as researchers used to do. While, Hofstede (1984p.82) defines culture as 'the collective programming of the mind which distinguishes the members of one human group from another. There are several definitions for culture, but everyone agrees that culture is a complex multi-level construct, that is shared between people in the same group or society' (Dan, 2020p.227). This characteristic of culture allows people to predict, with some limitation, other people's behaviour. When this kind of behaviour becomes repetitive, it leads to shaping individual behaviour, and it will transfer to be the right thing to do and be accepted by society (Bonder, Martin and Miracle, 2004; Andreatta and Ferraro, 2012). These behaviours and beliefs can evoke a behavioural response, affecting their health (Spencer-Oatey, 2008). Hofstede (1984) argued that societies differ along four major cultural dimensions, including power distance, individualism/collectivism, and masculine/feminine and uncertainty avoidance. Uncertainty avoidance is defined as the extent to which individuals within a culture are made nervous by unstructured and unclear circumstances

and the extent to which these individuals attempt to avoid uncertainty. The other dimensions are explained in relation to the two countries being explored in this study.

The Kingdom of Saudi Arabia is called the Arabian Peninsula and is located in the Middle East in Asia. The location is considered the capital of Islam; Makkah is located in the Western parts of KSA – one of Islam's most holy sites. The location of KSA and the existence of Makkah have shaped KSA's culture and law. The law in KSA is based on the Holy Quran. In the 1940s, women traditionally occupied the domestic sphere, working as farmers or tailors while cooking and looking after their children. Their husbands tended to travel abroad to other Arab countries to earn money. Women were married at a young age and usually gave birth at home, assisted by one of their family members or a midwife (who got her title by experience only, not by a degree) if the delivery was complicated. Women were usually considered shy, reserved, and seen to walk and act in a 'decent' way (Song, 2019). KSA remains a conservative country, where women struggled to go to school before 1940 because of the influence of strong religious conservativism at the time. More recently, in particular, under the influence of the American Petrol Company, the government started to open female schools and women were allowed to study from 1940 onwards (Hamdan, 2005). Hence, women have been more able to access education and can now work in the private and public sectors. Nonetheless, most of the governmental sectors have a female branch, where women are expected to deal with women only. KSA has a traditionally patriarchal society, where men are often seen as superior to women, in contrast to Western culture.

More recently, in 2019, women have been allowed to drive, travel and finish their governmental services without requiring permission from their closest male relative (Rizvi and Hussain, 2021). This gave women the freedom to work in different sectors, gain money, seek medical treatment and help independently. Saudi women have been raised to share decisions with their family, mainly male figures; as a result, they can be reluctant to make independent decisions. This might be particularly common among elderly and illiterate women. These factors may influence their decision to seek help or medical advice (Xianhui and Shah, 2021). Women from different ethnic minorities living in the West, who hold similar spiritual beliefs, may share the same features as Saudi women. According to Hofstede's theory, Saudi society is based on a collectivist, masculine and power distant society. The power distance is the extent to which less powerful individuals within the community will accept inequality in power and accept it as the norm.

For instance, Saudi Arabian managers are expected to tell subordinates what to do; team hierarchy is likely to be accepted. In this way, collectivist cultures expect individuals to belong to one or more 'in-groups', such as extended families and clans, from which individuals cannot separate themselves. The 'in-group' protects the interests of its membership and in turn, assumes their permanent loyalty (Hofstede, 2003). Furthermore, masculinity is how individuals in a community expect women to care for children and provide non-material quality of life. The masculine culture expects men to be assertive, competitive, and able to strive for material success.

On the other hand, UK culture is diverse; it is shaped by human rights rather than religion. Historically, English culture and values were shaped by the Christian faith, which may still influence current beliefs, particularly regarding tolerance and acceptance of multiculturalism. According to Hofstede's theory, the UK society is based on individualism, defined as societies where individuals are primarily concerned with the interests and motives of their immediate family. Women's rights in the UK have improved exponentially, especially since women were officiallyly granted the right to vote in 1918 and the effect of the Second World War where women were given more prominence within society. Currently, the laws protecting women's rights include the 1975 Equality Act which introduced women's right to equal pay and status in the workplace and the 2010 Quality Act which makes it illegal to discriminate on the basis of a range of protected characteristics including gender, age, religion and sexuality (Wadham et al., 2010). Women have autonomy and freedom in most, if not all, areas that affect their lives, but especially regarding health care decisions. The UK is a diverse country with some areas of the country having a high proportion of ethnic minority populations, such Leicester (50%), London (40%) and Birmingham (30%) (Gov.UK, 2020). Hence it is important to understand the needs for ethnic minorities.

The National Health Service (NHS) implements strategies to maintain cultural competency and ensure equality in health delivery within ethnic minorities (Haugevik and Bratberg, 2011; Luquis and Pérez, 2021). This research was undertaken in two different countries, as it was assumed that culture may affect how physiotherapists reach their decisions, due to the differences in societies and organisational culture. The meaning of this is discussed in the next section.

2.4 Organisational culture

Siehl and Martin (1989) argue that culture has an influence on the attitudes of individuals and employees within a company and consequently can influence organisational effectiveness. Organisations are made up of individuals but have distinct organisational cultures; they are based within societies and countries with a wider cultural context. Consequently, organisations themselves create their own organisational cultures within the context of broader cultural contexts (Willcoxson and Millett, 2000). Organisational culture (OC) encompasses the values and beliefs that employees of an organisation share, and is defined by Sun (2008p.138) as 'the pattern of common beliefs and values that give fellows of an institute meaning, and provide with the rules for behaviour in their organisation'. Schein (1983p.14) explains that organisational culture can be understood as 'a pattern of shared basic assumptions that a group learns as it solves its problems of external adaptation and internal integration, that has worked well enough to be considered valid and, therefore, to be taught to new members as the correct way to perceive, think and feel in relation to those problems. It is clear that there are a number of different definitions of organisational culture within the literature. The type of definition chosen impacts how organisational culture is examined. In practice, an organisation may have its own cultural explanation and interpretation within a given environment and as Alkhoraif and McLaughlin (2018) suggest, organisational culture impacts the way in which people behave; therefore, it is necessary to understand the culture of an organisation.

In relation to decision making, Alkhoraif and McLaughlin (2018) suggested that an individual's decision making process can be affected by their cultural background according to what is considered 'the correct way or the norms', which highlights the significance of individual culture and organisational culture. Also, organisational culture differs from one institute to another (Alkhoraif and McLaughlin, 2016). Fischer and colleagues explored the potential impact of cultural dimensions norms across different contexts. They found that most cross-cultural research has treated norms as abstract and content-free, and has several limitations (Fischer and Mansell, 2009). There are possibilities that norms may vary across cultural groups; the effects of norms may be different across variable behavioural domains and can be affected by situational context. For instance, the extent to which flexibility and wealth affect the strength of norm-

behavioural intentions and norm-behaviour relationships across national cultures (Lefringhausen, Spencer-Oatey and Debray, 2019).

Overall, different cultural norms exist within other countries. These factors may influence physiotherapy education and training, which may affect the decision making and clinical reasoning of physiotherapists. Different organisational bodies in the UK and KSA manage physiotherapists' education and training, as outlined in the subsequent section.

2.5 Pelvic health physiotherapists' education in the UK and KSA

An overview of the regulatory bodies of physiotherapy education and practice in both KSA and the UK is presented in this section. In addition, the relationship between physiotherapy undergraduate curriculum content, international guidelines and pelvic health physiotherapists' competency are outlined.

2.5.1 Physiotherapy education and the regulatory bodies in KSA and the UK

More than 16 universities in KSA offer a bachelor's degree in physiotherapy, but there is no uniform regulatory body for the education and curriculum (Alghadir *et al.*, 2015). The Saudi Health Commission conducts theoretical and practical exams for all physiotherapists before they practise to ensure high-quality patient care. In the UK, on the other hand, the educational curriculum is regulated and approved by the professional body- the Chartered Society of Physiotherapy (CSP). However, the CSP do not specify curriculum content; instead, they prescribe essential learning outcomes and the broader topics that must be included, as well as a recommendation of '1000 hours of hands-on clinical education' (Bithell, 2007).

2.5.2 Pelvic health physiotherapists' competency frameworks

Competency frameworks vary within different professions and nations, as suggested by Lester and Religa (2017). Within physiotherapy, there are other competency frameworks in the core areas of practice at the undergraduate level, such as sports and orthopaedic fields (Bulley and Donaghy, 2008). While these core areas are not directly related to pelvic health physiotherapists' unique practice – as this area is not taught at the undergraduate level – it would give clinicians basic knowledge on how to practise in this area after

graduation. There is a lack of a pelvic health physiotherapist competency framework in the literature (Frawley, Neumann and Delany, 2019).

In the UK, one university provides an advanced postgraduate course in women's health and continence management. In addition, beginner and advanced courses are provided continuously by the Pelvic Obstetructive Gynaecology Physiotherapy association to ensure pelvic health physiotherapists' competence before they practice to advance their members' skills. In KSA, on the other hand, there are limited entry-level and postgraduate pelvic health physiotherapist courses at universities; the Saudi Physiotherapy women's health Association also provides beginner-level courses at the post-graduate level.

The International Organization of Physical Therapists in Pelvic and Women's Health (IOPTPWH) recommends that the knowledge of entry-level physiotherapists should include basic and clinical science aspects related to the management of women's health problems (Frawley, Neumann and Delany, 2019; McPherson, Nahon and Waddington, 2020). Educators from the ICS developed a guideline for advanced skills and knowledge on pelvic health physiotherapist education and recommended using it in the educational curriculum of pelvic health physiotherapists (Bakker *et al.*, 2018). This may help in creating create minimum standards at the graduate entry level, resulting in structured competencies within pelvic health physiotherapists courses.

McClurg *et al.* (2013) conducted a UK-wide survey of continence education within undergraduate programmes, including medicine, nursing, midwifery, physiotherapy, and occupational therapy. The survey results demonstrated that, contrary to international recommendations by IOPTPWH and ICS, there is limited evidence of the essential inclusion of a specified number of hours of incontinence education in the UK. The survey reported that the average number of hours across all programmes and professions was 4.7. The response rate in the survey was 81%, with the highest response rate being achieved from the adult nursing programmes, which might not be representative of the other undergraduate programmes and potentially limits the possibility of generalising the results across all professions within the UK. In addition, in the survey, there was no distinction between faecal and urinary incontinence; thus, responses were interpreted as relating to both. The limited distinction between the two incontinence problems may lead to different answers from the participants. In this survey, the question about learning and teaching strategies was used to promote continence learning, including various types of teaching strategies such as seminars, lectures and practice-based learning. While, seventy-eight percent of the programmes surveyed continued to rely heavily on lecture-based teaching, which might influence the participants reply on the average number of hours spent in continence education. However, the survey findings may highlight an area of concern given that examples of poor quality care and the number of patients at risk of incontinence are set to increase over the following decade (McClurg *et al.*, 2013). Moreover, there is evidence that if patients receive proper continence management, cost and social care savings can be achieved (Williams *et al.*, 2005).

2.6 Varying degrees of difficulties related to digital palpation

Digital palpation or vaginal examination is considered a vital assessment method for patients with UI. The vaginal examination provides information about the strength of the pelvic floor muscles using the Modified Oxford Scale (MOS), which guides the physiotherapy treatment plan (Bo, 2005). Performing digital palpation in a clinical routine is essential to assess if the patients are contracting the right pelvic floor muscles, show patients' progress and is a method that does not require any equipment (Laycock, 1991).

Other options are available to assess pelvic floor muscles, such as ultrasound, electromyography and perineometer. The validity and reliability of transperineal ultrasound in assessing pelvic floor muscles are already established in the literature (Van Delft, Thakar and Sultan, 2015). However, there is the possibility of cross assessment from surrounding muscles in electromyography and perineometer, thus decreasing the reliability of the data (Bo et al., 2014). The transabdominal ultrasound is a gold standard that images the base of the bladder during pelvic floor muscles contraction, resulting in a vertical displacement of the base of the bladder, representing pelvic floor muscle contraction. Imaging of transabdominal ultrasound does not provide physiotherapists with a muscle strength grade but instead a visualisation of a patient's ability to activate the pelvic floor muscles adequately to elevate the bladder neck (Bo et al., 2014). There are different advantages and disadvantages to transabdominal ultrasound; one advantage is that it allows for lower extremities movement, which might be helpful in pelvic floor muscles training; also, it offers non-invasive pelvic floor muscle assessment of patients with contraindication to digital palpation such as sexual disorders, virgin and teenage girls with UI (Dietz, 2011; Bahrami et al., 2021). The disadvantage includes advanced training

and limited ability to assess each levator ani muscles individually, that is only provided by digital palpation.

Most of the entry-level physiotherapy courses do not include pelvic floor examinations; they may not be formally taught or included as a core skill in the UK and/or KSA in section 2.5.2 (Frawley, Neumann and Delany, 2019). The literature suggests that digital palpation is an intimate examination requiring specific sensitivities since patients might find it challenging to disclose their symptoms (Lawrence *et al.*, 2008). Patients with UI have been found to show different levels of stress and anxiety during consultations, which may lead to varying degrees of discomfort related to obstetric history (Fitzgerald *et al.*, 2011; Kavvadias, Baessler and Schuessler, 2012; Bloomfield *et al.*, 2014). As a result, the decision to conduct digital palpation must be undertaken carefully. Assessment and treatment of an intimate body part require an awareness of cultural, religious and contextual factors and communication skills to assess the patient's understanding of the treatment plan (Delany and Frawley, 2012).

Hence, there are differences between the UK and KSA in terms of the organisational body overseeing the curriculum content and the competency frameworks in the core areas of practice in physiotherapy as well as at pelvic health physiotherapy undergraduate and post-graduate levels. The influence of pelvic health physiotherapists' education and competency in clinical reasoning and decision making will be explored further in this study. The following section details the health care systems in the UK and KSA.

2.7 Health care systems in the UK

In each of the UK countries, the National Health System (NHS) has its structure and organisation, but overall, healthcare comprises two broad sections: one dealing with strategy, policy and management, and the other with actual medical/clinical care. In the UK health vision 2025, Public Health England (PHE) has substantial scope to say what assistive technology will be used to protect and improve the public's health and reduce health inequalities over the following five years (Public Health England, 2019). Inequalities in health are seen by region, gender, socio-economic features and ethnicity. Limited access to healthcare can be a problem both internationally and in the UK (Ham, 2009). The health care system in the UK is divided into primary, which includes some community services, General Practitioners (GPs), and Dentists and Pharmacists. The first

contact practitioner roles are a new model evolving in the UK, which involves placing experienced physiotherapists with advanced skills directly into GP practices to assess, diagnose and recommend appropriate treatment or referral without the need for any initial assessment by a GP.

Secondary care includes hospital-based care,often in district general hospitals accessed through GP referral, whilst tertiary care involves specialist hospitals that treat particularly complex or rare Conditions (Boyle, 2011).

2.8 Health care systems in KSA

The KSA health system is divided into primary, secondary and tertiary, similar to the UK. In KSA, the Ministry of Health (MOH) is the primary government provider and sponsor of health services. The MOH has undertaken many steps to reform the Saudi health care system. However, some challenges still exist, particularly concerning the workforce, financing and expenditure, utilisation of electronic health strategies and accessibility to health care services (Almalki, FitzGerald and Clark, 2011).

Regarding the broader context, Mousa and Aldehayyat (2018)found that health care efficiency in KSA differed across selected regions and between private and public sectors. Mousa and Aldehayyat (2018) used data from the Health Statistical Annual Book (2014) collated by the MOH in Saudi Arabia. The study included the health-care services provided in medical centres and hospitals in Saudi Arabia, including those governed by MOH and the private sector in 13 regions of Saudi Arabia. Mousa and Aldehayyat (2018) used different human and physical capital inputs to assess the efficiency of healthcare services, such as the number of physicians, nurses, and beds. Mousa's quantitative study using data envelopment analysis (DEA) showed that the Riyadh region, which employs a disproportionately large amount of health resources compared to other regions, is efficient in both sectors.

Mousa and Aldehayyat (2018) also found that less developed regions within KSA show a relatively high level of efficiency. In contrast, others, such as Makkah and the Eastern regions, seem to be the least efficient in both sectors. As part of KSA's Vision 2030, the Ministry of Health (MOH) has begun developing a new healthcare system that aims to help the nation socially, mentally, and physically through a new patient-centred Model of

Care (Rahman and Al-Borie, 2020). In KSA, there are different government hospitals, such as university and military hospitals. Staff training and research funding may vary from one sector to another ,which may influence, the staff level of certainty in decision making. This is supported by Mirzaei *et al.* (2019), who conducted research in a developing country assessing the possibility of applying for international accreditation in a university and military hospital.

The structure of the health care organisation in the UK and KSA are pretty similar. However, there is an apparent discrepancy in health access in most of the regions in KSA and the lower socio-economic areas of the UK. The high demands on the MOH in Saudi have led to the introduction of autonomy in financial and administrative services in hospitals in KSA. It could be argued that autonomy in management may create differences in organisational culture. Recruiting participants from different sectors and regions in KSA and the UK may be essential to understand the influence of these factors on clinical reasoning and decision making. An overview of the healthcare system in the UK and KSA is critical because this study compares the physiotherapists' clinical reasoning and decision making model while managing patients with UI in both countries.

2.9 Understanding clinical reasoning

Clinical reasoning was defined in section 1.2.1. Clinical reasoning is essential for the autonomy of different health professions (Higgs, 2008). In 1977, the Department of Health memorandum HC (77)33 established professional autonomy for UK physiotherapists, permitting them to accept referrals from individuals themselves (self-referrals) and make their own diagnoses and treatment decisions. With greater autonomy (Øvretveit, 1985), came an increased demand for effective clinical reasoning to support practitioners' understanding of the required knowledge as well as cognitive and reflective skills that need to be developed to make high-quality assessments and treatment decisions (Higgs, 2008). Physiotherapists need to clinically reason about likely causation and impact and make decisions about decide what treatments to apply. Understanding how they clinically reason and make these decisions is essential for ensuring good practice, practical outcomes, and efficient services. Therefore, a scoping review and narrative synthesis of the literature were undertaken.

2.10 Scoping review and narrative synthesis of Clinical Reasoning in Physiotherapy

This scoping review, originally conducted in 2016 and updated in 2021, sought to understand pelvic health physiotherapists' decision-making processes in assessing and treating patients with UI following the procedure outlined by Arksey and O'Malley (2005). A systematic search of clinical reasoning in physiotherapists was carried out using different key words, for example: 'physiotherapy' AND/OR 'urinary incontinence' AND 'clinical reasoning' AND 'decision making'. Arksey and O'Malley (2005) suggested using different research objectives developed in support of the current research question or aim as a basis to create different key words of the search objectives was to explore the models of clinical reasoning among physiotherapists treating patients with UI (Appendix A).

The retrieved studies were scanned for the relevant references and processes of narrative synthesis to identify research gaps and to inform the rationale of the current study (Grant and Booth, 2009; Tong *et al.*, 2012). To understand the factors that may contribute to the similarities and differences of clinical reasoning in the UK and KSA, further keywords were added, including 'social' AND 'ethnicity' AND 'cultural diversity' AND 'organisation', in addition to other keywords. More details about the keywords, different databases, including grey literature, and inclusion or exclusion criteria are explained in Appendix A.

The scoping review demonstrated that, in total, 1189 papers were retrieved. After reading article titles and abstracts, 1029 studies were excluded because they did not address either clinical reasoning in physiotherapy for UI or clinical reasoning in physiotherapy more generally, in addition to other exclusion criteria mentioned in Appendix A. Those initially included (n= 28) were then read in full and described in a summary chart represented in Table 7-6 in Appendix B. No studies focussing on clinical reasoning and decision making in the field of physiotherapy for UI were identified at either time point – see Appendix A, and Appendix B. The twenty-eight studies related to physiotherapy in clinical reasoning more generally provided new insights and are discussed in the narrative in sections 2.11 and 2.15. To summarise, no papers reporting on clinical reasoning in physiotherapy for UI were found at either time point highlighting the need for this study. However, the broader literature on clinical reasoning in physiotherapy identified a range

of considerations of value to this study. The outcome of the 2016 and 2021 scoping review is presented in the following Figure 2-3.

Figure 2-3: The selection processes of different studies in 2016 and 2021 in clinical reasoning in pelvic health physiotherapy and other physiotherapy area using the Preferred Reporting Items for Systematic Review Chart (Moher *et al.*, 2009).



The quality of the qualitative studies was identified using both a checklist and framework, as suggested by (Williams, Boylan and Nunan, 2020). One of the most widely used appraisal tools or checklists is the Critical Appraisal Skills Programme by CASP (2022), which judges aspects of recruitment, sampling, data collection and analysis in qualitative research papers. In addition, the framework focuses on the overarching concepts of quality in qualitative research, including transparency, reflexivity, dependability, and transferability. The quality of the qualitative studies was discussed with the supervisory team, and only high-quality qualitative studies were included (Mays and Pope, 2020). Physiotherapy specialities treating acutely ill people, such as end-of-life disease management and cardio-respiratory care, were excluded from the search because clinical reasoning in those fields of physiotherapy is unlikely to be similar to that used in UI (Smith, Higgs and Ellis, 2008; Evans, Sharp and Shaw, 2012). The subsequent literature review discusses the narrative synthesis of the scoping review findings by describing the common clinical reasoning models and factors that might influence decision making in physiotherapy.

2.11 Clinical Reasoning in physiotherapy

The most common clinical reasoning models discussed within the literature have been explored in relation to different physiotherapy specialities, such as musculoskeletal, neurology and domiciliary. The following sections present the findings of the scoping reviews and narrative synthesis for 2016 and 2021, which include different clinical reasoning models, such as hypothetico-deductive reasoning and pattern recognition, narrative, ethical and dialectical reasoning, in addition to bio-psychosocial reasoning.

2.11.1 Hypothetico-deductive clinical reasoning and pattern recognition

In the hypothetico-deductive (H-D) model, physiotherapists start with a hypothesis regarding a patient that develops as they question and examine the patient. This process of questioning and examination is instrumental to the physiotherapist's acceptance or rejection of their original hypothesis (Elstein, Shulman and Sprafka, 1978). H-D reasoning is part of the cognitivist model of clinical reasoning. It involves information processing, memory representation and problem-solving (Case and Bereiter, 1984). The H-D reasoning model is frequently found to be used in physiotherapy and medicine within the diagnostic process (Payton, 1985). It has been suggested that the clinical reasoning of

experts and novices differ (May *et al.*, 2008a). Novice physiotherapists commonly use the H-D model, while experienced physiotherapists tend to use it in more complex cases or when facing a difficult diagnosis (Collins, Johnson and Masaracchio; May *et al.*, 2010a; Dierckx *et al.*, 2013; Karvonen *et al.*, 2017).

It has been suggested that as physiotherapists gain experience, they are more likely to use metacognitive skills and knowledge (Øberg, Normann and Gallagher, 2015; King *et al.*, 2018). Pattern recognition is a combination of cognitive and metacognitive skills as supported by Flavell (1979) and is often explicit in terms of intuitive reasoning (Marcum, 2013). It is characterised by speed and efficiency. It occurs in familiar clinical cases where physiotherapists use 'illness scripts' to aid rapid and accurate retrieval of information from a well-structured knowledge base. New cases are classified, and similarities are acknowledged in relation to previously experienced clinical cases (Schmidt, Norman and Boshuizen, 1990; Schuwirth, 2002). Novices may require experience and knowledge to use metacognitive reasoning (Schuwirth, 2002; Vedin, 2021).

Prokop (2018p.356) examined the use of the dual-processing theory in physiotherapy and asserted that 'a crucial element to accurately manage the use of both types of processing is the awareness of the accuracy of the thought process.' According to Croskerry (2009b) this theory proposes two general kinds of cognitive operations to explain and understand human reasoning. Often, the two processes consist of an implicit, more intuitive, unconscious process (Type I) and an explicit, more analytical, conscious process (Type II). This theory could explain how diagnostic failure occurs (Croskerry, 2009a).

The awareness of one's own cognition, known as metacognition, is challenging to teach to physiotherapy students (Kosior, Wall and Ferrero, 2019; Daluiso-King and Hebron, 2020). Hence, it is important to consider the use of techniques that encourage metacognition, such as case-based learning, simulated and standardised patient interactions, and experiential learning to provide tools that promote metacognitive awareness in health profession students (Kosior, Wall and Ferrero, 2019).

2.11.2 Narrative reasoning

Narrative clinical reasoning is defined as a reasoning strategy that involves the clinician engaging themselves in gaining an understanding of the unique lived experience of the individual patient (Edwards *et al.*, 2004a). Hence, clinicians obtain information regarding patients' stories, experiences of illness, meaningful perspectives, contexts, beliefs and cultures (Mattingly and Fleming, 1994). Narrative reasoning helps to individualise and personalise the approach to each individual patient's needs (May *et al.*, 2008b; Foster and Delitto, 2011; Holdar, Wallin and Heiwe, 2013). A number of studies have shown that experienced physiotherapists used narrative clinical reasoning, with Edwards *et al.* (2004a) demonstrating that neurological physiotherapists use only narrative clinical reasoning in the assessment and treatment of neurological patients; and Mattingly (1991) reporting that occupational physiotherapists use narrative reasoning as a central model of clinical reasoning.

The use of the narrative model is in contrast with the biomedical model, in which therapists mainly focus on the pathology, included the physical symptoms and impairments, assessment goals and treatment modalities and strategies. Narrative reasoning includes story telling or story-making. Physiotherapists often make metaphors and analogies to make the exercise easier for the patients to learn and often they will model the movement or exercise for the patient. They will frequently depend on exercise modelling and movement in their clinical reasoning and decision making. For instance, they rely on their bodies and hands as tools to gather data and demonstrate information used in their clinical reasoning process (Edwards, Jones and Hillier, 2006; Øberg, Normann and Gallagher, 2015). However, it is more challenging for physiotherapists to use movement as a sensorimotor tool to demonstrate pelvic floor muscle exercise in comparison to other physiotherapy specialities, as a result this may influence their clinical reasoning and decision making. For instance, they may decide to use biofeedback to help patients visualise pelvic floor muscle contraction and relaxation, but this may be limited due to contextual factors. The therapeutic efficacy of narrative reasoning interventions is about much more than meeting specific treatment goals; it is about creating an experience that gives the participants a vision of themselves in the world (Mattingly and Fleming, 1994; Higgs, 2008).

The narrative model of reasoning has been presented as a core component of the clinician's competencies towards patient centred care (Kumagai, 2008); however, it has been shown that novice physiotherapists have difficulty in engaging in narrative reasoning compared to experienced physiotherapists (May *et al.*, 2010b; Cruz, Moore and Cross, 2012b). As such, this study will examine whether the physiotherapists included in this study use narrative reasoning within their clinical reasoning and decision making processes.

2.11.3 Ethical and moral reasoning

There is a general lack of understanding regarding ethical clinical reasoning and decision making in rehabilitation. Ethical reasoning allows a clinician to use similar principles of inquiry, CR and validation in order to understand and make decisions regarding ethical issues. Walker (2002 p.353) has identified moral reasoning as the psychological process which 'entails deliberation regarding the various consideration relevant to different courses of action and making a judgement regarding which of the available actions would be most morally justifiable'.

Furthermore, in a qualitative study using hermeneutic phenomenology, physiotherapists described a recent ethically based clinical decision (they were asked to describe scenarios where they considered ethical values' importance in their decision making) (Finch, Geddes and Larin, 2005). After analysing the interview transcripts, it was revealed that physiotherapists had a clear understanding of professional ethos, valued patient autonomy and often explored a variety of alternative actions before implementing one. However, it was also found that physiotherapists were not always fully aware of certain ethical principles and often did not use a theoretical ethical approach to analyse the ethical problems. Other qualitative research has also reported similar findings, stating that physiotherapists often consult colleagues for advice on ethical dilemmas rather than referring to the evidence based ethical literature (Doherty and Purtilo, 2015).

A recent study by Delany, Edwards and Fryer (2019) explored physiotherapists' responses, interpretations and perceptions of ethical challenges in different work contexts, such as MSK, palliative care, continence and women's health. They found ethical challenges from the working environment, such as healthcare policies, ethical codes of conduct and pressure from hospital administrators and others to discharge patients that might be against the physiotherapist's decision. In addition, it was challenging to balance

diverse needs and expectations of physiotherapists, patients or their family, and other ethical obstacles that arise when providing services to diverse groups of people. This tends to happen where differences in language and cultural views about health and wellbeing make it ethically challenging to modify treatment to meet a patient's requirements, especially in end-of- life situations. The authors explored ethical reasoning within different work contexts but did not explain the influence of managing and making decisions on diverse groups of patients in detail.

Ethical reasoning is affected by a physiotherapist's values, universal ethical principles, codes of ethics, previous experiences, moral virtues, culture, patients' stories and ethos (Edwards, Braunack-Mayer and Jones, 2005). Pelvic health physiotherapists deal with intimate body parts and can be exposed to ethical or sensitive problems experienced by patients; hence this study focuses on the clinical reasoning and decision making processes of physiotherapists and aims to gain an insight into the impact of ethical challenges on their clinical reasoning and decision making.

2.11.4 Literature for an integrated approach: Bio-psychosocial clinical reasoning model

The existing physiotherapy clinical reasoning and decision making literature explores the use of bio-psychosocial clinical reasoning in detail; it is clear that there is an overlap between bio-psychosocial, collaborative, and shared decision making (SDM). In a recent concept analysis exploring the meaning and implementation of bio-psychosocial reasoning in musculoskeletal physiotherapy, Daluiso-King and Hebron (2020) mentioned that psychological and social elements are important with respect to biological causes in the treatment of chronic low back pain and chronic illness. Physiotherapists were aware of the importance of bio-psychological reasoning in gaining active patient engagement, and reported improved self-management adherence and better outcomes when compared to paternalistic clinical reasoning and decision making (Sanders *et al.*, 2013; Hall *et al.*, 2016). When the physiotherapists see the world through their patients' lens to understand the relationship between patients' beliefs, behaviour and psychological factors, the individual reported feeling heard after active engagement with their lifeworld (Solvang and Fougner, 2016).

Shared decision making is when physiotherapists and patients collaboratively make health-related decisions, discuss the treatment options and harms of each option, and consider the patient's values, preferences, and circumstances (Hoffmann *et al.*, 2014).

Hence, collaborative decision making and shared decision making often overlap conceptually and contextually within the literature; yet, the importance of collaboration and involving the patient, family, and the multidisciplinary team in the decision making process is often emphasised in clinical reasoning literature (Edwards et al., 2004a; Furze et al., 2015; Øberg, Normann and Gallagher, 2015). Patient-centred care or person-centred care is manifested by treating the patient as a person that may include sharing power, giving greater responsibility, and sharing the decision making with the patient to promote the self-management approach. The person-centred -care and evidence-based practice somewhat overlapping Fields (Ekman et al., 2011). Treatment guidelines supported by evidence are diagnosis based and contrast with the ideals of person-centeredness. They did not accommodate the heterogeneity in patients' initial presentation, recovery rate, and the reasons that people chose specific treatment options (Kittelson et al., 2020). Scholars have reported that physiotherapists may experience challenges integrating patient-centred care elements within their clinical practice. Shared decision-making is one of the components of patient-centred care and there have been calls for a move toward personcentred physiotherapy care that includes a shared decision making (Dukhu, Purcell and Bulley, 2018; Hall et al., 2018; Hutting et al., 2020). Although shared decision-making is essential to the provision of patient-centred care and evidence-based practice, the integration into physiotherapy management of the chronic condition is low (Hoffmann, Lewis and Maher, 2020). The principles of shared decision making are well documented in the literature, but there is limited guidance about how to achieve that in daily clinical practice (Elwyn et al., 2012). Hence, it is appropriate to be incorporated into managing long-term health conditions in people with multi-morbidity (Hoffmann, Jansen and Glasziou, 2018).

Dierckx *et al.* (2013); Jones *et al.* (2014) conducted observational studies to assess SDM in physiotherapy and the level of patient involvement during the clinical reasoning and decision making process. In these studies, the Observing Patient Involvement (OPTION) instrument was used to measure shared decision making by observers. Dierckx *et al.* (2013) used participant observations and agreement between therapists and patients on the 'Control Preference Scale' to assess the patient's involvement in decision making. The results of both studies indicated that shared decision making was underdeveloped in the observed consultations. It was indicated that most patients preferred to give their thoughts about the assessment and treatment rather than delegating the decisions to physiotherapists. The reason for the low level of patients' involvement in the above studies

might be because physiotherapists mistakenly think that they can estimate the patient's preferred level of involvement without asking (Hirsch *et al.*, 2012). In addition, physiotherapists in Dierckx *et al.* (2013); Jones *et al.* (2014) did not attend training in shared decision making, while the OPTION scores of health professionals who had been trained in shared decision making were significantly higher than scores of untrained health professionals (Hirsch *et al.*, 2012).

To date, most studies that aim to increase the uptake of shared decision making in physiotherapy have been conducted with therapists who work with musculoskeletal patients. The focus of shared decision making is on the interaction between the patient and physiotherapist, and to be part of the clinical routine, and it needs to be embedded within an organisation where it is encouraged and supported through leadership and adequate resources (Coulter, 2018). To implement patient-centred care, clinicians need to reflect on their own beliefs and practice, encourage patients' involvement and facilitate disclosure of beliefs, fears, and concerns. This collaborative decision making process promotes self-management and ultimately patient-centred care (Hutting *et al.*, 2020).

2.11.5 Dialectical reasoning

A 'dialectic' is a discourse between two or more individuals with differing points of view that intend a reconciliation through a dialogue of reasoned debate (Jones, 1995; Edwards, Jones and Hillier, 2006). Hence, dialectical reasoning is a term used to describe the process of arriving at a clear understanding by comparing and contrasting various solutions and viewpoints (Merriam and Baumgartner, 2020). Within clinical reasoning, Edwards *et al.* (2004a p.323) defined the dialectic reasoning model as an 'interplay between every recognised reasoning component of clinical practice, namely: diagnosis, procedure, interaction, teaching, collaboration, prediction, and ethical decision making.

In a qualitative study involving 12 physiotherapists, Edwards *et al.* (2004a) argued that the dialectical reasoning model fits well with recent models of health and disability, such as the 'biopsychosocial model' as well as the 'World Health Organisation (WHO), International Classification of Functioning, Disability and Health (ICF). Physiotherapists often use dialectic reasoning to assist with the clinical reasoning and decision making process; however, it has been criticised by some researchers as being over-simplistic and

limited because it does not include the role of movement (Øberg, Normann and Gallagher, 2015; Huhn *et al.*, 2019).

2.11.6 Limitations of the bio-psychosocial model in clinical practice

There is limited standardisation in implementing bio-psychosocial and shared decision making models in the clinical practice (Daluiso-King and Hebron, 2020). Qualitative studies using semi-structured interviews aimed to explore physiotherapists' approaches to patients' psychosocial concerns. Physiotherapists felt undertrained to handle patients' psychological elements related to their chronic pain or illness, and some of them felt these problems were outside their scope of practice (Sanders *et al.*, 2013). Pelvic floor health depends on life events and predisposition such as biological factors (aging and genetic factors) and external factors such as cultural, socio-economics, lifestyle and life events (Brandt, 2021). Ultimate or optimal pelvic floor health is understood to be a concept that encompasses the physical and functional integrity of the pelvic floor structures, in addition to the effect that pelvic floor muscle dysfunction has on an individual's psychological, social and quality of life context (Pierce *et al.*, 2015).

A range of contextual factors, such as time and resource and lack of relevant academic and professional training, have been cited as limiting the application of a biopsychosocial approach (Foster and Delitto, 2011; Singla et al., 2015; Daluiso-King and Hebron, 2020). Others argue that the prominence of biomedical and hypotheticodeductive reasoning models in physiotherapists' student training methods have inhibited the integration of the shared decision making and bio-psychosocial models into clinical practice (Domenech et al., 2011).

2.12 Summary of clinical reasoning literature

Physiotherapists move from one clinical reasoning model to the other based on their experiences, knowledge, training, and patients' cues. The presence of different types of reasoning in response to different factors is suggestive of the physiotherapists' clinical reasoning and decision making. Physiotherapists' reasoning incorporates a synthesis of patients' information, ongoing process, reflection, and reassessment of ideas in the reasoning process and decision making (Furze *et al.*, 2015; Nicola-Richmond, Pépin and Larkin, 2016; Elvén and Dean, 2017). There is a range of clinical reasoning approaches:

some suggest that physiotherapists may prefer the use of pattern recognition and biopsychosocial models, but the evidence is weak (Slade et al., 2020). Despite mentioning bio-psychosocial, the focus of clinical reasoning appears to be more often on biology rather than psychology or social factors. Social factors are an important aspect when considering UI literature because patients with UI may reveal social withdrawal, whereby one cannot consider the impact of social factors in clinical reasoning and decision making without understanding those which affect women. It is also reasonable to assume that these factors will vary depending on the culture. To date, there has been little to no consideration of the impact of culture on physiotherapists' clinical reasoning and decision making.

Section 2.11 explored the existing literature on clinical reasoning in physiotherapy and evaluated the areas where the theories of reasoning may or may not be applicable for this current study. However, it also identified gaps in the literature and their ability to explain the complexity of physiotherapists' clinical reasoning and decision making while managing patients with urinary incontinence across two different countries and cultures. Therefore, there is a need for broader literature that considers various theories on decision making and reasoning beyond physiotherapy, UI, western cultures, and organisations. The literature on these potentially useful theories was not located through a single predefined systematic search strategy. Instead, it was found by scanning text (in textbooks and online) to gain a broad conceptualisation of the decision making literature, reading in further depth about the reasoning and decision making theories cited in the most relevant studies. Sufficient literature and theories were explored to address the conceptual complexity and will be discussed in the next section.

2.13 Theories in clinical reasoning and decision making

There is a wide range of clinical reasoning and decision making theories in different disciplines, including economics, psychology, medicine, marketing, computer sciences and organisational sciences. A brief overview of the various theories will be presented.

2.13.1 Classic decision making theory

The classic decision making theory is a normative approach that assumes health practitioners make decisions based on statistical probabilities. It assumes that "decision makers are objective, have complete information and consider all possible alternatives and

their consequences before selecting the optimal solution" (Huczynski and Buchanan, 2001p.630). There is a strong argument that this may not be how people make real life decisions.

2.13.2 Dual Process Theory

According to Croskerry (2009a), this theory proposes that there are two general kinds of cognitive operations to explain and understand human reasoning. Often, the two processes consist of an implicit, more intuitive, unconscious process (Type I) and an explicit, more analytical, conscious process (Type II). In some circumstances, an intuitive approach may be applicable, while in others, an analytical approach might be preferred as suggested by Simon (1990), and a blend of the two may be sometimes optimal. Hammond (2000) believed there to be a continuous alternation between the two modes. The dual process theory has been extensively researched within the fields of reasoning, judgment, and decision making (Evans, 2008). It is widely recognised and applied in philosophy, psychology, neurology, neuroanatomy, neurophysiology, and genetics (Lieberman, Jarcho and Satpute, 2004; Buschman and Miller, 2007).

Croskerry (2009a) proposes that the dual process theory particularly applies to diagnostic reasoning within a medical setting. It can be used to understand and explain how diagnostic reasoning skills are acquired, how they optimally function and how diagnostic failures occur. At the beginning of the assessment, the health care professionals access features of the medical problem either from the patient or a colleague. Some conditions may be diagnosed on visual signs, such as facial palsy. Still, others will need additional evidence, such as an explanation of symptoms, quality of life or other severe aspects of the presentation. Early in the process, it will be obvious whether the condition is recognised or not. If it is, Type I (intuitive) processes will rapidly make the diagnosis, and nothing further may be required. If it is not, then analytical type II reasoning will need to be involved instead. Dynamic fluctuation may occur between the two systems (Hammond, 2000).

Different places in the dual process theory where diagnostic failure might occur. Most errors occur in type I processes and may be expected, while type II errors are infrequent and unexpected but may be substantial when they take place (Dawson, 1993). This model has precise applications in different areas. Such as across the work of variable health care fields, research in clinical decision-making, and education in all fields. Also, nursing traditionally emphasises 'intuitive' approaches in decision making (Thompson and Yang, 2009).
However, physicians might blend two methods depending on the time and resources, and sometimes an optimal calibration may be needed for specific clinical situations under particular conditions. It is unclear if physiotherapists are using dual process theory within their decision making processes or not, and this is explored within this study.

2.14 Perspective Theory

The classic decision making theory and dual process provide possible solutions to less than optimal decision making and clinical reasoning. The goal of decision making in prescriptive theories is to shift decisions toward a normative world. There is an explicit limitation in the above theories because they do not provide an understanding of how people make decisions in real clinical life. In contrast, the following section considers theories that provide a good explanation of how decision making happens, concentrating on naturalistic decision making, institutional theory and sense making.

2.14.1 Naturalistic Decision making

Naturalistic Decision Making (NDM) is "the way people use their experience to make decisions in the field settings (Zsambok and Klein, 2014p4). NDM is different from other traditional decision models because it considers the decision making in real-world behaviour and settings. There are four markers of NDM, which include task and setting factors, research participants (experiences versus novices), the aim of the research and the locus of interest within the decision episode (Zsambok and Klein, 2014). The NDM researcher studied people in different fields, such as Navy commanders, jurors, nurses, and engineers (Klein, 2008). NDM models are more convenient for explaining experts' reasoning in different domains, such as dynamic situations, time pressure, and multi-players.

NDM has different decision models; for example, the recognition-primed decision is one of the models of NDM as suggested by Klein, Calderwood and Clinton-Cirocco (1986). The recognition-primed decision (RPD) describes how people make decisions by matching the patterns they have learned from previous experience (Klein, Calderwood and Clinton-Cirocco, 1986). The RPD model includes a combination of intuition and analysis.

NDM has changed and established new models, such as the RPD and critical decision method (Mosier *et al.*, 2018).

In a study informed by the critical decision method (CDM), interview data were collected to understand challenging surgical moments for 32 surgical cases. Purposeful sampling was used to include surgeons with a variety of experiences. The result showed that surgeons begin each surgery with a planned course of action (pattern) and continue to review that course of action during the surgery in response to the emerging situation (analytical) and the perceived difficulty level. The proposed model was created based on existing theories of naturalistic decision making (Cristancho *et al.*, 2013). Furthermore, in a study by Reiter-Palmon *et al.* (2015) to understand how naturalistic decision making can be utilised as a learning tool to decrease hospital errors. They collected 226 falls over a certain time and implemented the naturalistic decision making theory. They found that the types of errors identified as contributing to patient fall changed, with a decrease in the number of errors. This study illustrates the role of NDM in health care and real-life settings.

NDM has been criticised because it focuses on experts' behaviour and decision-making processes (Elstein, 2001). For instance, expert decision makers usually generate an initial practical course of action and then assess the consequences of that action; if that action works as desired, they act. NDM did not consider other factors in the decision making processes, such as individual, social, and psychological factors; as a result, it might not be applicable in this study because of the complex nature of physiotherapists' decision making while managing patients with UI.

2.14.2 Institutional Theory and Institutional logics

Comparing clinical reasoning models and decision making processes between two different cultures and healthcare institutions required an understanding of institutional theory. Institutional Theory can be understood as policy-making emphasising the substantial and legal aspects of government structures (Kraft and Furlong, 2019). Socially constructed belief and rule systems exercise enormous control over organisations—how they are structured and carry out their work (Scott, Scott and Meyer, 1994). Over decades there has been an evolution in the institutional theory; there is a variety of social norms and values constitute larger social institutions, which put the organisation under pressure. Friedland (1991) attempted to offer a new explanation of institutional theory by introducing institutional logics as "a set of material practices and symbolic constructions that constitute organising principles for institutions or supra-organisational pattern of human activity" (Friedland, 1991, p243) Institutional logic can be defined as the organisational values that underlie belief

and practices within an institutional setting, playing a traditional role in determining an individual's explanations and legitimatising their activities (Scott, 2013). The work of Thornton, Ocasio and Lounsbury (2015) proposes six forms of institutional logics: the market, professions, corporations, the state, the family and religion. Each has distinguishing features that are useful for studying multiple logics that might contradict each other. As this study sought to understand how different countries, cultures and organisations impacted clinical reasoning, it was anticipated that an understanding of institutional logics would be necessary.

The linkage between institutional logics and practices is well established; organisational practices are seen as manifestations of the institutional logics (Greenwood *et al.*, 2010). It is expected that institutional logics affects organisational decision-making by moving the attention of decision-makers. For example, Thornton, Ocasio and Lounsbury (2015) stated that when one logic is dominant, the concentration of decision-makers is directed toward issues and practices consistent with this logic. Often organisations experience multiple and sometimes conflicting institutional logics (Thornton and Ocasio, 2008). Health care is highly fragmented; as a result, health care organisations depend on many actors with possibly different logics (Pache and Santos, 2010). Other authors indicate shifts in institutional logics in the health care field from a professional logic to a business-like logic. In addition, it is acknowledged that multiple institutional logics might coexist (Kitchener, 2002; Reay and Hinings, 2009; Scott, 2013). As suggested by Goodrick and Reay (2011), autonomy is an essential aspect of professional logic, the core parts of the professional logic are high quality of care, sufficient time to spend directly on patients and autonomy. On the other hand, the business-like logic ascribes importance to practices that could lead to cost reduction (Reay and Hinings, 2005).

In a study Productive Ward: Releasing Time to Care by Robert *et al.* (2011), this programme was developed in 2006 by the NHS Institute for Innovation and Improvement in the United Kingdom. The core assumption of the programme is that nursing staff organise their own ward and improve processes themselves. This may increase the time spent on patient care, resulting in a higher quality of care and more patient and nurse satisfaction. One of the programme's core elements was to increase the nurse's empowerment and autonomy. How the programme is communicated throughout the organisation refers to multiple institutional logics. The labelling of the programme, 'Productive Ward: Releasing Time to Care, suggests that this could be an example of a hybrid practice that incorporates both the professional

nursing logics (Releasing Time to Care) and the business-like logic (Productive Ward)(Robert *et al.*, 2011).

Institutions enable organisational identities and their implementation and enactment (Scott, 2013). Thus, institutional-based categories can explain cultural ranges of meaning that organisations can fit to address the question of 'who we are (Greenwood *et al.*, 2010). The answer simultaneously characterises their central features and classifies them as a member of one organisational field but not of others. This category distinguishes them from other organisations or social communities (Greenwood *et al.*, 2010). Cerderbom, Bjerk and Bergland (2020), in a qualitative study exploring physiotherapists' views on their role in working with fall prevention services, found that a favourable policy environment is required to support the physiotherapists' multifaceted role in fall prevention. A favourable policy environment involves person-centred communication and collaborative practices between professionals, focusing on patient cues and concerns and family interactions (Cerderbom, Bjerk and Bergland, 2020; Manhas *et al.*, 2020). Using institutional theory in this study is important because it has multiple logics and could help understand the possible differences and similarities between physiotherapists' clinical reasoning and decision making in the UK and KSA.

This study aims to explore physiotherapists' clinical reasoning and decision making. At the same time, managing patients with UI, the institutional theory may be sought by physiotherapists, but it does not consider different factors such as psychological and social factors. Sense making theory is one such theory, and the following sections describe this theory and present relevant studies.

2.14.3 Sense making processes

Sense making has been adopted in the work of Weick (1995). Sense making provides a lens to explore clinical reasoning and decision making among other health practitioners; it also has the potential to help with understanding physiotherapists' clinical reasoning while managing patients with UI. The sense making theory has been used widely in the nursing literature to account for nurses' decision-making processes. This section presents a review of the theoretical background of sense making, followed by a discussion of its findings from selected studies.

2.14.3.1 Theoretical background

Sense making is an organisational theory with its origin in social psychology. It is a process by which people give meaning to their ongoing experiences. Sense making focuses on human agency, ambiguity, and relationships (Weick, 1995). The human agency concentrates on the actions people take based on their explanation of ambiguous circumstances that generate sense making (Weick, 1995; Weick, Sutcliffe and Obstfeld, 2005). Relationships refer to the social process of sense making, and consider the actions people take based on their explanation of an event, the social reactions to dealing with the event, and the subsequent sense making of the event (Weick, Sutcliffe and Obstfeld, 2005). Sense making occurs retrospectively and prospectively and reduces the ambiguity from an unpredicted event, allowing a health practitioner to carry out daily tasks within an ultimately complex system (Battles *et al.*, 2006). The seven properties of sense making, explained by Weick (1995), are presented in Table 2-1

Properties of sense making	Meaning
Grounded in identity construction	The process of figuring out what is going on is a product of and a process based on who the sense maker is and who they are becoming
Retrospective	Making sense based on data that has occurred and trying to justify it. Shaping experience into meaningful patterns based on our memory of experience
Enacts sensible environments	This happens when deciding on a domain of activity and then taking action. In sense making, an individual creates their own environment for future action.

Table 2-1: Weick (1995) describes seven properties that are part of the sense making processes

Properties of sense making	Meaning
Social	Considering sense making primarily as an introspective, intrapsychic process. Making sense of things while in conversation with others, reading and exchanging ideas with others.
Ongoing	Constantly bracketing moments in the flow of life as we seek to reflect on and codify the meaning of things. Bracketing means identifying a structure of information so it can be labelled to overcome ambiguity
Based on extracted cues	From actions around us, we tend to bracket and extract certain elements, which become the targets of the sense making process.
Based on plausibility, rather than accuracy	Organisational decision making is often based on odds and can involve intuition and careful analysis and systematic elimination of suboptimal choices.

2.14.3.2 Dimensions of sense making

Unanticipated events and workflow disruptions generate sense making. To resolve the circumstances and continue working, the individual tries to fit the issue into a familiar situation based on their personal knowledge, experiences, and knowledge, as well as the organisational rules and norms (Weick, 1995). When the issue is unfamiliar, an information gap occurs. The individual may seek out help, look for additional clues and take an action. The decision to seek help, improves the communication that may add perspective to the circumstances (Weick, 1995). Communication allows individuals to interpret and construct shared meaning that shapes future action. Recurring sense making from the individual health practitioners can also shift to include sense making at an organisational level (Weick, Sutcliffe and Obstfeld, 2005; Battles *et al.*, 2006).

2.14.3.3 An overview of the existing literature in sense making among different health practitioners

Studies have used sense making theory to implement change in technology, business, and nursing (Paul, 1996; Battles et al., 2006; Kristiansen, Obstfelder and Lotherington, 2015). Other studies have used the role of sense making processes to understand participants' performance, as well as to consider how to eliminate hazards or risks that are a threat to patient safety (Battles et al., 2006; Rosness et al., 2016). There are seven properties of sense making listed in Table 2-1, with much of the literature including all the properties of sense making or only some of the properties (Weick, Sutcliffe and Obstfeld, 2005). For instance, Rosness et al. (2016) studied the prospective and retrospective properties of sense making and tried to implement the use of these properties in the performance of participants in the operating theatre. On the other hand, Battles et al. (2006) mentioned that the participants' conversation about patients' safety incidents encouraged them to start to make sense of the cause of those hazards. This occurred by looking retrospectively and analysing the incidence of accidentally swapping the medications of two patients with the same name. An ongoing process of extracted cues guided the participants to take an action that helped participants to overcome uncertainty prospectively. Hence, sense making theory can be used to overcome ambiguity, to understand participants' performance, risk management, or to make sense of the influence of organisational culture on decision making. Within the existing physiotherapy literature, there is limited use of sense making process, therefore the use of sense making processes to overcome uncertainty in physiotherapists' clinical reasoning and decision making process is explored in this study. This was deemed important as the existing models of clinical reasoning used by physiotherapists who manage patients with UI, as proposed by Slade et al. (2020), did not consider or place adequate emphasis on the effect of culture or social context on clinical reasoning and decision making. This is discussed further in the following section.

2.15 Factors that might influence clinical reasoning

The literature identifies several factors that can influence physiotherapists' clinical reasoning and decision making processes; these may relate to the physiotherapist, the patient or the context. Clear differences in reasoning have been demonstrated between experts and novices (Higgs, 2008; Ajjawi and Higgs, 2012; Cruz, Moore and Cross, 2012a; Holdar, Wallin and Heiwe, 2013), with novices using H-D reasoning, while experts use

pattern recognition. Expertise may be related to years of experience (Doody and McAteer, 2002; Wainwright *et al.*, 2011; Elvén and Dean, 2017), but Slade *et al.* (2020) also demonstrated, that in pelvic health physiotherapist, expertise related to the availability and engagement with continuous specialist pelvic health education and the ability of physiotherapists to interpret and critique research findings and clinical guidelines.

2.15.1 Physiotherapist factors

Expert continence clinicians have been found to highlight the importance of effective communication skills in identifying problems and discussing different treatment options in UI (Slade *et al.*, 2020). Communication is a key factor of bio-psychosocial and shared decision making models that enables 'patient centred' management, by incorporating the patient's narrative (including goals, values, beliefs and choices) (Solvang and Fougner, 2016). Ways of embracing this have been emphasised through physiotherapists' personal characteristics, motivational interviewing (involving open questions), attentive listening and giving patients time to discuss their story; these are common features within the bio-psychosocial model, which encourages active participation by patients (Cowell *et al.*, 2019; Langridge, 2019).

Higgs (2008) also argued that physiotherapists' emotional intelligence (EI) which includes, but is not limited to, active listening, empathy, reassurance, encouragement and using simple language influences clinical reasoning (Marcum, 2013; Aljuraifani *et al.*, 2019). Furthermore, Slade *et al.* (2020) identified EI as a key feature of expert pelvic health physiotherapists. A study by Langridge, Roberts and Pope (2016)) involving MSK physiotherapists suggested that external pressures (such as time restrictions) impacted their clinical reasoning. This caused them to have less consideration when it came to their feelings, emotional factors and physical responses, in particular when dealing with difficult and vulnerable patients. This concept was also discussed in clinical medicine (Marcum, 2013). Patients with UI have been shown to experience a range of negative emotions associated with their incontinence and may have experienced problems, such as sexual abuse that may lead to increased vulnerability; hence, the health practitioner may need time to build trust and rapport that will allow patients to expose some ethical or sensitive issues related to their UI and enable effective clinical reasoning.

2.15.2 Patients' factors

As well as factors associated with the physiotherapist, patients' factors have been shown to influence clinical reasoning (McGinnis *et al.*, 2009b; Amundadottir *et al.*, 2018). The biopsychosocial model of clinical reasoning asserts that physiotherapists take account of biological, psychological and social factors to reach their clinical decisions. However, Dune *et al.* (2021), Mescouto *et al.* (2020) and Brandt (2021) have indicated that social factors are known to impact an individual's health, such as class, gender, family relationships, macro socioeconomic and political contexts, as well as access to healthcare, may be given less consideration than biological or psychological factors. A patient's physical and psychological state is vital to consider in the clinical reasoning and decision making process (Holdar, Wallin and Heiwe, 2013), in addition to patients' goals, motivation, beliefs and their expectation of the physiotherapist and their treatment. All these common factors influence physiotherapists' decision making (Chapparo and Ranka, 2008; Wainwright *et al.*, 2011; Holdar, Wallin and Heiwe, 2013; Hall *et al.*, 2016; Elvén and Dean, 2017).

Patients' belief and culture can affect physiotherapists' clinical reasoning and decision making, since, Stewart and Bennett (2011) found that cultural competence was low among newly qualified UK physiotherapists, while Te *et al.* (2020) found that Australian newly graduated physiotherapists felt challenged when dealing with patients from different culture. Patients' belief and culture were also included as lists rather than in depth factors that influence physiotherapists' clinical reasoning and decision making in MSK, cardio-pulmonary, pelvic floor disorder and chronic illness. Yoshikawa *et al.* (2020) also found that discordance might affect the relationship between physiotherapists and patients which may affect patients' progress. However, it is important to understand how a patients' culture shapes the way health practitioners make decisions.

2.15.3 Contextual Factors

There is limited research regarding the effect of cultural, racial, language and/ or religious factors on clinical reasoning and decision making in physiotherapy in general (Mescouto *et al.*, 2020). Organisational culture has been shown to influence decision making in acute patient care, orthopaedic, medical, neurological and end-of-life diseases in Holdar, Wallin

and Heiwe (2013); Elvén and Dean (2017), but less consideration has been given to the patients or physiotherapists' cultural background.

A recent scoping review synthesised the sociocultural factors, such as cultural and linguistic diversity, affecting chronic pain management between patients, and their physiotherapists. The authors reported culturally generated discordance between patients and physiotherapists (Yoshikawa *et al.*, 2020). They indicated that although some physiotherapists were accepting of the differences in patients' beliefs, they did not necessarily apply that difference in culture into their management plan, which sometimes created stressful relationships between patients and physiotherapists. Yoshikawa *et al.* (2020) suggested that it is vital to adopt cultural competency in physiotherapy and provide physiotherapists with opportunities to improve their sociocultural awareness, with the possibility of practising in culturally diverse societies.

Mescouto *et al.* (2020) conducted a recent critical review of 66 studies in bio-psychosocial model of lower back pain care. They found that wider features of care, such as culture and power dynamics, received little attention. The results of the review suggested that multiple important aspects, such as interpersonal or institutional power relations, cultural considerations, as well as ethical and social factors of health, are not being integrated into physiotherapy research and practice when taking the biopsychosocial approach while working with people who have chronic pain and illnesses (Holdar, Wallin and Heiwe, 2013; Mescouto *et al.*, 2020; Brandt, 2021). Therefore, it can be suggested that the current clinical reasoning models may be inadequate when it comes to addressing patients' cultural factors. Additionally, given the complexities and sensitivities of treating women with UI, a reworking of the existing model may be necessary.

Moreover, most of the research on physiotherapy clinical reasoning has been conducted in Western democratic countries, such as Australia (Edwards *et al.*, 2004b; Higgs, 2008; Slade *et al.*, 2020), the UK (Jones *et al.*, 2014; Langridge, Roberts and Pope, 2015; Thackray and Roberts, 2017) and the USA (Payton, 1985). Relatively little is known about clinical reasoning in other cultures or how culture may shape clinical reasoning. It cannot be assumed that clinical reasoning in Western countries will be equivalent to that used in other cultures. This premise is explored in this study by including participants from two different countries, i.e., the UK and KSA.

One study that partly addresses the issue of culture within the treatment of UI is a systematic review of qualitative research by Siddiqui et al. (2014) that aimed to understand women's perceptions of UI in different racial and ethnic populations. The 23 studies included in the review were undertaken with women across different racial and ethnic groups such as mixed populations (white, black, Hispanic, Asian) in Western countries, Women of Asian and Southeast Asian backgrounds (Bangladeshi, Pakistani, Indian) and Chinese women backgrounds in addition to studies of Arab women included Syrian immigrants in Sweden and Moroccan/Turkish immigrants in the Netherlands. The included studies had varying levels of quality, with different methodologies, including mixed methods and qualitative studies. Siddiqui et al. (2014) found two main themes, UI management and UI experience, and were similar across racial and ethnic groups. In the studies included in the review, women from different races had difficulty communicating with health practitioners regarding UI and preferred to discuss UI with other women. There were some apparent variations in UI experiences within the diverse group, nonwhite women stated self-blame and perceived UI as a negative consequence of previous experiences such as giving birth and intercourse, while Muslim women had additional disruption from UI because of the need to retain purity during prayers. In addition, Latina women showed more secrecy around UI issues, even among close family members. Given these cultural differences in patient experience, it is likely that differences also exist in clinical reasoning.

Significant differences likely prevail between the UK and KSA regarding dominant models of medicine, patient preferences, physiotherapy education and culture. Although Saudi women play an essential role in society and can work in many sectors and fields, they are not necessarily able to hold the same positions as men. They are limited regarding certain decisions that require authorisation from a male relative; for instance, some women are not allowed, due to certain cultural beliefs, to travel abroad without the permission of a male relative. This may influence the health and well-being of Saudi women. For example, Pirotta (1994) claimed that assessing a female Saudi patient, even when a female staff member implements the assessment, can sometimes be problematic. This is because some female patients are hesitant to expose certain parts of the body, given the strictly held religious and cultural beliefs. In addition, Muslims in KSA pray five times a day and must perform a ritual ablution before prayer. Throughout the prayer, they must maintain that state of purity, which might be challenging for individuals suffering from UI. Muslims also fast during the holy month (Ramadan), fasting for ~15 hours (dawn to sunset) per day

for 30 consecutive days. Fasting can be very effective in weight loss, controlling diabetes and decreasing the risk of heart disease (Wei *et al.*, 2017; Malik *et al.*, 2021). However, fasting may affect the body's hydration level and voiding time and could change the routine of medication times. A review that assessed the risk of fasting on chronic kidney disease suggested that clinicians should perform a thorough risk assessment for patients within a month before Ramadan, as they may require medication changes and a plan for regular monitoring of renal function to fast safely (Malik *et al.*, 2021).

In recommendations established by the International Diabetes Federation and the Diabetes and Ramadan International Alliance, it is suggested that patients in the high-risk and very high-risk categories should be encouraged to explore alternative options to fasting, while those in the low-moderate category may be able to fast safely with guidance from their clinician (Hassanein et al., 2021). Physiotherapists, therefore, may need to consider providing patients with alternative options to bladder training and lifestyle modifications during Ramadan. During fasting, Muslims are avoiding anything that breaks their fast, which includes having sexual intercourse during the day. As a result, some women attending physiotherapy sessions may not accept digital palpation. There is a difference of opinion among the scholars of Islam regarding this issue, some physiotherapists consulted different Mosques' Imams about the effect of digital palpation on a fasting woman, and they reported that digital palpation would not break women fast because it is done for a medical reason (Malik et al., 2021). Physiotherapists managing patients from the ethnic minority may need to be aware of their patient's cultural needs, which is part of cultural competency; however, other physiotherapists might try to classify ethnic minority patients, potentially leading to stereotyping. As a result, cultural competency may be inadequate to manage patients from ethnic minorities.

2.15.3.1 Cultural humility'

Cultural humility', introduced by Tervalon and Murray-Garcia (1998), gives a different and deeper perspective on cultural competence through a continuous reflection of one's self and acknowledgement of the power imbalances between patients and healthcare providers and the challenges of institutional-level barriers. Also, it considers the fluidity of culture and challenges to individuals and institutions to address health inequalities (Fisher-Borne, Cain and Martin, 2015). It is not about a particular ethnicity, race, behaviour, or belief but about integrating cultural perspectives into management decisions. It requires physiotherapists to be open about patients' identities, giving it more weight than their own experience.

Overall, the two approaches – cultural competency and cultural humility – differ in their cultural perception, organisational assumptions, elements, interested parties and limitations, presented in Table 2-2

Table 2-2: shows the differences between cultural competency and cultural humility adapted from Tervalon and Murray-Garcia (1998) and Fisher-Borne et al. (2015

Cultural Competency		Cultural Humility	
Perception	 Recognises the layers of cultural identity. Challenges stereotypes. May encourage systematic disparities. 	 Recognises the layers of cultural identity. Acknowledges that working with different cultures is a lifelong and continuous process. Emphasises understanding ourselves and others as well. 	
Organisation assumptions	Develops values, knowledge, awareness and skills to work within diverse cultures.	 Develops self-awareness and attitudes toward diverse patients. Includes understanding of ourselves, colleagues and communities to understand our patients. Involves humility and acknowledgement of power imbalance between patients and health care providers, in addition to the relationships with society. 	
Elements	Cultural knowledge.Skills and behaviours.	 Questions power imbalances. Institutional responsibility. Integrated within management process. Ongoing self-reflection and critique. 	
Interested parties	Health practitioners	Health practitioners.Patients and community.Organisation.	

Cha	nter	2
Chia	μιει	2

	•	Regarded as a checklist		
		approach.		
	•	Focus on knowledge, leads to	•	Lack of pragmatic data.
Limitations		stereotyping the 'other' and	•	Limited conceptual framework.
		does not encourage ongoing		
		learning.		

These cultural factors may influence the clinical reasoning and decision making of physiotherapists. Consequently, the impact of culture on clinical reasoning for physiotherapists will be explored in more detail. This concerns Saudi physiotherapists and the clinical reasoning of UK physiotherapists when working with ethnic minority patients with urinary incontinence.

2.15.3.2 Studies in culture as a factor that influences health practitioners clinical reasoning and decision making

Managing ethical problems may vary based on different beliefs and religions. For instance, in Muslim majority countries, such as KSA, ethical issues are often solved within the family and not via higher authorities (such as social services or the police), though a generalisation cannot be made among all Muslims (Gill and Harrison, 2019; Rollston *et al.*, 2020). Delany, Edwards and Fryer (2019) explored how physiotherapists perceive, interpret (i.e., clinical reasoning) and respond to the ethical dimensions of physiotherapy practice in different settings. They found ethical challenges in the working environment, such as health care policies. In addition, they found balancing diverse needs and expectations to be ethically challenging. This included differences in language and cultural views regarding health (Delany, Edwards and Fryer, 2019). In a study by Sawrikar and Katz (2018) a number of barriers were found to reporting child sexual abuse among ethnic minorities in the UK, with the most significant barriers including the need to protect the family honour and name, as well as fears regarding being stigmatised by the community; this was more common among people from collectivist cultures, such as KSA.

Mobeireek *et al.* (1996) investigated the health system in KSA and commented that in traditionally conservative societies with strong traditional cultural identities, medical professionals are often considered to be figures of authority who must be ultimately

respected. This can lead to a power shift from the patients to the health practitioner, which can undermine the patients' autonomy. Therefore, be surmised that clinical reasoning in KSA may be influenced by this and might be mostly clinician-centred in nature (Simmens, 2009). Furthermore, Alshehri *et al.* (2018) undertook a study investigating physicians' attitudes and opinions towards physiotherapists' services in KSA. They identified that the physicians held superior attitudes towards physiotherapists with ambiguity about physiotherapist's identity. For instance, some of them believed that physiotherapists in KSA did not create good awareness about what they could offer patients, while physicians admitted that they did not have any adequate knowledge about physiotherapist services. The physicians reported that lack of knowledge and skills were a common feature within KSA physiotherapists, and there was limited cooperation between both professions.

AlHaqwi *et al.* (2016) is one of very few studies exploring decision making in KSA. In a study conducted within family medicine to assess patients' preferences regarding either shared decision making, paternalism or informed decisions in KSA, AlHaqwi *et al.* (2016) found that 57% of patients preferred shared decision making with the treating physicians, 28% preferred the paternalistic decision making, and 14% of patients preferred the informed approach. Most of the participants who preferred a paternalistic approach were over the age of 50, were female and suffered chronic conditions. Most of the participants who preferred shared decision making who preferred shared decision making were highly educated and male. However, this difference might not be representative of all male KSA patients as this study took place in one of the university hospitals where most patients were either working there or relatives of the employee. In addition, 60% of the participants who took part in the study were male, which might bias the result of shared decision making towards male patients only.

In addition, Lewis *et al.* (2021) reported that self-management and patient autonomy in decision making may not be culturally accepted by the patient. Self-management is an essential component of patient-centred care and part of physiotherapy management of patients with UI. However, qualitative research identified that clinicians currently address self-management in people with chronic pain conditions, such as low back pain, is insufficient, because self-management strategies were not addressed (Hutting *et al.*, 2020). There are challenges in implementing patient-centred care in the world's middle- and low-income countries. This study will be conducted in Saudi Arabia, and although it has a high income, it is still recognised as a developing country. In a study by Oleribe *et al.* (2019); Alluhidan *et al.* (2020) Alsadaan *et al.* (2021) on the challenges facing health care professionals in the implementation of patient-centred care in developing countries, they

found that there are other challenges, including inadequate human resources, budget allocation to health, unique Arabic culture and poor leadership and management. The awareness of shared decision making is limited in low- and middle-class income countries, and there are multiple cultural and operational barriers to shared decision making happening (Sam *et al.*, 2020). Another study using a self-completion questionnaire by Mobeireek *et al.* (2008) identified that experienced Saudi physicians favoured individual patient autonomy in decision making, while patients and novice physicians preferred family involvement in the decision. This contrasts with practice in Western countries, where discussing serious medical issues with family rather than with patients alone and with their overt consent may be considered against patient autonomy (Bone *et al.*, 1990).

In terms of how clinicians are educated in KSA, Elzubier (2002) reported a lack of communication and skills training in many undergraduate and postgraduate medical curricula in KSA. This led Zaini et al. (2011) to establish a new framework for the curriculum of undergraduate medical students, which placed more emphasis on patientcentred care and physician-patient communication. Furthermore, Alshehri (2017) reported that there is a lack of clinical reasoning training within physiotherapy curricula in KSA universities and suggested that clinical reasoning might be introduced within clinical practice modules when students are required to see patients at the hospital. The learning system in KSA is more didactic in nature, which may limit physiotherapists' ability to learn and solve problems independently, in addition to limiting critical thinking and reflection skills in practice. Similarly, Alshehri (2017) suggested the significance of adding clinical reasoning into the curriculum to improve physiotherapists' clinical reasoning skills and increase their understanding of biopsychosocial reasoning. In addition, a study by Fielden (2012) reported similar findings, i.e., that newly graduated Saudi nurses have the inadequate critical thinking, reflection abilities and poor communication skills, with a tendency to focus on clinical understanding without considering patients' factors or characteristics.

These findings suggest that there is room for improvement within the KSA healthcare education system, especially with regard to critical thinking. The current limitations within the existing Saudi curriculum may affect clinical reasoning and decision making for physiotherapists, although there is no existing literature that examines this. Given the researcher's own experience of completing a master's degree in the UK, she was familiar with the differences in education and aware of the different cultures between the UK and KSA. This experience raised questions in the researcher's mind about how physiotherapists

make sense of the information, and the factors that influence them will depend on each of the two different cultures. Having knowledge of the existing Western literature and contemporary Western culture, as well as experience of studying in the UK, in addition to personal experience of KSA, meant the researcher was acutely aware of how culture could influence decision making. These assumptions and biases may be affected by the researcher's experiences working as a lecturer at KSA. The educational system in KSA encouraged students to memorise the subjects rather than critical thinking. The authoritarian system might drive this in KSA that does not promote critique and debate. Students sometimes found difficulty in reflecting on previous clinical training or providing honest feedback.

My experience working with Saudi students may influence the literature chosen because of the lack of critical thinking among the students I have taught. In addition, from my experience working as a physiotherapist with Saudi patients, I remember that some patients struggled to set their own goals and were happy to be informed about the management decision; they looked at me as an experienced clinician who should decide on their behalf, that might impact on the literature chosen because of my assumption that the KSA health system is clinician-centred rather than patient-centred care. Being aware of my experiences, background and beliefs, I tried to be open to all the literature. I found that male patients were delighted to share the decision with the health professionals in one of the university hospitals in KSA. In contrast, others prefer to be informed about the management decision, this findings supported my assumptions that some patients chose to be inform about the best decision related to their health (AlHaqwi *et al.*, 2016). It was also interesting that elderly and low-educated patients preferred informed decision making in different countries and not only in KSA. That put me at ease in trying to avoid biases in my literature review and findings.

2.16 Summary and Study rationale

A literature search was undertaken using various search terms. This included physiotherapy, pelvic health physiotherapist, women's health physiotherapists, urinary incontinence, clinical reasoning, and decision making. The literature search in 2016 revealed no research into clinical reasoning and decision making by women's health physiotherapists or pelvic health physiotherapists, with most of the existing clinical reasoning and decision making literature was undertaken in the fields of musculoskeletal, cardiorespiratory, acute care and neurological physiotherapy (Case, Harrison and Roskell,

2000; Edwards *et al.*, 2004b; Smith, Higgs and Ellis, 2008; McGinnis *et al.*, 2009a; Haas *et al.*, 2012). The review of the existing literature identified biomedical, H-D and pattern recognition models of clinical reasoning and a range of factors that might influence decision making; however, it did not cover patients' sociocultural factors. Moreover, there are very few studies that address socio-cultural factors among ethnic minorities or examine organisational factors or explore how the vulnerability of patients with UI will influence physiotherapists' clinical reasoning and decision making. It is essential to consider the sensitive nature of UI and the fact that women from ethnic minorities often seek help only when they have severe symptoms.

Moreover, due to the substantial impact of UI on women's lives, it is reasonable to assume that the lifestyles that women lead will affect how UI is experienced. Consequently, it is likely that women in different countries and cultures experience UI differently, and thus there may be differences in pelvic health physiotherapist's clinical reasoning and decision making. However, most of the existing guidelines have been conducted in Western societies, with very few studies exploring the influence of culture and social factors in physiotherapists' clinical reasoning and decision making. Therefore, this thesis addresses an important gap in the literature, i.e., understanding what clinical reasoning models and processes are used by experienced physiotherapists in two countries with very differing cultures and identifying what factors shape their reasoning and decision making.

2.17 Research question

What clinical reasoning and decision making processes are used by pelvic health physiotherapists while assessing and treating women with UI in the UK and KSA ,and what factors influence their clinical reasoning and decision making?

2.18 Study Aim

To explore pelvic health physiotherapists' clinical reasoning and decision making processes while assessing and treating patients with UI in two countries with differing cultures (UK and KSA).

2.19 Objectives

- To identify and understand the similarities and differences in clinical reasoning and decision making processes between pelvic health physiotherapists in the UK and KSA.
- To identify the clinical reasoning models being used during decision making processes in the UK and KSA.
- To explore the factors influencing clinical reasoning and decision making processes in the management of UI.
- To contribute towards the understanding of future development of theoretical frameworks of clinical reasoning specific to pelvic health physiotherapists in the UK and KSA.

Chapter 3 Methodology

The previous chapters outlined the decision making and clinical reasoning processes used by physiotherapists and the study rationale. This chapter presents the research methodology, including the research design, data collection used in this study, as well as data analysis. The methods used to ensure the rigour and credibility of the study, as well as the researcher's reflexivity, are also discussed.

3.1 Research Paradigm

This study aimed to explore physiotherapists' decision making and clinical reasoning whilst assessing and treating patients with UI in two countries, the UK and the KSA. The ontological position or perspective of the researcher in the social world was to understand people's experiences, thoughts, views, feelings, memories, interpretations of actions and cultural practices (Mason, 2017). Quantitative research methods, such as surveys, randomised controlled trials and other methods commonly used within the positivist paradigm, are more frequently focused on identifying cause and effect relationships, proving hypothetical assumptions, and producing cumulative generalisations based on a data set. This approach is less useful for listening to and collecting participants' opinions and views on the research topic (Silverman, 2015), which is part of the aim of this research. There is a possibility that these aspects are interrelated and interconnected in the social worlds. Hence, an interpretivist paradigm using qualitative research was selected as the most appropriate method to help achieve the research aims in this study. It is a phenomenon linked to the early writings of Kant (1908), who published the Critique of Pure Reason in 1781. He suggested that knowledge of the world is based on understanding, which is a result of reflecting on what occurs, not just from having had certain experiences. Hence, the epistemological position that this research expresses or implements is interpretivist, as it views information as being created through a direct relationship and interaction with the participants (Charmaz, 2013). It places emphasis and value on human interpretation of the social world and the importance of both the participants' and the researcher's understanding and interpretation of the topic being studied (Ritchie et al., 2013). In the interpretivist tradition, 'researchers see, hear and

construct meaning depending upon their prior interpretive frames, participant biographies and interests, as well as the research context, concrete field experience and nature of data' (Denzin and Lincoln, 2008 p.206).

The conceptualisation of the research process in the interpretivist tradition begins with what the researcher brings to the inquiry (Ritchie *et al.*, 2013). Often, researchers bring their perspectives and experiences such as personal history, views of themselves and others, as well as ethical and political issues. The researcher also brings to the inquiry certain philosophical assumptions. These assumptions are in turn often applied in research using paradigms and theories. Paradigms are the researcher's beliefs and views that guide action (Creswell and Poth, 2017). For instance, the researcher's views, opinions and beliefs cannot be isolated during the interviews and focus groups; hence, in the current study, the researcher's epistemological position is interpretivist. This influences the way the researcher asks questions and therefore the participants' response. In this study, the researcher is a Muslim woman wearing a head scarf who encourages participants to talk about the effect of Muslim culture and beliefs on UI. The researcher aims to take a reflexive stance, but cannot avoid bias through their beliefs, which can guide their actions as a researcher (Creswell and Poth, 2017).

Nevertheless, the interpretivist approach was considered the most appropriate and suitable method as it allowed the gathering of rich and insightful data; it also enabled the exploration and understanding of the social world of physiotherapists, with a focus on their experiences and interpretations of their decision making and clinical reasoning processes, as suggested by Holloway and Galvin (2016). This paradigm also enabled the exploration of the connections between the social and cultural aspects of physiotherapists' decision making and clinical reasoning and the context in which certain actions occur.

This interpretivist philosophical assumption underpinned the research processes from research questions, data collection methods as well as the relationship between the participants and the researcher (Denzin and Lincoln, 2011).

3.2 Methodology and Research Design

The methodology of any research should relate to its research questions and objectives (Denzin and Lincoln, 2011). To understand the process of clinical reasoning and explore any differences in decision making and clinical reasoning between UK and KSA physiotherapists, qualitative research was used to provide the necessary in-depth exploratory tools for obtaining a clear picture of clinical reasoning processes (Creswell and Creswell, 2017). Another objective of this study was to identify the clinical reasoning models used by physiotherapists and to contribute towards the understanding of the future development of theoretical frameworks of clinical reasoning specific to physiotherapists in the UK and the KSA.

The framework approach is a qualitative method developed by two qualitative researchers, Jane Ritchie and Liz Spencer, in 1994. It is a deductive form of analysis that is widely used in healthcare research. Framework approach tends to emphasise greater sample sizes and saturation of themes, and it is epistemologically grounded in the interpretivist tradition. Framework approach is suitable to applied or policy research in which the objectives of the investigation are typically set in advance rather than created from a reflexive research process. It starts deductively from the aim and objectives that have been set for the study (Pope, Van Royen and Baker, 2002). This approach was deemed suitable for the present study as it helps identify if the existing clinical reasoning theoretical models are applicable to physiotherapists in both the UK and KSA and highlights the similarities and differences between clinical reasoning and decision making in the UK and the KSA.

3.3 Methods

Semi-structured interviews and focus groups (FGs), which are particularly appropriate for eliciting people's thinking and rationales, were used to collect data in this study. Hennink (2007) suggested combining multiple qualitative methods in a study design, as each method brings a particular insight to the research problem.

3.4 Other Methods Considered

There are several other methods that can be used to collect data within qualitative research. One of the most common one used in qualitative studies is observation methods. It was initially considered for the current study as it would have enabled the observation of physiotherapists' clinical reasoning, as suggested by Higgs (2008). However, after careful consideration it was deemed unsuitable. This was because physiotherapists often use digital palpation to assess pelvic floor muscle power - a part of patients' biofeedback to show them if they are contracting the correct muscles or not. Patients often feel uncomfortable during this process and might not accept a researcher observing the procedure. Therefore, it might have been difficult to observe physiotherapists whilst assessing and treating patients with UI. This was the reason why observation as a data collection method was not selected for use in this study. This research was interested in gaining insights into how people think and not necessarily on what they do, so it was considered most beneficial to access this information through discussions with physiotherapists as individuals during interviews and during group discussions in the focus groups. The aim of the interviews conducted in this study was to discuss physiotherapists' clinical reasoning process, whilst focus groups were conducted to discuss the factors that influence physiotherapists' clinical reasoning and decision making during the management of UI.

3.5 Interview method

Semi-structured interviews were selected as the most appropriate method for this study. They enable the researcher to explore the participants' perspectives, knowledge, interpretations and experiences, which are meaningful properties of the social reality and can be used to clarify the reasoning process behind their decisions during the evaluation and treatment of UI (Myers and Newman, 2007; Rubin and Rubin, 2011).

3.5.1 What are semi-structured interviews?

Qualitative interviews can be defined as a conversation that has a 'certain purpose of gaining a description of the life world of the interviewee in order to interpret the meaning

of the described phenomena' (Brinkmann and Kvale, 2008 p. 3). Interviews come in different forms, ranging from structured to semi-structured interviews; most qualitative interviews, however, are semi-structured (Mason, 2002). A semi-structured interview is a qualitative method of enquiry that attempts to understand the themes of the lived daily conversations from the participants' own perspectives by using open-ended questions (Mason, 2002). The researcher provides some structure based on the research topic and interview guide, but enables the participants to be spontaneous and provide narrative description (Brinkmann, 2014).

3.5.2 Key features of and rationale for the use of interviews

The interviewer and the interviewee are not only discussing a topic but are also reacting to one another's appearance, identity and personality (Hennink, Hutter and Bailey, 2010). The specific characteristics of in-depth semi-structured interviews involve using prompts to enable in-depth discussions, establishing rapport between the interviewer and the interviewee, utilising a pre-determined set of open-ended questions and motivating the interviewee to elicit his/her story by probing for more depth and details (Kvale, 2008; Silverman, 2015).

A particular advantage of interviews in the present study was that they provided an environment that facilitated the participants to discuss different decision making and clinical reasoning processes within their unique culture and context (Corbin and Strauss, 2008). Higgs and Jones (2008) suggested that studies in clinical reasoning research have started to use the interpretive paradigm that seeks to interpret human phenomena. Interpretive approaches use interviews and observations to record physiotherapists' perspectives and descriptions of their clinical reasoning. Many authors have used these methods of data collection in physiotherapy (Edwards *et al.*, 2004a; May *et al.*, 2010a; Ajjawi and Higgs, 2012). One advantage of these approaches is that they increase the likelihood of the research revealing physiotherapists' reasoning as used in practice. Also, these approaches help demonstrate that clinical reasoning is a complex, multidimensional, integrated, task- and context-dependent process.

Creswell and Poth (2017) highlighted the importance of good preparation for the interview and the development of effective research questions to ensure a productive and beneficial interview. McNamara (2009) applied several principles to the preparation phase of interviewing, which included the following elements: choosing a location where interruptions and distractions are minimal; explaining the aims and objectives of the interview; addressing terms of confidentiality; indicating the expected duration of the interview; providing the researcher's contact details; and affording the participants the opportunity to ask questions before the commencement of the interview. McNamara (2009) also suggested using open-ended questions that are as neutral as possible and that avoid influencing the answers.

3.5.3 Limitations of semi-structured interviews

There are some limitations to conducting semi-structured interviews, such as the possibly limited feedback of the respondents, thus requiring a skilful interviewer who can establish rapport and listen to and interact with the participants. There may also be a need to change the order of the interview questions after listening to the participant's discourse, and this requires a high level of skill and expertise, as well as high familiarity with the content (Hennink, Hutter and Bailey, 2010; Creswell and Poth, 2017). In addition, the interviewee might assume that the interviewer knows the answer because of their experience in pelvic health physiotherapy and this might decrease the depth of the discussion. Further, the participant might be worried about judgement from the researcher so they may not provide detailed answers; this was more apparent with the KSA participants. Nevertheless, interviews have the potential to provide the researcher with a rich source of insightful data into the participants' views, opinions, and experiences.

3.6 Focus Group Method

Focus groups were also selected as a data collection method in this research to complement the insights gained from the semi-structured interviews with further nuanced information.

3.6.1 What are focus groups?

Wilkinson and Silverman (2004 p 177) defined focus groups as 'a way of collecting qualitative data, which essentially involves engaging a small number of people in an informal group discussion "focussed" around a particular topic or set of issues. Focus group research differs from other qualitative methods in terms of purpose, composition, and the process of data collection. The purpose of focus groups is not to collect the personal stories of participants but to obtain a range of these participants' opinions and/or experiences. Discussions take place regarding people's collective experiences as participants in the group rather than their individual experiences. Furthermore, focus group participants react to one another, and these reactions can be prompted by the opinions of others, which may not occur in an interview situation.

3.6.2 Key features of and rationale for the use of focus groups

An advantage of using focus groups is that they tend to be more natural than interviews, and they yield a large volume of data from many participants within a short period (Wilkinson and Silverman, 2004). One of the unique features of focus groups is their group dynamics; hence, the kind and range of data generated through the social communication and interaction of group members can often be deeper and richer than those from data obtained through face-to-face interviews (Thomas *et al.*, 1995). Focus groups can provide a range of information regarding an individual's ideas and feelings, and participants may highlight issues between and among themselves with limited researcher influence (Hennink, 2007); they are therefore able to react to the comments of others in the group. Focus group discussions may lead to reflection, refinement or justification of the issues raised, which can, in turn, provide deeper insights into the context studied (Hennink, 2007). Similar to the results of semi-structured interviews, those from focus groups can be illustrated in simple ways using lay terminology, reinforced by the participants' quotations (Thomas *et al.*, 1995).

3.6.3 Group homogeneity

The distinctiveness of focus groups is their ability to create data based on the cooperation and interaction of group members. Therefore, group members need to feel comfortable with one another and participate in discussions (Krueger and Casey, 2000). Both homogenous and heterogeneous groups have advantages and disadvantages. However, Krueger and Casey (2000) believed that a homogenous group could generate rich data and encourage participants to be fully engaged with one another. As a result, the authors recommended spending time and effort choosing a group member. The authors indicated that participants might have similar features, such as gender, age-range and ethnic or social backgrounds. Other authors stated that if group members do not know one another, this situation may lead to an honest and natural discussion of opinions and stimulate a broader range of reactions. It may also prevent some behaviours regarding previous relationships and forms of control and leadership in the group (Thomas et al., 1995). The physical environment of all the focus groups was quite similar, there was a table in the middle of the room, and the participants were sitting around the table; the facilitator was seated next to one of the participants but at the head of the table. The audio recorder was in the middle of the table and visible to the participants. The physiotherapists in the current study may likely have known some of the other physiotherapists attending the focus group as they could have been their colleagues or managers. In this case, the role of the group facilitator is significant, and this will be discussed in the following section (Krueger, 2014).

3.6.4 The role of the focus group facilitator and the observer

A facilitator needs to create a relaxed environment in the group, manage the current relationship with and between participants, and encourage them to engage and discuss their feelings, opinions and thoughts about the topic explored. In this study, an observer was present in the focus group discussion, and her role included taking down notes; observing non-verbal communication; identifying the influence of group dynamics; documenting the general content of discussions; and writing reminders on the statements made by each individual in the group, thus adding to the transcripts and allowing for a more comprehensive analysis of the data (Kitzinger, 1994). The observer sat outside the discussion circle to address any disturbances or late arrivals. The observer also reminded

the facilitator if any critical issues or topics had been overlooked during the discussion. The observer was a colleague of the researcher and was known by some participants, especially in KSA.

The role of the facilitator might be similar to that of an interviewer in a semi-structured interview in terms of establishing rapport and remaining focused on the research topics (Hennink, Hutter and Bailey, 2010). However, a focus group facilitator's role might be more challenging and require additional skills in terms of managing the group members. A facilitator might use a directive approach at the beginning of a focus group discussion to provide focus and direction to the research topic. Then, during the central part of the discussion, he/she can apply a less directive style so that group interaction is allowed, and the discussion flows more naturally, allowing spontaneous views to emerge.

3.6.5 Limitation of focus groups

A limitation of using focus groups is the number of non-attendees; hence, it has been recommended to over-recruit by 10% to 25%, based on the research topic and the participants' background (Rabiee, 2004). Agreeing on the days/times that are convenient for the participants via email or phone well in advance of the planned focus group is essential to increase participation, and reminders can be sent a few days before the focus group discussion. Another limitation of focus groups is that the facilitator must have appropriate skills to deliver the questions, facilitate the discussion and manage any discomfort or disagreement within the group (Patton, 2002). In focus groups, it is also natural that some participants might dominate the discussion, and as the venue of focus groups involves a less-controlled environment, hierarchies might develop; the presence of a skilled facilitator to manage the focus group discussion is therefore important. Focus groups are also less confidential, so measures must be in place to ensure confidentiality

(see Section 3.13.2). Despite these limitations, focus groups were used in the present study, as they have the potential to generate a large amount of data from many participants.

3.7 Procedure

The following sections outline in detail the procedures used to collect and analyse the data in this study.

3.7.1 Sampling

Purposeful and snowball sampling were used to gain variation in the sample. Purposeful sampling involves recruiting physiotherapists with a range of characteristics such as years of qualification, years as a pelvic health physiotherapist and postgraduate qualifications in pelvic rehabilitation. Snowball or chain sampling was used in combination with purposeful sampling, as it helped identify cases of interest from physiotherapists; they may know other professionals in the field who were aware of interesting and relevant cases (Creswell and Poth, 2017). Kuzel (1992) stated that the validity and significance of the views generated from qualitative enquiry are more likely to be related to the informational richness of the cases selected, and to the observations or analytical abilities of the research, than to the sample size. Pelvic health physiotherapy is considered a rare specialisation, and few therapists work in this field. Therefore, it was difficult to recruit many participants for the study. There were over 30 registered members of the Saudi Physiotherapy Women's Health Association (SPTWHA) at the time of recruitment, and their contact details were available by contacting the SPTWHA secretary.

3.7.2 The participants

3.7.3 In the semi-structured interviews

Seventeen pelvic health physiotherapists were selected to participate in the semistructured interviews from the KSA, but two participants were excluded due to lack of clinical experience in treating patients with UI. In addition, thirteen participants from the UK were recruited for the semi-structured interviews. Four more pelvic health physiotherapists in the UK contacted the researcher in order to take part. However, in the end, they could not participate because they lived far away. When the researcher suggested a Skype interview, they did not reply, which could have meant that they did not wish to participate in a video conference or were unaware of Skype programme. Overall, 28 participants were recruited from both the UK and the KSA for the interviews.

3.7.4 In the focus groups

In KSA, eight participants agreed to take part, but on the day of the focus group two participants dropped out. As a result, six participants took part in the KSA focus group whereas in both UK focus groups, eight participants took part.

3.7.4.1 Inclusion criteria

This study sought to obtain the views of physiotherapists with enough women's health experience in their clinical reasoning. They were required to meet one or more of the following criteria:

- Physiotherapists were included if they had been OR are currently registered with the Health and Care Professional Council (HCPC) in the UK or with the Saudi Commission for Health Specialities (SCHS) in the KSA.
- Qualified physiotherapists who had worked with patients with urinary incontinence for a
 minimum of one year and had treated a minimum of ten patients with UI in the past year;
 the reason for this was that physiotherapists might also have orthopaedic and neurological
 caseloads due to the low number of patients with women's health issues compared with
 those who have other diagnoses.
- Qualified physiotherapists with more than two years' women's health experience in either the UK or the KSA but who did not necessarily treat ten patients with UI per year at the time of the study.
- Academic physiotherapists who did not necessarily treat patients with UI in the past year but had undertaken research and/or taught students about UI.

3.7.4.2 Exclusion criteria

- Physiotherapists were excluded if they were unable to speak English. The interviews and focus groups were conducted in English, in both the UK and the KSA. Saudi physiotherapists study in English at both the undergraduate and postgraduate levels, and they document and communicate in English during their daily clinical practice. The English proficiency levels of the physiotherapists in this study were therefore enough for them to communicate in English during the interviews and focus groups.
- They were also excluded if they were unable to provide informed consent because of mental health issues.
- Those who never worked as a pelvic health physiotherapist in the UK or the KSA, as appropriate where also excluded.

3.8 Settings

The semi-structured interviews were conducted in both the UK and KSA and were in a convenient location to both the participants and the researcher. The researcher travelled to meet the participants in their workplace, as agreed. In KSA, most of the participants live in the central part of KSA. This means the researcher did not have to travel long distances. However, a few participants were from the Eastern and Western provinces of KSA; whilst the researcher was able to travel to the some of these parts, the remaining two participants were invited to a videoconference instead of a face-to-face interview, which they declined. This could have been due to a cultural issue. As a result, the researcher decided to communicate with them through a phone call only, which was accepted. In the UK, most of the participants lived-in South-East England and the researcher decided to travel to the participants who lived within a 130 miles radius. For those who lived further than that, the researcher suggested a video conference using Skype instead of a face-to-face interview. Thus, three participants from the UK attended video conferences and two from KSA attended phone call interviews.

Selecting settings from a wide range of regions in the UK and KSA allowed for a nuanced understanding of the clinical reasoning models and factors influencing decision making, thus rendering findings transferable to a wider range of locations. In KSA, focus groups took place in a private hired conference room. Two focus groups took place in the UK, one of them in a hotel conference room and the other in a meeting room in one of the London hospitals. The associated costs for the room hires were funded by the Saudi Cultural Bureau. The researcher travelled to the participants' location, except in cases where the participants preferred to travel to the researcher's location for the interviews and/or focus groups. In these instances, a reimbursement for their expenses was made.

All data collection was undertaken either in the participants' working place or in a meeting room in either the UK or the KSA, given that these were the most convenient locations for the participants and the researcher. Care was exercised in following lone-worker safety guidelines before any data collection sessions were conducted, which involved giving a sealed envelope containing both the locations and times of the data collection sessions, as well as the study participants' details, to a responsible third party before conducting any data collection. The researcher also carried a charged mobile phone, took the appropriate public transportation, and prepared a list of emergency telephone numbers. However, as each participant was a health care professional, it was anticipated that there would be no problems encountered.

3.9 Ethical approval

Ethical approval (Appendix C) was sought from the University of Southampton (Ethics and Research Governance), and full approval was obtained in July 2018. As the participants were recruited from different professional institutions (and not from the NHS), the Health Research Authority (HRA) confirmed that HRA approval was not required for this study. Additional ethical approval from King Saud University in the KSA (Institutional Review Board / College of Medicine) was sought, and full approval was obtained in October 2018 (Appendix D)

Ethical considerations were factored into the design and implementation of this study. Matters of confidentiality and anonymity were addressed throughout the research. The associated costs were funded by the Saudi Cultural Bureau (from the data collection fund). The nature of the focus group discussions meant that whilst the researcher could ensure the participants' confidentiality, the researcher was unable to guarantee that participants

would do the same. The consent form, the participants' information sheet (Appendix K, Appendix L, Appendix I) and the ground rules discussed before the focus groups, highlighted this issue to the participants and asked them to maintain the confidentiality of others. Confidentiality and anonymity in the study were maintained by anonymising both the participants and the physiotherapy services discussed in the transcript.

3.10 Recruitment

3.10.1 UK

The research protocol was presented at a conference organised by the UK-based Pelvic Obstetric and Gynaecological Physiotherapy (POGP), a professional network affiliated with the Chartered Society of Physiotherapy. During the conference, the head of the POGP group was asked about the possibility of cooperating in recruiting participants from their members; the head of the POGP agreed to help. The researcher then sent an invitation letter (Appendix G) to the secretary of POGP to be distributed among their members. Physiotherapists who were interested in taking take part contacted the researcher via email to show their willingness to do so. In addition, the International Uro-gynecological Association, and the International Continence Society, who agreed to help, by sending invitations to their members in exchange for a small fee. Likewise, the researcher contacted the Association for Continence Advice (ACA), which encouraged the researcher to be a member of ACA and attend one of their online forums to write about the study and distribute the poster and information sheet (Appendix H and Appendix I retrospectively) in order to raise interest in the study and recruit participants. However, only POGP members showed interest in taking part. Few UK physiotherapists registered with other associations when compared to POGP, which might be the reason for little interest shown by associations.

3.10.2 KSA

An e-mail was sent to the general secretary of the Saudi physiotherapy women's health association to ask them about the possibility of helping in recruiting participants from their members; the general secretary agreed to help.

3.10.3 Recruitment and data collection processes

Recruitment processes are described below and summarised below:

- An email (Appendix F) was sent to the secretaries of pelvic obstetric and gynaecological physiotherapy association, in addition to the Saudi physiotherapy women's health association and other organisations. They were asked to send to their members the invitation letter (Appendix G), the poster (Appendix H) and the participant information sheet that contained the contact details of the researcher (Appendix I).
- The researcher attended an online forum of the association for continence advice and uploaded the poster and information sheet.
- In order to target a wider pelvic health physiotherapist audience, the researcher used Twitter to invite physiotherapists to participate in the study. The researcher sent a tweet to the above organisations' Twitter account (Appendix J). The tweet provided the researcher's email and links to the invitation letter, and the participant information sheet.
- An additional invitation email (Appendix F) was sent to a SOTON (Southern and London) representative at pelvic obstetric and gynaecological physiotherapy after a period of approximately four to six weeks. This was done in an attempt to increase the response rate in places that were closer to the University of Southampton. The second email included the original study information and the researcher's contact details.
- Respondents from Saudi physiotherapy women's health association and pelvic obstetric and gynaecological physiotherapy association interested in participating in this study were able to contact the researcher directly via email.
- The researcher contacted eligible participants by email and discussed the study information sheet. This provided them with an opportunity to ask further questions.
- After the researcher applied the inclusion and exclusion criteria, the eligible respondents were asked to participate in the interview and/or the focus groups.

- If the respondents were not eligible to participate in the study, they were also informed, and their personal details were deleted.
- The researcher asked the eligible participants for a convenient time and location for the data collection. Following this, the semi-structured face-to-face interviews and focus group appointments were arranged.
- The researcher determined that a venue at the university, a private research laboratory or the participant's home could be used, if the physiotherapists' workplace was not appropriate.
- The researcher booked the room for the interviews and focus groups.
- An email was sent to the participants to notify them about their data collection appointment and to remind them of their right to withdraw from the study.
- To ensure safety, confidentiality and compliance with lone-worker safety guidelines, the researcher followed certain procedures as outlined previously.
- A thank-you email (Appendix Q) was sent to the participants after the data collection was completed. They were also asked if they were interested in receiving a copy of the study findings, which are likely to become available in two years.
- At the end of the interview, the researcher asked the participants if they could suggest any physiotherapists who might also be interested in taking part in the study (snowball sampling). The participants were told that if they knew of a relevant physiotherapist, they were requested to forward the invitation email (Appendix F), along with the participant information sheet and poster, to that person on behalf of the researcher. The researcher did not obtain these new contact details. Instead, if new persons wished to participate in the study, they were requested to contact the researcher directly. The researcher followed the same process as above for new contacts.
3.11 Pilot interviews and focus groups

To overcome the limitations of a researcher's lack of experience in conducting interviews and focus groups, Patton (2002) suggested conducting pilot interviews and focus groups. It is often difficult to anticipate how interviewees interpret questions in an interview or a focus group guide. Therefore, Hennink, Hutter and Bailey (2010) recommended assessing several different issues during pilot testing, such as participants' understanding of the questions, adaptation of the concepts and words to the context of the interviewee, rephrasing questions and assessing their logical organisation, as well as determining the practical length of the interview or focus group guide. Hence, the first two interviews conducted by the researcher were considered pilot interviews. This allowed the researcher to improve her experience, confidence and time management during the data collection sessions and make modifications to the interview schedule; for instance, one of the interview questions was misleading: Question "how did you confirm the diagnosis?" led pilot participants to give responses related to hypothesis testing and was reframed to "how did you come to this decision?" For more details, check Appendix Appendix O. The data collected from the pilot interviews and focus groups were included in the study in addition to the main study participants.

At the end of the planned questions, the researcher asked the pilot participants for their feedback regarding the interview and focus group questions, as well as the delivery and structure of the interview, to assess the appropriateness of the questions and identify any areas of improvement (Kvale, 2008). The importance of conducting a pilot study has been highlighted in the literature; Kvale and Brinkmann (2009) explained that it is fundamental to any research, as it assists in determining possible flaws and areas of improvement. The literature highlights the importance of skilled facilitation. It offers extensive advice on conducting focus groups, including preparing for the session, managing the discussions, encouraging quiet participants to share their thoughts and probing for participants' reactions (Millward, 1995 p.281-285; Finch and Lewis, 2003p. 180-189).

3.12 The interview and the focus group topic guide

It has been suggested that a novice researcher could use a topic guide to assist in the flow of discourse and increase confidence during interviews and focus groups (Turner III, 2010). Using the framework approach in the present study suggested that the topic guide tends to be slightly more structured than would be the norm for most qualitative methods (Pope, Van Royen and Baker, 2002). There are contradictory opinions in the literature as to how structured focus group discussions and interviews should be conducted. In the present research, the interviewer followed the interview and focus group topic guides (Appendix Appendix O and Appendix Appendix P), which consisted of general areas for discussion, prompts and a possible order in which topics might be discussed, although this was considered flexible and not rigid. The questions in the interview were initially general and open ended (e.g., what assessment did you do for your UI patient?), followed by prompts (e.g. How did you come to this decision?). During the focus group discussions, the questions were also initially broad and open ended (e.g., in your opinion/experience, what are the main differences between the factors that affect physiotherapists' decision making compared with other physiotherapy specialities?), followed by prompts (e.g., Why are there differences?) to track factors not routinely raised. During the focus groups, this allowed the facilitator to both identify the physiotherapists' obvious explanations and to explore issues in depth.

3.13 Conducting the interviews and the focus groups

Different methods might help approach the research questions from different perspectives or in greater depth, and they may enable some form of methodological triangulation (Mason, 2002). The participants chose whether they wanted to participate in both the interview and the focus groups or just one of them.

The interviews and focus groups were recorded using two digital audio voice recorders (in case one of the recorders was not working during the data collection). An external disk mic with built-in memory was used for the group interviews. These audio recordings were used to make verbatim transcriptions of the interviews and focus groups. The researcher ensured that the digital recorders were working. At the beginning of the interviews and the focus groups, the participants were greeted by the researcher and/or note taker, were given a name badge and were offered refreshments (Millward, 1995). The study was explained, and the participants were encouraged to ask questions. The participants were also reminded of their right to withdraw from the study up until the data analysis of the interviews or until seven days into the focus groups.

It was considered important to establish rapport and avoid rushing when starting the interviews and the focus groups. Rapport was established through small talk, chatting about the weather, drinking coffee or tea, or engaging in other types of conversations. As the researcher is from the KSA who is studying in the UK, it might have been deemed an outsider by the UK participants; hence, at the start of the UK interviews and focus groups, the researcher stated that she was from a different country and culture whilst also explaining that she was partially familiar with their own norms because her education in the UK. Likewise, as a physiotherapist practitioner, emphasising existing commonalities (insider–outsider considerations are discussed in detail in section 3.22 and 6.5). Therefore, an atmosphere was created in which the interviewees were encouraged to talk more about clinical reasoning and their work culture. When rapport is established, interviewees can feel more comfortable sharing their opinions or personal feelings.

The aim and objectives of the interviews and/or focus groups were explained, and ethical considerations were reiterated, including obtaining consent to participate, ensuring confidentiality for participation as well as data use and storage. The researcher confirmed that all participants were fully informed of the study and had all their questions answered. Each participant then signed a consent form, which they received a copy of an Appendix Appendix K and Appendix Appendix L. Participants also completed a demographic data form (Appendix Appendix M and Appendix Appendix N) before taking part in an interview or in the focus groups. Skype and FaceTime were considered for interviewing participants in case of difficulties travelling around the UK or the KSA (Iacono, Symonds and Brown, 2016). This was useful because three participants were living more than 130 miles away from the researcher. Iacono, Symonds and Brown (2016) Comparing face-to-face interviews and Skype interviews they stated that Skype and FaceTime could not

replace face-to-face interviews. They found that it might be difficult to establish rapport, follow ethical considerations, observe non-verbal cues such as body language, and clarify the participant's identity in a Skype interview. However, Iacono, Symonds and Brown (2016) claim that voice-over-the-internet protocol worked well as a practical substitute or complementary data collection instrument for qualitative researchers. The advantages of Skype interviews were time efficiency, maintenance of visual cues and financial affordability, which could increase the variety of participants.

3.13.1 During the interview

The interviews focussed on asking participants to describe the processes they had used when assessing and determining treatment choices for a specific patient from their recent caseload.

The participants in the UK were asked either to recall a typical case or complex case. In the KSA, experienced participants were asked to recall a complex case while participants with a few years of experience were asked to recall a typical case to allow them to be comfortable and confident during the discussion. I surmised that participants in the KSA may feel that, as the researcher, I was assessing their knowledge, so it was explained upfront that the aim of the discussion is to understand the participants thinking process and not to assess their knowledge level.

Participants were asked to recall their process of managing one of their patients with urinary incontinence and to explain their clinical reasoning processes to the researcher. The interviewer asked the participants a number of questions regarding the process of clinical reasoning that they had used during the assessment and treatment of the specific patients with urinary incontinence (the interview guide, including the questions asked, is found in Appendix Appendix O. I tried to follow the assessment and treatment order but sometimes participants answered the questions before I had the opportunity to ask. In these instances, awareness of the topic guide helped in asking other questions relevant to participants' answers and avoiding repeating the same idea.

The dialogue within the interviews allowed for depth discussion of a very specific case to facilitate understanding of the complexities of clinical reasoning processes, encouraging

complex topics to be discussed, such as addressing ethical dilemmas and discussing sensitive issues that may be difficult to raise in a focus group setting.

Open-ended questions and prompts were used to encourage the free flow of information and avoid pre-conceived answers (Corbin and Strauss, 2008). Appendix Appendix O and Appendix P presents the different open-ended questions, follow-up questions and prompts used during the interviews and focus groups; these were based on those used in previous physiotherapy clinical reasoning studies. It was also useful to apply different motivational probing techniques, such as the use of 'ah-ah', 'mmhm' or 'OK', to encourage the participants to continue talking. In addition to body language, nodding and eye contact were important. Another strategy applied to motivate the participants was reflective probes, which involved repeating or paraphrasing an interviewee's comments to ask for further clarification (Silverman, 2015).

A substantial characteristic of qualitative data collection is that it facilitates identifying the key issues emerging from the interview to refine questions and topical probes in the subsequent interviews. Using this approach, the researcher was able to make inductive inferences and thus go deeper into the topic in the subsequent interviews (Hennink, Hutter and Bailey, 2010). Field notes were also used to record the nonverbal interactions, techniques, positions or actions that the participants showed during the interview (Birks and Mills, 2011). In addition, field notes were used to write feedback after each interview related to what kind of questions needed to be added or changed and if there were any positive or negative thoughts that could help in improving the subsequent interviews. This type of field note seemed useful during the KSA data collection due to the fast nature of it, and it helped to improve the following interviews (Birks and Mills, 2011).

Each interview was transcribed after interview completion to recognise key issues and make inferences, which were used to enrich subsequent interviews (Mason, 2017). Making inferences and analysing notes after each interview helped the researcher make small changes in the interview guide (Hennink, Hutter and Bailey, 2010; Silverman, 2015).

3.13.2 During the focus groups

In this study, the goal of the focus groups was to obtain a broader range of information to address the research questions and objectives, including discussing the factors that might affect the participants' clinical decision making, as well as the problems they faced during their assessment and treatment of patients with urinary incontinence. Focus groups were conducted in both the UK and the KSA to gather data for this research.

Prior to conducting the three focus groups, participants were offered light refreshments. This was a good opportunity to talk to the participants, establish rapport and be familiar with the dominant and/or quiet participants. The researcher helped to ensure the participants felt welcome. At the beginning of each focus group, the observer was introduced, and her role was explained. A Power Point presentation was used to remind the participants that the discussion would be digitally recorded. Ground rules were established which included that everyone should have a chance to speak and give their opinions on the issue raised. Participants were asked to speak clearly, avoid talking at the same time and feel free to contradict with one another's opinions. It was important for the researcher to manage group dynamics (for example, quiet and dominant participants). The researcher made sure to talk to the quiet participants during the dinner to encourage them to feel more confident to speak later on. The quiet participants were directly asked to give their opinions, and the researcher kept making eye contact with them (Goldenkoff, 2004). While body language was used to manage the dominant participants, probing and listening was used to gain in-depth responses on issues raised to prompt the discussion. The researcher also ensure the discussion remained on the research topics, whilst monitoring the time and pace of the discussion (Hennink, Hutter and Bailey, 2010).

Goldenkoff (2004) suggested the use of tools that help the researcher and the participants to think deeply in the focus group topic. For instance, using games or writing on a board. The participants were asked to take a minute to think about the factors that influence their decision making as a pelvic health physiotherapist and write them down. Participants then used those points during the discussion; at the end, the participants were asked to handle their notes to the researcher in order to see if missed any discussing points were missed.

A range of probes was used during the focus group discussions, like the ones applied in the interview. In this group setting, the researcher was able to use different types of probing in order to obtain additional information. Examples include using explanatory probes where participants were asked to explain the reasons behind their opinions; using participants' gesture probes; and probing for diversity by asking if anyone has a different opinion. A focus group guide was used to direct the discussions (see Appendix Appendix P). The participants were invited to talk about their multidimensional knowledge base, their experiences, their understanding of their patients' problems, patients' goals, preferences and perceptions towards the assessment and treatment, and the cultural constraints or considerations they faced. This opening invitation was broad to encourage the participants to talk about the most important issues to them. The discussion was directed to include all the key parts of the topic guide, but it did not follow any specific order of questioning. A less rigid focus group topic guide enabled the discourse to flow more naturally and to focus on matters of particular significance to the group. The focus groups had low facilitator involvement which gave the researcher an opportunity to change the contents of the questions as the discussion progressed (Morgan, 1996). The degree of diversity and consensus was explored, and participants were asked to clarify details and to reflect on interesting issues raised during the discussion, where necessary (Finch and Lewis, 2003).

Each focus group typically lasted around 1 to 2 hours, depending on the complexity of the research topic, the number of questions raised and discussed, and the number of participants. At the end of the discussion, the digital recorder were switched off, the participants were thanked for their contribution and time, and asked if they had any questions. Appropriate salutations were given to show appreciation for their time spent during the focus group discussion (Morgan, 1996; Green, 2007). Following each focus group, the focus group discusse were discussed immediately with the observer and reflective field notes were written. These recorded the main topics discussed and areas of interest to delve into deeper in the following focus groups. The main researcher and the observer also noted the communication style, the language used and the emotional disposition of the participants. Reflection on the pros and cons of the facilitation

techniques and possible adjustments to them were undertaken after each focus group. The observer likewise wrote field notes, and these were added to those of the researcher.

It was clear that culture was not discussed in detail in the KSA focus groups and the 1st UK focus group. Therefore, the main researcher decided to expand on it in the follow-up groups. The definition of culture was included in the Power Point presentation at the beginning of the 2nd focus groups and participants were encouraged to give their opinion on culture as a factor that can influence their decision making. Using the participants' opinions on culture from the previous focus groups were useful to elicit the discussion and motivate the participants to talk. This pattern of reviewing the direction of the research as the data gathering and analysis continued is in line with the general approach of grounded theory (Charmaz and Belgrave, 2007).

During the focus groups in the UK, the researcher compared the factors that affect the decision making and clinical reasoning process in the UK and the KSA. The participants contributed to the data analysis during the focus groups by providing their views and opinions regarding the differences between UK and KSA clinical reasoning among physiotherapists prior to the actual data analysis. In addition to establishing the factors that influence the decision making and clinical reasoning process, this technique allowed for further exploration of the cultural impact on clinical reasoning among physiotherapists.

3.14 Different Phases of Data Collection

The first phase of data collection involved interviews conducted in the UK see Figure 3-1. Figure 3-1: Different Phases of Data Collection in the UK and the KSA



The second phase comprised interviews and focus groups that took place in the KSA following ethical approval (Appendix Appendix DError! Reference source not found.), granted by King Saud University (where Saudi physiotherapy women's health association is registered). The third phase of data collection involved focus groups that were conducted in the UK. Beginning the interviews in the UK meant the researcher was in close contact with the supervisory team and colleagues when advice or guidance was needed. This gave the researcher experience and confidence in conducting interviews. Also conducted interviews and focus groups in the KSA, which allowed the researcher to practise the skills required to conduct productive and useful focus group discussions with Saudi physiotherapists first.

The researcher was an integral part of pelvic health physiotherapy professional networks in KSA. In contrast, the researcher did not know any of the participants from the UK prior to the study. There were advantages to be an insider within KSA institutions. For example, the researcher could arrange and organise the focus groups easily because of the participants' superior understanding of the local culture and ability to interact fluently within the group. In addition to these advantages, the researcher had greater relational intimacy with the group from the KSA (Bonner and Tolhurst, 2002). However, the researcher did acknowledge the need to maintain a facilitator relationship with the participants in order to avoid leading them to specific answers and to encourage them to talk freely. This was a good opportunity to explore the cultural effects of UI on KSA patients and increase the researcher's experience before conducting the focus groups in the UK.

By concentrating on the interview and focus group processes, the researcher avoided reflecting upon the personal nature of the collected data (Kanuha, 2000). As such, the

researcher was engaged in reflexive processes, allowing a critical self-evaluation of the researcher's positionality (Bonner and Tolhurst, 2002).

3.15 NVivo

There are many tools used to assist the data analysis process of qualitative data and could be used for data management and provide easier access to codes and transcripts rather than pen and paper. These tools can vary from Microsoft word, Excel spread sheets or involve other qualitative data analysis software, such as NVivo or ATLAS. Each one of the tools has advantages and disadvantages, for instance, the limitations to using Microsoft word and Excel are the spreadsheets can be unwieldy when working with 48 participants and unwieldy when trying to review the coding of the entire dataset. In a study by Woods *et al.* (2016) ATLAS and NVivo were used to conduct a pilot analysis of previously coded articles to determine which program better supported the qualitative analysis approach and overcome the limitations of Microsoft word and Excel. Woods and colleagues found that coding with NVivo provided different advantages such as indexing of data categories and charting of the data, which is part of framework analysis. In addition, the University of Southampton offered training on how to use NVivo and it was a good opportunity to practice in a safe environment and with excellent support. As a result, NVivo was chosen to support the data analysis in this thesis.

NVivo is a computer-assisted qualitative data analysis software that can be used for a wide range of qualitative research designs and data analysis methods and is not methodologically specific. It is used for qualitative data processing, management, analysis, and demonstration. The main function of NVivo is not to analyse data but to help in the analysis process, whilst the researcher remains in control of the qualitative data. Often, qualitative data are manually handled by pen and paper. NVivo aids in easily finding answers to research questions in the qualitative data. In manual qualitative analysis using a large sample size, the probabilities of obtaining an accurate answer within a short period of time are limited. It is difficult to obtain insightful answers with manual analysis. Carrying out an electronic research using NVivo can lead to more reliable results because of the elimination of human error (Zamawe, 2015).

In the present research, NVivo facilitated the researcher, after importing data, to classify similar information to different nodes and apply coding, bringing the information either manually or automatically together under common themes. The software also allowed the researcher to write memos in the programme rather than in a notebook, and different pieces of data were linked together through electronic memos that were used in building themes within the data (Welsh, 2002). The searching process in NVivo added rigour and credibility to the analysis process by enabling the researcher to carry out fast and precise searches of a particular type, and it enhanced the credibility of the results by ensuring that all cases of a certain participant were found; this searching process might need to be combined with manual search techniques so that the data are examined thoroughly (Hinchliffe *et al.*, 1997). It is imperative that researchers identify the importance of both electronic and manual tools in qualitative data analysis and management, as well as avoid relying on one tool over the others; researchers need to be open to and make use of both tools (Morison and Moir, 1998).

3.16 Framework analysis

Framework analysis was therefore selected for the data analysis process. The framework analysis method, often similar to thematic analysis or qualitative content analysis, is part of broader analysis methods (Ritchie *et al.*, 2013). However, it is different to thematic analysis because it tends to be more explicit, more strongly informed by prior reasoning and development of the matrix-based method of analysis (Pope, Van Royen and Baker, 2002). In addition to the different steps of analysis, it includes the creation of a theoretical framework that is usually inductive and driven by the data rather than being deductive and driven from the literature review (Gale *et al.*, 2013). In addition, there is summarisation of the codes and themes and subthemes, which may include a description of what the participants have mentioned in that code rather than using the participants' quotes only. This is useful in finding the meaning of the codes and subthemes in the early stage of the data analysis that helps in building a good understanding of the data.

The framework analysis method identifies similarities and differences in qualitative data before concentrating on the relationships between various parts of the data, thereby trying to make explanatory and/or descriptive conclusions grouped into themes. This method is

flexible during the analysis process in that it allows the researcher to both collect all the data and then analyse them, or to conduct data analysis during the data collection, which was the case for the semi-structured interview transcripts in this study. However, framework analysis can also be utilised for other data sources, such as focus groups, field notes or other documents. The framework analysis method also allows the use of constant comparative techniques by reviewing the data across a matrix. The other unique aspect of framework analysis is that whilst it uses a thematic approach, it enables themes to develop both from the research questions and from the research participants' narratives. Moreover, the findings derived from framework analysis are reflective of the participants' true beliefs and values. More details regarding how framework analysis was applied in the current study are presented in the next section.

3.17 Data Analysis

Framework analysis was used to analyse the data following the process outlined below in Figure 3-2 adapted from (Ritchie *et al.*, 2013).

Figure 3-2: Framework analysis phases outlining the processes of data analysis



The initial interview and focus group data were transcribed verbatim. Interactive methods of data generation, i.e., interviews and focus groups, were used. It was important to consider the role of this interaction within the context of the analysis because of its effects on data collection.

In the formal analysis, the researcher was conscious of the analytical process, using concepts and themes directly from the data and using ideas from the literature. The rationale for the analysis was to examine the transcript and identify any patterns that existed, as well as to compare these with those in existing clinical reasoning models in physiotherapy (Gale *et al.*, 2013; Bonello and Meehan, 2019). Comparing and contrasting data were important in the qualitative analysis of this study, with a particular focus on comparing findings within and across individual cases, which was a part of the framework method. This type of analysis cannot accommodate highly heterogeneous data, which means the data must have similar subjects or content so that they could be categorised (Heath *et al.*, 2012). This substantive and cross-sectional approach involved interpreting meaning in the data, with a focus on what the transcripts said. The framework analysis procedure (Figure 3-2) involved familiarisation, identifying a thematic framework, indexing, charting data into the framework matrix, as well as mapping and interpreting the data.

The first stage of the process was referred to as familiarisation, which involved the researcher becoming familiar with the data by thoroughly listening to the interviews and focus groups on numerous occasions, reading transcripts and discussing emerging issues in the data with the supervisors. This stage of framework analysis aims to identify the data widely - from individual interview and focus group to its overall '*decision making processes*' (Ritchie *et al.*, 1994 p.179).

Transcripts were read whilst listening to the recording and referring to the field notes from the interviews (Phillippi and Lauderdale, 2018). This provided detail on the participant's body language, confidence level and researcher's responses. In addition, any negative and positive feedback after each data collection contributed to the change in interviews or focus groups topic guide. This may influence the depth of data, which changes from one interview to another. For instance, one of the field notes stated;

"I believe I was trying to encourage the participant to follow Hypothetico-deductive reasoning by asking her: How do you confirm the diagnosis? Some participants said by re-assessment, others mentioned I do not have to confirm it because I am treating the symptoms. I realise maybe I was influencing the participants to think about confirming the diagnosis, and I should try to reframe the question in the next interviews."

In this 'familiarisation' process, the data were coded using NVivo.12. This helped establish a group of initial codes for different aspects of the participants' clinical reasoning process and the factors that influence physiotherapists' clinical reasoning. After several interviews, the group of initial codes was reviewed. There were a group of 150 initial codes relating to the clinical reasoning process. At that time, the researcher felt submerged in the factors that influence physiotherapists' clinical reasoning and understood the significant issues that emerged when participants spoke about their assessment and treatment of patients with UI. For instance, where the researcher was unsure about the participants' model of clinical reasoning that lead to analysing individual participants to understand the process of their decision making. A diagram was made that represented each participant to justify the model of clinical reasoning used by everyone. Figure 3-3. Then the researcher returned to the transcript and read it through to justify the analysis process. The reasoning models were discussed with one of the supervisors when agreement was made to some of the models and asked to change some of them. Members checking is a method to check for rigour to ensure that the most suitable models of clinical reasoning for participants were found.

Figure 3-3: A diagram used to identify one of the participant's clinical reasoning models. The main researcher raised the questions, and the answers were given by the participant.



The second stage involved identifying an initial thematic framework. The purpose of this stage was to systematically organise the data in a significant and controllable way for subsequent retrieval and analysis in the final stage that included mapping and interpretation. This stage helped in hierarchically arranging the themes and subthemes, which allowed the researcher to get to grips with the overall structure instead of being lost in labelling (Bonello and Meehan, 2019). The researcher found it helpful to write the meaning of each theme and subtheme and how it could be used. During this step, descriptive statements were formed. Ritchie *et al.* (1994) recommended that the process of developing framework categories is informed by prior knowledge and emergent issues arising from the data in the familiarisation stage. The researcher used previous knowledge in clinical reasoning to guide the thematic framework and include issues emerging from the data to answer the research questions. Ritchie *et al.* (1994) stated that the thematic framework is tentative, encouraging further refinement at different stages of the analysis.

The benefit of using both prior knowledge and issues emerging from data is that it focuses the framework on the research questions but gives flexibility to integrate the researcher's interests, in addition to the disputes most relevant to the participants, the researcher decided to base any framework categories around the main areas of attention in the

interview guide (physiotherapists' central skills, physiotherapists' knowledge and experience) (Bonello and Meehan, 2019). As the framework need to be driven by the data and participants' positions and opinions, the researcher checked with the supervisory team on one random interview to refine the initial categories based on the data. Each member of the supervisory team coded the interview by hand, marking the categories that each part of the transcript applied to. The researcher used this to check against any initial coding and categorisation, and found them to be quite similar to the marking of the supervisors (Gale *et al.*, 2013).

Since there were some differences, the researcher and the supervisory team discussed the importance of revisiting the research questions and asking various questions: Why does a specific part of the transcripts interest the researcher? Is it related to the topic, and how can it fit the aims and objectives? This helped the researcher to revise the initial framework to ensure it fits with any emerging topics in the data (Gale *et al.*, 2013; Bonello and Meehan, 2019). For instance, the researcher found that physiotherapists participating in the study discussed at length their relationship with their patients, the effect of that relationship on them and the importance of debriefs. As a result, a category was added for 'physiotherapist being vulnerable'. This allowed the researcher to manage the complexity of the data while maintaining the ability to manage unanticipated new ideas.

The final framework used in this study about the factors that influence clinical reasoning is as follows:

- Patients' characteristics, beliefs, behaviours, compliance with physiotherapy programmes and consent for vaginal examination.
- Physiotherapists' knowledge, experience, central skills, encounters with ethical and sensitive issues and consideration of patient's culture.
- Organisational factors such as health care systems, resources, and time.

Based on analysing each participant's thinking processes, the researcher determined the most common models of clinical reasoning that were used (as reported by the individual). Then, looking back at the definition of each model, the researcher returned to the transcripts to find the most suitable model to best illustrate the thinking processes of each participant. This helped in understanding the most common models of clinical reasoning and contributed to the development of a theoretical framework.

Developing the framework, though time-consuming, but was important because it started with category development, followed by members checking the categories on a small part of the data to refine them, in parallel with the newly emerging data (Gale *et al.*, 2013; Bonello and Meehan, 2019). When analysing the UK and KSA data, the challenge of generalising the framework to suit both countries became evident. Despite this, after reading a few of the KSA interviews, the researcher decided to use the same framework for both countries but be open to any emerging issue in the data. The supervisory team was involved in developing the framework through regular meetings to test and refine the framework.

The third stage of the process was indexing. Indexing aims to arrange the transcripts into the theoretical framework. This is the first stage of coding an interview and focus group transcript before moving to 'charting' the interview. This involves applying the framework to each interview or focus group transcript systematically. It means showing the themes or subthemes related to a specific data section (Gale *et al.*, 2013; Bonello and Meehan, 2019). Ritchie *et al.* (1994) suggested using software for indexing references. NVivo, a qualitative data analysis tool, was used in the present study.

While doing this, the researcher found herself coding data to a new category to capture the interaction between the researcher and participant, which was not caught elsewhere in the framework but could reveal something interesting about physiotherapists' clinical reasoning model. For example, physiotherapists are sometimes exposed to horrible patients' stories, where they find themselves vulnerable and need time to debrief and talk with other colleagues. It was essential to capture this, resulting in the addition of a category labelled "being vulnerable". Thus, during this process, the researcher was open to unexpected issues that emerged from the data and used the supervisory discussions to determine how to respond to these issues. The development of the theoretical framework is an ongoing process that may continue to be modified, even at the later stages of the framework analysis (Bonello and Meehan, 2019). There is a risk that the indexing stage (Figure 3-4) can become an automated process, especially when having both UK and KSA participants. Further refinement can ensure that the framework remains a thoughtful process. It was helpful to write the analysis diary Table 3-1 to document the changes made

to the theoretical framework and the researcher's feelings toward indexing as an automated process.

Figure 3-4 An extraction from NVivo to show the indexing process



Table 3-1: the data analysis diary

Progress of data analysis diary that helped in organising the researcher's thoughts and

report

16/07/2019

I printed the theoretical framework and made sure that I covered all the categories and subcategories with each participant. This will help me to do the charting and indexing. If participants did not cover explaining how significant internal examinations were to patients, then I wrote a memo that participants did not mention it. This helps to save time when it comes to reading the transcripts again, looking for certain categories and subcategories. Also, my supervisor encouraged me to write the assessment process for each participant.

22/08/2019

I decreased the number of codes but until this point, I did not get to the final step of realising the categories and themes. I worked on KSA interviews and UK focus groups. In order to understand the process of clinical reasoning, I wrote the process of clinical reasoning for each participant in the KSA, and I am planning to do the same for the UK. Then, at the end, I'll be able to understand the clinical reasoning model for all the participants in general. I must understand the factors that influence clinical reasoning from the focus groups first; then add the other factors from the interviews.

The fourth stage involved charting the data into the framework matrix. The purpose of the charting is to arrange the data into a more organised format to assist in the analysis process at a later stage of framework analysis. It included summarising and indexing the data that had been working on and arranging the summaries in chart form (Appendix Appendix E). The chart columns show the categories from the framework, with participant data shown in raw format (Gale *et al.*, 2013; Bonello and Meehan, 2019). This summarises each participant's thoughts that relate to each category. NVivo allows summarising each category related to individual participants automatically. This is not always relevant and might not include all the critical issues. However, this feature helped link participants' quotes in the primary transcripts to a summary of the main points that reflect the code using a link (See Figure 3-5 below). This was undertaken manually to ensure that the most relevant ideas were covered (Bonello and Meehan, 2019).

In some instances, participants' quotes illustrated the categories, eliminating the need to summarise them. In this case, italics were used to highlight that the summary was the participant's quotes. The researcher was aware that some of the participants' reasoning might be lose, so the supervisors were asked to compare the chart to the original text to ensure that the original, whole meaning was not lost. At the same time, it was necessary to make sure that the summaries were not simply duplicating all the text in the transcripts as it would defy the aim of reducing the data into a more controllable way (Gale *et al.*, 2013; Bonello and Meehan, 2019), useful for analysis of the 48 participants. Although this data management process can be slow and time-consuming, having well-labelled and refined data provided a strong foundation for starting the interpretive analysis.

Figure 3-5: Framework matrices in NVivo that summarise each participant's quotes related to each category and subcategory

	Home Import	Create	Evolute Share Energy Main					
ste Clipt	X Cut Rh Copy Doard	Format	Speling Annotatio Editing Annotatio	New Summary Link Summary	Associated View * It Layout *	Ri S	wes & Columns - Coding Highlight Ge Also Links Stripes - Coding Highlight	^
: 1	Cubre 😠	Echare 😿 to explore the clinical reasoning in Saudi.						^ Rel
F +		ense	C : Considering culture again that's very different from living at the culture within a country rather than a migrant	D: Culture can be different within the sa me indigion comparing imigrant Muslims to Saudi Arabia Muslim culture is completely different	E: Different ethnic group so, you know, I always want to be sure that people are absolutely and the majority of people are, there involved in (MASE OF	F	Intenviewee: When we were at the WCDT Congress in Cape Town last year still last year and we did a semirar and (name of person)) id share the semirar spoke and that was (name of person) did a bit about cultural differences and it was interesting it's actually it's play although WCPT to you might still be on the WCPT webster you could see it and there were but we were talking about things and barries to treatment	ie
	1 : Interview 1 júpdate	d)	population in another country which I fully accept		When Y NUMBER IN (VALUE) UN CITY for a lot of my career, umm, about 25% of the population are either erm, Pakktani or Bangladeshi Muslim origin and I probably throught when I worked there at first err, you know, that they'll be more of an issue there about examination but actually it was probably no different from anybody elso, you know, se there's different people come, you know(NAME OF CITY) are poor city, outide London It's probably one of the poorest		95 and I know when (name of person) was in I don't know if it's the same in Saudi but when (name of person) was in (NAME OF CITY) at that stage and that's a long time ago it was, you know, female physics treat women and gay male physics treat men and when I was in (name of City) they were saying that if a woman's not married you wouldn't do an internal examination yeah so that's intercenting yeah but obviously I was not impleed but, you know, if Safariy typken how in the Lik but cortainly working in (NAME OF CITY) that was a an introduction to it but it but you can also get I mean and again that's very different from Jimg at the outbure within a country rather than a migrant population in another country which if tub, account and addition and the sound provident and the were Midlement but is for a low would be a sound but the to you know, companying Pakitani Muellins to Saudi Arabia Muellim culture is completely different as wells to things that things are very different but is more than your know, you do have that we have the part of the part warp	
	2 : Interview 10 transco	भंतन					tecon who did the guidelines and loe worked _ 	
F	4					w b	97	•
In	Nodes			•	Code At Enter node name (CTRL+C	2)		© X

The fifth stage was abstraction and interpretation, which involved analysis of the key findings. The purpose of this stage of framework analysis is to move away from data organisation to understanding it. In this stage, the researcher began to identify what might become the main findings of the research. The mapping process typically involving from the superficial features of the data to more analytic properties (Ritchie *et al.*, 1994; Ritchie *et al.*, 2013). This stage allowed the researcher to make sense of the data by moving back and forth between the different stages of abstraction – without losing sight of the raw data or the cross-case and within-case analyses – to create more analytical concepts or themes. At this stage, if the data supported it, the researcher developed specific forms of classification, such as typologies (Bonello and Meehan, 2019). Ritchie *et al.* (2013) described typologies as a classification system that generates categories and classes, a particular type of classification where categories are independent of each other (so that each case can only fall into one type).

The next part of the abstraction and interpretation process was mapping. This involved comparing the responses to understand the relationship between the quotes and the links between the data, as well as creating a set of categories or classes about key analytic themes. In this process, the investigator might want to explore how these separate features of the data are interrelated (Dey, 2003). Mapping and interpretation included finding patterns and expressing the researcher's sense making of the data, about the research question. This step was the most challenging part, however, since the process of interpretation is often gradual, the researcher tried to make sense of the data gradually after discussing it with the supervisory team over time (Gale et al., 2013; Bonello and Meehan, 2019). There were limited descriptions of the interpretive process in the literature. It was helpful to use a schematic diagram (Figure 3-6 below) to guide any data interpretation, as Srivastava and Thomson (2009) suggested. Also, Swallow et al. (2011) found the work very useful in helping the researcher move from abstraction to more interpretive ways of thinking. This stage took longer than expected; and required some modification of the main findings quite a bit by going back and forth to the raw data. Writing memos and diaries were very helpful in keeping track of any changes that have been made in Table 3.1

The fifth stage was abstraction and interpretation, which involved analysis of the key findings. The purpose of this stage of framework analysis is to move away from data organisation to understanding it. In this stage, the researcher began to identify what might become the main findings of the research. The mapping process typically involved moving from the superficial features of the data to more analytic properties (Ritchie *et al.*, 1994; Ritchie *et al.*, 2013). This stage allowed the researcher to make sense of the data by moving back and forth between the different stages of abstraction – without losing sight of the raw data or the cross-case and within-case analyses – to create more analytical concepts or themes (Gale *et al.*, 2013; Bonello and Meehan, 2019). At this stage, if the data supported it, the researcher developed specific forms of classification, such as typologies. Ritchie *et al.* (2013) described typologies as a classification system that generates categories and classes, a particular type of classification where categories are independent of each other (so that each case can only fall into one type).

The next part of the abstraction and interpretation process was mapping. This involved comparing the responses to understand the relationship between the quotes and the links between the data, as well as creating a set of categories or classes about key analytic themes (Gale et al., 2013; Bonello and Meehan, 2019). In this process, the investigator might want to explore how these separate features of the data are interrelated (Dey, 2003). Mapping and interpretation included finding patterns and expressing the sense of the data, about the research question. This step was the most challenging part. However, since the process of interpretation is often gradual, the researcher tried to make sense of the data gradually after discussing it with the supervisors over time. There were limited descriptions of the interpretive process in the literature. It was helpful to use a schematic diagram (Figure 3-6 below) to guide the interpretation of the data, as Srivastava and Thomson (2009) suggested. Also, the work by Swallow et al. (2011) was handy in terms of helping to move from abstraction to more interpretive ways of thinking. This stage took longer than expected; and required modification of the main findings quite a bit by going back and forth to the raw data. Writing memos and diaries were helpful in keeping track of any changes that have been made in Table 3.1.



Figure 3-6: A schematic diagram illustrates the process of data analysis in interpreting and abstracting the findings.

The final part of abstraction included explanation and occurred in the subsequent stages of the analysis process when most of the descriptive and typological work was already carried out. When explanations were established, instead of hypothesis testing like in a quantitative study, the researcher looked for the key factors or processes that reflect a pattern of data connection and attempted to find common-sense and fitting features between the data (Gale *et al.*, 2013; Bonello and Meehan, 2019). In addition, the researcher looked at other high-quality qualitative studies or existing theory within the literature for inspiration (Ritchie *et al.*, 2013).

During the data interpretation process, the researcher considered the following headings as a framework for interpreting coded data: words, context, internal consistency, frequency and extensiveness of comments, specificity of comments, intensity of comments and big ideas (Krueger, 2014). It was important to consider the actual words used and their meaning, as well as to reflect on the context of the language of the facilitators' enquiries and the consequent comments made by others in the group, which might affect the context within which the comments were made. Likewise, it was necessary to consider the repetition and breadth of comments. Repetition or frequency relates to how often a view or comment was made. The intensity of comments in the transcripts was also observed by looking at the depth of emotions in which views were said or feelings were made evident (Gale et al., 2013). Internal consistency was tracked by considering any changes in views or positions by the participants. The specificity of the participants' responses was noted by placing greater attention to responses relating to personal experience (Ritchie et al., 2013). Big ideas related to the larger theories or models that appeared from an accumulation of signals and evidence across the transcripts. The analytical processes described above were presented with respect to the data generated through the face-to-face interviews and focus groups. However, focus groups might have distinctive characteristics that warrant additional consideration during the analysis process. The different types of focus group analysis will be discussed in the following sections.

3.18 Analysis of Focus Groups

A one-hour focus group discussion could easily take five to six hours to transcribe in full, generating approximately 30 to 40 pages of transcript. Thus, a central goal of data analysis, according to Robson and McCartan (2016), is to reduce data volume needed to carry out

analysis. It is important to begin by going back to the purpose of the study. Following this is helpful for managing the data and discarding irrelevant and additional information (Krueger and Casey, 2000).

3.18.1 Two main ways to analyse focus group data

A whole-group analysis was used in the present study, which considered the data created by the group without explaining individual contributions. The group became the basic unit of analysis and was handled in the same manner as a unit of individual data. During the data management stage, data were abstracted and indexed in the form of raw in the matrix for each group (Appendix Appendix E). If there was a debate between the participants within the group, then an individual participant was highlighted.

There are advantages and disadvantages to each type, and there is debate on the validity of each type. The advantage of doing participant-based analysis is that it is easy to observe the similarities and differences between the members of each group and across all groups in the sample. In terms of disadvantage, whole-group analysis might take longer to carry out because the points made by each member needs to be traced throughout the transcript. For a focus group, it is not like a one-to-one interview, as it does not allow for full exploration of a participant's views and opinions; the data might therefore be insufficient for data analysis, as discussion needs to be placed in context of how other group participants have responded (Ritchie *et al.*, 2013)

The decision on which approach to use depends on the level of detail available for each participant, in addition to the objectives of the research and the types of outcomes required. In this study, the aim of the focus groups was to explore the factors that might affect clinical reasoning and decision making among physiotherapists. Therefore, it was deemed useful to analyse the data produced by the whole group in order to understand the different factors that can affect their clinical reasoning and decision making. However, in some cases, the researcher identified a unique factor, such as patient culture, which was raised by one participant only. In this case, it was considered interesting to explore this participant individually and evaluate the influence of the participant's experience, knowledge and other factors that might have affected her clinical reasoning and decision

making (Ritchie *et al.*, 2013). It was important in both types of analyses to consider the interactions and collaboration between group members, such as their areas of conflict, non-verbal communication, the characteristics of the participants who contributed more or less than the average, and the construction of opinions over the course of the group discussions. Data from the interviews were combined with that of the focus groups using NVivo following the processes outlined above.

3.19 Methods to ensure rigour

Researchers who use interpretivist frameworks engage in reflexive analysis to ensure rigour. Lewis-Beck, Bryman and Liao (2003) stated that credibility, confirmability, and transferability are the three main standards of rigour that are common to qualitative methods, in general. Within interpretive approaches, a 'reflexive stance' is recommended because it informs 'how the researcher conducts his/her research, relates to the research participants and represents them in written reports' (Charmaz and Belgrave, 2007).

3.19.1 Credibility

Credibility is sometimes used in qualitative research. Hammersley (1990 p. 57) stated that 'credibility means the extent to which an account accurately represents the social phenomena to which it refers'. Credibility relates to the trustworthiness of the findings of a study (Barbour, 2001). Ritchie et al. (2013) explained that 'credibility in qualitative research has focused on evaluating how well a participant's meaning has been captured and interpreted'. There are several different approaches to improving the credibility of qualitative data, such as triangulation and members checking. Triangulation is the use of different sources of data, which helps improve the clarity and confirmability of a research outcome. In the present study, triangulation involved two different methods of data collection that will answer the research question, for instance, interviews and focus groups.

Patton (2002) highlighted the importance of triangulation in data analysis: it provides a comprehensive and diverse way of exploring the same phenomenon, and its use strengthens confidence in the conclusions and thus adds to the credibility of the findings. The assumption is that if the findings obtained with all these methods are consistent and draw the same or similar conclusions, then the credibility of the findings and conclusions

has been established (Moisander and Valtonen, 2006). However, the absence of similar findings from different approaches does not necessarily provide grounds for disproof of the results. This is because each method used in qualitative research can provide a limited view of the whole picture (Barbour, 2001). In the present research, credibility centred on whether clinical reasoning was accurately identified and described from the perspective of physiotherapists.

There is some argument about the use of triangulation in qualitative research; a number of researchers argue that triangulation can play some part in validating findings (Barbour, 2001; Flick, 2004; Fusch, Fusch and Ness, 2018). However, the aim of this qualitative study is to explore participants' viewpoints and opinions. Physiotherapists' opinions related to the factors that influence their clinical reasoning can vary from one participant to another. Using different data collection methods in this research enabled the researcher to dig deeper and understand the factors influencing the physiotherapist's clinical reasoning and the participants thinking processes. It is not essential to validate the findings or test their probability like a quantitative study. Proving the rigour of the findings could be guaranteed through different aspects, such as using data collection methods that answer the research question and including clear data analysis processes in addition to members checking. Hence, triangulation adds a depth of understanding and adds to the credibility of the findings rather than validation.

Barbour (2001) claimed that there are different ways to promote rigour in a qualitative study, such as purposeful sampling, multiple coding, triangulation, and respondent validation. However, Barbour (2001) argued that these techniques might strengthen the rigour of the research only if they are embedded in a wider understanding of the qualitative research rationale, design and data analysis. Barbour (2001) suggested using a constant comparative method to compare the opinions and experiences of purposely selected participants to highlight important differences and decrease potential research bias. This is in addition to the use of multiple coding that concerns the same inter-rater confirmability in qualitative studies. Although multiple coding does not necessarily mean accurate replication of coding, it involves the cross-checking of coding approaches and clarification of data by independent researchers; in this study, these were the researcher supervisors. However, the level of concordance between researchers was not really significant; what

was of importance is the content, which provoked disagreements and the insights that resulting discussions can yield for refining theoretical framework (Barbour, 2001; Silverman, 2015).

3.19.2 Confirmability

Confirmability, also known as dependability or auditability, is sometimes a more preferred term in qualitative research over reliability. It refers to the ability of another researcher to follow the methods of a study and reach the same conclusions as the original researcher (Patton, 2002).

The confirmability of the study can be achieved by following a transparent research process—describing the research strategy and a clear process of data analysis—so that the procedure is clearly documented and understood. This step would allow another researcher to verify the findings. Furthermore, it is important to show explicitly the theoretical stance of the researcher in which the data interpretation takes place (Moisander and Valtonen, 2006). Regarding confirmability in studies involving interviews and focus groups, Silverman (2015) highlighted the importance of each participant's understanding of the questions in the same way so that answers can be coded without ambiguity. This was accomplished through several resources, such as pilot testing the interviews and focus groups, improving the interview skills of the researcher by attending specialised training courses, and conducting inter-rater confirmability checks on the coding of the transcripts with the researcher supervisors.

Nonetheless, when an interview or focus group is digitally recorded and transcribed, the confirmability of the transcripts' interpretation might be compromised by a failure to transcribe critical pauses and overlaps. Silverman (2015) suggested using transcripts that do not try to eliminate such structures of spoken English as hesitations ('er'), breaths ('hh'), non-dictionary words ('yer') or pauses. It would results in a more reliable recording of the data because of the amount of detail given by the method (Heritage, 2013). This method involves observing the small and less obvious details of naturally occurring talk, which may reduce the interpreter's possibility of misunderstanding the transcript and may improve research rigour, therefore this method was used in this study.

3.19.3 Transferability

This is also referred to as resonance and generalisability. Transferability pertains to the probability that the research findings hold meaning to others in similar situations (Patton, 2002; Charmaz and Belgrave, 2007). Transferability is not easily achieved in qualitative research. It requires strategic planning throughout the research process and not just during the write-up at the end. The methods used to organise the data can influence transferability. Many researchers use different strategies to promote transferability. The type of analysis has a significant impact on better supporting the development of an argument. This might contribute to the transferability of the results and can improve the rigour of the study. On the other hand, using a sampling strategy (for instance, the number of participants and a representative sample of the population) to allow empirical generalisation is not supported in a qualitative study. However, theoretical generalisation can be supported by purposeful sampling. The sampling strategy could provide an important background in which the researcher can read and interpret (Mason, 2002). Furthermore, writing a detailed description of the sample's demographic information is important. The aim of purposeful sampling is to reflect diversity in a given population. For example, in the current study the UK participants came from different regions, such as the Southwest and East, North of England, Scotland and Wales. It also included different settings, such as private, NHS and community services. On the other hand, in KSA, twelve participants came from the Central region only, where all the big tertiary hospitals with good facilities are located. Therefore, another three participants were recruited from the Eastern and Western regions of KSA to ensure the transferability of the data within Saudi.

Furthermore, deviant case analysis involves actively recognising cases that do not fit with the research question and objectives. These cases can lead to a change of initial ideas but, more frequently, to a deeper analysis that accounts for a wider variety of conditions (Seale, 2012). Complete data coverage with attempts to consider deviant cases altogether through a clear and systematic analysis can increase transferability (Ritchie *et al.*, 2013). Furthermore, showing the participants' own accounts can assist other researchers in judging how closely the interpretations reflect the data. This can enhance transferability by making effective use of the original data that support the research questions studied (Ritchie *et al.*, 2013).

Using computer programmes, such as NVivo, helps to manage data and thus helps qualitative data analysis and supports the inclusion of typical and deviant instances of data, potentially improving the systematic analysis of data (Weitzman and Miles, 1995; Miles, Huberman and Saldana, 2013). Furthermore, recording data accurately and comprehensibly, such as through the use of audiotapes and variable levels of detail in the transcription of data can increase research rigour (Silverman, 2015).

3.20 Ethical Considerations

If sensitive issues are raised, an interviewee may become emotional; therefore, the interviewer needs to show empathy, offering the interviewee a break from the interview so that he/she can gain his/her composure, stopping the interview altogether if needed (Hennink, Hutter and Bailey, 2010; Creswell and Poth, 2017; Mason, 2017). In the present study, the researcher mentioned to the participants that their information, obtained through their participation in either the interview or the focus groups, would not be disclosed. However, as focus groups are group discussions, other physiotherapists in the focus groups would evidently know the participants who attended and would hear any comments made during the discussions. Nevertheless, all physiotherapists who attended the focus groups were asked to respect the confidentiality of the other participants.

It was difficult to end some of the interviews after establishing rapport with the interviewees, so repeating some of the research aspects covered in the introduction part was useful. Examples are repeating what will be done with the data or emphasising the clinical implications of the research, as these helps create some distance between the researcher and the interviewees; consequently, the researcher felt comfortable leaving the interviewees.

3.21 Data management and confidentiality

Unidentifiable data will be stored for 10 years in accordance with the University of Southampton's research data management policy. Personal data, on the other hand, are to be stored until completion and award of the researcher's PhD degree. Data collected in the UK in paper format (demographic details and consent form) were scanned as PDF files and uploaded to a secure file store in line with the University of Southampton guidelines;

these data can only be accessed by the researcher and her supervisors. Written and digitally recorded data were stored in a password-protected computer that was kept in a secure environment. The original written papers were shredded by the researcher. In the KSA, the researcher accessed the University of Southampton's Windows-based desktop, which can be done remotely from any device in any location via a password (Southampton Virtual Environment). The researcher followed the same processes regarding data management for both the UK and KSA participants.

Anonymised electronic data were transferred to NVivo for analysis by using the University of Southampton's secure transfer system 'drop off'. These electronic data did not include identifiable information and instead used identification numbers to anonymise. At the end of the study, these data—both physical and electronic—were stored according to the University of Southampton's data management policy (Southampton, 2021).

As for confidentiality, the researcher's contact details were provided to the participants should they have any questions or wish to withdraw from the study. Confidentiality was maintained throughout the interviews and focus groups. For instance, at the beginning of each focus group, the researcher asked all the participants to be mindful of the importance of confidentiality and requested them not to discuss outside what was said during the focus groups. All physiotherapists who attended the focus groups were also asked to respect the confidentiality of the other participants.

Finally, the interviews and focus groups were coded, and real names and places were altered to preserve confidentiality. Any e-mails sent to the research participants had their content coded and preserved in a password-protected Word document, but the contact details and original emails were deleted. All participants who took part in the research were given a code, which was used instead of their name, to identify them during the analysis and write-up of the study.

3.22 Reflexivity

Parahoo (2014) defined reflexivity as the extended procedure of the researcher's reflection on his/her principles and views, presumptions, behaviour or position and those of the participants, which can influence the clarification of responses in the research or with participants. Awareness of the reciprocal effect of the participants' and researcher's relationship with the research process and outcome is an important part of ensuring rigour and credibility in qualitative research. Reflexivity is critical in physiotherapy research in which the researcher often knows the participants. Primeau (2003) indicated that reflexivity increases the quality of research through its capability to prolong our understanding of how our values, beliefs, experience, personal involvement and positions as researchers influence all steps of the researcher process. Dowling (2006) clarified how undertaking the reflexive process helps researchers develop insights into how physiotherapists make their clinical decisions, explaining that it involves being aware in the moment of what is affecting the researcher's internal and external responses while at the same time being conscious of the researcher's link to the research topic and the participants. The key is to be explicit regarding the influence of the research process on the relationship between the researcher and the participants in a detailed reflexive stance at the beginning of the data collection process and during the data analysis.

3.22.1 Engaging in critical self-reflection

Koch and Harrington (1998) believed that a researcher should engage in continuous selfcritique and self-appraisal, as well as explained that previous experiences may or may not influence the research process. I realised that my experiences as mentioned in section 1.2.2 were focused on dedicated teaching about the clinical reasoning process, and my practice was based on the model that I believed physiotherapists should use in 1.2.2. Reflecting on my previous experiences, I asked myself the following question: Why did I value the HD clinical reasoning approach? I then began to reflect on how my past experiences as a student, pelvic health physiotherapist and lecturer affected my future research work. I decided to accept that my past would guide my interpretation of a specific reality. Knowing more about myself helped clarify my expectations and biases, which was essential in the interpretive process.

This perspective changed the way I saw the participants in my study. I believed that building a good relationship with them during the data collection process instead of considering them as data sources is important. During my interaction with expert physiotherapists, I encouraged them to bring life to the experiences that we were exploring; they should move from certain parts of their experiences to the entirety of it in order to increase the depth of engagement involved.

My personal experiences as a physiotherapist and an educator provided me with important knowledge on clinical reasoning in pelvic health physiotherapy, which was used to inform my study. This knowledge represented previous experiences at the start of the study. New knowledge was gained through the interpretation of the data collected, which eventually led to an in-depth understanding of current clinical reasoning among physiotherapists using a methodological framework. I used an interpretive, framework analysis approach to discover meaning within clinical contexts of practice in the UK and the KSA.

3.22.2 Positioning the researcher in the study

Reflecting on my past is essential to the success of the entire process. For instance, awareness of my previous experiences, beliefs and prejudices was necessary. Knowledge of pre-understandings helped in accessing my participants' world and allowed such understandings to be argued with the use of the participants' stories (Dowling, 2006).

My role as the researcher was dominated by the ontological and epistemological assumptions taken from the interpretive position, as stated in Section 3.1. I conducted the interviews and asked questions without taking a privileged position. My previous experiences in physiotherapy practice and reasoning may have influenced both the content and the process of enquiry. Therefore, open-ended questions and prompts were used to address this issue. Furthermore, having solid knowledge of what the participants were talking about enriched the discussion without influencing their opinions or committing interviewer bias. I ensured that I did not search for a particular type of answer. A researcher's knowledge should not create researcher bias and result in leading questions.

In this way, a collaborative, non-hierarchical relationship between me as the researcher and the participants was encouraged, and the participants played an active role in the creation and justification of knowledge (Fontana, 2004). In this study, I, the researcher, was a physiotherapist and a colleague of the participants. This factor facilitated the achievement of the research outcomes. Familiarity with one another allowed us to create a relaxed

atmosphere in which we could comfortably communicate. As a result, we were able to explore the subject deeply.

Furthermore, exploration of my personal beliefs made me more aware of the possible judgments that can occur during data collection and analysis based on my belief system instead of the actual data collected from the participants. This process was used to separate my personal opinions and presumptions from the phenomenon under study. This process is called bracketing and requires reflexive thinking (Jootun, McGhee and Marland, 2009).

Rolls and Relf (2004) suggested that a principal researcher needs to bracket or to set aside his/her assumptions and previous experiences during data collection. In the present study, this process assisted me in understanding how my assumptions may have influenced the data collection and analysis process. Furthermore, it was important to ensure that I did not assume that I knew what the participants were talking about. Doing so might have led to losing opportunities to obtain valuable insights from the participants.

During the data analysis process, the data gathered from the participants may have been combined with my experiences and placed in a specific context. To minimise this, I stepped back during data analysis to avoid having my opinions and beliefs bias my interpretation of data from participants' transcripts. The interpretation of the data involved a combination of data analysis techniques. In the interpretation, I consciously used my experiences as a reflexive tool to acquire new knowledge of and insights into the clinical reasoning of physiotherapists in the UK and KSA.

3.22.3 Reflection in action

Data were collected using semi-structured interviews and focus groups, which were conducted in the UK and the KSA. To determine the social significance of actions in a study, a researcher must create an account of how participants see the phenomenon in question, which needs analysis to understand participants' perceptions (Schatzman and Strauss, 1973). The production of the participants' viewpoints and perceptions depends on the researcher's knowledge of the social setting. In the present study, I was, to some degree, familiar with the setting in which the physiotherapists socialised. I had practised as a pelvic health physiotherapist and worked as a physiotherapist–lecturer, so I was familiar with physiotherapists' jargon and hospital practices and techniques, in general. Although the
approach I used was interpretivist, I was aware of the potential biases that could occur during this study. I found it useful to keep research diaries to raise awareness of the effects of bias on my interpretation and clarification of the data and on my relationship to the research topic and the participants (Koch and Harrington, 1998).

I used the research diary to log decisions made and write down reflections on the research process to document changes in the thinking (Silverman, 2015). A research diary aims to document all easy-to-forget feelings and decisions made at a particular time in the past. It served as an anchor for my thoughts, feelings, and conclusions. The research diary helped me to realise how my knowledge is created. Examples of reflections on the research processes are narratives on learning points, interesting observations, useful reading texts and responses to actions and events such as resolving data collection anxiety, dealing with difficult feedback, and writing up (Engin, 2011). I found that the diary also became emotional support. The emotional aspect of carrying out research processes. The content of the research diary was used to remind me about my position and relationship to the topic during data collection, analysis and writing up.

3.23 Summary of the research paradigm, methods of data collection and analysis

An interpretivist paradigm was used as a philosophical underpinning to guide the collection and analysis of the data. The researcher encouraged participants to give their own opinions and views on their assessment and treatment decisions while reassuring them that there were no wrong or correct answers. This was important as some participants came from positivist paradigm backgrounds and were used to answering survey questions rather than qualitative open-ended questions.

To understand the participants' thinking processes, observation of the physiotherapists' management processes while assessing and treating patients with UI was considered as the optimal data collection method; however, it was assumed that this method would not be accepted by the KSA patients due to cultural issues. As a result, semi-structured interviews were used as a method of data collection to understand the participants' clinical reasoning and decision making processes. Focus groups, on the other hand, were used to

understand the factors that influenced physiotherapists' clinical reasoning and decision making.

The processes of data collection started with semi-structured interviews in the UK; then, a quick analysis of those interviews took place using the first few phases of framework analysis. For the practicality of data collection, the researcher travelled to KSA to collect data there using semi-structured interviews first, which was followed up with a focus group discussion. A summary of the UK interview findings was presented to the KSA participants towards the end of the focus group in order to obtain their views and opinions on the findings. This further encouraged the participants to provide rich and insightful data in order to help the researcher to gather meaningful data. After the data collection processes in KSA, a quick data analysis was conducted to understand the KSA participants' clinical reasoning and decision making using the first few phases of framework analysis. Later on, two UK focus groups took place to understand the factors that influence the UK participants' clinical reasoning and decision making. Towards the end of the first UK focus group, a summary of the KSA interviews findings were reported to the participants and they were asked to provide their opinion on these findings. The aim of reporting the findings in both focus groups, was to increase the depth of the analysis and to understand both countries' participants' views and opinions about the similarities and differences between the UK and KSA clinical reasoning and decision making. At the end of data collection, a theoretical framework was created and checked by the researchers' supervisors to improve the credibility of the codes. Further data analysis including charting and matrices were employed to understand the patterns that exist within the data. After long processes of comparing the charts, matrices and the common patterns within the data and the existing literature, different themes and categories were identified. The next chapter presents the participants' demographics and the findings from the current study.

Chapter 4 Factors that influence Pelvic Health Physiotherapists' Clinical Reasoning and Decision Making

4.1 Introduction

This chapter presents the findings of the analysis and interpretation of the transcripts of 28 semi-structured interviews and three focus groups with physiotherapists from the UK and KSA. During the analysis, it became clear that the process of clinical reasoning and the factors influencing the decision making among physiotherapists were both varied, complex, and overlapping. As a result, one of the challenges in analysing, organising, and presenting these data was to create a precise, useful, and meaningful interpretation of decisions influenced by such a broad range of factors. The findings are derived using the perspective provided by the study aim and objectives as outlined in Sections 2.17, 2.18 and 2.19

As described in Sections 3.5, individual interviews were conducted to understand the process of clinical reasoning by asking each participant to recall one of their patients and to describe the assessment and treatment processes. They also served to explain how participants reached their diagnosis, and how they confirmed the diagnosis – more detail is available in the interview guide in Appendix Appendix O. Interviews were the most appropriate way to understand the models of clinical reasoning used by physiotherapists while assessing and treating patients with UI. In the focus groups, participants were asked to discuss the factors that influenced their decision making and clinical reasoning. The data from these two methods are presented.

This chapter starts with an overview of the demographic details of the study participants, including details of their qualifications, post-graduate courses and number of years worked in pelvic health physiotherapy. The second part of this chapter presents the interpretation of the data through the development of three key themes and associated sub-themes (derived from analysis) related to the factors affecting reasoning and their

associated sub-themes derived from analysis. In addition, one theme on the sense making processes is discussed in the next chapter. A framework of inter-relationships between the themes is presented in Figure 4-1, and the findings for each theme are discussed in turn. Themes are explored in detail with supporting evidence for the interpretations provided via quotations from the participants, as well as how each method (interviews and focus groups) contributed to the theme.

4.2 Themes and Categories

The data from both interviews and focus groups were used to generate the themes. The interviews mainly generated data on the thinking processes of physiotherapists while managing patients with UI, whilst the focus groups primarily generated data on the factors that influence physiotherapists' decision making. However, given some of the overlaps between the data generated from both methods, both contributed towards the themes identified, and were therefore used to understand the objectives of this research. The process of coding, analysis and interpretation described in the method chapter led to the identification of four themes and seven categories of factors and processes that affect the decision making of physiotherapists (Table 4.1). The fourth theme on the sense making processes will be discussed in the next chapter.

Table 4-1: Themes and categories of the factors and processes that influence pelvic health

Themes	Description	Categories
Contextual factors	Factors include cultural context, health systems and other factors that affect physiotherapists' decision making and clinical reasoning.	 Cultural context Resources Organisational culture
Physiotherapists' factors	Pelvic health physiotherapists' factors that influence decision making and clinical reasoning.	 Physiotherapists' multifaceted knowledge, training, and experience. Physiotherapists' emotional intelligence
Patients' factors	Patients' factors that is affecting physiotherapists' decision making and clinical reasoning.	Patient characteristics such as age, gender, and comorbidity.
Making sense of physiotherapists' decision making and clinical reasoning	Physiotherapists' sense making of different factors using different properties of sense making theory that guide the decision making and clinical reasoning	Sense making process

There is interplay and overlap between many of the themes and categories above. For instance, the decision to provide patients with different assessment options might be linked to a physiotherapist's training or a patient's individual culture and resources. The categorisation of factors and clinical reasoning models into themes in this thesis provides a way to help simplify and organise the data. This in turn helps the reader to understand the multiple and complex influences on physiotherapists. To reflect this complexity, where there are apparent interactions and patterns between themes, these are highlighted and discussed. Figure 4-1 below shows the correlation between the four themes identified.

Figure 4-1: Illustrates that contextual factors are a strong recurrent theme with pelvic

health physiotherapist decision making and clinical reasoning.

Physiotherapists' characteristics and level of training interact with their consideration of the patient in front of them, and their awareness of the available resources, health care systems and cultural context. These factors were utilised in a sense making process to reach a clinical decision for each patient. Different types of clinical reasoning models were evident, with different cues employed from each theme.



4.3 Interview Participants

Table 4.2 and Table 4.3 below provide the demographic details of the 28 participants who attended the semi-structured interviews in the UK and KSA. The data are categorised by

demographic factors deemed most likely to impact their decision making: level of education, years of experience in UI, specialised UI courses attended and clinical settings. Participants who attended the semi-structured interviews had various qualifications and experiences, and held different post-graduate courses in pelvic floor rehabilitation. All

participants from the UK and KSA were female. KSA participants had an average of six years of experience; most of them were junior, whilst the UK participants had an average of 18 years of experience. The average age of participants in the UK was 52 years old, while the average age in the KSA was 32 years old. Most of the participants in the UK and KSA had achieved a post-graduate certificate in continence care, except for five KSA participants who had obtained the shorter Introductory Postgraduate Continence Care (basic courses). Pelvic health physiotherapy speciality started relatively recently in 2005 among Saudi physiotherapists. This could perhaps explain the differences in the years of experience between the UK and KSA participants. In both countries, most participants were working in government-funded hospitals or clinics, except for three participants in the UK, and five in KSA - they were working in private settings.

No.	Age	Qualification	Number of years working in UI	Post-graduate courses	Clinical settings
1	65	Master's degree	31	Certificate in continence care*	Private clinic
2	55	Master's degree	20	Advanced courses**	NHS hospital
3	45	Bachelor's degree	14	Certificate in continence care	NHS hospital
4	55	Bachelor's degree	18	Advanced courses	NHS hospital

³ * Certificate in continence Care or women's health is a post-graduate program incontinence management for one year that is provided by the University of Bradford or East of London/UK.

^{**} Advanced courses in continence care and/or women's health that is provided by POGP.

No.	Age	Qualification	Number of years working in UI	Post-graduate courses	Clinical settings
5	56	Master's degree	25	Advanced courses	Private hospital
6	45	Bachelor's degree	8	Advanced courses	Community service
7	55	Bachelor's degree	25	Certificate in continence care	NHS hospital
8	55	Master's degree	12	Certificate in continence care	Community service clinic
9	56	Diploma Physiotherapy	24	Certificate in continence care	Private clinic
10	30	Bachelor's degree	3	Certificate in continence care	NHS hospital
11	56	PhD in Physiotherapy	20	Certificate in continence care	Community service clinic
12	45	Bachelor's degree	16	Certificate in women's health*	NHS hospital
13	55	Diploma Physiotherapy	25	Certificate in continence care	NHS hospital

Table 4-3: Demographic detail of KSA interview participants⁴

⁴ * Post-graduate diploma in pelvic rehabilitation is an intensive training from a USA institute on pelvic floor function, dysfunction, and treatment. It includes a pregnancy and postpartum course.

No.	Age	Qualification	Number of years working in UI	Post-graduate courses	Clinical settings
1	36	Master's degree	2	Post-graduate dinloma*	Tertiany Hospital
-	50		5		
2	26	Bachelor's degree	6	Post-graduate diploma	Tertiary Hospital
3	36	PhD level	10	Post-graduate diploma	Tertiary Hospital
4	35	Bachelor's degree	8	Post-graduate diploma	Tertiary Hospital and
5	46	PhD level	13	Post-graduate diploma	Tertiary Hospital
6	35	Bachelor's degree	7	Post-graduate diploma	Private clinic
7	26	Bachelor's degree	3	Basic courses**	Tertiary Hospital
8	26	Bachelor's degree	2.5	Basic courses	Tertiary Hospital
9	35	Bachelor's degree	6	Post-graduate diploma	Tertiary Hospital and
10	26	Bachelor's degree	3	Basic courses	Tertiary Hospital
11	35	Bachelor's degree	1	Advanced courses***	Tertiary Hospital

^{**} Basic courses on pelvic floor muscle assessment and treatment.

^{***}Advance courses on pelvic floor function, dysfunction, and treatment course in addition to physiotherapy's role in women's health.

^{****} Certificate in continence care is a post-graduate program in continence management for one year provided by one of the well-known universities.

No.	Age	Qualification	Number of years working in UI	Post-graduate courses	Clinical settings
12	26	Master's degree	10	Certificate in continence care****	Private clinic
13	38	Bachelor's degree	5	Basic courses	Tertiary Hospital and
14	30	Bachelor's degree	2	Basic courses	Private clinic
15	30	Master's degree	6	Post-graduate diploma	Tertiary Hospital

4.4 Focus Group Participants

Table 4.4 and Table 4.5 show the demographic details of the participants in the UK and KSA focus groups, respectively. Twenty participants attended the UK and KSA focus groups, three of the participants in the second UK focus group participated in both a focus group and an interview. Participant recruitment within the UK and KSA was quite challenging because pelvic health physiotherapy is an emerging discipline. As a result, the target number of participants was not reached, particularly in KSA. The original plans were to conduct 2-3 focus groups in KSA, but they needed to be modified because there is a limited number of pelvic health physiotherapists in KSA. As a result of this issue, the number of focus groups was decreased to only one.

Table 4-4: Demographic details of the two UK focus groups Table 4-5: Demographic details of the focus group participants in the KSA⁵

⁵ * Post-graduate diploma in pelvic rehabilitation is an intensive training from a USA institute on pelvic floor function, dysfunction and treatment, and postpartum course.

^{**} Basic courses on pelvic floor muscle assessment and treatment.

^{***} Advance courses on pelvic floor function, dysfunction, and treatment course in addition to physiotherapy's role in women's health.

No.	Age	Qualification	Experiences in UI management	Post-graduate Courses	Clinical settings
1	35	Bachelor's degree	6	Post-graduate diploma in pelvic rehabilitation*	Tertiary Hospital and private clinic
2	32	Bachelor's degree	8	Post-graduate diploma rehabilitation	Tertiary Hospital and private clinic
3	26	Bachelor's degree	3	Basic courses**	Tertiary Hospital
4	30	Bachelor's degree	6	Post-graduate diploma rehabilitation	Tertiary Hospital and private clinic
5	26	Bachelor's degree	1	Advanced courses***	Tertiary Hospital

4.5 Theme 1 - Contextual / Organisational factors

The data clearly identified that it is not only patient health factors that shape clinical reasoning; contextual factors including cultural context, resources and health system are important considerations. Participants discussed the impact of these factors on their assessments and treatment decisions. These are presented in Figure 4-2 and then explored in detail below.

Figure 4-2: Contextual factors impacting clinical reasoning/decision making



4.5.1 Cultural Context

Culture is defined as a group of people who share the same patterns of behaviour, such as language, communication, customs, beliefs, and values, in addition to how this behavioural response is essential to their health (Spencer-Oatey, 2008). This section will present the findings in two sections: first, the influence of the culture of individual physiotherapists and patients; then the wider context of social and religious culture. The supporting quotes are listed in Table 4.6.

4.5.1.1 Impact of individual culture on clinical reasoning/decision making

When referring to individual culture, this includes patients and their family or caregivers as well as physiotherapists. Focus group participants were asked to elaborate in more detail on the effect of culture on clinical reasoning and decision making processes, as the interviews revealed little about this area.

UK participants appeared more aware of multicultural diversity amongst patients and are therefore better able to identify cultural differences and appreciate their impact on clinical decision making. This may reflect the multiculturalism in parts of the country as a whole. It could also link to physiotherapists' own existing experiences in specific settings. For instance, some who worked in multi-diverse hospitals were able to recall different individual cultural issues.

KSA participants mentioned that their patients tended to be from the same culture, i.e., monoculture, but the main cultural influences on clinical reasoning were health illiteracy and the practice of family centred care. UK participants on the other hand, reported awareness of the effect of managing patients from multicultural background on their decision making by asking questions differently to overcome the language barrier.

Table 4-6: Themes include contextual factors, with a category on individual culture, and different subcategories derived from the data and reported by physiotherapists' participants 6

Category	Subcategory/	Quotes
Individual Culture. When referring to individual culture, this includes patients and their family or caregivers as well as physiotherapists.	Awareness of different cultures	FG2 UK (P5) "regarding culture and one of my patients is over the years, with female genital mutilation, that has an impact on treating a patient because there's often shame behind that. Particularly in cultures here where they feel "Oh if they realize that, people will think of me differently."
	Family centered care : Family gatherings involving drinking coffee were important among KSA patients. Involving the patient's partner or another family member in decision making can increase adherence to the management programme.	 P12 KSA "Our culture is a very group-based culture and it is a very family-based culture." FG1 KSA (P3) "When I ask them to try to do something new, I mean add some other herbs that will not affect their bladderdrink a cup of mint tea or any other herb, it is still a hot drinkand stay with your family".

⁶ P: Participants, FG: Focus Group, KSA: Kingdom of Saudi Arabia, UK: United Kingdom.

Category	Subcategory/	Quotes
	Patients' beliefs in KSA: Considering UI as a normal process of aging and after giving birth. Patients think that UI can be cured by herbal remedies.	P15 KSA 'Concerning herbal treatment, I try to talk to them on their level [patient's level of understanding] and tell them that it is an external treatment, like treating the area [vagina] from the outside in order to alleviate the irritation in the area [not decreasing UI]'.
	Language barrier: Patients speaking different languages, required interpreters. This service is limited and often available via telephone only. UK participants who identified with the same cultural background as their patients noted that it makes them more confident and more equipped to understand their concerns.	FG2 UK (P8) 'If I were going to see someone from an Asian culture, I feel like I would understand that better because I have worked with Asians and I have an Asian background.'
Social and religious aspects of culture: Contrary to the UK, religion plays a dominant role in KSA culture, which in turn influences healthcare delivery.	Asking questions differently when dealing with patients from different religious or cultural backgrounds, particularly when dealing with sensitive issues and topics.	P4 UK 'The Trust has a policy of asking for FGM, so we word that [ask questions]. We use the language differently according to who we are speaking to, and we use our common sense to advise us on how we will discuss it.'

Category	Subcategory/	Quotes
	Sensitivity about digital palpation: Participants in KSA also reported that they expect their single patients to be a virgin because women are only expected to have sexual intercourse once they are married. Participants in the UK were aware of this issue, but only concerning those who had never been in a sexual relationship before, such as teenage patients.	FG1 KSA (P5) 'We face some patients that are virgins, and we cannot do the internal examination on them. Also, younger women are often more open to the assessment and easier to communicate with than older women'.

4.5.1.2 An Impact of Heterogeneous culture in the UK

Participants in both UK focus groups were aware of the importance of cultural competencies while managing patients with UI. The participants in the first UK focus group were working with mainly white British patients whilst the second group were working with multicultural patients from different ethnicities. Some of the participants in the first UK focus group mentioned that they had barely any issues relating to culture while managing patients with UI due to working with monoculture patients. Participants in the second focus group however, reported different issues impacting their clinical reasoning and decision making while working in a multicultural area. For instance, language barriers, family centered care and patients' beliefs were mentioned. Table 4.6 explains each point and supports these with the relevant quotes.

UK participants living in culturally diverse places were aware of the significance of considering culture in their management decisions, based on their training, knowledge, and experiences of working within a multicultural context; this point is covered in detail in Section 4.5.1. Some participants in the UK also attended a cultural competency program.

However, they questioned the benefit of that programme in helping them deal with different ethnicities.

FG1 UK (P1) 'I think that it is left to us as individuals to understand the population and the cultural issues that are related to them. It will vary from workplace to workplace and depend on the demographics in that area, but I do not think that it is taught well within the NHS. We have a self-appraisal system and part of our appraisal system covers certain topics, such as equality and diversity.'

4.5.1.3 Homogenous culture in KSA

Conversely, according to KSA participants, culture had little to no effect on their decision making, as they believed they worked in a monoculture, with slight variations if the patients came from a rural or urban area. Patients' health illiteracy is a common issue in the KSA, and a further influencing factor on physiotherapists' clinical reasoning and decision making. This illiteracy is evident in patients' beliefs that UI is a normal aspect of ageing or after giving birth for example or thinking that herbal remedies can help cure UI. Physiotherapists mentioned that an aspect of their clinical reasoning was attempting to change patients' beliefs without being judgemental. Patients' beliefs are also influenced by socio-economic factors, covered in detail in Section 4.7.

In KSA, participants mainly used their intuition and awareness of their own culture by considering patients' beliefs and the importance of family-centred care only. However, KSA participants who had worked in different cultures were able to understand the effect of culture in decision making in a more explicit way (Table 4.6).

Physiotherapists in KSA used their intuition to manage their patients from the same culture. For instance, they were aware of the importance of family gatherings and drinking coffee, while simultaneously wanting their patients to maintain a healthy diet with a low caffeine intake. They encouraged their patients to drink alternative warm drinks instead of coffee while maintaining their social gathering traditions (Table 4.6).

4.5.1.4 Impact of social and religious factors on decision making

Compared to the UK, religion plays a dominant role in KSA culture, which in turn influences healthcare delivery. In the UK however, the religious background of patients does influence how physiotherapists decide to provide healthcare to certain patients. The findings suggest that some patients from different ethnic groups or religious backgrounds might be sensitive to digital palpation, and the way physiotherapists ask questions might vary due to different beliefs or understandings. In addition, UK physiotherapists integrated patients' spiritual rituals in their clinical reasoning (Table 4.6).

The findings identified that whilst religion does play a role in clinical reasoning and decision making, it should not be singled out as a key determining factor, as even among Muslim population, religious preferences will still vary. Effective clinical reasoning and decision making is therefore more about understanding what is important for an individual patient and finding an optimal treatment solution that works, while also respecting the patient's beliefs.

The data clearly shows that the cultural context is a broad but influential factor in physiotherapists' clinical reasoning and decision making. Lack of knowledge or awareness regarding a patient' different culture or influence of culture on that patient (especially in a multicultural context such as the UK) is a concern since it might influence the patient's outcome. This is especially the case if physiotherapists are working in a certain organisation that has different policies. This issue of organisational culture is discussed in detail in a later section.

4.5.2 Healthcare system/ Organisational culture

The data identified that UK and KSA healthcare systems are quite similar in terms of the availability of primary care (GP) and secondary or tertiary care hospitals with more advanced specialities. Physiotherapists are only available in territory hospitals in the KSA but are available in both primary and secondary care services in the UK. Physiotherapists are available in private settings in KSA and the UK in both NHS and private primary and secondary care services.

The result of data analysis identified that culture within an organisation may be related to the way health practitioners, patients and administrations behave, act, communicate with colleagues and make decisions. A country's religious background, patients' beliefs, physician's hierarchy in decision making, family-centred care and health practitioners' identity shape the culture within an organisation in the KSA. The organisational culture varied between two organisations in the KSA, seemingly based on hospital guidance and the socio-economic background of their patients. For instance, the participants working in university hospitals mentioned that they follow international clinical guidelines compared to other military hospitals with a big research centre where they created their own policy and guidance. The data identified that the UK's organisational culture had clear guiding principles that lead participants to make similar decisions to one another, but this was the opposite in KSA. The key differences between the UK and KSA's organisational culture were limited national clinical guidelines and teamwork in KSA, physicians' hierarchy and socio-economic factors. These led to a sense of uncertainty among KSA physiotherapists' decision making compared to the UK physiotherapists. This suggest that organisational culture is an influential factor in decision making. The participants' quotes are listed in Table 4.7.

Table 4-7: Themes include contextual factors, with a category on individual culture, and different subcategories derived from the data and reported by physiotherapists' participants6⁷

Category	Subcategory	Quotes
Healthcare system/ Organisational culture: The culture within an organisation may be related to the way health practitioners, patients and administration behave, act, communicate with colleagues and make decisions.	Limited national clinical guidelines: KSA participants used clinical pathway or guidance based on highly available research evidence and agreement from referring physicians to guide their decision making. They used international clinical guidelines.	P1 KSA "I use NICE guidelines for UI and American physiotherapy for post and ante-natal patients" KSA FG "We created our own clinical guidance based on high quality research and hospital policy; it will be nice to have our own guidance [national guidelines]."

⁷ P: Participants, FG: Focus Group, KSA: Kingdom of Saudi Arabia, UK: United Kingdom. PHPTs: Pelvic Health Physical Therapists. DM: Decision making, UI: Urinary incontinence. Niqab: is a face cover used by some Muslim women for religious reasons.

Category	Subcategory	Ouotes
	Limited multi-disciplinary team: a lack of teamwork and communication within the multi- disciplinary team, leading to KSA physiotherapists feeling lost and unable to find answers. This was not the case among the UK physiotherapists.	 P1 KSA "A lack of multi- disciplinary teamwork is an issue, so the urologist sees the bladder point of view and the surgeon from the surgical point of view." P9 UK "If I face any difficulties, I usually ask my colleague. We have good, understanding consultants and nurses, and one of them is a specialist Uro-gynecologists Also, I can research on the internet"
	Physicians' hierarchy and limited awareness of physiotherapists: the hierarchy of physicians and their limited awareness of physiotherapists' role influenced the culture within the KSA healthcare service.	P15 KSA "I gave lectures to the gynaecology team, and this was the reason that I did not receive any patients. They did not believe in it, and they just gave the patients instructions and medications".
	Organisational culture may affect KSA physiotherapists' decision making, and some participants were keen to prove the significance of their management programme to change the physicians' and patients' view of physiotherapists' role and identity.	FG1 KSA (P3) "Some of the doctors don't feel that physical therapy is playing a role especially with the Obstetrics and gynaecology but for the Uro- gynecological they refer all the patients to us. But for the Obstetrics and gynaecology they do not understand what we will do, and they always prefer surgery for those patients, and they don't believe in us."
	Physiotherapist administration in KSA may play a significant part in providing financial support and equipment for physiotherapist management.	P1 KSA "I combine the plan of care something active with passive, the passive thing increases adherencethe evidence is not very supportive to the electromagnetic chairbut still it makes patients come and we want the patients to come. We have let say a contract between physio and urology, after threemonthsfollow-up

Catagory	C-hastagen	Orester
Category	Subcategory	appointment in the urology it should be as scheduled, as the physical therapy have set their plan."
		FG1 KSA (P3) "You're not well supported by your admin, and they are not providing you a supply, area and equipment and staff. You cannot treat all those patients, so for me it was a big factor, the priority for the neuro and Ortho cases because they are the majority of our patients."
	Socio-economic factors: related to urban and rural areas, transportation and limited specialised services in rural areas with significant deprivation, especially in KSA. Limited awareness of physiotherapists' role was a common issue in the two countries,	P4 KSA "In the military hospital, some of my patients can not adhere especially if they are not from the "big cities". If they have transportation issues, some patients said give us the program and we will try to do it at home."
	which led to high or low expectations from physiotherapy.	FG1 UK (P8) 'Sometimes, they have no expectations because they have no idea about what we can offer. Sometimes, they come with high expectations Therefore, it might be a lack of understanding or a reluctance to talk about it'.
	KSA physiotherapists uncertain in decision making: Uncertainties regarding decision making when receiving self-referral within private settings. KSA physiotherapists were also possibly unsure of follow-up treatments due to limited clinical guidelines, and sometimes ended up changing the management programme without proper justification. Some KSA participants who worked with elderly and illiterate patients, usually faced disapproval with regards to vaginal examination. This created uncertainties when it came	 P6 KSA 'The Saudi healthcare system does not support self-referrals. However, the patients are still allowed to come to private clinics using self-referrals. I am not happy to receive patients having a contagious disease. I prefer to rule them out because I like to do dry needling. But I would not like to send her back because it is unlikely that she'll seek help again.' P9 KSA "she was very old lady from a village, during the session, she even refused to take off her Niqab. Some of them [patients] were very

Category	Subcategory		Quotes
	to recommend	appropriate	religious, they refused it [vaginal
	management options.		examination] from a religious point
			of view. [Some of them] they refused
			totally, from a religious background
			or from village or very old patients."
			P12 KSA 'If that physiotherapist is
			not confident or comfortable, then
			the patient will pick up on that and
			might not accept the vaginal
			examination'.

UK health care was reported to be more uniform with national guidelines, and to consist of more consistent policies and processes. It was clear that UK physiotherapists' clinical reasoning was influenced by this more unified guidance. This meant their practices appeared more standardised across the UK, whereas in KSA, policies and procedures, and therefore clinical reasoning, varied across regions. The data suggests that the organisational culture in which the participating KSA physiotherapists operate, is influenced by individual patients and religious culture, communication within a multi-disciplinary team and physicians' hierarchy. In contrast, in the UK, it is shaped by individual culture, National Institute for Health and Care Excellence guidelines, communication within a multi-disciplinary team and patient-centred care. Resources shaped the organisational culture in both countries. This is discussed in the following section.

4.5.3 Resources

The data identified clear impact of the available resources on physiotherapists' clinical reasoning and decision making. Resources may include, but are not limited to the availability of equipment, management time and waiting lists.

4.5.3.1 Influence of the availability of assessment and treatment options on decision making

Participants from both countries mentioned limited treatment and assessment options for patients with urinary incontinence in physiotherapy in general. Many potential treatments were not always available to prescribe or offer to each patient.

KSA participants mentioned that they regularly used equipment such as electrical stimulation, a functional magnetic chair and biofeedback. They reported that most patients were referred to their service as a last resort instead of the first line of treatment. These patients may have been suffering from severe weaknesses, resulting from pelvic floor muscles; this may relate to the physicians' lack of awareness of the physiotherapists' role, which in turn could contribute to referring patients as a last resort option. Some physiotherapists in KSA reported that their patients were sometimes passive and preferred equipment instead of active exercising. Despite this, physiotherapists usually provided patients with equipment in addition to exercise to increase their adherence and confirm that they are following the treatment programme to see improvements.

4.5.3.2 Differences between the UK and KSA in terms of resources

Physiotherapists in KSA reported that they may provide their patients with up to 12 sessions, and sometimes more, depending on their patient's conditions and hospital policy. Patients may only attend their sessions to receive electrical stimulation, biofeedback, or Posterior Tibial Nerve Stimulation (PTNS) Some of the equipment, such as vaginal cones and dilators, are not easily found in KSA and some patients were reluctant to use them due to religious culture.

On the other hand, the UK physiotherapists mentioned using equipment less frequently, based on the NICE guidelines. For instance, sometimes they use educators as a biofeedback tool and other affordable equipment that patients could purchase or borrow themselves, in some situations, such as very weak pelvic floor muscles, or to improve patients' motivation to continue doing the management program.

P1 UK "It's nice to know about everything that's available [equipment] and, if people ask about it or if you think this woman's not progressing or she could do with something to motivate her or she wants a gadget."

UK participants who worked in private settings sometimes suggested that patients purchase equipment, while the NHS had biofeedback units in a clinic, and had stimulators which women could take home with them, with vaginal cones were given out free of charge, in case they were needed. This varied from one NHS hospital to another.

UK participants mentioned that their patients preferred to do pelvic floor muscle exercises rather than using equipment. They reported providing up to six sessions because UI was considered a chronic problem, which required self-management, as suggested by National Institute for Health and Care Excellence guidelines. There was variation between UK participants when it came to their ability to provide more than six sessions. This depended on their clinical settings and patients' conditions. Participants who worked in private settings mentioned that they could not exceed four to six sessions due to insurance requirements. Another participant, working in the NHS, mentioned a lack of control over her diary, with patients sometimes waiting up to eight weeks until the next available appointment. This could affect the patients' conditions, such as vaginismus and UI, where the physiotherapist needs to build rapport before conducting the vaginal examination. Some UK physiotherapists stated that doing urinalysis was a routine practice to rule out Urinary Tract Infection (UTI) and haematuria in postmenopausal age. This helped in providing certainty and confidence to proceed with vaginal examination and the pelvic health physiotherapist management process rather than having to wait for the urine lab test result. In contrast, KSA participants stated that they did not do urinalysis routinely, which could have led to uncertainty and delays in patients' management (Table 4.7).

Resources play a significant part in decision making and clinical reasoning process. The resources in the UK and KSA vary based on organisational and individual culture, clinical guidelines, and the severity of patients' conditions, as well as the need to establish rapport. The next sections consider pelvic health physiotherapist and patients' factors.

4.6 Theme 2 – Pelvic health physiotherapist's factors

Several physiotherapists-related factors were identified as influential to decision making and clinical reasoning when assessing and treating patients with UI, including (1) multifaceted knowledge based on postgraduate level of training and experience; and (2) emotional intelligence. These primary and secondary factors were explicitly discussed in the interviews and focus groups and are shown in Figure 4-3 below.

Figure 4-3: Factors influencing physiotherapists' decision making and clinical reasoning. PG: postgraduate, DM: decision making, CR: clinical reasoning.



4.6.1 Physiotherapists' multifaceted knowledge, training, and experiences

Despite the importance of this training, physiotherapists only commence training in this specialism at the PG level, even though physiotherapists acknowledge the significance of theoretical and practical knowledge. Within the literature and from the participants' responses it was clear that physiotherapists view their knowledge and experience as unique because of the sensitivity of the UI diagnosis, which can depend on several elements. This includes conceptualisation and implementation of EBP, and reflection-in action skills. Notably, participants from both countries reported that this results from their experience of doing postgraduate training; this means there may be differences in participants' decision making based on their levels of experience and PG training. This will be explored in detail in the next section. A description of each category and subcategory along with relevant exemplar quotes are listed in Table 4.8 below.

Table 4-8:	Theme	includes	physiother	apists'	factors,	with a	category	on phys	siothera	pist's

Category	Subcategories	Quotes
Physiotherapist's multifaceted knowledge, training and experiences: physiotherapists' knowledge and experience were unique because of the sensitivity of the diagnosis; it depends on an art of PHPT PG training, reflection skills and conceptualisation of EBP	Attending PGT: gives novice physiotherapists confidence and advance her clinical reasoning skills. Expertise and decision making and clinical reasoning: Expertise in physiotherapy is a combination of multi-faceted knowledge and the ability of the experienced physiotherapists to recall the similarities in patients' current situation compared with previous ones.	P11 KSA Novice physiotherapist's who attend PGT "Attending advanced courses in pelvic rehabilitation, gave me the confidence of judgement, especially for POP and different level of pelvic floor muscle contraction."
	Conceptualisation of EBP	 P10 UK (experienced mentor with PGT) "I think we accept that's [EBP] not real in terms of what we do in clinical practice so actually the research is more substantial if you have someone [colleague] that's experienced doing a pragmatic approach with that patient." P9 KSA attended introductory PGT "I started with pelvic floor muscle training and bladder re- training for 6 sessions, then I started posterior Tibial nerve stimulation for 12 sessions then biofeedback. I used articles to support my treatment plans"

⁸ P: Participants, FG: Focus Group, KSA: Kingdom of Saudi Arabia, UK: United Kingdom, PHPTs: Pelvic Health Physical Therapists, DM: Decision making, UI: Urinary incontinence, EBP: Evidence based practice, USA: United States of America, PGT: Post graduate training, POP: Pelvic Organ Prolapse, PFM: Pelvic Floor Muscle, PFMT: Pelvic Floor Muscle Training and CR: Clinical reasoning.

Category	Subcategories	Quotes
	Influence of PG training on decision making	FG1 UK (P1)" I think [that], in the UK, there's not much undergraduate training for pelvic health, so sometimes people come into your rotation, and they've actually never done anything. And it's from the professional guidelines from the Chartered Society of Physiotherapy that vaginal examination, it is a skill that is learned during postgraduate training".
	Hesitancy in decision making: KSA physiotherapists who attended introductory PG courses and had minimal experience were quite hesitant to make decisions and usually referred patients for further investigation to overcome any uncertainties or to clarify the diagnosis.	P7 KSA "If the patient needs to do the urodynamic or to double- check the diagnosis, I will send the patient to the urology and suggested doing urodynamic study."
	UK physiotherapists confident in decision making: UK participants were confident in making management decisions based on national clinical guidelines. These instilled them with confidence with regards to deciding on the appropriate management, number of sessions, possibilities of getting self-referral and progressing the treatment plan. They were also confident in saying they could not offer something else and discharge patients.	P4 UK "I am confident enough to say to the patient who is confusing me, this is a new one, and I have to go away and find a friend and come back to you."
	Impact of reflection on decision making and clinical reasoning: differences between the UK and KSA participants' reflection levels	P1 UK "We do the practical [vaginal] examinations on each other so. It is that's a very personal experience, and you learn from reflecting on that with them." P10 KSA (attended PGT in the USA) "I am single, and It was kind

Category	Subcategories	Quotes
		of my colleague in the USA to let me do the practical [digital palpation] examinations on them"
4.6.1.1 Differences in th	e decision making and clinical	reasoning capabilities of

4.6.1.1 Differences in the decision making and clinical reasoning capabilities of participants in relation to introductory and advanced post-graduate training and experiences

All the participants in this study attended either introductory postgraduate (basic courses) or advanced PG courses in pelvic rehabilitation, with the majority from both countries holding postgraduate courses. A small number of participants from KSA attended introductory postgraduate courses only and had limited years of experience. It was evident from the participants' responses that a combination of both factors, rather than a single factor, may have affected their decision making and clinical reasoning.

During both focus groups discussions, it was evident in terms of the group dynamics that novice physiotherapists were less confident about their decision making and clinical reasoning when compared to more experienced and senior colleagues. Physiotherapists from the UK and KSA with advanced experience, training, or a postgraduate certificate, such as an MSc or PhD, showed a deep understanding of pelvic health rehabilitation practice and were more able to articulate the factors they considered in decision making and clinical reasoning. In addition, an understanding of evidence-based practice (EBP) enabled them to overcome any uncertainties and influenced their daily clinical decisions. For example, a UK participant who attended several advanced postgraduate courses, and held a lead position, found that with experience and knowledge, she did not get stressed if she did not know something or was uncertain, and she felt confident in telling her patients that she had not come across this before Table 4.8.

The UK participants also reported that high quality research evidence to provide clear and pragmatic information on correct exercise intensity and the number of repetitions of exercise per day was limited, which may indicate that the available evidence is not applicable within clinical practice.

4.6.1.2 Influence of postgraduate training on decision making

Pelvic health physiotherapy is a rare speciality in both countries. In KSA, most of the physiotherapists were junior and lived in bigger cities. Some of the UK physiotherapists had many years of experiences with limited up-to-date postgraduate training. These participants might make their management decisions based on routine practice using previous experiences with similar cases. On the other hand, one of the novices KSA physiotherapists who attended advanced courses in the USA was very confident to talk about her patient and gave detailed assessment on pelvic floor muscle and Pelvic Organ Prolapse (POP). Another UK interviewee mentioned that attending postgraduate courses influenced her decision on the timing of digital palpation. Her tutor mentioned that doing digital palpation in the second session can help patients understand the reason for the examination, and as a result, the patient can be more prepared. This positively enhanced the practice in terms of patient experience and the clinic changed their policy to reflect this. It was clear that physiotherapists' level of postgraduate training influenced their level of knowledge and confidence in their decision making and clinical reasoning. Furthermore, participants suggested that advanced PG training played a more significant role than years of experience. EBP also played a significant role in physiotherapists' decision making and clinical reasoning, but participants from both countries questioned the applicability of it within clinical practice. Some KSA physiotherapists reported that their reason for using certain management decisions was because they were following highly available research evidence rather than following EBP, thus excluding their patients' opinion -a key component of EBP. Sackett et al. (1996) reported that EBP involves a combination of highly research evidence, with the health practitioners' experiences and patients' opinion, while some KSA physiotherapists stated that they did not consider patients' opinions.

4.6.1.3 Impact of reflection on decision making and clinical reasoning

The participating physiotherapists' reflection skills influenced their decision making and clinical reasoning processes. In the UK, participants reported performing vaginal assessments on each other in introductory or advanced postgraduate courses in pelvic health rehabilitation. This provided them with an opportunity to reflect on their experience, highlighting reflection as an important tool for learning and advancing knowledge, and in turn, affecting decision making. Despite this, the data from the current study suggested that

it is a combination of years of experience and postgraduate training that primarily affects decision making.

In contrast, KSA participants had limited access to advanced postgraduate courses in pelvic rehabilitation. This may be due to the nature of digital palpation itself or religious culture. Furthermore, it may be unacceptable for physiotherapists in KSA to practise on each other. This limits access to advanced postgraduate courses for KSA physiotherapists, as they are not provided with the opportunity to reflect on their skills, which in turn, may affect their decision making.

In summary, physiotherapists' knowledge, level of postgraduate training and years of experience play a vital role in their assessment and treatment decisions. It appeared that participants who attended advanced postgraduate courses and had vast clinical experience reported a deeper understanding of UI and engaged in reflective practice more than those who had not. There were differences between KSA and the UK when it came to decision making and clinical reasoning while managing patients with UI. This might be due to the level of postgraduate training, limited reflection skills, and individual or religious culture. Participants from both countries stated that they used readily available evidence to make decisions regarding treatment and to manage patients' conditions. The way they interpret the evidence, however, appeared to be somewhat different. As pelvic health physiotherapist is a relatively new speciality in the KSA, it is not surprising that they rely on evidence rather than clinical experience to make decisions. Very few KSA participants mentioned the importance of reflection, which may be due to the structure of their undergraduate training. In contrast, UK physiotherapists appeared to use their reflective skills, as well as clinical experience, and knowledge when making decisions.

4.6.2 Physiotherapists' Emotional Intelligence

Participants from both countries showed emotional intelligence (EI), which might be related to their unique personal characteristics, life experience and level of postgraduate training. EI may include being empathetic and providing a safe environment that help patients to disclose ethical and sensitive issues (Marcum, 2013).

P1 UK "I think sometimes you just need to have a bit of empathy with people"

The elements that demonstrate physiotherapists' emotional intelligence and influence their ethical and sensitive decision making and clinical reasoning will be discussed in detail with the relevant quotes (Figure 4-4).

Figure 4-4: the elements that demonstrate physiotherapists' emotional intelligence and influence their ethical and sensitive decision making and clinical reasoning



4.6.2.1 Pelvic health physiotherapists' personal characteristics and life experiences

According to physiotherapists emotions were a key part of decision making. Participants mentioned that their patients disclosed sensitive and ethical issues to them due to the amount of time spent with them (follow-up sessions can be up to thirty minutes in length), and due to personal characteristics. This includes EI, which can help to build trust and rapport, and differentiate them from physiotherapists in other areas of practice.

The longer duration of physiotherapy sessions enabled patients to disclose issues, such as faecal incontinence and sexual abuse, that they did not mention to the referring physician – perhaps due to embarrassment or the perception that it is unrelated. Participants stated that trust was important as some patients felt inadequate and were only able to talk about their problems with someone they trusted. For instance, one participant mentioned that her patient

had vaginismus, which is associated with UI. She had this problem for a long time, and the pelvic health physiotherapist was the first one to discover that sexual abuse was behind this problem. This suggests that physiotherapists' personal characteristics influence their decision making and clinical reasoning. The following section goes into more detail on the specific interaction skills used by physiotherapists and its significance for decision making and clinical reasoning.

4.6.2.2 Pelvic health physiotherapists' interaction skills

The participants used specific interaction skills to guide and teach patients. The uniqueness of pelvic floor muscle anatomy and function required participants to spend time in their first session with the patient, explaining the relationship between pelvic floor exercises and UI; this required unique communication skills due to the sensitivities of UI and the complexities of pelvic floor muscle, which are discussed in the following sections.

4.6.2.3 Communication between physiotherapists and patients

The majority of participants from both countries considered specific communication styles to be significant to decision making and clinical reasoning. In particular, it was essential to maintain eye contact and talk to the patient like a friend by considering the patient's emotions and using their own words. Several UK participants working in private NHS and government hospitals reported providing patients with their email address; in KSA, some participants provided 'professional' phone numbers to patients in order to enable patients to ask questions or report any concerns at a later date or between appointments. This facilitated an open line of communication.

4.6.2.4 Complexity of pelvic floor muscles and lifestyle modification

Participants from both countries commented that it was challenging to explain pelvic floor muscle exercises to patients, which differs quite significantly from MSK problems, as patients with urinary incontinence cannot observe the movements required. Patients might require several sessions, or they may need biofeedback or analogies and metaphors to explain the exercise adequately, especially with less educated patients or the elderly. These options entailed different forms of decision making and clinical reasoning, such as narrative reasoning.

From the focus groups discussions with the participants, it was clear that interpersonal skills played an essential part in physiotherapists' decision making and clinical reasoning. Specifically, participants reported that lifestyle modification helped their patients to follow healthy bladder behaviour, such as voiding time, frequency, healthy drinking, eating habits, in addition to addressing some wrong beliefs. Physiotherapists' ability to interact with patients to increase their understanding to contract pelvic floor muscles correctly as well as lifestyle modifications were reported to result in improved patient adherence and outcomes. Physiotherapists' unique communication and interaction skills allowed them to provide a personalised management plan, which in turn, affected their decision making and clinical reasoning processes.

4.6.2.5 Variations in managing ethical or sensitive issues between the UK and KSA

Physiotherapists from both countries reported that their patients sometimes disclosed sensitive or ethical issues. It might have depended on physiotherapists' personal characteristics or hospital policy, which led to encourage or discourage patient disclosure. Some of the KSA participants reported that they did not encourage patients to talk about sensitive issues because they did not know how to handle it (Table 4.9).

Others did encourage patients to talk about sensitive topics. In the UK, participants had clear clinical pathways to manage ethical issues. In KSA, it varied from one hospital to another.

Most of the UK participants reported that they discussed the decision making and clinical reasoning of sensitive or ethical cases with a senior colleague or referred these to one of the safeguarding teams. In KSA, physiotherapists reported different decision making and clinical reasoning due to unclear clinical pathways to follow. KSA physiotherapists reported either referring the patient to a social worker or psychologist or signposting them to a domestic abuse hotline in line with the hospital policy. Other hospitals would require them, with the patient's permission, to document the incident and refer them to the emergency department. In KSA, patient resistance regarding disclosure was witnessed and could be due to cultural issues surrounding shame. This may be related to the fact that abuse can be considered to bring a family shame. Therefore, it is apparent that in the UK, physiotherapists had clear uniform clinical pathways to guide their decision making and clinical reasoning, whereas in KSA, physiotherapists had a lack of clear guidance, and they often had varied hospital policies, which in turn affected their decision making and clinical reasoning.

Participants from both countries mentioned that sometimes, patients did not like to formally document their problem in their medical file or share their issues with a professional body, such as their GP or the safeguard committees. This could be due to the nature of the problem; the victims could be worried that others would be influenced by it, and as a result they tried to hide the problem and suffer in silence. The difficulty in documenting ethical or sensitive issues, along with related pathological or psychological problems, could affect physiotherapists' decision making and clinical reasoning and may restrict their ability to work in multi-disciplinary teams. Participants from both countries were in contradiction regarding the influence of patients' sensitive and ethical stories on their own wellbeing and when making decisions. This will be explored in the next section.

 Table 4-9: Theme involving pelvic health physiotherapists' factors/ Emotional intelligence

 and description of subcategories89

Category	Subcategories	Quotes
Physiotherapists' Emotional Intelligence: this may include being empathetic and providing a safe environment that help patients to disclose ethical and sensitive issues.	Physiotherapists' personal characteristics and life experiences: a unique set of skills based on their experience and training, including: (1) tentativeness, (2) active listening, (3) empathy, (4) rapport building, and (5) providing a safe environment. Physiotherapists reported that these factors were influenced by their life experiences, such as their age, marital status, being pregnant or having children.	 FG1 UK (P8) "You have to be open- minded and non-judgemental; the key difference is empathy, and you need the ability to communicate and not be embarrassed easily." FG1 UK (P6) "I had an experience with the patient once who wouldn't do anything I said and she said, 'one day you'll know how I feel because you do not have had children'. And I said, well I have had children, I've got three of them. And suddenly, for her, she seemed to relax a bit and that made me more comfortable being around her."

⁹ PHPTs: Pelvic Health Physiotherapists, FG: Focus Group, UK: United Kingdom, P: Participant, KSA: Kingdom of Saudi Arabia, PFMs: pelvic floor muscles

Category	Subcategories	Quotes
	Differences between the UK and KSA: KSA participants reported that patients might find it difficult to discuss sexual disorders with single physiotherapists. This was not an issue for participants in the UK. In the UK, older physiotherapists were more positively perceived by patients than younger physiotherapists.	P5 KSA "Patients might find it difficult to talk about their sex life, and the first question they would ask me was, whether I was married. If I said yes, they would feel comfortable enough to talk about their sex life." P10 UK "One of my male patients said, 'what is a lovely young lady like you doing putting your finger up men's bottoms for a living?'"

Category	Subcategories	Quotes
	Physiotherapists' interaction skills: Communication between physiotherapists and patients	P4 UK "I rely on talking to them, hearing them and caring about them. I give them my email and work on my instinct rather than anything precise. Being interested in patients will validate their symptoms and make them feel differently."
	Complexity of PFMs and lifestyle modification: It is difficult for patients to visualise PFMs.	FG1 KSA (P4) "[A challenge is] educating patients on how to perform the exercises properly. In other specialities, you can show them how to do the exercises, but in our speciality, it is challenging to explain it to them. Therefore, it needs more than one session, and they might need to use a mirror or a finger as feedback".
		FG1 KSA (P2)" I explained that their body is like a hammock and that they need to support it." FG1 KSA (P4) "I told a patient that it is abnormal to go (to the toilet) during the night more than once because she was going three or four times. However, I found out that she was taking many fluids before bedtime. [Teaching patient lifestyle modification]."

Category	Subcategories	Quotes
Physiotherapists' Emotional Intelligence	Variations in managing ethical or sensitive issues between the UK and KSA	P1 KSA "I never ask questions that I do not know what to do with their answer."
	Unclear clinical pathways to manage	
	ethical or sensitive issues in KSA	FG1 KSA (P3) "We have committees for abuse, and there is a number in our hospital that you can use to contact them. I faced that once with my patient: her husband abused her, and I asked her whether she wanted me to deal with the problem. I said that I could give the social worker her number, but she refused"
		P3 KSA "Patient said that she had been abused for 5 years old. She could not give detail because she is having many influences that would really put her in a big trouble if she talk about it. [Regarding documenting that] no, because I have not been allowed, I have not been given the permission from her side to talk about it."
	Using other colleagues to solve ethical dilemmas in the UK: Using safeguarding teams and following a clinical pathway to manage these problems.	FG2 UK (P4) "There is an element of what happens in the room remains confidential in the notes. However, we also have a duty, and I think we are much more likely than any other physiotherapy group to be aware of that. We are the first protocol for that: for something like domestic abuse."
Category	Subcategories	Quotes
----------	---	--
	Novice Physiotherapists and patient relationships: KSA novice participants reported having a formal relationship with patients, rather than a mutual one. Furthermore, novice physiotherapists commented that they were concerned about patients sharing additional unrelated problems that fell outside of their practice.	P8 KSA "To follow my instructions, my exercise program, to ask and cooperate with me if she has something, to understand what I will do and why I will do that because understanding and being part of the treatment and then understanding what will happen and why we will do that, it is a main rule."
	Impact of patient disclosure: patients disclosed sensitive and ethical issues to physiotherapists due to different factors.	P10 UK "She disclosed that she had been used within a drug circle where she had grown up and her ex- husband had abused her she had never discussed it with anyone before."
	Physiotherapists' own wellbeing: participants expressed intent to empathise with their patients, whilst at the same time trying to manage their own emotions and prevent emotional trauma. One participant described an experience with two known patients.	FG2 UK (P3) "It all became murky because I knew what was happening. Someone confessed it to me, and then I knew that the other one was being cheated on, and it was dreadful. People do say things that are sometimes intimate, and it can have an impact on us."

4.6.2.6 Impacts of patient disclosure

This subcategory also includes consideration of how physiotherapists manage their own emotions. When participants were asked how they dealt with emotions after being exposed

Chapter 4

to their patients' sensitive and ethical stories, most of the participants in KSA and in the first UK focus group said that keeping professional boundaries helped them to leave everything related to the patient at the clinic door. For example:

FG1 UK Senior PHPT (P1) "So if there is somebody who disclosed, sexual abuse to me, if I stopped and thought about it, yes, it could hugely impact on me. But if I know that I've been able to signpost them in and arrange for them to have some follow up contact that they've agreed to, I can close it off my perspective. Dealing with sort of the relationship between you and the patient would be the professional boundary. But in terms of us mentally, for me, personally, it gets left at the clinic door."

In the first UK focus group, the participants appeared to show unanimous agreement with a more senior pelvic health physiotherapist in the group, who stated the importance of maintaining a professional distance. However, it was possible that participants may have held a different opinion but felt reluctant to raise this due to the circumstances and not wanting to disagree with a more senior pelvic health physiotherapist. The body language of some of the participants showed hesitancy to speak about this issue (for more information on the focus group homogeneity see Section 3.6.3. This hesitancy to admit the impact that patients' stories can have on them personally may have been due to a workforce dynamic with participants unwilling to disagree or show their vulnerability in front of a more senior pelvic health physiotherapist. Only a few experienced physiotherapists admitted that they were influenced by patients' emotional stories from the past when they were novices. They further explained that through experience they had become desensitised, so that hearing an emotional or sensitive story no longer affected them too greatly.

FG1 UK (P3) "It definitely gets easier. I find that at the start, it's very hard and it gets easier, I think once you get more experienced to kind of leave it at the door."

However, in the second focus group, there were some conflicted opinions between physiotherapists with regards to the influence of patients' vulnerability on their own wellbeing and decision making. Some physiotherapists stated that they were affected by patients' disclosure. The quotes of the participants are listed in Table 4.9 above. Differences in experience levels may explain the variation between participants in both countries; the lack of transparency from participants in the focus groups due to seniority issues may also play a role. These differences could be due to individual cultural issues in KSA in addition to the lack of clear and uniform clinical pathways. Participants who shared the effect of patients' disclosures on their personal or emotional life, highlighted the importance of formal and informal debriefing, which varied across the UK and maybe do not exist in KSA.

Most of the novice physiotherapists in both countries created professional boundaries to protect themselves from emotional distress and to ensure they had enough time in each session to treat and manage the patient. In the second UK focus group, participants admitted that patients' vulnerability could influence their own wellbeing, decision making and clinical reasoning.

This highlights that physiotherapists were not only doing cognitive work, but also doing personal work. More specifically, participants were required to think about how to manage the patient's emotions to help them achieve their own goals, while simultaneously managing their own feelings as well as the patient outcome targets. This could add additional pressure on physiotherapists to manage patients' time and caseload and may present challenges which in turn may impact physiotherapist decision making and clinical reasoning. In addition, it could influence the way physiotherapists strike a balance between being emotionally available to form a bond with patients that enables more effective treatment, and not taking too much of the emotional burden of patients on so that it negatively impacts on their own wellbeing, decision making and clinical reasoning.

Patients' different characteristics could also influence their relationship with physiotherapists and decision making; this will be discussed in the next section.

4.7 Theme 3: Patients' factors

The participants in the focus groups and interviews identified the effect of UI on the patients' lives and explained how this could influence their decision making and clinical reasoning. Although it would be preferable to ask patients with urinary incontinence for their opinions directly, it was challenging to recruit patients from KSA due to the

sensitivity of the UI topic. As a result, the patients' factors discussed in this study have been collected and are based on the reports and statements of physiotherapists and not patients themselves. Hence, the participating physiotherapists in both countries reported the effects of the patients' factors on their assessment and treatment decision. Physiotherapists reported that the patient's behaviour, beliefs, and characteristics are essential factors that can affect the physiotherapists' assessments, treatment decisions and actions. Patients' beliefs and behaviours were covered previously in the individual culture section 4.5.1.

Patients' factors that influence physiotherapists' decision making and clinical reasoning shown in Figure 4-5 varied depending on the patients' demographics, such as age and gender, as well as comorbidities associated with UI, such as diabetes and hip osteoarthritis. Additionally, patients' cognitive and psychological characteristics were also considered during decision making and clinical reasoning. Also, the participants in both countries mentioned the relationship between understanding patients' factors and projecting how those characteristics would influence UI management decisions. Physiotherapists stated that UI had unique features, such as chronicity, that required self-management and required patients' compliance with suggested daily activities, such as exercise and lifestyle modifications.

Figure 4-5: Patients' factors may include patients' demographics, associated morbidities, uniqueness of UI and socio-economic features



4.7.1 Patients' demographics and socio-economic features

The participants from both countries mentioned that they treated patients with different demographics such as age, gender, married status, which could affect their decision to conduct a digital palpation. Socio-economic factors can also affect decision making and clinical reasoning, although this was mentioned specifically by the KSA participants. It was related to the variety of regions from which patients originated and included elderly people with low levels of education as well as more traditional and conservative values and beliefs. For instance, KSA physiotherapists mentioned the need to take into account within their clinical reasoning that they could only do external examinations to assess pelvic floor muscle contractions using specific bony landmarks on some elderly patients from rural areas. As a result, the participants tried to simplify the instructions to ensure that the patients understood them.

P10 KSA 'I tried to simplify the instructions and encourage the patient to use their imagination when I prescribed the exercise, especially with the elderly'.

4.7.2 Acceptance of digital palpation

As mentioned, physiotherapists in KSA stated that some of their older patients from rural areas did not accept digital palpation and had low adherence to the management programmes. While some KSA physiotherapists chose to adopt a more prescriptive, less flexible approach, other physiotherapists accounted for these patients' factors within their clinical reasoning in order to provide alternative treatment options.

P9 KSA "A very old woman from a village did not accept vaginal examination. Therefore, the social worker tried to help me by explaining the significance of vaginal examination to the patient, but she did not accept that. I gave her the exercises and I explained to her daughter to help her."

In the UK, the participating physiotherapists showed awareness of the sensitivity and intimacy of digital palpation among Asian/Muslim patients, which affected their decision

making and clinical reasoning since they reported providing different management options.

FG2 UK (P3) 'During the consent process, I talk to them and explain what we can do. It enables you to offer them a scan through the groin rather than the perineum, which means that they can keep their knickers on. Therefore, I do not think about it, and during the consent process, they can then change the treatment plan or assessment plan'.

4.7.3 Associated comorbidities

Physiotherapists from both countries reported that some patients with associated problems, such as brain tumours, multiple sclerosis, and strokes, might need extra time to understand the physiotherapist's instructions due to cognitive problems, which were the results of the pathological deformities. In some instances, they also needed a family member or carer to assist them while doing their exercises. The participants sometimes used different tools to assess pelvic floor muscles while maintaining a safe environment for these types of patients.

P1 KSA 'My patient, was disoriented after a brain tumour. I decided to maintain this trustworthy and safe environment and spend more time with the patient to allow her to talk. I did not do the vaginal examination because that might make the patient feel unsafe'.

Other physiotherapists from both countries reported that patients' psychological problems, such as depression, could influence their management decisions. One KSA participant reported that anti-depressant medications might contribute towards her patient's muscle weakness and her patient did not declare this until the last few sessions as a potential cause of limited improvement in pelvic floor muscles strength.

KSA FG "One patient did not mention taking anti-depression medication until the 10th session, she just said, 'Oh I didn't improve. Unfortunately, it was not included in the electronic file system that she was taking those medication from [another] private clinic."

Therefore, it is clear that when patients have a comorbidity it can affect physiotherapists' decision making and clinical reasoning.

4.7.4 Uniqueness of urinary incontinence

Participants in both countries mentioned that UI required a commitment from patients for a long time to show improvements, but patients usually wanted quick results. They mentioned that UI is a recurrent problem and encouraged their patients to continue doing pelvic floor muscle training and follow a healthy lifestyle to avoid recurrence of UI. Hence, the method of communication of physiotherapists with their patients could be affected, which could impact on their decision making and clinical reasoning.

Furthermore, due to the chronicity of UI, physiotherapists from both countries reported that patients' anxiety and psychological symptoms usually increased their urgency and UUI. When they were asked about the significance of assessing psychological symptoms routinely and using validated outcome measures, some of the participants mentioned that it is highly important to know how psychological symptoms influence UI as it may affect their decision making and clinical reasoning. For example, they mentioned that working with psychologists in the same team would be beneficial for patients especially for those with sexual issues in addition to UI.

P10 UK "in the waiting area there is a group of iPads and every patient that comes in will have to fill it in ... then that will come up on our system. we have an indication of how their psychological state is I also think it's really important (to assess psychological state) with these patients that we're doing internal examinations on, some people are happy and used to them and some people really find it uncomfortable."

On the other hand, some physiotherapists from both countries reported that it was not their responsibility to assess psychological symptoms using validated outcome measures. But rather, they claimed that they relied on their experience to decide on the relationship and influence of psychological issues on UI. Then, they would refer the patient to a psychologist if needed.

P6 KSA "I don't confirm [psychological problems], it is not my job to do, but sometimes we notice some red flags like for example she is still repeating her birth story for example, I don't know how to react to this unless that I try to tell that I understand and I can't help them in the psychological part, only by referring them and that's all."

Regardless of the method used to assess the impact of psychological issues on UI, either with validated outcome measures or through experience, it is clear that it affected their decision making and clinical reasoning because it was considered when making decisions about assessment and treatment.

In summary, it is clear from the participants' responses that patients' factors can influence physiotherapists' decision making and clinical reasoning, and can be closely related to patients' demographics, socio-economic factors, and biomedical problems such as cognitive and psychological levels, in addition to the unique features of UI disease. In KSA, physiotherapists may need to modify their decision making regarding management programmes due to different patients' socio-economic and cultural factors.

4.8 Summary of the findings of the factors that could influence physiotherapists' decision making and clinical reasoning in the UK and KSA

This chapter identified the factors relating to patients, individual physiotherapists, and the organisational/country-wide context that inform and influence the decisions and reasoning of physiotherapists. The impact of these factors on decision making and clinical reasoning varied slightly between the UK and KSA participants due to differences in the cultural and religious context, resources, organisational culture, patients' characteristics, and physiotherapists' training levels. Further factors that showed similarities and differences between the countries included patients' demographics, socio-economic status, comorbidities and preferences for treatment options (for example, equipment versus exercise).

Culture was the biggest differentiation between the two countries. Religion played a key role that tied into traditions and norms to be respected in order for patients to adhere to

treatment. It is important however, not to single it out as a key determining factor in influencing decision making and clinical reasoning. The more diverse cultures in the UK throws up more challenges, such as language barriers and accessibility. Hierarchies within the organisational culture and lack of collaboration between multi-disciplinary teams were seen as more of a challenge in providing effective treatments to patients in KSA when compared to the UK.

It was challenging to separate these overlapping factors. This might be due to the methods of data collection that were used, i.e., semi-structured interviews and focus groups. Concurrent think-aloud interviews could have helped participants to recall data easily and could have decreased the possibility of mixing patients' data. However, in this study most of the interviews took place in the participant's clinic where they were able to access patients' files to remind themselves of details (while strictly maintaining patients' confidentiality), which negated concerns regarding recall. Nevertheless, there can be some limitations with using these data collection methods as it can cause this overlap between the factors and the model of decision making and clinical reasoning. This will be discussed further in the limitations outlined in Section 6.7.

The factors identified are broad and complex and clearly indicate that the decision extends beyond a biomedical understanding of the body, with reasoning influenced by patients, physiotherapists, and organisational factors. How these factors come together to generate a decision is explained in the next chapter.

Chapter 5 Theme 4: Making sense of physiotherapist's Decision making and Clinical Reasoning

5.1 Introduction

The previous chapter explored the factors that influenced and informed physiotherapists' clinical reasoning regarding patients with UI. The general physiotherapy literature tends to focus on the roles of decision making and clinical reasoning in i) confirming the diagnosis, ii) involving patients in the decision making aimed at improving patients' condition and iii) reducing diagnostic error. The accounts of physiotherapists suggest a somewhat different approach of decision making and clinical reasoning. A combination of factors help physiotherapists to come to a decision. Elements of clinical reasoning models are used, but sense making by (Weick, 1995) appears to provide a useful framework for understanding the overall clinical reasoning process. This chapter discusses the fourth theme: making sense of physiotherapists' decision making and clinical reasoning. It examines how sense making concepts provide a framework and presents vignettes to demonstrate 1) How different factors and clinical reasoning models were combined in different situations and 2) How the sense making process is used by physiotherapists to make and justify their decisions. The theme and categories as well as how participants used them during their decision making and clinical reasoning process are described in detail. Evidence for the interpretation is provided through real-life vignettes from individual interviews.

5.2 What is sense making?

Sense making by Weick (1995), was described in Section 2.14.3. Sense making simply means the process of making sense of what is occurring. It is the human ability to retrospectively find patterns in the continual flow of clinical events that health practitioners experience daily in order to give those clinical events meaning (Dixon, 2003). The patterns clinicians identify are strongly affected by their experience and knowledge. Health practitioners do not make sense of events once only, but instead engage in a continuous revision of their understanding based on the interpretation of others (social

context) and on subsequent events (historical). To make sense is not to figure out the right or wrong answer, but to find a pattern that gives meaning to individual health practitioners or groups, reflecting in a manner that provides sense to what has happened as sensible (Dixon, 2003).

The seven properties of sense making defined by Weick and shown in Table 2-1 (Chapter 2) provide a framework for understanding the clinical reasoning process. Each property has practical implications for clinical reasoning, and incorporates action and context, which are vital aspects of sense making. All the properties can be represented as a sequence; people, concerned with identity in the context of others, engage in ongoing events. From these they extract cues and make plausible sense retrospectively, while, enacting more or less order to those ongoing events. This sequence is crude because it omits feedback loops, simultaneous processing and the fact that over time, some steps may drop out (Weick, 1995). In sense making, the 'work' of decision making and clinical reasoning is less of an individual cognitive activity, and more of a collection of individuals¹⁰ carrying out components of that activity between themselves in a dynamic fashion (Weick, 1995). The next section explains and provides examples from the data of the seven properties.

5.3 Examples of how the seven properties of sense making constructs are used by physiotherapists, including quotes/ descriptions

5.3.1 Plausibility rather than accuracy

The meaning of plausibility rather than accuracy is "in an equivocal, postmodern world, infused with the politics of interpretation and conflicting interests and inhabited by people with multiple shifting identities, an obsession with accuracy seems fruitless, and not of much practical help, either" Weick (1995 p.61) .Pelvic health physiotherapist clinical reasoning and decision making were based on an evaluation of 'possibilities' such as patient's symptoms rather than accurate diagnosis.

¹⁰Individuals: people who talk with each other about what is going on – the patient, consultant, peers – but also the collective imagined group of individuals whose judgement of 'is this a good decision' is part of the grounded in identity, in addition, to other properties of sense making.

P12 UK "'I, as a physio, treat symptoms. So, somebody can have whatever diagnosis somebody else wants to put on them. But I actually treat symptoms. So, if somebody comes with frequency, even in the absence of urodynamic tested overactive bladder, I will treat the symptoms, because that's considered to be problematic for the patient."

Most of the physiotherapists, in both countries, reported that their patients' psychological problems caused or increased the severity of UUI. They used their intuition to support their assumption of the existence of psychological problems. They mostly did not use a validated questionnaire to support their hypothesis. They were using plausibility rather than accuracy, such as pelvic health physiotherapist instinct or intuition, rather than questionnaires to validate their patient's psychological symptoms.

P4 UK "It was probably an anxiety and behaviour-related overactive bladder. I know there is questionnaires but the time constraints.... I think I am generally quite good at picking up patients with these sorts of issue and different personalities. I rely on talking to them and hearing them and working on instinct in that case rather than anything more precise."

To summarise, physiotherapists used plausibility rather than accuracy to make sense of patient's UI problem. The participants from both countries were thinking about eliminating the suboptimal choices related to the assessment and treatment using patients' factors, organisational factors as well as their own experiences and knowledge. Sense making was not used to name a medical diagnosis but rather to guide the selection of assessment and treatment approaches (Weick, Sutcliffe and Obstfeld, 2005).

5.3.2 Social context

Sense making is a social activity and the narratives are both individual and shared – an evolving product of conversation, reading and exchanging ideas with others (Currie and Brown, 2003).

Chapter 5

In the case of physiotherapists, this included sharing the sense making with the patient. Most of the UK participants avoided imposing their values on their patients. P13 UK encouraged her patient to talk openly about the effect of UI on her life. She said that her patient's anxiety increased her UI symptoms, which in turn was very distressing and affecting her quality of life.

P13 UK 'As an experienced therapist, you get quite a bit of an idea when you very first meet patients of how they interpret their individual situation....And my colleagues often laugh at me and say that I treat a lot more from the neck up than the waist down. And for a lot of it is giving people confidence because they have had that confidence completely knocked by this problem'.

The participant was making sense of factors that were important to the patient during her conversation with the patient, i.e. social sense making. The participant used her emotional intelligence to gain the patient's trust and instil confidence in her to speak up. She also tried to reinforce positivity, motivate and empower her patient through conversation.

Most UK and KSA participants reported making conversation with their patients to establish rapport and provide them with a safe environment. In this way, physiotherapists make sense of the other factors that are related to bio-psychosocial issues and could influence their clinical reasoning and decision making.

P6 KSA "I work on the biopsychosocial way because it is not just the physical and she is not just a cadaver, so I need to see her beliefs as well".

In addition, physiotherapists sometimes have conversations with their multi-disciplinary team to make sense of a patient's problem and make joint decisions. Sense making is a social activity in that physiotherapists have conversation with patients and others in order to make consensual, joint decisions (Currie and Brown, 2003).

5.3.3 Extracted cues

Participants extracted cues based on a patient's history in order to help them decide on what information is related and what explanations are suitable (Brown, Stacey and Nandhakumar, 2008). Extracted cues provided a point of reference for physiotherapists, helping them to link ideas of meaning from broader networks (Weick, 1995).

P12 KSA felt that her patient's vaginal tissue did not stretch very well, and she used her experience and ability to recognise cues when considering the patient's complaints of general systematic problems; this led to a discovery of abnormal vaginal tissue. When the participant discussed these issues with the referring physician, he confirmed the same suspicions and did further tests that showed that the patient had cervical cancer.

P12 KSA "The complicated part was whenever I asked her questions about general health or systematic conditions, she would give me pointers to tell me that there is something else going on. After I assessed her, I felt that there is something that did not feel right, that something did not feel right with her tissues. They did not stretch in the right way; they felt a little hard in certain areas, so I called the Uro-gynecologists and he said that he kind of has the same thing."

Physiotherapists from both countries were extracting cues from the patient's history, main complaints and further tests to make sense of a patient's condition and to guide the clinical reasoning and decision making of UI managements.

5.3.4 Ongoing

'Ongoing' is constantly bracketing moments in the flow of life as we seek to reflect on and codify the meaning of things (Weick, 1995).

For example, P12 KSA used 'ongoing bracketing' of her patient's psychological and social issues by making sense of conversations with her patient and exchanging ideas regarding the best ways to implement changes in her patient's behaviour related to UI. For instance, she said that her patient wanted to continue her sunset family gathering that included drinking coffee (Chapter 4, Section 4.5.1.3). This is linked to the family-based culture in KSA, where families make mostly joint decisions. P12 KSA's awareness of her own family-centred care culture informed and shaped her clinical reasoning. However,

Chapter 5

she did not force her patient to follow her recommendations; instead, she encouraged her patient to adopt the decision to incorporate pelvic floor muscles exercises and lifestyle modification into her daily routine. Thus, she applied ongoing bracketing to consider her patient's social, emotional, and environmental needs. This helped her to make sense of the best way to implement constant lifestyle changes. This approach allowed the patient to comply with the treatment programme later on.

Participants from both countries used ongoing sense making through constant bracketing of extracted cues. Participants observed their patients for few sessions. This in turn allowed them to constantly evaluate and update the treatment plan based on changes that occur in the patient's condition.

5.3.5 Enactment of environmental context

When people talk, and build narrative accounts, it helps them understand what they think, and predict events. This happens when deciding on a domain of activity and consequently taking action. In sense making, they create their own environment for future action (Weick, 1995).

One of the participants mentioned that she received many self-referrals. It was therefore essential for her to rule out serious problems such as cauda equina.

P3 UK "I go on courses, and I say, what about red flags and what about cauda equina patients that you might see. I have to think about that kind of stuff all the time because we are the first line. People are self-referring, and they are saying, I have found a lump. It could be a simple [prolapse]. It could be something not simple at all."

The UK pelvic health physiotherapists understood the responsibilities of her environmental context of being the first point of contact as a health practitioner who is managing patients with urinary incontinence. This narrative helped her to predict that patients might have serious issues (or not). This meant she recognised the importance of using ultrasound in order to help her to rule out red flags and take relevant action. Most of the physiotherapists from both countries were making sense of the environment, and then taking action.

5.3.6 Retrospective

A retrospective view affects what people notice and decide (Dunford and Jones, 2000). Physiotherapists were making sense of patients' data by asking about their main complaints, aggravating factors, and past medical history. Going back in time helped physiotherapists notice the pattern of the disease and guide the management decision.

One of the participants reported that her patient was worried about having serious medical problems, which, in her view, may have led to UI. When the physiotherapist dug deeper, and gained the patient's trust and confidence, the patient reported that her best friend died from bowel cancer. She was therefore worried that she might have the same problem. The discussion between the physiotherapist and her patient as well as a retrospective view on the patient's medical history helped her to understand the reason for the low score in the quality of life questionnaire.

P7 UK "A friend of hers had died recently from cancer and so it transpired that she was very anxious that she might be riddled with cancer. She was insightful to say that she feels that she is hyper alert to her bladder and her back pain. She was probably lonely. All her children have not long left home. So, it felt like I was somebody that she wanted to talk to. That also maybe explain why she got in the Quality-of-Life Questionnaire 12 out of 21."

Physiotherapists from both countries used retrospective sense making of their patients' problems. This included understanding patients' previous psychological issues, injuries and lifestyles, which may be the predisposing factors that increase severity of UI. Where a previous experience was relevant, physiotherapists incorporated this into their sense making of UI problems and management decisions.

Chapter 5

5.3.7 Identity

"Who are the sense makers? In addition, how do they interpret events?" (Weick, Sutcliffe and Obstfeld, 2005). "From the perspective of sense making, who we think we are (identity) as organisational actors, shapes what we enact and how we interpret. This affects what outsiders think of us (image) and how they treat us, which stabilises or destroys our identity" (Gilliland and Day, 1999 p.334).

Physiotherapists from both countries reported a clear sense of their collective identity; and a recognition that they have special characteristics that differentiate them from other physiotherapy specialities, i.e., they are dealing with intimate body part and sensitive issues. They have learnt to show empathy, establish rapport and be active listeners. Their emotional intelligence helps them to gain patients' trust and confidence, which is important for providing patients with a safe environment to talk about personal and sensitive issues such as faecal incontinence and sexual problems. The identity of KSA physiotherapists was clear due to the differences in different organisational cultures. For instance, participants working in military hospitals were quite firm and followed their hospital guidance even if it was against patients' preferences. While other participants in the university hospitals opted for external assessment if patients did not accept vaginal examination. The hospital policies shaped their identity and practices.

Participants from both countries explained how their identity as physiotherapists helped them make sense of patients' factors and handle the sensitivity of UI appropriately. Their unique characteristics and above-mentioned traits helped them create a safe environment for patients to talk openly and influenced their management decision.

The preceding sections have defined the different properties of sense making and provided evidence of the participants' clinical reasoning and decision making demonstrating these properties. The participants' clinical reasoning showed all or some of the properties of sense making. Vignettes will be used in the following sections to demonstrate the processes of sense making.

5.4 Vignettes to demonstrate physiotherapists clinical reasoning using sense making processes

Sense making serves as a framework for showing physiotherapists' approaches to clinical reasoning and decision making regarding patients with UI. These vignettes may include patients' background, key influencing factors, clinical reasoning models and how the decision was reached (sense making process). These vignettes are used to clearly demonstrate different properties of sense making, and to show how factors (from Themes 1-3) and other clinical reasoning models combined, are used in sense making. The sense making properties are demonstrated in italics. The vignettes presented are a mixture of both typical and complex cases. However, it is uncommon to have a typical and straightforward case in the pelvic health physiotherapy clinics.

5.4.1 Vignette 1: Sense making to overcome uncertainty and ambiguity - Social, retrospective and plausibility rather than accuracy ¹¹

P1 KSA with a post-graduate certificate in women's health and continence care.

Overview of the case (paraphrased account from the physiotherapist's perspective)

X is a 56-year-old patient. Referred with MUI. She had surgical removal of brain tumour a few months ago. She was under urinary catheterisation at hospital and had been discharged with a continence product. She did not acknowledge the problem of MUI when she attended her physiotherapy session and she thought that the brain tumour was more serious than incontinence. She was wearing the continence product during the session. It was challenging to communicate with her and understand her main problems; her family carer was not sure about her physical and mental abilities. The medical file did not cover cognitive impairment or limited mobility as a reason of MUI.

The treatment decision: Follow-up for more than four sessions to ensure adherence to general upper and lower limb exercises and bladder training. Discussion of the treatment plan with the referring physician, and suggestion to the patient to have Botox injection in the bladder

Theme 1-3 factors: the key influencing factors

Patient's factors: Multiple co-morbidities such as progressive loss of vision, obesity, limited mobility, unaware about her condition due to mild disorientation and health illiteracy. She was living with a family carer at her home.

¹¹ MUI: Mixed urinary incontinence, P: Participants, KSA: Kingdome of Saudi Arabia

Continue Vignette 1/ Theme 1-3 factors: the key influencing factors

Organisational culture:

Limited teamwork because P1 KSA was trying to communicate with the referring urologist and he was not sure about the main reason of MUI. Neither he nor the neurosurgeon were able to report on patient's cognitive level. The information was not clear in the patient's medical file.

Pelvic health physiotherapist's knowledge, experience and personal characteristics:

P1 KSA tried to simplify the instructions and treat the patient as a neurological case rather than urology specific. Her goal was improved mobility, fall prevention, bladder training and scheduled voiding every 2 to3 hours. Her patient liked the gym and the social environment that is missing at her home; P1 KSA tried to build trust and good relations with her patient. Attending the gym improved her patient's adherence to attendance and gave P1 KSA a chance to make sure she understood the bladder training. The participant also encouraged her to use poem recital as an urge suppression technique and recommended lifestyle modification. P1 KSA would have liked to integrate cognitive behavioural therapy but did not have enough training and knowledge to do that.

How the physiotherapist reached the management decision? P1 KSA used analytical, narrative and bio-psychosocial model.

Analytical: P1 KSA was collecting all the available cues such as patient's disorientation, MUI mainly urge, patient's main priority, which was mobility rather than UI, to make sense of the reasons for UI and to help the patient.

Narrative: The participant took the patient to the gym to see how significant exercise could improve mobility by letting the patient observe other patients and helping her visualise her own success story if she were to adhere to the exercise programme.

Bio-psychosocial model: P1 KSA avoided doing digital palpation because of the patient's cognitive level and ability to give full consent, thus providing the patient with a safe environment. The participant did not like to break the trustful relationship with her patient by using digital palpation. P1 KSA treated her patient from head to toe and did not concentrate on the MUI only. She figured out that her patient liked the gym as this gave her a chance to improve her mobility and see other patients there.

5.4.2 Sense making process

This vignette demonstrates the participant's uncertainty and *ambiguity* about her patient's medical condition.

The features of *social*, *and plausibility rather than* accuracy in sense making were observed in this case. These helped to overcome uncertainty, in addition to different factors, such as patient, pelvic health physiotherapist and contextual factors. Figure 5-1 illustrates the process of sense making using different properties in addition to the factors that influenced physiotherapists' decision making. This will be discussed in detail in the following sub-section:

5.4.2.1 Social context

Social sense making in the above vignette is about talking to patients and making sure the female patient understood the urge suppression technique using poetry recital or other methods from the patient's own culture. Further, the participant supported her patient socially and emotionally by letting her use the gym, which the participant knew motivated her patient to move and exercise. Although the patient was keen to go to the toilet to empty her bowel for cleanliness and prayer reasons, she was unable to do so independently. Her incontinence was partly caused by her immobility; the pelvic health physiotherapists understood that this needed to be treated as well as the direct urinary symptoms. Ongoing interaction (see Figure 5-1) between the patient and the physiotherapist provided an opportunity, over time, to understand the patient's needs, overcome ambiguity, and provide individualised care. However, the physiotherapist was still uncertain how to best manage MUI.

Figure 5-1: this is a pictorial image of what is going on: Sense making process using different proprieties of sense making that include extracted cues, and identity.

It is an ongoing process and based on plausibility rather than accuracy, where social communication is circular and leads to an action. Through sense making, decision making is seen as the result of an interplay between pelvic health physiotherapist perception, patient and contextual factors where decisions occur over time to overcome ambiguity, through a process of multiple interactions (Goodwin, 2014).



5.4.2.2 Plausibility rather than accuracy (Figure 5-1)

The physiotherapists used analytical, narrative and bio-psychological clinical reasoning. She eliminated the sub-optimal choices by making sense of her patient's cognitive level, psychosocial needs, and approach to managing UI –though she was uncertain about the cause of UI. Analysing the case was therefore not enough; yet, making sense of organisational culture, she reported limited communication between multi-disciplinary

Chapter 5

teams. The culture within the organisation was divided, with physicians often having a hierarchical relationship with other health practitioners (Weick *et al.*, 2004). She made the decision to discuss the case with the urologist and suggested that the patient might benefit from Botox injection. In this way, the physiotherapist was making sense of the patient's factors as well as individual and organisational culture to overcome uncertainty and make management decisions.

This is an example where the physiotherapist used plausibility rather than accuracy, highlighting that in sense making there is no one right or accurate answer. P1 KSA considered improving the patient's mobility as it was considered an indirect cause for UI. This decision, though perhaps not the ultimate one, made sense of the patient's factors (i.e., disorientation, limited mobility) as well as organisational and individual culture. This in turn led to plausibility in decision making rather than accuracy. For instance, the choice to improve the patient's mobility and encourage bladder training is the result of identifying the patient's priority instead of focusing on managing the UI only. Further, by sharing the decision with the referring physician (organisational culture) a joint decision was made to provide the patient with Botox injection to control UI.

In summary, this vignette (1) demonstrates how the participant overcame uncertainty using the properties of sense making, particularly plausibility and social cues. Rather than just using the diagnosis and proceeding with digital palpation, the participant used sense making of the patient's factors. The participant's acute awareness of the limitations of her organisation's culture, i.e., poor communication among multi-disciplinary teams meant that she used plausibility to recognise these shortcomings and focussed on treating the symptoms rather than merely the diagnosis. The clinical reasoning and decision making of physiotherapists is a process of sense making where both the decisions undertaken (to gather cues and test them) and the process of clinical reasoning are not only individual but also context-dependent (Weick *et al.*, 2004). The second vignette will show another property of sense making.

5.4.3 Vignette 2: Properties of sense making: Extracted cues, 'social' and 'grounded in identity' ¹²

P10 UK with a post-graduate certificate in women's health and continence care.

Overview of the case (paraphrased account from the physiotherapist's perspective)

The patient was referred by the urologist who had done many urodynamic studies and flow rate tests; the result had been normal so the diagnosis was a functional voiding problem. The patient seemed a bit nervous when she first saw P10 UK who went through the urodynamic assessment report and said: "There's nothing they can see that is blocking you voiding, so the reason they have sent you here is that they think it might be to do with increased tension in your muscles, which are not relaxing for you to go for a wee,"

The PHPT asked the patient whether there was anything that she thought contributed to the issue, i.e. "Do you suffer from any stresses or anxieties? Have you had any bad experiences before?" In response, the patient revealed a long, complicated history ... she had been used within a drug circle...where she grew up, she has had an abusive ex-husband, though she is now in a more stable relationship. She also was not sure about her sexuality, but had never discussed it with anyone.

The treatment decision: P10 UK spoke to the patient about voiding techniques and positions, as well as about breathing and relaxing into her pelvic floor. After a few sessions, the PHPT managed to build rapport with the patient, who agreed for the PHPT to write to her GP and the urologist, and seek referral to a psychologist regarding her sexual abuse.

¹² P: Participant, UK: United Kingdom, PHPT: Pelvic Health Physiotherapist, GP: General Practitioner

Continue Vignette 2/ Theme 1-3 factors: the key influencing factors

Patient's factors: Had a psychological problem because of sexual abuse. This factor contributed to developing tension in PFMs and functional UI.

Organisational culture: Patient was exposed to many urodynamic tests and no one asked her if she had any history of sexual problems. The focus had been on physiology and no one had undertaken a broader, holistic assessment.

PHPT's factors: PHPTs required many sessions to build rapport, then referred the patient to a counsellor.

P10 UK: "We're compassionate and we listen but we don't have the skills specifically to know how to support that patient [psychologically]"

PHPT's knowledge, experience and personal characteristics: PHPT related her patient's PFMs tension to anxiety and stress. She had the emotional intelligence (EI) to build a sensitive rapport with the patient, which allowed the patient to disclose the reason for her tension in PFMs, which she had not disclosed to other health professionals.

How the physiotherapist reached the management decision?

P10 UK used bio-psychosocial and ethical reasoning. The physiotherapist explained: "the patient cried a lot the first time we spoke, and then she said that she never wanted a vaginal examination, which was fine. She came back, she was better, and she was like 'I think I just need to talk about it.' P10 UK asked her to tell the GP but she said no. The patient needed professional help from a counsellor who could understand the complexities of how this problem might affect her in the future. P10 UK eventually managed to build rapport with her after a few sessions and told her "Look, are you sure you don't think speaking to someone professionally will be of more help?" Eventually she agreed for the physiotherapist to write to her GP.

5.4.4 Sense making process

This vignette demonstrates participant's (pelvic health physiotherapists) uncertainty about the reasons for the patient's pelvic floor muscles tension.

The features of *extracted cues, social and grounded with identity were observed in this case* (Figure 5-1). The definition of each property is inTable 2-1. The participant's use of the proprieties of sense making in her decision making will be discussed in detail in the following section.

5.4.4.1 Extracted cues (Figure 5-1)

Extracted cues are derived from actions around us, we tend to bracket and extract certain elements, which become the targets of the sense making process (Weick, 1995).

Weick and colleagues (2005) argue that the first question of sense making is "what is going on here?" (Weick, Sutcliffe and Obstfeld, 2005). During routine assessments, the pelvic health physiotherapist noticed a range of different cues such as the patient's low mood, pelvic floor muscle tension, a negative result from all the urodynamic procedures, and the fact that something was still missing.

She used 'noticing and bracketing', an initial state of sense making. In this context, sense making means "inventing an interpretation for something that has occurred but is not yet identified, has never been recognised as a separate process, event" (Magala, 1997 p.324). The mental models she had gained through training, life and work experiences guided the physiotherapist's bracketing and noticing (Weick, Sutcliffe and Obstfeld, 2005). Such mental models might be elicited by her patient's low mood and pelvic floor muscle tension that allowed her to notice and make sense of her patient's condition. Sense making is about assumptions (Weick, Sutcliffe and Obstfeld, 2005). P10 UK asked herself what was going on, she assumed that there was something missing which could be the reason for pelvic floor muscles tension. She explicitly stated:

P10 UK "What if I'm missing something? Or she is of low mood? I think I pick up cues."

5.4.4.2 Social (Figure 5-1)

The 'social' property of sense making primarily considers it an introspective, intrapsychic process. Making sense of things while in conversation with others, reading and exchanging ideas with others (Weick, 1995).

The P10 UK's sense making was influenced by various social factors. The social factors may include previous discussions with the physiotherapist's colleagues about the patient's concerns, making sense of it collaboratively. For example, the physiotherapist in the above vignette was aware of the Organisational culture in the UK and the responsibilities of the GP, which includes referral for psychological treatment. P10 UK understood the importance of distributing medical sense making across the multidisciplinary team, i.e., if knowledge about the cause of UI unfolds and is documented in the medical file, then knowledge of this unfolding event does not just stay in the head of the physiotherapist. Instead, the knowledge becomes located system-wide, and understood by interdependent healthcare professionals. This in turn can influence future sense making(Weick, 1995).

The other important question of sense making is "what do I do next?" (Weick, Sutcliffe and Obstfeld, 2005). This question is about action, as is shown in this vignette, where the physiotherapist's intuition is interwoven with the essential task of encouraging the patient to agree to document the sensitive issue so that the physiotherapist can take action. The conversation leads to a continuous, iteratively developed, shared understanding with the physiotherapist's colleagues. This means that discussions with the patient can then lead to action, demonstrated by the "documentation to GP" which is part of organisational action. In sense making, talks and actions are treated as a cyclical sequence – occurring in both early and later stages. Action occurs continuously in sense making until talk brackets it and gives it some meaning (Weick, Sutcliffe and Obstfeld, 2005). Healthcare sense making is a way of thinking that occurs through conversation in the medical world as it is related to the physiotherapist's knowledge and techniques. When facing routine cases, it is sufficient for physiotherapists to use their knowledge to interpret these; however, when confronted with uncertainties, i.e., complex, unusual cases, the physiotherapist may require new interpretations. These truths of the moment develop and take shape over time. It becomes increasingly evident that the changes that were correct "back then" are incorrect actions "now" (Weick, Sutcliffe and Obstfeld, 2005). The close fit between processes of organising and processes of sense making illustrates the recurring argument that people organise to make sense of equivocal inputs and enact this sense back into the world to make that world more orderly (Weick, 1969 p.40-42).

The patient in Vignette 2 went through invasive urodynamic procedures that were all negative" back then", but no one had asked her before if she had sexual problems that might affect the decision to undergo urodynamic tests.

P10 UK "no one had asked her in urology. She had all invasive tests that she was quite traumatised by, and they were all negative,"

5.4.4.3 Grounded in identity

Identity in sense making is the process of figuring out what is going on as a product of, and a process based on who the sense maker is and is becoming (Weick, 1995).

P10 UK was able to gain her patient's trust and confidence. She reported that some physiotherapists are better at this than others are.

P10 UK "Pelvic health physiotherapists feel qualified to decide on psychological issues. I think some of us are better at it [addressing psychological issues] than others just depending on your personality and ability to gain patient trust as well."

The physiotherapist acted as part of her organisation, i.e., physiotherapy department, and this shaped how she enacts and interprets what her patient thinks and how they react during the treatment session. The physiotherapist's personal characteristics created an image in front of her patient, which established her identity as a trustworthy professional, who demonstrated high emotional intelligence.

In contrast, some of the KSA physiotherapists mentioned that they do not have clear guidance in their organisation in terms of ethical or sensitive issues. This meant they avoided asking sensitive questions that could lead to disclosure of sensitive issues, indicating that their organisation and the culture within it shaped their identity and their sense making.

P1 KSA "I avoid asking questions that I don't know what to do with their answers; if there is, for example, sexual abuse, then what's the next step? If they talk, I listen, but I never ask questions."

Sense making, filtered through issues of identity, is shaped by the recipe "how can I know who we are becoming until I see what they say and do with our actions?" (Weick, Sutcliffe and Obstfeld, 2005 p.416). The patient had gone through different investigations to determine the cause of her UI; no one asked her if she had sexual problems. The pelvic health physiotherapist participant was the only one who discovered this problem. She was making sense of the cues that she extracted from her patients' low mood, the Organisational action and the importance of communication with the GP in order to understand the patient's problem. This process of sense making was ongoing, and influenced by the physiotherapist's identity, which was shaped by emotional intelligence, and clear clinical pathway in managing ethical issues, that encouraged the patient to document the contributing factors to pelvic floor muscles tension. This action helped the patient to get the treatment she required.

In summary, this vignette (2) highlights how the pelvic floor muscle training used sense making properties to reduce diagnostic ambiguity, notably by adopting the sense making properties of 'extracted cues', 'social' and 'grounded in identity'. Rather than just relying on the referral by the urologist and the multiple urodynamic tests, which revealed the tensions in pelvic floor muscle, the participant applied sense making to understand the underlying issues. Instead of being content with improvements seen after a few sessions, the pelvic health physiotherapist in Vignette 2 used 'extracted cues' bracketing to build rapport with the patient and extracted social cues that pointed to deeper psychological

issues. These actions, on top of the participant's emotional intelligence and ability to successfully build a trustful relationship with her patient (grounded in identity), meant that the patient was able to open up, reveal her complicated sexual history and agree to be referred to receive further support.

5.4.5 Vignette 3: Ongoing extracted cues, enactment of sensible environment and sense making process to overcome uncertainty and ambiguity13

P11 UK with a post-graduate certificate in women's health and continence care.

Overview of the case (paraphrased account from the physiotherapist's perspective)

This patient is 84-year-old; a widow with SUI, proven by urodynamic studies. She had complained of SUI for 15 years and it was self-managed while her husband was alive. Her symptoms increased after her husband died. She had two normal vaginal deliveries. The urodynamic study showed no evidence of urgency, UUI and detrusor muscle over activity. The urology surgeon offered a bulking agent, but the patient preferred to manage the problem conservatively as she did not want a general anaesthetic. She admitted to faecal incontinence that happened twice in the last year. She also reported constant seepage, i.e., leaking that she could not feel was happening until she stood up and saw her clothes were wet. In addition, she had bed-wetting (nocturnal enuresis) since her husband died. Perineal ultrasound revealed a poor PFM lift the first time; the patient performed it incorrectly, with her legs going up, her pelvis tipping back and little use of many accessory muscles. After helping the patient to calm down and instructing her to avoid moving her legs, she got a better result in ten seconds – though on the third one, she started moving her legs again.

Treatment decision: Perineal ultrasound as a biofeedback to do PFM exercise. In addition, advice was given to overcome faecal incontinence, which entailed starting to use public loos if needed to relieve herself and use of air fresheners to overcome embarrassment when using public loos.

¹³ P: Participant, UK: United Kingdom, SUI: Stress Urinary Incontinence, UUI: Urge Urinary Incontinence, nocturnal enuresis complaint of intermittent incontinence (wetting) that occurs during periods of sleep, NICE: National Institute for Health and Care Excellence guidelines. Knack: is part of functional PFM training.

Continue vignette 3/ Theme 1-3 factors: the key influencing factors

Patient's factors: An elderly woman who prefers conservative treatment. She is widowed but had been married for over 60 years. She is now on her own. There is probably some anxiety related to this.

Organisational culture: NICE guidelines were guiding PHPT's decision.

Resources: The treatment session took up to 30 minutes to encourage the patient to talk openly about her medical concerns and embarrassing issues such as faecal incontinence that she had never exposed to the doctor before. Perineal ultrasound is an option for the patient who do not prefer digital palpation.

PHPT's knowledge, experience and personal characteristics: Having a good grasp of evidence – based practice to overcome uncertainty. Patient came back to the second session and said that she was feeling much better: P11 UK"I am not sure if it is because of the talk with me or because of the knack."

How the physiotherapist reached the management decision?

Analytical reasoning and pattern recognition

P11 UK explained that "seepage had been due to some detrusor over activity. Therefore, P11 UK did not guite know why the patient was seeping when she was sitting still. Pattern recognition: It was clear that she had SUI when she coughed or sneezed. She was also wet, though the physiotherapist was unsure about that. Analytical reasoning: The enuresis overnight could have been an open bladder neck at night that was working she was losing quite a lot. Physiotherapist was not able to identify this without ambulatory urodynamic tests but did not want to put the patient through that. Therefore, the only test possible was to try some medications; that also applied to the enuresis at night. The faecal incontinence sounded like urgency faecal incontinence, lack of sensation and proprioception, not being present, all tied up with the anxiety about being out and not wanting to use a bathroom".

5.4.5.1 Sense making process

The above vignette demonstrated uncertainty about the cause of urine seepage.

The features of ongoing, based on extracted cues and enacting of sensible environment were observed in this case (Figure 5-1). The definition of each property is inTable 2-1. The participant's use of the sense making proprieties in her decision making will be discussed in detail in the following section.

5.4.5.2 'Ongoing' and based on extracted cues

The pelvic health physiotherapist undertook ongoing analysis of the patient's problems that included 'bracketing and noticing'. P11 UK expected the cause of seepage to be due to opening of the bladder neck and detrusor over activity. However, urodynamic tests did not prove that. The physiotherapist considered the use of ambulatory urodynamic to know the reason for seepage. However, she also considered her patient's age and decided not to put her through such an invasive procedure. This 'noticing and bracketing' is an emergent state of sense making (Weick, Sutcliffe and Obstfeld, 2005). In the context of sense making 'bracketing' means interpreting an occurrence that has happened to the patient but has not necessarily been identified or concretely determined yet. For example, in this instance, seepage with an unknown cause (Magala, 1997 p.324). The physiotherapist's 'noticing and bracketing' are guided by mental models that she has adopted throughout her life, work experience and training (Weick, Sutcliffe and Obstfeld, 2005). The combination of mental models and ongoing extracted cues drew her attention to seepage in addition to SUI, though she could not confirm this without invasive procedures.

5.4.5.3 Enacting sensitive environment

In sense making, they create their own environment for future action (Weick, 1995). The physiotherapist's thinking process attempted to unfold the cause of seepage through a series of approximations and interpretations. She was making sense by thinking, while also using her knowledge within a trusted framework (Weick, Sutcliffe and Obstfeld, 2005). Sense making is about continued analysis of an emerging narrative so that it becomes more inclusive and incorporates the emerging data. The physiotherapist used analytical tools to overcome uncertainty related to the cause of seepage, taking account of

her knowledge, experience and a good grasp of EBP. These tools helped her to make sense of the patient's problems by identifying the most plausible causes and line of action regarding the causes of seepage. In this way, she took account of the possible diagnosis whilst also minimising intrusive procedures. Consequently, the physiotherapist decided to start pelvic floor muscle trainings for three months, based on NICE guidelines, with the option of sending her patient back to the doctor to start medication or carry out ambulatory urodynamic studies, in case the patient's condition did not improve.

P11 UK "the leakage when she sits down could be stress urinary incontinence due to an open bladder neck or it could be detrusor over activity, I wouldn't know that without ambulatory urodynamic, [but] wouldn't want to put her through that, so the only test would be to try some medication."

In summary, this Vignette further consolidates findings emerging from the first two vignettes, which both demonstrated the strong case for adopting sense making as a framework to explain physiotherapists' clinical reasoning and decision making. Vignette (3) reinforces the importance of 'ongoing' extracted cues to help overcome uncertainty in treating a patient's condition, using enacting of sensible environment to take an action in order to test her decision. In this vignette, the participant used continued conversations with the patient and extracted cues from the patient to understand the latter's embarrassment regarding her condition, including faecal incontinence, and reluctance to undergo invasive procedures. Incorporating these sense making properties helped the pelvic health physiotherapist to overcome some of the organisational barriers, to involve the patient in the decision making process and to suggest a multiple-step plan for treating the patient. Several options were providing, i.e., by suggesting starting with pelvic floor muscle training for a certain period, with the option of going back to the doctor to take medication if required.

To conclude physiotherapists, use sense making processes to overcome ambiguity: it starts with 'noticing and bracketing'; it is a combination of retrospective and prospective; it depends on assumptions to guide action; it is embedded in social, shared decisions; and it settles for plausibility (instead of accuracy). Answers to the question "what is the narrative?" arise from retrospective of past experiences, and conversation among multi-disciplinary teams who act on behalf of larger social divisions (Weick, Sutcliffe and

Obstfeld, 2005). Answers to the question "now what?" arise from assumptions about the future, articulation coexisting with action, and the patients' cases that become clear with the unfolding of events (Weick, Sutcliffe and Obstfeld, 2005).

Sense making theory was used to interpret the participants' thinking process. It was evident that physiotherapists were making sense of the different factors to make decisions. However, organisational culture was a recurrent factor that shaped physiotherapists' identity and guided their decision making. Discussions lead to continual understanding of sensitive issues, which is so central to organisational action (Weick, Sutcliffe and Obstfeld, 2005). In sense making, talks and actions are a cyclical process rather than a linear sequence. Both are significantly important and could occur early or late; action is a central factor in the understanding of sense making (Weick, Sutcliffe and Obstfeld, 2005). Organisational cultures were different between the UK and KSA that may lead to the decision that sense making theory might not be enough to interpret the participants' clinical reasoning and decision making. This point will be discussed in detail in the next chapter.
Chapter 6 Discussion

6.1 What this research has identified

The aim of this study was to better understand the clinical reasoning and decision making processes of physiotherapists undertaking the assessment and treatment of women with UI in two countries. Models and theories provided useful tools to understand and explain clinical reasoning and decision making. Analysis of the data collected for this study revealed that commonly used clinical reasoning models such as pattern recognition, dual process and hypothetico-deductive (H-D) reasoning partially help to explain some of the participants' decision making and clinical reasoning processes. These models and theories were useful in part for understanding how physiotherapists attempted to make diagnoses and to decrease diagnostic error. However, they did not address many causes of the patient's problems and different factors, including individual and organisational cultures and sense making properties such as plausibility rather than accuracy and others. Further theoretical concepts and models were therefore explored in order to explain and interpret the findings; this included cultural humility and institutional logic, which were useful alongside sense making theory to understand the decision making and clinical reasoning processes of the UK and KSA physiotherapists who treat patients with urinary incontinence.

A novel finding of this research is that individual and organisational cultures were more influential in clinical reasoning than had previously been considered in the general physiotherapy clinical reasoning and decision making literature. This finding regarding the importance of individual and organisational cultures was made possible by comparing the UK and the KSA. National and regional identity as well as religion, shaped patients' individual culture in both countries, but there was also organisational or institutional culture that shaped the physiotherapist's identity and influenced how they made decisions in the face of uncertainty.

The arguments to support this assertion are laid out in this discussion in the following way. Firstly, the chapter reiterates how the study findings help to explain physiotherapists'

thinking processes and decision making compared to the available literature in clinical reasoning. Secondly, this chapter considers how institutional theory provides additional insight alongside sense making theory to explain how differences in institutions influence participants' sense making processes. Thirdly, comparing two different cultures provides an insight into the value of cultural humility theory alongside sense making theory. Finally, in order to understand how clinical reasoning happens in different settings it is critical to consider culture and institutional logic alongside sense making proprieties, which is discussed in detail later in this chapter.

6.2 Insufficiency of clinical reasoning models

This is the first study to the researcher's knowledge which explores the clinical reasoning process and decision making of physiotherapists while managing patients with UI in two different cultures. The available models and theories of clinical reasoning and decision making in physiotherapy helped to explain some of the physiotherapists' thinking processes. The participants in the current study reported using pattern recognition, bio-psychosocial models, experience and their ability to treat patients holistically. This supports findings from previous studies exploring the process of clinical reasoning in outpatient musculoskeletal, neurological physiotherapy ,and a recently published study in Australia and New Zealand exploring the attributes of physiotherapists (Edwards *et al.*, 2004a; Holdar, Wallin and Heiwe, 2013; Bernards *et al.*, 2014; Slade *et al.*, 2020; Brandt, 2021). The latter found that physiotherapists have special attributes that are important for being an expert continence clinician. For instance, high levels of research literacy and the ability to generate and translate research into practice; managing patient-centred care and consumer expectations; advanced clinical reasoning; participation in research teams; and highly developed inter-personal skills and mentorship.

However, these models did not fully describe the clinical reasoning processes; the findings of the current study suggest that the process of pelvic health physiotherapist clinical reasoning and decision making for patients with urinary incontinence does not appear to be following a solely cognitive model of clinical reasoning, carried out by physiotherapists and patients only. Instead, physiotherapists' reasoning and decisions around UI management need to acknowledge that UI management is carried out by a range of individuals over time within an organisational culture that is socially embedded (Weick, Sutcliffe and Obstfeld, 2005). For instance, the process of UI management took place in a

context which is dependent on effective communication; where the patient needs to be able to communicate their UI symptoms and where the physiotherapist is able to receive such communication. This provides particular difficulties for patients with a history of, for example, sexual abuse or co-morbidities, including immobility or vaginismus (in addition to UI). Depending on their level of impairment, they might be unable to communicate distress if this is not encouraged by physiotherapists. In this context, UI quality of life assessment tools, by themselves, did not appear to provide the type of information physiotherapists used in order to allow effective UI recognition and management. Rather, organisational, and individual context such as type of health care system, resources and culture were mediating factors in the recognition and management decision of UI. For instance, one of the UK participants reported that her patient who had tightness in pelvic floor muscles only admitted to sexual abuse after encouragement by the physiotherapist. This participant was 'making sense' of certain types of patients using cues such as the patient's low mood to prompt her to encourage deeper discussions, which in turn led to the disclosure of sexual abuse. This physiotherapist was making sense of her organisational culture and policies. She was also aware that gaining the patient's trust to document the sexual abuse incident is the only way to help the patient to get access to psychological support. It is clear that the discussion of sexual abuse was something that was personally, institutionally, and culturally acceptable to this physiotherapist and allowed her to better understand the UI problems and management.

Conversely, in KSA, some participants were reluctant to discuss sexual abuse with their patients because there were unclear clinical pathways and national clinical guidelines to follow. The culture in KSA is a collectivist one where ethical and sensitive issues are solved within a group rather than a higher authority (Rollston *et al.*, 2020). When they did discover an ethical issue, they reported uncertainties on how to make sense of that issue.

Given that clinical reasoning in physiotherapy seems to miss potentially important issues, like organisational and individual context, it is important to look at other models of reasoning and decision making that might be useful. One such model is sense making (Section2.14.3), which is discussed in the next section.

6.2.1 Reworking of the physiotherapists' thinking processes: using sense making theory

The existing physiotherapy clinical reasoning literature concentrates on biomedical clinical reasoning models (Edwards *et al.*, 2004a; Croskerry, 2009b), in addition to limited clinical implementation of psychological and social parts of bio-psychosocial reasoning processes. This is supported by Brandt (2021), who explored the implication of bio-psychosocial models in physiotherapists while dealing with patients with pelvic floor disorders using a systematic search of the literature. The author found that most of the physiotherapist literature considers the biological effect of pelvic floor disorder rather than the psychosocial effect.

The theory of sense making offers a framework to help better understand the complexities and uncertainties of physiotherapists' clinical reasoning and decision making. The process of decision making is often characterised as plausible rather than accuracy-focussed, where knowledge and experience about the specific patient provide potential solutions to patients' problems, with physiotherapists trying different solutions until one appears to work (Dowding et al., 2016). Many of these characteristics in the process of decision making can be found within the sense making theory by Weick (1995). It suggests that individuals use different properties of sense making to underline relevant cues and identify plausible goals. The participants in this study also mentioned using a "trial and error" approach when managing UI, using pelvic floor muscle training and bladder retraining as a way of seeing if UI symptoms improved and changing treatment plans if it did not work. This is supported by National Institute for Health and Care Excellence guidelines Smith et al. (2013); guideline (2021) and other international UI management guidelines by Bernards et al. (2014); Brandt (2021). It identified pelvic floor muscle training as the first line of treatment, followed by possible adjunct therapy, such as biofeedback, depending on pelvic floor muscle power. This could help increase motivation in the patient and consequent improvement of their UI symptoms (Abrams et al., 2018; Peate, 2019). Sense making theory was used to interpret the participants' thinking processes: participants in this study overcame uncertainty by using plausibility and social cues, which are part of the properties of sense making, rather than just justifying the diagnosis and avoiding making diagnostic errors as suggested by dual process theory (Croskerry, 2009a), in addition, the classic decision making theory and dualprocess section 2.13 were not appropriate because it could not be applied in people actual life (Huczynski and Buchanan, 2001).

Furthermore, the naturalistic decision making theory did not include different factors that could influence the decision making process, such as patients' social factors; because of that, it cannot help in interpreting the participant's clinical reasoning and decision making. As a result, it was clear that the participants used sense making of the patient's factors, organisational factors, as well as their own experiences and knowledge. Sense making theory, compared to other theories, was not used to name a medical condition but rather to guide the selection of assessment and treatment approaches (Weick et al., 2004).

Another property of sense making that was apparent from the findings in this study is cue acquisition. During the process of sense making, physiotherapists identified specific cues or pieces of information, which have been referred to by Weick, Sutcliffe and Obstfeld (2005) as noticing and bracketing. This involves the physiotherapists noticing small signs and singling out these signs for interpretation (bracketing), using mental models based on past experiences. This process might be cited as being similar to the process of the pattern recognition model; it is a model of clinical reasoning and thinking processes where physiotherapists use their knowledge and information extracted from patient presentations to make a clinical diagnosis. Miller, Rivett and Isles (2009p.29) defined pattern recognition as "patterns derived from experience with similar patients or conditions which form a prototype in a clinician's non-propositional knowledge base. The pattern is triggered when similar case features are confronted, and a hypothesis relating to the presenting case is consequently formed". Pattern recognition is a diagnostic clinical reasoning as supported by Edwards, Jones and Hillier (2006), associated with the term forwards reasoning as suggested by Higgs and Trede (2018) and refers to the movement from cues to the hypothesis.

However, sense making distinguishes itself through the focus of physiotherapists on past experience and mental models, which have been developed through the continual, iterative, shared understandings between physiotherapists within a particular organisational or social culture. Sense making is an ongoing process that depends on different properties; it is not only cue acquisition, but it is a combination of cue acquisition of the patient's signs and symptoms, social and psychological background, patients, and physiotherapist's identity (Weick, Sutcliffe and Obstfeld, 2005). It was apparent from the participants' statements that many physiotherapists in both the UK and KSA used cue recognition within their practice. This is a relevant finding because UI management is about gathering information and adding meaning to this information (Brandt, 2021); cue acquisition is a way of doing this. It is a process of sense making where both the actions assumed, for example, to gather information and to perceive cues, and the process of interpretation is not only individual but also context dependent (Weick et al., 2004). This demonstrates the importance and significance of sense-making to physiotherapists' practice within various patients' contexts.

Through the lens of sense making, decision making is seen as the result of the interplay between pelvic health physiotherapist's cognition and the social/organisational context, where decisions occur over time, through multiple interactions across a number of individuals, such as multi-disciplinary teams (Goodwin, 2014). Probably, sense making may also occur during a single meeting between two people. In this approach, whether or not a physiotherapist identifies the cause of a patient's UI symptoms depends on the expectations and previous experiences of physiotherapists (through the development of patterns or mental models) and the social context of the environment within a specific organisational culture, as well as patterns of behaviour for when and how UI management occurs. The findings of this study, alongside other research, highlight the complexities and uncertainty that physiotherapists often face in attributing meaning to the information they find in patients with UI, which is where sense making properties, such as cue recognition and plausibility rather than accuracy, can help to overcome these uncertainties and address the complexities during decision making and clinical reasoning processes (Toye and Barker, 2020; Brandt, 2021). The participants' sense making of their patients with urinary incontinence psychological aspects is a good example to demonstrate the different properties of sense making. This will be explored next.

6.2.2 Treating from the neck up rather than the waist down: How the women's emotions and psychological state affect their bladder, and on a wider level, as a human being

There is a relationship between psychological problems, the bladder and UI. There are clear biological reasons why emotions affect the bladder: through the sympathetic and parasympathetic control of micturition. For instance, the sensitive nature of certain body parts, the taboo around them and women's lives, particularly in Muslims culture, may all play into the causes and consequences of UI. There is a recognition in the literature of the importance of bio-psychosocial factors in clinical reasoning, but this appears much stronger in the current study than in the rest of the clinical reasoning literature.

In the current study physiotherapists reported that they carry special inter-personal characteristics, such as empathy, self-emotional expression and optimism; this is indicative of emotional intelligence (EI), as suggested by Bar-On and Parker (2000). It can be suggested that due to the personal and intimate nature of UI, physiotherapists have characteristics that differentiate them from other physiotherapy specialities, and which inherently shape their identity. This is in line with findings stated within the literature (Hall *et al.*, 2016; Cerderbom, Bjerk and Bergland, 2020). Exceptional communication skills that use cue acquisition can build trust and good relationships, which can help to create successful interactions, which in turn achieve better treatment outcomes and experiences.

Emotional intelligence is one of the key requirements to implement a bio-psychosocial approach; however, it is unclear how EI is applied within this approach. In the current study, participants were aware about the importance of EI in UI management, but it was unclear how they developed EI and if it varied between different institutions and cultures. Further studies are required to answer those questions. Physiotherapists' clinical reasoning models were found to be impacted by the individual and organisational culture and patients with urinary incontinence were often reliant on physiotherapists' ability to identify patients' behavioural characteristics that may indicate the presence of ethical or sensitive issues.

The physiotherapists in this study exhibited different degrees of emotional intelligence due to the sensitivity of their speciality. In this study, how emotional intelligence was expressed, varied between the UK and KSA participants due to different factors and may be directly related to the differences in organisational and individual cultures between the two countries.

Organisational routines of professional roles provide a framework within which sense making can occur. Using sense making theory is a step forward that helps better understand some of the complexities of physiotherapists' clinical reasoning and decision making. However, the limitation of sense making theory relates largely to its lack of ability to recognise the impact of an organisation on clinical reasoning and decision making, and on an individual's identity. This can be considered a limitation of sense making theory

because this theory tends to focus on micro-level actions, but organisational culture is not attributed to individuals only but also to institutional controls, organisational places, and traditions. For instance, some KSA participants followed their hospital policy rather than patients' preferences and decided to discharge their elderly and illiterate UI patients with pelvic floor muscle training only because they did not accept digital palpation.

On the other hand, some of the KSA participants were aware that their patients' culture is collectivist, as supported by Hofstede (2003). They, therefore, recognised that their culture depended on "in-group" decisions rather than individual decisions. As a result, some participants reported making their management decisions based on patients' family preferences to increase their adherence to pelvic floor muscle training. For instance, replacing coffee with a herbal warm drink to manage urgency and frequency while maintaining family gathering traditions as mentioned in Section 4.5.1.3 and Chapter 4.

Consequently, comparing clinical reasoning models and decision making processes between two different cultures and healthcare institutions required the addition of institutional theory in parallel to sense making theory to improve the understanding of physiotherapists' decision making and clinical reasoning (Goodwin, 2014; Cerderbom, Bjerk and Bergland, 2020), which will be explored in the subsequent section.

6.3 Institutional theory: how the differences between institutions influence decision making

Sense making theory is useful but not sufficient in fully acknowledging the broader role of context and the influence of the organisation on clinical reasoning and decision making. It may be better understood by including institutional theory. The use of institutional theory 2.14.2, in addition to sense making, is important for interpreting the participants' findings in this study; it has been chosen because it has multiple logics and comparing physiotherapists' decision making and clinical reasoning in two different countries showed that different logics influence decision making. The organisational logics were not found in the other theories and explored in detail in section 2.14.2.

The KSA participants' sense making varied based on their organisational core characteristics and social categories, while the UK participants' identities were more

homogenous and based on national guidelines and institutional logics (of the NHS) that justify action. In KSA, the culture might be more homogenous (mostly based on one religion and minimal ethnic diversity). Nonetheless, KSA's institutional policy is heterogeneous and depends on multiple logics such as religion, hospital policy or the state and family. In contrast, in the UK, national guidelines and policies create greater unity of practice. Often organisations experience multiple and sometimes conflicting institutional logics (Thornton and Ocasio, 2008). Health care system is fragmented especially in KSA, as a result, health care organisations are dependent on a high number of actors with possibly different logics (Pache and Santos, 2010).

If sense making is affected by the organisations' culture, this can also affect the individuals' characteristics, shared beliefs and norms which may affect clinical reasoning and decision making processes (Schein, 1985; Parmelli *et al.*, 2011).

The management of organisational culture is increasingly viewed as an essential part of the health system reform (Parmelli *et al.*, 2011). The current study showed the importance of considering the impact of organisational cultures on physiotherapists' clinical reasoning and decision making. This might also be applicable in other physiotherapy specialities that treat chronic health conditions. This is supported by Cerderbom, Bjerk and Bergland (2020) in a qualitative study exploring physiotherapists' views on their role in working with fall prevention services section 2.14.2. The precise nature of organisational culture in healthcare policy and its influence on physiotherapists' clinical reasoning often stays underspecified. The benefits and possibility of implementing strategies have also been limited (Parmelli *et al.*, 2011).

Nevertheless, the importance of organisational culture in healthcare policy and its impact on clinical reasoning and decision making of physiotherapists have been shown within this study and within the literature; it would certainly be a relevant area of future research in order to further understand the extent to which organisational culture can affect clinical reasoning and ensure good practice, efficient services and positive patient outcomes.

6.3.1 Factors affecting pelvic health physiotherapists' decision making and clinical reasoning

After introducing the concept of institutional theory and the need to use that theory, in addition to sense making, to understand participants' clinical reasoning, this section explains the relationship between sense making, institutional theories and the factors that influence decision making. It was impossible to explain what was happening using one of these theories alone. Instead, what became apparent through the findings of this study is that there was a clear interaction between cues at the patient level and the social and organisational logics that shape what is appeared to and acted up on. Pelvic health physiotherapists' clinical reasoning and decision making are formed by three main elements, including 1) the decision-makers, who are the physiotherapists and patients 2) Physiotherapists' sense making; 3) institutional properties of the organisation, which include individual and organisational culture, resources, and socio-economic factors.

Sense making varies among organisational contexts; institutional logic can be implemented as an outcome of institutional properties. In physiotherapists' sense making processes, institutional logic can be considered as one input for physiotherapists and patients' sense making which may be a factor in guiding their decision making and clinical reasoning process (see Figure 6-1). The cues generated to further understand patients and physiotherapists will also be factors in decision making. Hence, it is not only the institutional logic that forms the patients' and physiotherapists' sense making, but it is also shaped by the practice that can be seen as an outcome of institutional logic. Thus, physiotherapists' decision making and clinical reasoning processes are a blend of cues from the institutional logic and practice that in turn shapes the sense making of extracted cues (Linderoth, 2017). This notion is presented in Figure 6-1 below, which shows how physiotherapists' sense making process over time and institutional logics shape their decision making and clinical reasoning and facilitate the identification of a flow of extracted cues from interactions with patients and multidisciplinary teams. These cues relate to parts of physiotherapists' institutional logic, identity, and create meaning and guide further action. However, using this might be inadequate in patients from ethnic minorities since it does not take into account the impact of cultural diversities among ethnic minorities. The variation between the UK and KSA institutional logics and properties and how it may impact decision making and sense making of ethical or sensitive issues was apparent in the findings. This was clear in the way UK and KSA pelvic health

physiotherapists made decision around regulating patients' emotions reflectively. This will be discussed in more detail in the subsequent section.

Figure 6-1: This figure shows how pelvic health physiotherapists' decision making and clinical reasoning use sense making processes and institutional logic as well as the factors that influence decision making over time to guide action.

Sense making properties include extracted cues, social context, grounded with identity and plausibility rather than accuracy adapted from Linderoth (2017).



6.3.1 Institutional logics shaped KSA physiotherapists' decision making

In KSA, institutional logic and organisational culture could shape physiotherapists' decision making do its deeply entrenched cultural heritage, which shapes much of society. The participating KSA physiotherapists in this study acknowledged that elderly, illiterate and religious patients might be hesitant to accept digital palpation, regardless of whether the physiotherapists had explained the significance of that assessment to the patient. As a

result, physiotherapists appeared divided into two types: one accepted patients' preference (interpretive decision making), while the other focussed on their own opinion because it was driven by highly available research evidence, further supported by their hospital policy and guidelines (paternalistic decision making). However, excluding patients from decision making may create a power imbalance between patients and physiotherapists, as some KSA physiotherapist participants might consider themselves as an authoritative decision-maker. Physiotherapists who used paternalistic decision making instead of joint decision making with their patients, were trying to change patients' individual or religious beliefs instead of accepting and working with them. KSA participants reported that they were making sense of their hospital policies and guidance (institutional logics) rather than making sense of the extracted cues from patients and creating a culture of curiosity towards patients' beliefs and values. Considering patients' culture in the management decision will be explored in the next section.

6.4 Cultural humility theory: how to incorporate patients' cultural needs into the management plan

Institutional logic might vary between the two countries based on the patients' ethnicity and cultural background. For instance, some UK participants reported that the NHS encourages them to document patients with female genital mutilation (FGM). This created some tension for UK physiotherapists as they were torn between following NHS guidelines (which is mandatory) and following a patient-centred approach. Participants respected their patients and wanted to provide them with an emotionally safe environment, where patients could feel safe and trust physiotherapists. However, both sense making and institutional logic do not always fully consider how to apply patients' cultural needs and beliefs into physiotherapists' clinical management. This was evident in the findings of the current study, which demonstrated that participants in both countries exhibited different understandings and incorporation of culture into their management plans. Some of them did implement this, while others were only showing awareness of those cultural needs and beliefs without considering the impact on management. However, considering this is extremely important. For instance, patients with urinary incontinence from ethnic minorities whose religious beliefs render the subject taboo may feel shy and reluctant to talk about intimate issues. This in turns means they have often been found to seek help only when their symptoms were extremely severe (Zhou *et al.*, 2011).

The present study is the first study to explore the influence of culture on physiotherapists' clinical reasoning while treating UI patients in different countries. The UK participants in this study mentioned how the culture of UI patients from ethnic minorities, such as those practising the Islamic faith, influenced their decision making and clinical reasoning. Cultural humility theory is seen as particularly relevant to this study. As such, it has been incorporated into the theoretical understanding of the findings because it acknowledges the importance of power imbalance and avoiding stereotyping compared to the other existing theories. In addition, it incorporated both an institutional level and an individual one (Tervalon and Murray-Garcia, 1998). The meaning of cultural humility theory and the differences between cultural competency and humility were discussed in section 2.15.3.1; the way participants apply culture in their management plan is discussed in the following section.

6.4.1 Cultural competency in physiotherapy and this study

In physiotherapy practice, the focus is on developing cultural competency in decision making and clinical communication. This method enables effective work in cross-cultural conditions, which focuses on increasing cultural knowledge as one of its goals; however, it has been criticised by others as a checklist approach that may lead to bias and stereotyping of individual's patient (Jenks, 2011). There is a growing body of literature that has challenged the explicit and implicit assumptions of cultural competency (Beagan, 2015; Cox and Simpson, 2020; Tascon and Gatwiri, 2020). Some limitations of the cultural competency approach include an over-emphasis and rigidity when it comes to an individuals' health practitioners' own opinions. This approach fails to wholly and comprehensively value and incorporate the needs and wishes of an individual patient from a different cultural background (Fisher-Borne, Cain and Martin, 2015).

An example of cultural competency was apparent in the current study, where some physiotherapists mentioned that they were showing empathy and awareness towards patients' values and beliefs but did not integrate their understanding into their management program. One of the UK physiotherapists reported an awareness regarding the need for cleanliness for a Muslim patient during bladder re-training. The physiotherapist showed awareness and curiosity and asked detailed questions to help the patient carry out training but neglected to fully implement a practical plan that would help the patient with the issue of cleanliness during prayer. Most of the UK participants were aware about cultural competency as a method of understanding patients' needs from different ethnicities, but only a few physiotherapists were using cultural humility instead. However, cultural competency does not provide a completely tailored solution that fully encompasses and incorporates an individual's culture, so an increased awareness and understanding of cultural humility among physiotherapists may be necessary to help resolve this particular issue. Hence, the following section discusses the use of cultural humility in patients' management plan.

6.4.2 Participants' consideration of culture in the management plan using cultural humility rather than competency

As mentioned previously, there are differences between cultural competency and cultural humility in the way that health practitioners involve their patients within the management program. Cultural humility encourages physiotherapists to place themselves in the position of a supporter rather than an authority. Further, it offers an alternative approach that focuses on knowledge of self in relation to others, recognising challenges and barriers that affect marginalised communities on both an individual and institutional level, especially within the context of sensitive, personal and intimate types of treatment of UI in two different cultures (Fisher-Borne, Cain and Martin, 2015). When health practitioners move from a cultural competence approach to a process-oriented approach, this is known as cultural humility.

Another important element of any physiotherapy management plan involves taking consent and educating patients on lifestyle issues related to UI. These are found to be an important part of the physiotherapist's management decisions. Cultural humility can impact the consent taking process, which was suggested by Bassett and Tango (2002) in a study that explored the culture of Maori patients (who are the indigenous people in New Zealand). It was found that Maori women found dis-robing and the close physiotherapist-patient contact concerning and uncomfortable. However, when the physiotherapist offered to leave the room and provided patients with convenient draping techniques, the concerns

were eliminated, and patients felt more comfortable with accepting treatment; hence, incorporating cultural humility by showing understanding and incorporating simple solutions into the UI management plan is necessary for gaining trust and improving adherence.

A potential concern raised by the participants in this study was the issue of some patients requiring family members to interpret in health consultations, which broke patient confidentiality and often prevented patients from discussing private issues (Lowe, Griffiths and Sidhu, 2007). However, physiotherapy students taking part in a community based clinical placement with indigenous people from Metis communities in a study by Oosman et al. (2019), found value in the roles of diverse community members involved in the health matters of their family members and friends. They noted the importance of incorporating family relationships into the management plan. The study reported that working with indigenous people helped the physiotherapy students to reflect on their negative stereotypes about ethnic minorities (Oosman et al., 2019). Participants could safely analyse their own worldview, lived experience and identify how these could lead to misguided generalisations about ethnic minorities. This highlights the importance of increasing genuine understanding of ethnic minorities with differing cultural backgrounds in order to remove stereotypes and instead attempt to turn potentially negative issues into positive ones. For example, when a family member interprets, physiotherapists could potentially involve the family member as an additional support for the patient. This member could then encourage adherence to the treatment plan. However, it is important to follow a patient centred care approach with women who need to be asked if they would accept having one of the family members as an interpreter. This is particularly vital when dealing with confidential issues, such as sexual problems and during digital palpation.

Within the literature, there are suggestions that some physiotherapists can be quick to assume that every patient from ethnic minorities have a preference and expectation for a passive "quick fix" management and this may be contrary to their own belief in active self-management (Dressler and Pils, 2009; Buddhadev, 2012). It is likely that preferences for passive treatment should not be considered as fixed among ethnic minority patients as these preconceived assumptions may lead to cultural stereotyping. In a UK based study by Singh *et al.* (2018) it was found that third-generation Punjabi speaking Indian patients

changed their focus from passive to active pain coping strategies over time. While common factors appeared among patients from ethnic minorities, such as communication challenges, different values and beliefs, it is important to consider that culturally and linguistically diverse communities are not a homogenous group, because there are considerable variation in the values and beliefs held within and across communities (Chan, Hamamura and Janschewitz, 2013).

There are limited studies that explore the use of cultural humility in pelvic health physiotherapy or in clinical reasoning in physiotherapy. Within the existing literature, it is apparent that the majority of physiotherapy research involving ethnic minorities were conducted around chronic pain. This research identified limited implementation of patients' cultural beliefs within the management plan. From the literature review and from the present work, the researcher of this study would argue that cultural humility complements rather than substitutes cultural competency. This is because there is limited structural framework that guides the implementation of cultural humility, although this is beyond the scope of this thesis and requires further investigation in a future study. Nevertheless, the shift towards cultural humility has the potential to increase the effectiveness of health professionals' decision making and clinical reasoning that fall along cultural lines and can increase the relevance of pelvic health physiotherapy as it develops globally. Cultural humility provides continuous self-reflection, self-critique, overcoming power imbalances by open discussion and using open questions rather than stereotyping. It encourages a more critical and effective approach to working with clients from diverse cultural backgrounds (Agner, 2020). Thus, individual culture is not the sole determining factor that influence decision making, rather institutional logics and other factors intersect to influence physiotherapists' decision making.

6.5 Reflection and reflexivity

It is suggested that the way the researchers present themselves could influence the participants' input in qualitative research (Holmes, 2020). The researcher presented herself to the UK participants as an outsider who was unfamiliar about the UK health systems and came from KSA. In addition, it was apparent during data analysis that culture and religion were a recurrent factor that influenced the participants' decision making. However, only Muslim culture was covered in detail; this may be because the researcher was wearing a head scarf, which encouraged the participants to discuss this culture specifically. There are

advantages to being either an outsider or an insider to the participants (Dwyer and Buckle, 2009). Being an outsider may have encouraged the participants to speak openly and discuss deeply about different factors. It also provided a trustworthy environment, facilitating participants to elaborate on sensitive and delicate issues related to decision making and clinical reasoning and remove feelings of judgement. However, some participants may have viewed me as an insider because I explained that I was an experienced and practising physiotherapist; consequently, they may have felt that I was knowledgeable in the field, and this may have encouraged them to speak freely and openly about their experiences.

Introducing myself as experienced in pelvic health physiotherapy may influence some participants' responses because they may assume that I knew the answer and they did not have to elaborate on it or explain it in detail; this was more obvious among the KSA participants than in the UK ones. Whenever I felt this may be a possibility, I quickly ensured that I followed the question with another "what, why and how" question to gain more depth in their answers. It might also have made some participants reluctant to express opinions or experiences that deviated from the norm. If one of the participants reported something opposite to norms, I tried to elaborate on her points of view. I encouraged her to add details, which encouraged other participants to report different unique experiences with their patients and colleagues.

The researcher would have been considered an insider and familiar to most of the KSA participants, which may also have created a comfortable environment for them to discuss their views and opinions. Although it is worth mentioning that some participants may have felt less at ease and were extra prepared to explain pelvic floor muscles anatomy and physiology because they thought I was testing their knowledge. At the same time, I stated at the beginning of the interview that I am not intended to test their knowledge and encouraged them to talk freely because there were no correct or wrong answers. Some of the participants were concerned about being judged by the researcher due to the experience I have obtained in the UK and since I am studying for a PhD at a British University. Despite this being a possible inhibitive factor or barrier to gathering data, upon reflection, I feel that the participants were open and provided rich and insightful responses during the focus groups and interviews, so it is anticipated that the possibility of being viewed as 'an

outsider' was more my worry than the participant's concern, because all of the participants were extremely helpful and willing to offer information freely and openly.

6.6 Summary of the discussion

This study aimed to improve understanding of the decision making and clinical reasoning processes of physiotherapists undertaking the assessment and treatment of women with UI in two countries. Models and theories provided useful tools for understanding and explaining decision-making and clinical reasoning. Analysis of the data collected for the current study revealed that commonly used clinical reasoning models were complex and insufficient because they do not address many causes of the patient's problems nor different factors, including individual and organisational cultures, nor sense making properties, such as plausibility rather than accuracy. This is the first study that acknowledges that decision making and clinical reasoning in UI management are affected by a range of factors over time, within an organisational culture that is socially embedded (Weick, Sutcliffe and Obstfeld, 2005). It is important to look beyond the current models in the physiotherapy literature at other models of reasoning and decision making might be useful. In such model is sense making theory, particularly the elements of 'plausibility rather than accuracy' and 'grounded with identity'. Moreover, sense making, institutional logic and cultural humility theories all needed to be considered to explain what was happening regarding decision making and clinical reasoning among the participants in this study. However, it is apparent that these theories in combination are relevant and useful in understanding how physiotherapists manage patients with urinary incontinence.

The current study also provides an insight into how physiotherapists' professional identities are shaped by their organisational culture and how these in turn shape their reasoning. A pelvic health physiotherapist's awareness of the organisational culture and institutional logic, which includes family, religion, guidelines and/or policy would help them to tailor the management programme (Thornton, Ocasio and Lounsbury, 2015). At the same time, physiotherapists' cultural humility may allow them to incorporate patients' identity and beliefs into the management process without merely categorising patients based on physiotherapists' own judgements and assumptions about their patients' cultural identity; this is the opposite of cultural competency (Nadan and Ben-Ari, 2013).

Furthermore, recent concept analysis of physiotherapists' bio-psychosocial model in chronic pain and illness has identified that physiotherapists paid limited attention to patients' social, cultural and psychological factors (Daluiso-King and Hebron, 2020; Brandt, 2021). Comparing two different countries has shown the impact of institutional logic and cultural humility, alongside sense making theory to understand the decision making and clinical reasoning processes of UK and KSA physiotherapists who treat UI patients. Physiotherapists' emotional intelligence was a strong and recurrent feature of the physiotherapists involved in the current study. In the future, it could contribute towards an advanced level of physiotherapists' bio-psychosocial consideration in the management plan of UI patients. It is, however, unclear how exactly pelvic health physiotherapist participants expressed emotional intelligence and how it influenced their interaction with their patients. There is also the possibility that it varies depending on the institution and culture. More research is needed to answer these emerging questions. The current research suggests that emotional intelligence exists within the participants, but we would need studies that specifically set out to explore the factors that contribute towards developing emotional intelligence among physiotherapists.

It is apparent from the findings of this study that physiotherapists' decision making, and clinical reasoning is a process-oriented approach, which can be interpreted using sense making, institutional logic and cultural humility theories.

6.7 Limitations

There is a possibility that clinical reasoning processes could be better understood through combining observation of physiotherapists' management processes of patients with urinary incontinence alongside interviews about those practices; this, however, was challenging as patients were likely to express reluctance in accepting a third party to observe an examination or treatment that involves intimate and private areas. For example, patients might be hesitant during digital palpation, carried out by physiotherapists. This would be particularly challenging in KSA, due to the patients' culture, in addition to challenges posed by research ethics.

Analysing participants' thinking processes and reasoning during the interviews and focus groups was challenging as they were recalling patients whom they had treated days or weeks ago in the interviews. A 'think aloud' style interview might have been a good alternative to the semi-structured interview, but this would have either interrupted the physio/patient interaction (if occurring in real time). Another option would have been a video recording of the encounter, with a subsequent 'think aloud' phase where the pelvic health physiotherapist would re-watch the video with the researcher and explain what they were thinking, doing and why. As above, this was not considered to be appropriate due to the intimate nature of the treatment. In addition, most of the interviews took place in the participants' clinics. This was helpful as physiotherapists had access to their patients' medical files, which were used as a prompt or reminder in some cases.

The extent to which participants' thinking and clinical reasoning processes can be accessed, might be restricted due to different factors. For instance, some of them might have had high reflexive, metacognitive ability, with the ability to justify how they came to certain decisions but felt unwilling to disclose due to social desirability or lack of memory -there is no ultimate way to understand physiotherapists' clinical reasoning and decision making. In addition, some of the participants, especially in KSA, had limited exposure to qualitative research, which meant that they were asking if their assessment and treatment processes were right or wrong. Further, the researcher was known by a number of KSA participants. This may have influenced what they were prepared to say and how they accounted for their practice. In the UK, my appearance, i.e., wearing a dress with a headscarf, as well as my oral delivery, i.e., not having a British accent, -may have affected the extent to which the focus was placed on cultural issues. It could also have meant that participants' presented views which they considered to be socially acceptable. Alternatively, this limitation could be seen as an advantage in that it might have encouraged participants to reflect on the impact of culture on their clinical reasoning in a way that they might not have done otherwise.

Chapter 7 Conclusion

7.1 Conclusion and implications

This thesis presents new, more nuanced understanding of clinical reasoning and decision making in physiotherapists, in addition to the factors that influence physiotherapists' clinical reasoning and decision making while managing patients with UI. The factors identified are broad and complex and clearly indicate that decision making extends beyond a biomedical understanding of the body, with reasoning influenced by patient, physiotherapist, and organisational factors. The findings in the current study show an overlap between the factors and the models of clinical reasoning, hence, physiotherapists' reasoning and decision making are complex and not straight forward and the existing clinical reasoning models are insufficient when used as stand-alone models. As a result, sense making theory is suggested as a framework that could be used to interpret and understand physiotherapists' clinical reasoning and decision making, also it is supported by a consideration of institutional logics and cultural humility theory.

Most of the existing clinical reasoning literature in physiotherapy is about the interaction between physiotherapists and patients and making the correct diagnosis. The findings from this research show that participating physiotherapists from both the UK and KSA used different properties of sense making, mainly plausibility rather than accuracy, extracted cues, grounded with identity and the social context to understand uncertainty and ambiguity in the clinical reasoning and decision making while managing patients with UI. Physiotherapists sense making is a framework and process, which acknowledges the complexity of patient and contextual factors that influence physiotherapists working within the clinical context of UI. A sense making consideration of clinical reasoning may also be useful for other physiotherapy specialisms dealing with chronic patients' and neurological cases.

By undertaking research in two countries, with different cultures and healthcare systems, this study was able to identify concepts not incorporated in other clinical reasoning models. In this study, organisational culture within sense making was suggested as an additional model to help understand physiotherapists' clinical reasoning and decision making. It was

found that the organisation shapes pelvic health physiotherapist's identity, behaviour, and characteristics, which in turn can influence their assessment and treatment decisions. The organisational culture in KSA was particularly shaped by religion, patients' culture, and physicians' hierarchies, while the organisational culture in the UK was found to be influenced by National Institute for Health and Care Excellence guidelines, NHS policies and patient centred care. Such concepts have not been identified before within decision making and clinical reasoning models in physiotherapists and nor other area of physiotherapy and as such this is a novel contribution to the literature.

Another concept that was found to affect physiotherapists' decision making and clinical reasoning was institutional logic. Since this PhD thesis was conceptualised to answer questions regarding the similarities and differences in decision making and clinical reasoning between physiotherapists in two different countries, this comparison enabled the creation of a unifying story regarding the importance of institutional logic that physiotherapist's use when clinically reasoning and making decisions. Using institutional logics, as suggested by Goodwin (2014), alongside sense making and organisational culture, helped in understanding the impact of different institutions on physiotherapists' decision making and clinical reasoning in two different countries. KSA participants were aware about the importance of family centred care and religious background, while the UK participants might view that as a barrier to patient centred care. These findings could be applicable to other physiotherapists who are treating patients from ethnic minorities. In the current study, institutional theory was used to support sense making theory to understand the participants' decision making and clinical reasoning. The precise nature of organisational culture in healthcare policy and its influence on physiotherapists' clinical reasoning often stays underspecified and the benefits and possibility of strategies to be implemented is limited (Parmelli et al., 2011). Nevertheless, its importance and impact on decision making and clinical reasoning of physiotherapists has been shown within this study and within the literature and it would certainly be a relevant area of future study in order to further understand the extent to which organisational culture and institutional logic can affect clinical reasoning and ensure good practice, efficient services and positive patient's outcomes.

Moreover, culture was found to be an important influence on clinical reasoning. There are limited studies that explore the influence of patients' culture on physiotherapists' clinical reasoning and decision making but, in this study, patients' culture was found to be a recurrent factor that did influence physiotherapists' clinical reasoning and decision making. Culture in the UK is largely heterogeneous, while it is more homogenous in KSA and is shaped by religion in addition to other elements. Participants in the UK reported awareness regarding the importance of cultural competency in management plans, although only a few of them mentioned including patients' beliefs and values in the management plans. The existing clinical reasoning models were found to be generally insufficient when used as stand-alone models; however, cultural humility theory was found to be relevant and useful in identifying the importance of culture. Most of the existing research has been conducted in western countries, which makes it difficult to assess the effect of patients' national/ regional culture on clinical reasoning and decision making processes for physiotherapists treating patients UI. Therefore, this study aimed to address the existing gap within the literature by involving participants from two different countries, which made it possible to draw comparisons that would not have been otherwise achievable. It is imperative that future research considers the impact of individual and organisational culture in any new research especially when used to inform clinical guidance.

Existing evidence appears to suggest that the way the bio-psychosocial model is used takes limited account of patients' psychological and social factors and is therefore, restricted in its utility. In this study it was apparent that physiotherapists were considering patients' psychological factors and that their emotional intelligence encouraged patients' disclosure related to their symptoms. It can be suggested that further studies are needed to understand how the physiotherapists develop their emotional intelligence and how that could contribute toward the way bio-psychological model is implemented.

Long-term UI has some similarities with other long-term conditions that require selfmanagement, for example chronic neurological conditions. Therefore, physiotherapists from other specialities might also use the sense making process, organisational culture, and institutional logic to understand uncertainties and ambiguity with their chronic patients who require long term self-management. This study showed that patients' individual culture, beliefs, and institutional logic, such as family centred care are significant factors that influence decision making, which may be a relevant finding for

physiotherapists treating various long-term conditions (and not just limited to physiotherapists treating patients with urinary incontinence).

The existing literature and the current study suggest that there are differences in clinical reasoning models used by novices in comparison to experience physiotherapists. When considering how to support novice physiotherapists to develop their decision making skills, sense making theory could contribute by raising awareness of the clinical reasoning processes that take place. Current models of clinical reasoning encourage students to use cognitive models of reasoning, while sense making theory involves different properties that work on novices' metacognition, which may be easier to learn than cognitive ones. This is an important finding that can contribute towards improving physiotherapists' clinical reasoning and decision making and may help to improve clinical outcomes for patients with urinary incontinence, as well as those with other chronic illnesses that require self-management. Therefore, there is potential for further research to test the suitability of using sense making and institutional theories in other physiotherapy areas and in undergraduate physiotherapy study. Further studies are needed to explore the effect of adding cultural humility to cultural competency in physiotherapists' management plans of patients with ethnic minorities within the UK.

7.2 Recommendations

The current study led to important findings, and it is highly important to take further step to implement the findings within clinical practice, education and within KSA and the UK health service. This could be achieved by applying the following recommendations.

Improving awareness and understanding clinical reasoning and decision making for physiotherapists using sense making theory, organisational culture and institutional logic is a nuanced approach that could be developed further by implementing it in physiotherapy undergraduate education and within clinical practice. This could help physiotherapists understand uncertainties and ambiguity within their decision making and clinical reasoning process in both the UK and KSA. This could be achieved by communicating the findings of this research with the CSP and world physiotherapy association to inform future national and international clinical and education physiotherapy guidelines.

Using sense making theory and institutional logics is important to understand physiotherapists' clinical reasoning and decision making in other areas, such as chronic MSK, neurological problems and chronic pain that required self-management. This may occur by conducting additional qualitative research exploring the use of sense making properties and institutional logic on physiotherapists' clinical reasoning and decision making while managing patients with chronic conditions.

Adaption of clinical guidelines for physiotherapists in KSA is highly important to unify clinical practice, encourage physiotherapists to provide patients with different treatment options, and to consider patients' beliefs and needs within their clinical reasoning and decision making. This could be achieved by using recent international high-quality clinical guidelines in pelvic health physiotherapist and assessing the feasibility of applying these guidelines among KSA patients considering the differences in culture, institutional logics and making sense of the factors that influence KSA physiotherapists' clinical reasoning and decision making. Furthermore, KSA participants showed different clinical reasoning and decision making when dealing with sensitive and ethical issues. It is important to adapt international clinical guidelines in ethical reasoning using EBP to provide physiotherapists in KSA with a clear clinical pathway to understand uncertainty and ambiguity in ethical issues. Also, collaboration with multi-disciplinary teams is highly important in order to develop national ethical guidelines/ethical code of practice.

Public health England promote health equality, but this is difficult to be achieved without proper understanding of patients from ethnic minorities and having representatives (advocates) within the health service who are willing to serve their communities. The writing up of the current study took place during the global COVID-19 situation, which highlighted the high level of health inequality around the world. It is, therefore, essential to introduce cultural humility programmes in addition to cultural competency in physiotherapy departments in different clinical settings. This may be achieved by conducting additional qualitative research to further understand the benefits and feasibility of using cultural humility as well as cultural competency programmes in physiotherapy department and other wider contexts while managing ethnic minorities' patients.

Emotional intelligence appears to be an important feature that can contribute towards the successful application of bio-psychosocial model for physiotherapists. However, it is unclear how physiotherapists in the current study develop their emotional intelligence features and what factors contributed to that development. Hence, further investigation may be required to understand the best way to teach undergraduate students how to be emotional intelligence and maintain professional boundaries at the same time.

The researcher will follow the implementation of the above recommendations with different organisations to increase the transferability of the current study within clinical practices and education

Appendix AScoping review of the decision making and clinical reasoningliterature in physiotherapy

Scoping Review

Arksey and O'Malley (2005) suggested a scoping review framework using a five-step approach as follows:

- 1. Identifying the research question
- 2. Identifying relevant studies
- 3. Study selection
- 4. Charting the data
- 5. Collating, summarising, and reporting the result

Step 1: Identifying the research question

What are the similarities and differences in the clinical reasoning processes of experienced physiotherapists in the UK and KSA while assessing and treating women with UI?

The objectives are:

To identify and understand the similarities and differences in clinical reasoning and decision making between physiotherapists in the UK and KSA.

To identify the clinical reasoning models being used during decision making in UK and KSA.

To explore the factors influencing clinical reasoning and decision making in the management of UI.

To contribute towards the understanding of future development of theoretical frameworks of clinical reasoning specific to physiotherapists in the UK and KSA.

Step 2: Identifying relevant studies

Developing search terms

The PEO format (Bettany-Saltikov, 2012) was applied Table 7.1.

Table 7-1: Developing search terms using PEO Format, which is Population, exposure and outcomes

P: Population and problems

E/I: Exposure / Issue

O: Outcomes and themes

As the objectives have been developed in support of the research question, different stages of searches were carried out using the search terms identified in the table below. The key words and related synonyms were combined using Boolean operators AND/ OR, and truncation '*

Stage	Objectives	Key words	Boolean Operators
One	To explore the utilization of clinical reasoning and decision making in women's health physiotherapy	P: physiotherapists E/I: urinary incontinence O: clinical reasoning, decision making	Physiotherapy*OR 'physical therapy*' AND 'clinical reasoning' OR 'decision- making' AND women OR woman OR female* AND incontinence OR 'stress urinary incontinence' OR 'continence care' OR 'continence management'
Two	To explore the utilization of clinical reasoning and decision making in pelvic health physiotherapy	P: physiotherapists E/I: urinary incontinence O: clinical reasoning, decision making	Physiotherapy*OR 'physical therapy*' AND 'clinical reasoning' OR 'decision- making' AND pelvic health OR pelvic rehabilitation AND incontinence OR 'stress urinary incontinence' OR 'continence care' OR 'continence management OR Urge urinary incontinence OR mixed urinary incontinence'
Three	To explore the decision making and clinical reasoning process among other physiotherapists' specialities	P: physiotherapists E/I: other physiotherapists speciality O: clinical reasoning, decision making	Physiotherapy*OR 'physical therapy*' AND 'clinical reasoning' OR 'decision- making' AND women OR woman OR female*
Four	To explore the model of clinical reasoning among physiotherapists treating patient with UI	P: physiotherapists E/I: urinary incontinence O: the model of clinical reasoning	Physiotherapy*OR 'physical therapy*' AND 'clinical reasoning model' OR 'decision making model' AND women OR woman OR female* AND incontinence OR 'stress urinary incontinence' OR 'continence care' OR 'continence management'
Five	To identify the factors that influence physiotherapists' clinical reasoning and decision making	P: physiotherapists E/I: different than other physiotherapists speciality O: the factors of clinical reasoning and decision making	'Physical Therapy' OR 'Physical Therapists' AND 'Decision Making factors' OR 'clinical reasoning factors' AND 'factors that affect clinical reasoning' Or 'factors that affect decision- making 'AND 'culture' AND 'social factors' AND 'cultural diversity' AND 'religion' AND 'ethnicity' AND 'knowledge base' AND 'education' AND 'patients' AND clinician's perspective'

Table 7-2: the objectives, related key words, and Boolean operators

Six	To identify the patient- centred clinical reasoning model and shared decision making	P: physiotherapists E/I: other physiotherapists speciality O: patients centre and shared decision making	'shared decision making' AND physiotherapy* AND physical therap* AND 'patient centred care'
Seven	To identify different clinical reasoning models	P: physiotherapists E/I: other physiotherapists speciality O: different clinical reasoning models (dialectic reasoning, biopsychosocial reasoning, pattern recognition, H-D reasoning and dual process, clinical practice	Physiotherap* AND physical therap* AND Dialectic reasoning AND Biopsychosocial reasoning AND Pattern recognition AND Hypothetico-deductive reasoning AND Dual process

Date of Search	2016 and updated in 2021
Inclusion Criteria	Studies with in-depth analysis and exploration of the clinical reasoning model and decision making, factors and strategies among physiotherapists and other disciplines, the clinical guidelines of physiotherapy in UI.
Exclusion Criteria	Articles were excluded if they did not involve an in-depth analysis and exploration of clinical reasoning and decision making.
	The following were excluded: articles on physiotherapy assessment and treatment; clinical guidelines; health promotion; professional competence; physiotherapy student's clinical placement; physiotherapy ethics, physiotherapy continuing education, clinical reasoning in acute care and in other disciplines such as nursing, medicine and midwifery, imaging techniques, surgical interventions, Delphi-survey and other health professionals' assessment, and treatment including drug prescription. In addition to low quality studies.

Table 7-3: Setting the inclusion and exclusion criteria

Appendix A

	Other physiotherapy intervention for SUI, such as electrical stimulation, and biofeedback, and vaginal cones.
	Physiotherapy assessment for SUI.
	Language: English
Limits	From: 1980-2016
	Another limit after updating the literature:
	2016-2021
	Female
	Human

Database selection

The following databases related to health, social care, psychology and sociology accessed via the University of Southampton library were searched: Cochrane, MEDLINE, CINAHL, EMBASE, Psych INFO, Web of Science (W.O.S), Scopus and MIDRIS. In addition to the grey literature.

Table 7-4: Database selection criteria

Cochrane Library The availability of systematic reviews, technology assessments, economic evaluations and individual clinical trials.

MEDLINE is the largest and most widely used database in the health sciences. It covers journal articles and other reference types in medicine, dentistry and nursing, including biomedicine, medicine, nursing, dentistry, allied health, pre-clinical sciences and psychology.

CINAHL Offers comprehensive coverage of journals in nursing, midwifery and allied health.

EMBASE Offers a range of journal articles in biomedicine

Psych INFO Contains journal articles, books, dissertations and theses in core psychology

Disciplines, behavioural sciences and mental health.

Web of Science (W.O.S) Provides access to a wide range of field particularly covering science, health, social science, humanities.

Scopus Is the largest abstract and citation database of peer-reviewed literature. It comprises the world's research in the fields of medicine, social sciences, and arts and humanities.

MIDIRS Reference Database collates information and resources relating to midwifery.

Evidence Search Health and Social Care (NICE):

https://www.evidence.nhs.uk/

The British Library

http://catalogue.bl.uk/primo_library/libweb/action/search.do?dscnt=1&dstmp=13847813 50374&vid=BLVU1&fromLogin=true

Open Grey - a multidisciplinary European database, covering science, technology, biomedical science, economics, social science and humanities.

http://www.opengrey.eu/

The King's Fund - The King's Fund is an independent charity working to improve health and health care in England. They help to shape policy and practice through research and analysis; develop individuals, teams and organisations; promote understanding of the health and social care system; and bring people together to learn, share knowledge and debate.

http://www.kingsfund.org.uk/

UK Clinical Research Network Study Portfolio - The UKCRN Study Portfolio is a repository for recording research taking place within the NHS in the UK. Details of clinical research which meet specific eligibility criteria, and which is currently taking place across the UK is recorded in the UKCRN Study Portfolio.

http://public.ukcrn.org.uk/search/

Step 3: Study selection

The study selection from 2016 and 2021 will be present. The different stage of the search strategy resulted in a variable hit (see Figure 2-3). Implementing the inclusion and exclusion criteria Table 7.3 some of the studies titles and abstracts were screened and duplicates. A total of 28 full papers were relevant to the study.

Step 4: Charting the data in 2021

According to the Arksey & O'Malley's framework for conducting a scoping review, the data charting process involves insertion of information from individual articles. Therefore, the following data was entered into a spreadsheet: author(s), year of publication, study location, aims of the study, methodology/ study design, and method of data collection, sample/participant, and key findings are illustrated in Table 7.6.
Appendix BCharting the data 2021 and 2016

A scoping review framework using a five-step approach was used to update the literature review in July 2021. The fourth step include charting the data as illustrated in the detailed table below.

Table 7-6: A detailed data extraction sheet of the literature review conducted in 2016 and 2021.

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
 1.Attributes of physiotherapy continence clinicians: a qualitative perspective (Slade et al.2020) Australia and New Zealand 	Explore the experiences and information needs of clinicians who used pelvic floor muscle training for UI management	Qualitative methods. Data collection: using focus group, purposeful sampling from private and public settings of varying level of experience in pelvic floor health and continence management.	Results showed three main themes: 1. Attributes that are important to be an expert continence clinician.2. High levels of research literacy and ability to generate and translate research into practice.3. Managing patient-centred care and consumer expectations. Important attributes of expert continence physiotherapists include postgraduate education, advanced clinical reasoning and decision making, participation in research teams and partnerships, highly developed inter-personal skills, and mentorship.
		Using a predetermined set of questions to guide the sessions.	Conclusion: A stronger foundation in clinical reasoning is essential and assisted in overcoming deficits in research reports and clinical guidelines. The need for sustainability by providing structured mentoring and training to extend continence knowledge, skills, capability. They suggested further research

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		Codes on participant's beliefs and experiences, themes describing clinician attributes.	including clinician from rural area and a broader geographical and cultural sample to gain into practice similarities and differences.
2. A critical review of the biopsychosocial model of low back pain care: time for a new approach?(Mescouto et al.2020) Australia	Aim: To investigate how physiotherapy low back pain literature enacts the Bio- psychosocial model.	Methods: Conducting a critical review using discourse analysis of 66 articles retrieved from different data base	Results: showed dominant of biomedical model over bio-psychosocial model. Psychological aspects are conceptualised as cognitive and behavioural. Social context was rarely mentioned and other aspects of care such as culture and power dynamics received little attention within the texts. Conclusion: multiple important factors such as institutional power relations, cultural considerations, ethical and social aspects of health may not incorporate in physiotherapy research and practice when working with people with low back pain
3. Is the biopsychosocial model in musculoskeletal physiotherapy adequate? An evolutionary concept	Aim: To clarify the conceptual understanding of the bio-psychosocial model within musculoskeletal	Methods: Using concept analysis methodology by Rodgers 1989 to clarify how the bio-psychosocial is conceptualised in	Results: After applying search strategies and selection from different database. That result in 16 articles included, after thematic analysis. After analysis the main themes are; failure of biomedical model, literature and supporting guidelines, cultural change in physiotherapy and different master themes such

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
analysis. (Daluiso-king and Hebron,	physiotherapy, to understand	physiotherapy. The procedures	as, biomedical, psychological, social factors and communication and
2020) UK	the meaning and use in	involved four steps, collection of	individualised care. In addition to implementation failure of the bio-
	clinical practice.	data related to bio-psychosocial	psychosocial model
		models, analysing the data and	
		identifying implications for	Conclusions: A holistic framework for care is proposed, in which
		current and future practice and	communication and therapeutic alliance forms the scaffold for a humanistic
		researcn.	approach to therapeutic care. R conceptualising bio-psychosocial models to
			support patient's therapeutic care is needed
4. Physiotherapy and pelvic floor health	Aim: To address the gaps and	Methods: Critical review of the	Results: The resulted studies focus on the biomedical aspect of pelvic health,
within a contemporary biopsychosocial	controversies in research and	literature for studies published in	while psychological, cognitive, behavioural, social and occupational factors,
model of care: From research to	evidence on bio-psychosocial	the last ten years. Studies on bio-	individualised care, communication and therapeutic alliances are still under-
education and clinical practice. (Brandt	model within pelvic floor	psychosocial models,	investigated and not explored at a sufficient level into research, education and
C, 2021)	disorders.	communication and individualise	clinical practice in physiotherapy from physiotherapists perspective.
		care were retrieved.	
5. Sociocultural factors influencing	To synthesise the	Methods: Different data bases	Results: Key factors included: language competence, active vs passive coping
physiotherapy management in culturally	sociocultural factors	were searched until July 2018.	strategies, gender influences, cultural-spiritual beliefs, illness perceptions and
and linguistically diverse people with	influencing pain management	The methodological quality of	expression of pain, treatment satisfaction and barrier to access.

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
persistent pain: a scoping review. (Yoshikawa et al. 2020) New Zealand and Australia.	between culturally and linguistically diverse (CALD) patients with persistent pain and physiotherapists treating CALD patients.	the included studies was assessed using the Critical Appraisal Skills Program (CASP) and Mixed Methods Appraisal Tool (MMAT). Using a thematic synthesis to extract the common themes.	Conclusion: Discordant perspectives on patient autonomy are evident between CALD patients and physiotherapists. Such discordance create stress in the therapeutic alliance and influence the efficacy of pain management. To overcome such barriers, it is crucial to foster cultural competence in physiotherapy. There is limited implementation of patients' socio-cultural awareness in clinical practices.
6. Why Do We Do as We Do? Factors Influencing Clinical Reasoning and DM among Physiotherapists in an Acute Setting (Holdar et al. 2013) Sweden	To identify the factors that influence the physiotherapist's CR in specialist care.	Data collection: observations and interviews of eleven physiotherapists in acute care hospitals.	Data analysis showed six descriptive categories with five subcategories. Situational circumstances that could not be controlled by the physiotherapists/contextual factors (work culture and organisation), communication, acquired knowledge, physiotherapists and patients as persons and negotiation strategies to choose intervention.
7. A Clinical Reasoning Model for Manual Physical Therapy. Collins C;	To explore the literature on the clinical reasoning models used by physiotherapists	Narrative review of the existing literature in functional manual therapy	There are different models of clinical reasoning used by physiotherapists such as functional manual therapy clinical reasoning model for physiotherapy intervention based on the movement system and for the examination,

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
Johnson R and Masaracchio M (2017) USA	when developing a plan of care.		evaluation, and prognosis. Conclusion: This model of reasoning provides physiotherapists with a paradigm for the creation, implementation and improving the plan of care for patient management.
8. Physical Therapists' Clinical Reasoning and Decision-Making Processes When Mobilizing Patients Who Are Critically III: A Qualitative Study. Amundadottir et al. (2018) Iceland and Canada	To investigate factors influencing physiotherapists' clinical reasoning and decision-making processes when mobilising patients who are critically ill	Twelve physiotherapists were observed before, during, and after a mobilization session with one patient, followed by a semi- structured interview.	Six categories; the context of intensive care unit, patient, transfer, physiotherapists, and FITT parameters which includes frequency, intensity, type, and time and expected outcome. In addition to four factors that includes patients and physiotherapists' safety and well-being; continuous assessment and intervention intertwined; individualized and response-driven intervention; and barriers and solutions) which emerged as significant in influencing participants' clinical reasoning when mobilising their patients.
9. Clinical reasoning in osteopathy: Experiences of novice and experienced practitioners. King et al. (2018) Australia.	To understand clinical reasoning in osteopathy from the perspectives of both experienced and novice practitioners.	Semi-structured interviews and reflection on the clinical reasoning processes practitioners used in response to two case scenarios.	Two themes existed: (1) different approaches to clinical reasoning in experienced, and novice osteopaths, and (2) observation and palpatory findings from direct contact with the patient are considered experiential findings and were integral to clinical reasoning in osteopathy. Clinical reasoning in osteopathy is like that of other health professions and clinical experience were important factor in developing sounds clinical reasoning and decision making processes.

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
10. The clinical reasoning among master students specializing in	To describe the clinical	A semi-structured interview OF 19 master's students.	One main category: "A multidimensional picture of clinical reasoning" and three subcategories: 1) Confidence in the students' role as physiotherapist; 2)
Orthopaedic Manual Therapy. Vedin, Elin (2021) Sweden	reasoning processes of orthopaedic manual therapy physiotherapy students in a master's programme		Decision making, a cognitive analytical process 3) Creating therapeutic alliance and patient involvement in clinical reasoning and decision making.
11. The skills, knowledge and attributes	To understand the roles and	Eight health practitioners' views	The themes identified were medical assessment and systems knowledge; speed
needed as a first-contact physiotherapist	the clinical practice of	via a stage one think aloud	of thought in an uncertain environment; breadth of knowledge; people and
in musculoskeletal healthcare.	musculoskeletal	interview process. This was	communication skills; common sense/simplify; and responsibility and
Langridge N (2019) UK.	physiotherapists in the primary care.	followed by a stage two focus group involving physiotherapists and a general practitioner trainer.	experience.

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
12. Clinical reasoning and critical reflection in physiotherapists' examinations of patients with low back pain in its early phase: a qualitative study from physiotherapists' point of view. Karvonen et al. (2017) Finland.	To increase understanding of physiotherapists' clinical reasoning and clinical reflections on their decision making when managing low back pain patients.	Semi-structured interviews with six physiotherapists	The results showed that physiotherapists used a hypothetico-deductive reasoning model and reflected on their decisions related to their low back pain patient's examination, from the history to the diagnosis and on the classification of non-specific low back pain.
13. 13.Core components of communication of clinical reasoning: A qualitative study with experienced Australian physiotherapists. Ajjawi and Higgs, 2012 p.107)	- 'To explore the nature of communication of clinical decision making with patients and students in physiotherapy practice.	'N= 12 experienced Australian physiotherapists working in cardiopulmonary (Hopton <i>et al.</i>), musculoskeletal (MS) or neurological (N) physiotherapy-	Communicating clinical reasoning was found to be dynamic, complex, context- dependent and often automatic and habitual. 'Five themes were found: active listening, framing and presenting the message, matching the co-communicator, metacognition and decision making ability'.
(UK)	-To explore the perceived core processes of communication of decision	Experience= ranged from 6 to 26 years	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
	making by experienced physiotherapists.	Gender= Four of the participants were male, eight were female'	
 14. Clinical reasoning and patient- centred care in musculoskeletal physiotherapy in Portugala qualitative study Cruz, et al., (2012 p246) (Scotland) 	'To explore clinical reasoning processes by experienced Portuguese musculoskeletal physiotherapists. The focus is on clinician's interaction with their patients in order to define or manage clinical problems.	'N= Four physiotherapists with musculoskeletal background'. Designs: an interpretative hermeneutic research approach Data collection: observation, semi-structured interviews, patients 'chart, memos and researchers' field notes.	 'Findings showed that the clinicians were more likely to carry out their reasoning approach as a purely cognitive and clinician centred process. Perspectives of clinical reasoning therefore differs between cultures and contexts of practice and this has potential implications in the way physiotherapists interpret health and illness, their role as clinicians, as well as the patient physiotherapist relationship. five themes appeared: Finding the source of patients 'complaints; Taking control of patient's clinical history; Excluding the
		Thematic analysis to explore and interpret clinical practice and reasoning.	Patient from the decision- making process; A biomechanical under-standing of the patient's problem'.

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
15. Implementation of Shared Decision- Making (SDM) in Physical Therapy: Observed Level of Involvement and Patient Preference. Dierckx et al. (2013 p1321) Finland.	'To examine the status of shared decision making in physiotherapists, patients' preferred levels of involvement, and the agreement between physiotherapists perception and patient preferred level of involvement'.	N= 13 physiotherapists Design. This was an observational study of real consultations in physiotherapy Data collection: 237 consultations were audio recorded, and 210 records were analysed using the Observing Patient Involvement (OPTION) instrument was used to measure patient involvement during the decision making process. Before the consultation, the patient and therapist completed the Control Preference Scale (CPS) to assess therapist's perceptions about patient's preferences for decision	. 'The mean OPTION score was 5.2 (SD=6.8), out of a total score of 100. Female therapists achieved a higher OPTION score (b=-0.86, P=0.01). These authors found that female general practitioners have a different communication style and are more likely to talk about psychosocial problems and encourage questions from patients. In total, 36.7% of the patients wanted to share decisions, and 36.2% preferred to give their opinion before delegating the decisions'. 'In the majority of cases, therapists believed that they had to decide'. 'The kappa coefficient for agreement was poor at .062 (95% confidence interval=018 to .144)'. The majority of patients preferred to share decision making with their physiotherapists, and a substantial percentage of the patients wanted to express their opinion about treatment options before the therapists made a decision. In contrast, approximately, one third of the therapists assumed that patients preferred to delegate the whole decision making process to the therapist, whereas only 1 out of 6 patients reported this preference

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		making and patient's preferred	
		levels of involvement.	
		Data Analysis: Multilevel	
		analysis was used to study the	
		association between individual	
		variables and the level of shared	
		decision making. Agreement on	
		preferences was calculated using	
		kappa coefficients	
16. Clinical reasoning of expert and	'To investigate the decision	Ten experienced clinicians with a	. 'The subsequent analysis showed that all used an H-D reasoning process.
novice physiotherapists in an outpatient	making of expert and novice	minimum of ten years'	However, the experts and novices went beyond this essentially diagnostic
orthopaedic setting. Doody and	physiotherapists in an	experience (also had postgraduate	process to include reasoning focused on treatment. In particular, manual
McAteer, (2002 p258). (London)	outpatient orthopaedic	training in manual therapy) and	therapy treatment was used as a method of further hypothesis testing. In
	setting'. No studies to date	ten undergraduate students (5	addition to H-D reasoning the experts also made use of pattern recognition'.
	have investigated the clinical	fourth year and 5 third year also,	'The clinical reasoning of the physiotherapists in this study was found to be a
	reasoning of physiotherapists	they were taking orthopaedic	dynamic, cyclical process. The results of the study support the model of clinical
	as they have assessed and	outpatient rotation of their	reasoning proposed by Jones'.
		clinical experience program).	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
	treated a real and previously		
	unseen patients.	Data collection: The study took	
		place in two different hospitals	
		and six private physiotherapy	
		practices. Participants were	
		observed and audiotaped as they	
		examined and treated a real,	
		previously unseen patient. In	
		addition to field notes and think	
		aloud method was used	
		retrospectively with stimulated	
		recall to obtain the therapists	
		verbal protocol. A semi-	
		structured interview was carried	
		out where the audiotape was	
		stopped. This review session was	
		also recorded.	
		Data Analysis: The transcript	
		(audiotape, field notes and	
		comments) was analysed. They	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		use member check by physiotherapists who were not participants to check the inter and intra-coder reliability	
17. Clinical reasoning strategies in physical therapy. Edwards et al. (2004 p312) (Australia)	'To examine the decision- making of expert physiotherapists in 3 different Fields of physiotherapy: orthopaedic (manual), neurological, and domiciliary)' care (Home health)'.	Design: 'a grounded theory method, a multiple case study approach was used to study the clinical practice of the 6 physiotherapists in the 3 fields (primary sample). This approach is similar to that used by Jensen et al. (1992)' Data Collection: '79 hours of	They found that they are using a range of clinical reasoning skills or strategies representing a diversity of thinking and actions during diagnosis and management Within these clinical reasoning strategies, the Application of different paradigms is termed "dialectical reasoning'.
		observation to 6 physiotherapists were audiotape and transcribed. Then semi-structured interviews with secondary sample which involved 6 physiotherapists with	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		teaching experience. In addition	
		to field notes and they asked the	
		primary sample to write personal	
		reflection about the source of	
		knowledge they have (e.g.,	
		mentors, clinical and life	
		experiences) and factors and	
		people whom they considered	
		important influences on their	
		professional lives'.	
		Data analysis: 'comparative	
		analysis. Case studies were sent	
		to the respective therapist from	
		the primary sample for his or her	
		comments, a process known as	
		"member checks", on follow-up,	
		which was done by a telephone	
		discussion'.	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
18. Ethical reasoning as a clinical-	'This paper proposes a model	Similar to the study above	Similar to the study above.
reasoning strategy in	for integrating ethical		
physiotherapyEdwards and	reasoning into a wider		
Braunack-Mayer (2005, p229)	clinical reasoning framework		
(Australia)	without reducing the		
	complex, moral dimensions		
	of ethical reasoning to merely		
	logical and rational processes		
	of clinical reasoning'.		
	-To examine the nature and		
	scope of clinical reasoning of		
	expert physiotherapists		
	working in different		
	physiotherapy settings.		

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
19. Ethically-based clinical decision making in physical therapy: process and issues. Finch et al, (2005 p.147) USA	'To explore the way in which physiotherapists integrate ethical issues into clinical practice decisions and identify ethical themes used by physiotherapists.	 N=8 physical therapists using purposeful sample. Data collection and Data Analysis: Participants describe a recent ethically based clinical decision (they were asked to describe scenarios where they consider ethical value to be important in their decision making). -Transcribed interviews were coded and themes identified. Methodology: use of hermeneutic phenomenology, which interprets participant's descriptions of the 	The following categories were identified: 1) the integration of ethical issues in the clinical decision making process, 2) patient welfare including the themes of treatment effectiveness, patient autonomy and quality of life, 3) professional ethos of the physiotherapy including the theme of professional role and boundaries, patient advocacy, and professional collegiality, and 4) health care economics and business practices'. Including the themes of conflict of interest, funding, allocation of scares resources and endorsement of equipment 'Participants readily described clinical situations involving ethical issues but rarely identified specific conflicting ethical issues in their description. Ethical dilemmas were more frequently resolved when there were fewer emotional sequelae associated with the dilemma, and the PT had a clear understanding of professional ethos, valued patient autonomy, and explored a variety of alternative actions before implementing one'.

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		phenomenon of interest and reflects on it is essential meaning.	
20. Clinical decision- making in exercise prescription for fall prevention. Haas et al, (2012, p.666) (Australia)	'To describe the factors influencing the clinical decision making processes used by expert physical therapists to prescribe exercises for fall prevention'.	N=24 physiotherapists Design: This investigation was a qualitative study from a phenomenological perspective. Methods: 'Semi-structured telephone interviews were conducted with 24 expert physiotherapists recruited primarily from the Victorian Falls Clinic Coalition. They did five to six interviews in each group, and no additional themes were identified, and no additional themes identified means they reach saturation. Interviews	'Participants described highly individualized exercise prescription approaches tailored to address key findings from physical assessments of client's physical ability, cognition level and therapist time constraints. 'Dissonance between prescribing a program that was theoretically correct on the basis of physiological considerations and prescribing one that a client would adhere to exercise was evident and the level of motivation'. 'Safety considerations also were highly influential on the exercise type and setting prescribed'. In addition to research evidence and theoretical knowledge to rationalise the quantity of the exercise. Regarding external constraints, knowledge about the effective dosage for effective therapy was derived from clinical experience, research evidence and professional training

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		focused on 3 exercise	
		prescription contexts: face-to-	
		face individual therapy, group	
		exercise programmes, and home	
		exercise programmes.	
21.Shared decision making (SDM) in	'To measure the prevalence	Forty-two patients aged 18 years,	The mean OPTION score was 24.0 % (range 10.4–43.8 %). Shared decision
back pain consultations: an illusion Or	of shared decision making in	referred with back pain and 12	making was under-developed in the observed back pain consultations.
reality? .Jones, (2014 pS13) (UK)	clinical encounters involving	physiotherapists involving novice	Clinicians' strong desire to treat acted as a barrier to shared decision making
	physiotherapists and patients	staff and experts	and further work should focus on when and how it can be implemented'.
	with back pain'.		
		Data analysis: 'audio-recorded	-Physiotherapists did not explain the risk and benefits of the treatment options.
		were observed, transcribed	
		verbatim and analysed using the	-few physics asked patients about their preferred level of receiving information.
		12-item OPTION scale. (The	· · · · · · · · · · · · · · · · · · ·
		transcripts were analysed using	
		OPTION scale, which measure	-Patients views and expectation were not discussed at all.
		the overall shared decision-	
		making process from the	
		observer prospective in the initial	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		and follow-up). It measures the	
		overall shared decision making	
		process and is unique in	
		comparison with other	
		instruments as it scores the	
		clinician-initiated behaviour from	
		an observer's perspective'. 'The	
		revised instrument rates 12	
		behavioural items (mirroring the	
		core concepts of shared decision	
		making fundamental to good	
		clinical practice)on an ordinal	
		scale, ranging from zero—"the	
		behaviour is not observed", to	
		four—"the behaviour is observed	
		and executed to a high standard".	
		'Scores are summated and scaled	
		to give a percentage score.	
		The higher the score. the greater	
		is the shared decision making	

1 itle/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		competency attained, with 60 %	
		generally accepted to correlate	
		with the lowest meaningful	
		competency level by the shared	
		decision making'.	
		Example of OPTION scale	
		scores:	
		50005.	
		-The clinician explains the pros	
		and cons of options to the patient.	
		-The clinician explores the	
		patient's concerns (fears) about	
		how problems are to be managed.	
22. Factors that influence the clinical	'The aims of this study were:	Eleven therapists were	'A highly individualized approach to patient examination based on therapists'
decision making of physiotherapist in	to explore decision making	purposefully selected (6 from	
choosing a balance assessment	during examination of	outpatient offices. 5 from	practical knowledge emerged from the data, with limited influence of the
			provident knowledge enterged from the data, with initial initialities of the

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
approach. McGinnis et al., (2009 p233)	patients with balance deficits,	inpatient rehabilitation settings to	
(USA)	to understate the selecting of	participate in repeated interviews.	Literature'. 'Movement observation was the primary assessment and diagnostic
	assessment methods from the	Participants with one-year	tool'.
	clinician s perspective, and to	experience and former students	
	explore why specific methods	were excluded. Methods: 'A	
	were selected'.	qualitative design using a	When selecting assessment approaches for specific patients, the perceived
		grounded theory approach	validity of information gathered mattered more than testing time'.
		permitted exploration of clinical	
		decision making.	
		Data collection:	
		Repeated interviews (initial was	
		about examination of patients	
		with balance deficits, the second	
		interview focused on	
		participant's description of their	
		clinical decisions during a	
		specific patient case). In the	
		second interview participants	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		completed a demographic data	
		form that included questions	
		related to education, practice	
		experience, certifications,	
		memberships in professionals'	
		organizations, continuing	
		education, and identification of	
		any resources used (e.g.,	
		textbooks, journals) for balance	
		assessment.	
		Data analysis: constant	
		comparative analysis was used	
		whereby data collection and	
		analysis were ongoing. Data	
		sources included interview	
		transcripts, results of sorting	
		activities, field notes, reflective	
		memos, and comparison with	
		expert opinion. Each transcript	
		was reviewed line by line to gain	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		understanding of the raw data	
		(open coding)	
		'Credibility' of the findings were	
		established low-inference data,	
		member check, and triangulation	
		among participants and multiple	
		data sources'.	
23. Expert therapists use specific	'What are the key items in	N=Twenty-six experts in the UK	'These expert clinicians demonstrated the use of diagnostic pattern recognition,
clinical reasoning processes in the	the clinical reasoning process	(physiotherapists and	and H-D and narrative clinical reasoning and patient's centred and
assessment and management of patients	which expert clinicians	occupational therapists)	collaborative reasoning processes. 'The emphasis was on the history and basic
with shoulder pain: a qualitative study.	identify as being relevant to	consented to be involved and	physical examination procedures to make clinical decisions. The use of both
May et al., (2008 p261) (Australia)	the assessment and	were contactable, of whom 20	types of clinical reasoning by these clinicians confirms the dialectical model of
	management of patients with	contributed, with 12, 15, and 15	the two
	shoulder pain?'	contributing to the different	
		rounds. Design: Qualitative study	
		using a three-round Delphi	
		procedure.	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		Round one: unstructured and	
		allowed multiple open responses.	
		The clinicians were asked to	
		share the clinical reasoning	
		process by which they assessed	
		patients with shoulder pain and	
		arrived at a management strategy.	
		Data Analysis:	
		Qualitative themes analysis, in	
		which items were grouped	
		together through shared themes	
		relating to the narrative reasoning	
		process of shoulder assessment,	
		classification, and management.	
		In Round 2: respondents were	
		asked to rank the relative	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		importance of all the items using the scale (1Essential, 2very important, 3 important, 4less important, 5unimportant).	
		In Round 3, respondents voted on their agreement or disagreement with the ranking that emerged from Round 2. All correspondence was via e-mail.	
24. Limited clinical reasoning skills used by novice physiotherapists when involved in the assessment and management of patients with shoulder problems: a qualitative study. May et al. (2010 p.84) (UK)	'To explore the clinical reasoning process used by novice physiotherapist in specific patient problems.	N= Nine physiotherapist (three male and six female) in the UK from two NHS trusts in one city in the UK with limited experience of managing musculoskeletal problems were included Methods: 'Semi-	'Items mostly related to information gathering, although there was some use of H-D clinical reasoning there appeared to be limited understanding of the clinical implications of data gathered, and clinical reasoning through use of pattern recognition was minimal'. 'Major weaknesses were apparent in the clinical reasoning skills of these novice therapists compared to previous reports of expert clinical reasoning, indicating areas for development in the education of student and junior physiotherapists.

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		structured interviews were	
		conducted on how novice	
		Physiotherapist would assess and	
		manage a patient with a shoulder	
		problem'. They used a topic	
		guide from textbooks that include	
		history and physical examination	
		for musculoskeletal shoulder	
		problems.	
		Data analysis: interviews were	
		transcribed and analysed using	
		framework analysis. It was	
		decided before that in order for	
		themes or items mentioned by the	
		therapists to be included in the	
		final data analysis at least 50% of	
		participants has to have	
		mentioned the themes. In other	

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		words, the final themes would represent what the majority of novice physical therapists considered important in the evaluation of a patient with shoulder pain. For example, in history trauma or insidious onset were more common than time since onset and pain on resisted exercise in physical examination were less frequent	
25. Clinical reasoning process in physiotherapy. Payton (1985 p924) (USA)	to analyse physiotherapist's clinical Problem solving and compare the results with physicians' clinical problem solving. Descriptive study	N= Ten skilled (experts) physical therapy clinicians, suggested by academic faculty in physiotherapy Gender= 5 male and 5 females. One academic physio and others are clinicians. All are clinical educators Method: 'Participants were	 'The therapists defined their problem lists and developed treatment plans early in the interview, as they gathered data'. 'This clinical problem-solving sequence is comparable to a method reported in the literature that is used by physicians. 'This model of clinical problem solving based on actual performance of clinicians can be used to train physiotherapist students and, perhaps, to refine clinical evaluation skills.

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		observed as they performed an	
		initial interview with a patient.	
		Their performance was	
		audiotaped and later replayed the	
		tape to discuss it with each	
		participant. The therapist's	
		review of the patient's interview	
		was audiotape. In addition to	
		field notes and analysed'.	
26. Factors that influence the clinical	Aims: To identify differences	Three participant's pairs (each	The factors that influence clinical decision making were categorized as
decision making of novice and	in clinical decision making	pair consisted of one novice	informative or directive. Novice participants relied more on informative factors,
experienced physical therapists.	abilities and processes	physiotherapist and one	whereas experienced participants were more likely to rely on directive factors.
Wainwright et al. (2011) (USA):	between novice and	experienced physiotherapist are	
	experienced physical	purposefully selected from three	
	therapists' clinicians and to	inpatient rehabilitation settings.	
	develop a model of the	Design: Qualitative research	
	factors that influence clinical	methods and grounded theory.	
	decision- making		

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		Methods: Case summaries from each participant provided the basis for within and across-case analysis.	
27.Influence of the Therapist-Patient Relationship on Treatment Outcome In Physical Rehabilitation: A Systematic Review Hall et al. (2016)(Australia)	To investigate whether the working alliance (patients and physiotherapists relationship) is related to outcome in physical rehabilitation settings	Methods: A sensitive search of 6 databases identified a total of 1,600 titles. Prospective studies of patients undergoing physical rehabilitation were selected for this systematic review. Thirteen studies including patients with brain injury, musculoskeletal conditions, cardiac conditions, or multiple pathologies were retrieved.	 The results indicate that the alliance is positively associated with: (1) treatment adherence in patients with brain injury and patients with multiple pathologies seeking physical therapy, (2) depressive symptoms in patients with cardiac conditions and those with brain injury, (3) Treatment satisfaction in patients with musculoskeletal conditions, and (4) physical function in geriatric patients and those with chronic low back pain.

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
28. Clinical reasoning processes in	Explored the clinical	Outcome measure; working alliance inventory, the California scale and Vanderbilt scale N=14	'The findings revealed that physiotherapists used both hypothetic-deductive
physiotherapists' assessment of unilateral neglect (ULN): part two. Plummer et al.(2005 p533)(Australia)	decision-making processes used by 14 Australian neurological physiotherapists when assessing ULN after stroke.	<u>Methods:</u> Participants observed a videotape of an experienced therapist assessing a stroke patient with left ULN. The videotape was paused at intervals and participants recorded their thoughts pertaining to the patient at that time. Afterwards, physiotherapists were asked questions in relation to their clinical decisions and were instructed to classify the	and pattern recognition models of clinical reasoning'. 'Hypotheses concerning neglect were generated early in the assessment. Only a small number of physiotherapists were able to refine their neglect-related hypotheses in light of new information'. 'Types of neglect were rarely considered in the decision- making processes. A lack of specialized knowledge may limit the ability of physiotherapists to make detailed characterizations of neglect behaviour.' A glossary comprising definitions of the different types of neglect improved the ability of clinicians to accurately characterize neglect behaviour, suggesting that education may enhance diagnostic reasoning ability of physiotherapists in the assessment of ULN.'

Title/Authors/Year/Country	Aims	Methods	Results/ Conclusion
		behaviour of the patient using a glossary of diagnoses	

Appendix C Ethical approval in the UK

Ethical approval in the UK

Approved by Faculty Ethics Committee - ERGO II 31625

Submission ID: 31625

Submission Title: A qualitative study to explore the clinical reasoning processes among experienced women health physiotherapists in the UK and Kingdom of Saudi Arabia while managing patients with urinary incontinence

Submitter Name: Jawahr Alagil

Your submission has now been approved by the Faculty Ethics Committee. You can begin your research unless you are still awaiting any other reviews or conditions of your approval.

Comments:

Thank you for your clear and detailed responses. Good luck with this much needed study.

Appendix D

Ethical approval in KSA

Kingdons of Saudi Arabia King Saudi Universits (034) p.o., Box 7805 Ricadh 11472 Tel: +966 II 467 00 11 Fax: +966 II 467 19 92 http://mcdicalcity.ksu.edu.sa المملكة الغريبة السعودية جامعة الملك سعود (EE) ص.ب م NN الرياض TUBI هاتف: III.- VT3II TEP فاكس: TPPI VE3II TEP



17.10.2018 (08.02.1440) Ref. No. 18/0665/IRB

To:	Ms. Jawahr Hamd Alagil
	Department of Health Rehabilitation
	King Saud University College of Applied medical Science
	Email: jaqeel@ksu.edu.sa
	Principal Investigator
CC:	Dr. Hana Alsobayel
	Co-Investigator
Subject:	Approval of Research Project No. E-18-3348
Study Title:	"A Qualitative Study to Explore the Clinical Reasoning Processes among Experienced Women's Health Physiotherapist in the Kingdom and the Kingdom of Saudi Arabia while Managing Patients with Urinary Incontinence"
Type of Review:	Expedite
Date of Approval:	17 October 2018
Date of Expiry:	17 October 2019

Dear Ms. Jawahr Hamd Alagil,

l am pleased to inform you that your above-mentioned research project submitted to the IRB was reviewed and approved on 17 October 2018 (08 Safar 1439). You are now granted permission to conduct this study given that your study does not disclose participant's identity and poses no risk to the patients.

As principal investigator, you are required to abide by the rules and regulations of the Kingdom of Saudi Arabia and the research policies and procedures of the KSU IRB. If you make any changes to the protocol during the period of this approval, you must submit a revised protocol to the IRB for approval prior to implementing the changes. Please quote the project number shown above in any future correspondence or follow-ups related to this study.

This approval is for a period of one (1) year commencing from the date of this letter. If you wish to have your protocol approved for continuation, please submit a completed request for reapproval of an approved protocol form (KSU-IRB 017E) at least 30 days before the expiry date. Failure to receive approval for continuation before the expiration date will result in automatic suspension of the approval of this protocol on the expiration date. Information collected following suspension is unapproved research and can never be reported or published as research data.

We wish you success in your research and request you to keep the IRB informed about the progress and final outcome of the study in a regular basis. If you have any question, please feel free to contact me.

Thank you!

Sincerely yours,

Dr. Abdulrahman Alsultan Chairman of IRB Health Sciences Colleges Research on Human Subjects King Saud University College of Medicine P. O. B ox 7805 Riyadh 11472 K.S.A. Email: aalsultan1@ksu.edu.sa

in the second se	I Q I III	
Institution		
In Cost	Plage Medicinald	

/braezell

Appendix E Summary of the UK and KSA matrices

Table 7-7: illustrates charting of the data from the UK and KSA interviews and focus groups

This table illustrate the data in a form of data summary, coding, categories, and the themes of the UK and KSA interviews and focus groups transcripts

Pattern Recognition/UK Participants	Data summary	Dimension identified (Codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within
P8 UK	Patients subjective because she was leaking with exercise and if she got to go, she got to go that is a clear mixed incontinence.	Patient's symptoms	Straight forward case lead to clear diagnosis	Making diagnosis
P3 UK	It depends on the connection. So, there's certainly more to muscle health than just how long you can hold a contraction for. But there's no science behind that. That's what	Physiotherapist experience	physiotherapist experience influences clinical reasoning and decision making	physiotherapist experience influences clinical reasoning and decision making
P6 UK	Different between MSK and PHPT. The problem is mainly patient's symptoms. Asking questions help me to make diagnosis.	Patient's symptoms. Asking questions	Patients complains, symptoms and asking questions lead to the diagnosis	Making diagnosis is different in physiotherapist

Illustrates charting of the data from the UK and KSA interviews and focus groups in a form of data summary, coding, categories and the themes of the UK and KSA interviews and focus groups transcripts

Pattern Recognition/UK Participants	Data summary	Dimension identified (Codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P7 UK	I draw on my years of experience to solve uncertainties. I have been working in the field for about 25 years (participant's words)	Physiotherapist experience	physiotherapist experience influences clinical reasoning and decision making	making decision based on experience
P11 UK	Physiotherapists experience help her to decide that vaginal examination is not important in compared to ultrasound to assess patients	Physiotherapist's experience	Experience helps physiotherapists to make diagnosis.	Making decision based on experience
P13 UK	Physiotherapists are treating patient's symptoms whatever the diagnosis is.	Patient's symptoms	Treating patient's symptoms whatever the diagnosis	Treating the symptoms not the diagnosis

Pattern Recognition/UK Participants	Data summary	Dimension identified (Codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P4 UK	Patients is having features of both types of incontinence, in addition to obvious characteristics of an overactive bladder.	Features and characteristics	Different features and characteristics lead to the diagnosis	Making diagnosis
	I think I'm a lot more experienced. And it became obvious because I know my stuff, it became obvious that something wasn't auite	Physiotherapist's experience	Physiotherapist's experience leads to the	Physiotherapist's experience helps in making diagnosis.
P3 UK	Patient is restricting her fluids and had some night wetting.	Patient's symptoms	Patient's symptoms	Making diagnosis
P9 UK	Understanding the effect of episiotomy and forceps delivery on UI and this is because of her experience	Physiotherapist's experience helps in understanding the relationship between UI and other problems.	Physiotherapist's experience	Making decision based on experience

Pattern Recognition/UK Participants	Data summary	Dimension identified (Codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P10 UK	Patient almost tell you within the first period. Subjective assessment gives information for the clinical diagnosis.	Patient's subjective give clinical diagnosis	Patient's subjective	Making diagnosis
P12 UK	A lot of the symptoms are presenting that showed patients are getting stress incontinence.	Patient's symptoms lead to diagnosis	Patient's symptoms	Making diagnosis
FG1 UK	The decisions that are made by therapists vary depending on the experience and the knowledge that the therapist has. The more junior members that you might be working wouldn't have the same level of experience and the decision is more of a challenge. Junior staff might feel that treating pregnancy is everything is contraindicated; this is what they have been taught in the University and then we	Physiotherapist experience and clinical reasoning and decision making	physiotherapists experience	making decision based on experience
Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
---	--	---	--	--
PI UK	Patient was a complex case; the complexity came more from psychological than anything else. Doing training in psychosexual medicine help me to think about factors beyond the physical.	Psychological factor can be a complex factor.	Relation between identifying psychological factors and physiotherapist's training.	Patient's psychological factors and physiotherapist's knowledge.
	Patient is communicating with me via email to ask questions. I told them to do that Thinking of a bit of	Physiotherapist communicating with patients and building a good relationship with the patients.	Physiotherapist's skills and relationship with patients.	Physiotherapist and patients working alliance.

Illustrates charting of the data in a form of data summary, coding, categories and the themes of the UK and KSA interviews and focus groups transcripts

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P2 UK	I need to be aware of complex psychosocial issues, attending psychosexual medicine seminar help me to think about the dynamic type of treatment in relation to sexual	Complex psychosocial issues can be related to sexual problems.	Relationship between identifying psychosocial factors and physiotherapist's training.	Patient's psychosocial factors and physiotherapist's knowledge.
	problems. You need to give people the time to open up about sexual problems. (Participant's own words).	Patients need time to open up about sexual relationship.	Gaining patient's trust takes time.	Physiotherapist making sense of the important of gaining patient's trust to disclose intimate issue.
	Physiotherapist is being a good	Physiotherapist's personal	Physiotherapist's skills and	Physiotherapist and patients working alliance.
P3 UK	Physiotherapist was interpreting patient's perception of the world, anxiety and not feeling safe as driving her bladder. Physiotherapists think that she is treating the patient as a whole and not being vaginal	Complex psychosocial issues can affect patient's bladder.	Treating the patient as a whole and not being vaginal centric.	Not being vaginal centric.

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P4 UK	Patient was anxious and stressed at work, this might affect patient's bladder behaviour at work compare to home.	Complex psychosocial issues can affect patient's bladder	Physiotherapist depend on her instinct and sense making to identify patient's psychological problems.	Patient's psychosocial factors and physiotherapist's instinct or sense making
	There is lack of time to complete questionnaire to assess patient's psychological problem and I am good at picking up anxiety issues.	Lack of time to fill psychological questionnaire but relying on intuition.	Intuition	Organisational factors can be a barrier in decision making using intuition instead
	Relying on talking to them, hearing them and caring about them, give my email and working on instinct rather than anything precise. Being interested in patients will validate their symptoms and make them feel differently.	Communication Building a good relationship	Physiotherapist's skills and relationship with patients.	

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	I quite often have a chat with them. This will make them feel like they're sort of friends rather thansitting at the other side of the desk (Participant's own words).	Making patient feels sort of friend	Patient-physiotherapist relationship	Sense making and making patients feel included. Physiotherapist and patients working alliance.
P5 UK	Using questionnaires may help in picking up more of the emotional side, but it does not make difference to the treatment.	Psychological questionnaire	Psychological questionnaire will not affect physiotherapy treatment	Psychological factors will not affect physiotherapy treatment.
	Most of its talking and listening. (Participant's own words).	Communication Building a good relationship	Physiotherapist's skills and relationship with patients.	Physiotherapist and patients working alliance.

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P6 UK	Patient's job that involve heavy lifting with weak pelvic floor muscle might lead to pelvic organ prolapse and urinary incontinence.	Social factors can affect bladder function.	Patient's factors can contribute to pathological problems.	Patient's social factors related to pathological problems.
	I am always try to be reassuring, as I do with all patients there (Participant's own words).	Reassuring patients	Patients-pelvic health physiotherapist's relationship.	Physiotherapist and patients working alliance.
P7 UK	Patient anxiety level from social relationship and work stress affect her bladder symptoms. Physiotherapists think that attending training in sexual counselling, running a home program helped her to follow a collaborative and holistic approach.	Patient's anxiety Physiotherapist's knowledge	Relationship between identifying psychosocial factors and physiotherapist's training.	Patient's psychosocial factors and physiotherapist's knowledge.

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	I think she knows that I will listen to her. So I was quite keen to try and be empathetic (Participant's own words).	Listening and be empathetic.	Patients-pelvic health physiotherapist's relationship.	Physiotherapist and patients working alliance.
P9 UK	Trust is a mutual relationship between patients and physiotherapists. Patient gain confidence to talk and can burst into tears because physiotherapist can be the first person that is ever interested in their problems.	Mutual trust Disclosing personal and intimate issues.	Mutual trust lead to building rapport	Not being vaginal centric can help in building rapport.
	This might be because of physiotherapist's age and experience in addition to spending long time with patients compare to other health practitioners.	Physiotherapist's skills and time Building rapport.	Physiotherapist's skills and organisational factors lead to building rapport	

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P10 UK	Patient pelvic floor was high in tone and this might be due to psychological state.	Patient's factors can affect pelvic floor muscle function.	Patient's factors can contribute to pathological problems.	Patient's social factors related to pathological problems
	The hospital is setting a series of questionnaires about patient's	Psychological questionnaire	Organisational factors can facilitate physiotherapist's decision on patient's psychological factors.	Organisational factors as a facilitator
	psychological state and their mental health, patients can fill in this in the waiting area and this would allow me to know patients before walking through the door.	Building rapport.	Building rapport.	Awareness of patient's characteristics lead to sense making on the best way to build rapport.
	I eventually managed to build a rapport with her. So they need to have a trust in you (Participant's own words).		Patients-pelvic health physiotherapist's relationship.	

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	I've just realised how important it is to make sure you're listening. They need to make sure they feel listened to (Participant's own words).	Active listening		Physiotherapists and patients working alliance.
P11 UK	She's been much stressed in the last three years because she's lost her husband. That gives you an idea of how anxious she is. Being a widow make her ignore herself until the symptom become worse (participant's own words).	Social factors	Patients' factors can contribute to pathological problems.	Patient's social factors related to pathological problems.
	I think the talking actually, because I understand it now and somebody is listening to me, you've just got to keep listening to your patient. I was	Communication Building a good relationship	Physiotherapist's skills and relationship with patients.	Physiotherapist and patients working alliance.

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	able to email a patient with faecal incontinence the details of those sprays <i>(participant's own words)</i> .			
P12 UK	I would advise her to seek some psychological support for her to try and help to manage some of those other issues	psychological support	Communication with other healthcare practitioners.	Not being vaginal centric.
	It's only sometimes two or three appointments down the line when they trust you, when they will talk to you and they will open up that you start to unpick some of what's going on (participant's own words).	Intimate issues.	Gaining patient's trust take time.	

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	It's about giving them that opportunity to tell the picture from their perspective. And actually, giving them the space to talk back that far through. (Participant's own words).	Time to talk.	Patients-pelvic health physiotherapist's relationship.	Physiotherapist making sense of the important of gaining patient's trust to disclose intimate issue. Physiotherapists and patients working alliance.

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P13 UK	Sometimes you'll need to support somebody a lot more emotionally. And my colleagues often laugh at me and say that I treat a lot more from the neck up than the waist down	Supporting patients emotionally.	Considering patient's psychological issues.	Treat a lot more from the neck up than the waist down. Providing a safe environment.
	So, providing a safe environment both physically and emotionally for the patient to discuss their issues is important (participant's own words).	Disclosing important issues.	Disclosing important issues	
	She is the one who needs to be listened to. Its patients who present the problem. So, it's important to work around the patient's need	Active listening	Patients-pelvic health physiotherapist's relationship.	Physiotherapists and patients working alliance.
FG1 UK	I have to be open minded non- judgmental; the key difference is empathy and has the ability to	active listener	shared decision making	Plausibility of decision making

Collaborative sense making/ psychological issues/ UK Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	communicate and not embarrassed			
	easily. I have never had a patient say	showing empathy		
	to me you have changed my life like I			
	have in pelvic health. we are dealing			
	with patients who are exposed to a	Disclosing any sensitive issues		
	lot of trauma in this area either			
	childbirth or abuse, so we do need to			
	have more skills to deal with patient			
	than if you are working in sports,			
	more vulnerable area of the body.			
	We need to make the patient feel			
	more comfortable enough to be			
	happy to share what they would			
	consider be absolute secret thing			
	that have never shared with anyone			
	else.			

Collaborative sense making/ Discussing different options and harms to management /KSA Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P1 KSA	There are different options and I always encourage my patients to share what they think.	Being open to patient's preference.	Discussing different options.	Patients and physiotherapists working alliance.
P2 KSA	The most common problem with my patient that they don't want to have internal examination. So I accept their decision and I do external examination for them and I said to them that he can only offer exercise in general while if you accepted to have internal	Physiotherapist interpersonal characteristic accepting patient opinion	shared decision making	Sense making of clinical reasoning and decision making
	examination then I would be able to be more specific and give you extra			

Illustrates charting of the data in a form of data summary, coding, categories and the themes of the UK and KSA interviews and FGs transcripts

Collaborative sense making/ Discussing different options and harms to management /KSA Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	things in your pelvic rehabilitation			
	program.			
P3 KSA	She is sharing in decision making; she is not only a woman or a lady or an individual that you just give instructions without any feedback. We call it person-centred relationship, it is not patient- therapist relationship only, so it is a person (Participant's words)	giving instructions feedback from patient	Sharing in decision making.	Patients and physiotherapists working alliance.

Collaborative sense making/ Discussing different options and harms to management /KSA Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P8 KSA	I accept if patient do not like to have internal examination and I am encouraging them to do maintenance exercise and I progress the exercise later on. Some of the patients might accept later on.	Accepting patient's request adjusting the treatment program Patient's preference.	Changing the treatment program based on patient's needs.	Patients and physiotherapists working alliance.
P4 KSA	If patients want to continue with us, they have to remove the IUD and use any other contraceptive ways. Also, I explain to them that I cannot teach you the exercises until I do vaginal examination. So, when I make them afraid, they will do what	change their contraceptive treatment Have to have vaginal examinations.	Forcing the patient to follow physiotherapist advice	Paternalistic decision making

Collaborative sense making/ Discussing different options and harms to management /KSA Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P5 KSA	If patient does not accept vaginal	There are no other options to vaginal	Forcing the patient to follow	Paternalistic decision making in assessment and
	examination usually, I provide them	examinations	physiotherapist advice in	inform decision making in treatment.
	with exercise and discharge them.		assessment. Helping them to choose	
	This is an agreement between me	Detiont's culture	the treatment modalities.	
	and the referring physicians.	Fallent's culture		
	Regarding treatment options, I			
	inform them with the treatment	Trust health practitioner's decisions.		
	options, and I told them what is best			
	for them. Because most patients are			
	not confident to choose the			
	treatment modalities and they ask			
	me about my opinion.			

Collaborative sense making/ Discussing different options and harms to management /KSA Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
P6 KSA	I wouldn't do anything without their permission, never. They should be cooperating with me for example I want to use dry needling and I think needles are going to give her a good result but if she doesn't want it I can't force it. So, I have to use another treatment method. So, whenever you want me to stop [vaginal examination], I'd stop because all of them come with a bad history that I do the examination in the labour room (Participant's words).	Accepting patient's request Patient's preference.	Changing the management program based on patient's needs.	Patients and physiotherapists working alliance/ shared decision making
P7 KSA	Usually I do the vaginal examination in the second session because I am expecting the patient	Patient's preference.		

Collaborative sense making/ Discussing different options and harms to management /KSA Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	to be not ready or she had signs of infections but if I have time and patient accepted to assess her pelvic floor muscles in the first session. I'll do it.			
P8 KSA	If something is not giving harm to the patient, and is affecting the patient in a good way, just use it. Because you don't know what will work with the patient, either manual therapy, your instructions, the chair, patient's psychological aspect, something will work with her. Just do your best and give it a try with many chances. The good thing is that we as physical therapists we are not doing something harmful to the patient, so even if you are doing something and it is not working	different options different modalities limited experience and knowledge NOT involving patients in the decision.	What is next?	Informed decision making.

Collaborative sense making/ Discussing different options and harms to management /KSA Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	with the patient, it is not a big deal. You can fix it with another exercise or something. And the doctors are always here			
P9 KSA	I always give her the chance to change or to choose another thing and I give her alternatives. I'm not the one who is choosing for her, in order to make her feel comfortable and complied with what she does. Sometimes patients are accepting to insert the perineometer by themselves, some of them say no I don't want to, the very old ladies resist these things. So, I just do the external landmark, and start to encourage them to do the exercise properly while I'm just observing they are not substituting the other	different options patient's characteristics physiotherapist's knowledge Resources such as perineometer.	Relation between patient's characteristics and physiotherapist's skills in providing different options.	Patients and physiotherapists working alliance.

Collaborative sense making/ Discussing different options and harms to management /KSA Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	know (Participant's words).			
P10 KSA	Treating MUI by separating that to urge and stress incontinence. First treating urgency with posterior TPNS for 12 sessions then followed by EMG for another 12 session's then concentrate on the endurance training. Then I give patient one- month chance to return to physiotherapy in case she needs further management. This is because it is quite difficult to have another referral from the physicians to physiotherapy.	Providing patients with different options A lot of exercises. Limited experience and knowledge NOT involving patients in the decision.	What is next?	Informed decision making.
P12 KSA	If that physiotherapist is not confident or comfortable, they the	Physiotherapist skills	Physiotherapist skills can influence patient behaviour	Patient and physiotherapist factors

Collaborative sense making/ Discussing different options and harms to management /KSA Participants	Data summary	Dimension identified(codes)	Categories (labels) interpret data under broader heading	Clusters, patterns of association or linkage within the theme and across the data set.
	patient will pick up on that and might don't accept vaginal examination.	patient behaviour		

Appendix F Invitation email

Invitation email

Invitation e-mail will be send separately to General Secretary of the following:

Pelvic, Obstetric and Gynecological Physiotherapy (POGP)

Or

Saudi Physiotherapy Women's Health Association (SPTAWH)

Or

International Uro-gynaecology Association (IUGA) Or Association for Continence Advice (ACA) Or International Continence Society (Health et al.)

Delete as appropriate

I am a PhD student at the University of Southampton. For my thesis I am conducting a qualitative study to explore the similarities and differences in the clinical reasoning processes of experienced WHPTs in the UK and KSA while assessing and treating patients with urinary incontinence (UI). The study aims are to explore WHPTs clinical decision-making processes, in two countries (UK and KSA) while assessing and treating patients with UI. I would like to recruit physiotherapists with a minimum of one year experience treating urinary incontinence patients within your association. If this is possible, could you help me to do this by sending the invitation letter and the participant information sheet to your members please?

My ERGO/Ethics approval number is: 31625

Please do not hesitate to contact me for any other details. I appreciate your kind support. Sincerely, Jawahr Alagil PhD Student Members of SPTWHA Principal Researcher E-mail: <u>i.h.alagil@soton.ac.uk</u>

Supervisors:

Dr. Sarah Demain

Faculty of Health Sciences Student Office University of Southampton Highfield Southampton SO17 1BJ

Email: s.h.demain@soton.ac.uk

Room Number: 67/4061

Dr Ellen Kitson-Reynolds

Faculty of Health Sciences Student Office University of Southampton Highfield Southampton SO17 1BJ

Room Number: 67/3019

Appendix G Invitation Letter

Invitation Letter

Dear Members of the Charted Society of Physiotherapy (CSP) or

UK Pelvic, Obstetric and Gynaecological Physiotherapy (POGP)

I am a PhD student at the University of Southampton working on my thesis entitled: -

"A qualitative study to explore the clinical reasoning (CR) processes among experienced pelvic health physiotherapists (PHPT's) in the United Kingdom (UK) and the Kingdom of Saudi Arabia (KSA) while managing patients with urinary incontinence (UI)"

My ERGO/Ethics approval number is: 31625

The objectives of the study are

- To identify and understand the similarities and differences in clinical reasoning and decision making between pelvic health physiotherapists in the UK and KSA.

-To explore the factors influencing clinical reasoning and decision making in the management of UI.

- To identify the clinical reasoning models being used during clinical reasoning and decision making in UK and KSA.

-To contribute towards the understanding of future development of theoretical frameworks of clinical reasoning and decision making specific to pelvic health physiotherapists in the UK and KSA.

I would like to invite you to participate in this study if you are

- Pelvic health physiotherapist's who have been *OR* who are currently registered with the Health and Care Professional Council (HCPC) in the UK or the Saudi Commission for Health Specialities (SCHS) in KSA. This study seeks to hear the views of physiotherapists with sufficient pelvic health experience to reflect on their clinical reasoning. They must therefore meet one or more of the following criteria:
- Qualified Physiotherapists who have worked with patients with urinary incontinence for a minimum of one year and have treated a minimum of 10 patients with UI in the past year; this is because physiotherapists might also carry orthopaedic and neurological

caseloads due to a low percentage of patients with women's health issues compared to patients with other diagnoses.

• Or qualified pelvic health physiotherapist's with more than 2 years' women's health experience in either the UK or KSA, but who do not necessarily currently treat 10 patients with UI per year.

If you would like to participate, you will attend an interview and/or focus group to discuss your clinical decision-making process and the factors that might affect your decision during the assessment and treatment of patients with urinary incontinence.

For more details, please check the enclosed participant information sheet.

If you would like to participate in the study or need further details, please contact me using the contact detail below.

I appreciate your valuable time spent reading this information and I apologize to members who are not able to participate in this study.

Sincerely,

Researcher:

Jawahr Alagil PhD Student Members of SPTWHA Principal Researcher E-mail: <u>j.h.alagil@soton.ac.uk</u> Phone number: 07522110629 **Supervisors:**

Dr. Sara Demain

Faculty of Health Sciences Student Office University of Southampton Highfield Southampton SO17 1BJ

Email: s.h.demain@soton.ac.uk

Room Number: 67/4061

Dr Ellen Kitson-Reynolds

Faculty of Health Sciences Student Office University of Southampton Highfield Southampton SO17 1BJ

Room Number: 67/3019

Email: E.L.kitson-Reynolds@soton.ac.uk

Poster

How women's health physiotherapists Southampton (WHPT) make their clinical decisions?

(16.05.2018)(ERGO_number: 31625)

What is this study about?

Exploring WHPTs clinical decision-making processes, while assessing and treating patients with Urinary Incontinence.

What will happen during the study?

- The study involves individual interviews and/or focus group.
- The discussion will be about:
 - your views and experiences while managing patients with urinary incontinence.
 - Factors that might affect your decision making.
- The session will take 60-90 minutes respectively.

Where does the study takes place?

It will be in a convenient place to you, either in:

- A meeting room at the University of Southampton.
- A private practices.
- A privately hired venue (Such as a room in hotel or in library).

Are there any participation requirements?

You can take part in this study if you are:

 WHPT who have been OR who are currently registered with Health and Care Professional Council (HCPC)

AND at least one of the followings

- Treated a minimum of 10 patients with UI in the past year; or
- Qualified WHPTs with more than 2 years' experience in the UK; or
- Academic WHPTs, who did research and/or teach students about UI.

For Further I	information,	please conta	et Mrs.Jawa	ıhr Alagil, Phi	D student, Un	iversity of S	outhampton
		1000 C					
A. Televo A. Televo A. Televo	A. Nessa A. Nessa A. Nessa A. Nessa	A. Newson A. Newson A. Newson A. Newson	A North	A. More More	A. Teleso algebras A. Teleso	A Newson	A. Teleson all (Allowedd) A. Teleson
Constraint Constraint Constraint	Create An Create An Diverte An	Constraint Constraint Constraint	Constraint Constraint Constraint	Louir Mi Court Mi Louir Mi	Greate Ab	Constraint Constraint Lower Au	Constraint Constraint Constraint

Appendix I Participants' Information Sheet

Participant Information Sheet

Study title: A qualitative study to explore the clinical reasoning processes among experienced pelvic health physiotherapists (PHPTs) in the UK and Kingdom of Saudi Arabia while managing patients with urinary incontinence (UI). Researcher:Jawahr Alagil Ethics number: 31625

I would like to invite you to take part in a research study. Please take time to read the following information carefully. Talk to others about the study if you wish. Ask me if there is anything that is not clear or if you would like more information. Take time to decide whether you wish to take part. If you are happy to participate you will be asked to sign a consent form.

What is the research about?

This research is part of a PhD study. This study aims to explore the process of clinical decision making/ clinical reasoning among Saudi and UK physiotherapists in the management of urinary incontinence. The University of Southampton is acting as the legal sponsor for the research.

Why have I been chosen?

You have been invited to take part in this study because you are a Pelvic health physiotherapist (PHPT) who is currently or who has been registered with the Health and Care Professional Council (HCPC) in the UK or Saudi Commission for Health Specialities (SCHS) in KSA and who has either worked with patients with urinary incontinence for a minimum of one year and has treated a minimum of 10 patients with UI in the past year; or who has more than 2 years' experience but have not necessarily treated 10 patients with UI per year; or is an academic WHPT, who may not necessarily have treated patients with UI in the past year but has been involved in research and/or teaching students about UI. The interviews and focus group discussions will be conducted in English, so you would need to be a fluent English speaker and be willing and able to give signed consent.

Do I have to take part?

No, you do not have to take part. I will describe the study and go through the information sheet, which will also be given to you to read if you wish. If you agree to take part, you will be asked to sign a consent form to show you have agreed to take part.

What will happen to me if I take part?

If you would like to take part, then please contact the researcher using the contact details below to register your interest and to check that you are eligible to take part in the research. If you want to be included, then the researcher will contact you directly by telephone to schedule you for a focus group discussion and/or interview. You will then be asked to read through and sign a consent form saying you understand what the research involves and that you are willing to take part.

What will I have to do?

You will be asked to attend the scheduled focus group and/or interview. These will be scheduled at a time and place convenient to you. The researcher will ask you some questions about how you would manage previous cases with urinary incontinence focusing on your clinical decision making and clinical reasoning about that case.

In the focus group:

There will be a maximum of ten women's health physiotherapists in one focus group discussion. There will be one-two focus groups in the UK and one in KSA, each with physiotherapists from pelvic health specialty. It is therefore likely that you may know some of the physiotherapists attending the focus group; some may be your colleagues or managers. The focus group discussion will be led by Jawahr Alagil (PhD researcher) and will last from 90-120 minutes. The discussion will concentrate on the factors that affect women's health physiotherapists' clinical decision making. The discussion will be tape recorded and a second researcher will take notes of the session.

In the interview: The interview will take from 60-120 minutes and will be led by Jawahr Alagil (PhD researcher). The researcher will ask you to recall and describe the process of managing one of your UI patients (you will not be asked to provide names or any identifying information about the patient). Then the researcher will randomly allocate one of two specific case scenarios and discuss the clinical reasoning process that you as pelvic health

physiotherapists would use to assess and treat the patient in the case scenario. The discussion will be tape recorded and the researcher will take notes of the session.

What are the possible disadvantages and risks of taking part?

There are no known disadvantages or risks involved in taking part in this study. However, if it is possible that some of the topics discussed during the focus group or interview may cause distress (although this is unlikely to happen during a discussion about patients' treatment). If this should happen, the interview will be ceased, and you can take a break or withdraw from the study if this should happen. During the focus group, you can take a break or leave if you feel distressed.

What are the possible benefits of taking part?

Discussing the clinical reasoning regarding the case will give you an opportunity to reflect on your practice with these types of patients and cases.

You may feel that you will not benefit directly from taking part in this study but the information we get will help improve the understanding of UK and Saudi physiotherapist's clinical decision making in the management of people with urinary incontinence. This might improve the treatment of UI in the future and the training provided to students.

What if there is a problem?

Any complaint about the way you have been dealt with during the study or any possible harm you might have suffered can be addressed by contacting either of my supervisors. For contact details see the bottom of the information sheet.

Will my taking part in the study be kept confidential?

The researcher (Jawahr Alagil) will not disclose any information about you or your participation in either the focus group or interview and will not disclose any comments that you have made. However, as this is a group discussion, other physiotherapists in the focus group would know that you attended and would hear any comments you make during the discussion. All physiotherapists who attend the focus group will be asked to respect the confidentiality of the other participants.

Your data will be handled confidentially. Any data from this research will be stored according to the University of Southampton's Data Management policy. Written and digitally recorded data will be stored on a password-protected computer that will be kept in a secure environment. Interviews and focus groups will be coded and real names and places will be altered to preserve confidentiality. Any e-mails sent to research participants will have their content coded and preserved within a password protected Word document and contact details and original emails will be deleted. Any papers connected with this research will be scanned and saved in a password protected computer. The original papers will be shredded. All participants who take part in the research will be given a code and that code, rather than your name, will be used to identify you in the analysis and write up of the study.

What happens if I change my mind?

In the interview, up until data analysis and up to 7 days after the focus group you can decide that you no longer wish to proceed with this research, and you have the right to withdraw from the study.

Who has reviewed the study?

This study has been reviewed and approved by the ERGO: Ethics and Research Governance Online. University of Southampton.

Contacts for further information

Researcher: Jawahr Alagil

Email: j.h.alagil@soton.ac.uk

Supervisors

Dr. Sara Demain

Faculty of Health Sciences Student Office University of Southampton Highfield Southampton SO17 1BJ

Email: s.h.demain@soton.ac.uk

Room Number: 67/4061

Dr. Ellen Kitson-Reynolds

Faculty of Health Sciences Student Office University of Southampton Highfield Southampton SO17 1BJ

Room Number: 67/3019

Email: E.L.kitson-Reynolds@soton.ac.uk

Appendix J Twitter Message

Twitter Message

I am researching clinical reasoning and decision making among pelvic health physiotherapists and would be very interested to speak to you! If you'd like to be involved in the study, please contact me at j.h.alagil@soton.ac.uk. For more information

http://bit.ly/2mOTw9a

http://bit.ly/2LRjq6R

Appendix K Interview Consent form

CONSENT FORM

For Interview

Study title: A qualitative study to explore the decision making and clinical reasoning processes among experienced women's health physiotherapists in the UK and Kingdom of Saudi Arabia while managing patients with urinary incontinence.

Researcher name: Jawahr Alagil

ERGO number: 31625

Please add you're initial in the box (es) if you agree with the statement(s):

I have read and understood the information sheet (16/05/2018 /version no. 1 of participant information sheet)	
and have had the opportunity to ask questions about the study.	
I agree to take part in this research project and agree for my data to be used for the purpose of this study.	
I understand my participation is voluntary and I may withdraw (up to data analysis) for any reason without my rights being affected.	
I understand that my interview will be audio recorded and transcribed.	

I understand my responses will be anonymised in reports of the research.	
I understand that I may be quoted directly in reports of the research but that my name will not be used.	
I understand that information collected about me during my participation in this study will be stored on a password protected computer and that this information will only be used for the purpose of ethically approved research studies.	

Name of participant (print name)
Signature of participant
Date
Name of researcher (print name)
Signature of researcher
Date

Optional - please only initial the box (es) you wish to agree to:

I understand that the information collected about me will be anonymised and may be used in future ethically approved research studies.

Appendix L Focus Group Consent Form

CONSENT FORM

For Focus Group

Study title: A qualitative study to explore the decision making and clinical reasoning processes among experienced women's health physiotherapists in the UK and Kingdom of Saudi Arabia while managing patients with urinary incontinence.

Researcher name: Jawahr Alagil

ERGO number: 31625

Γ

Please initial the box (es) if you agree with the statement(s):

I have read and understood the information sheet (16/05/2018 /version no. 1 of participant	
information sheet) and have had the opportunity to ask questions about the study.	
I agree to take part in this research project and agree for my data to be used for the purpose of	
this study.	
I understand my participation is voluntary and I may withdraw (up to 7 days following focus	
group) for any reason without my rights being affected.	
I understand that focus group will be audio recorded and transcribed.	
I understand my responses will be anonymised in reports of the research.	
I understand that I may be quoted directly in reports of the research but that my name will not be	
used.	

I understand that information collected about me during my participation in this study will be
stored on a password protected computer and that this information will only be used for the
purpose of ethically approved research studies.

Name of participant (print name)
Signature of participant
Date
Name of researcher (print name)
Signature of researcher
Date

Optional - please only initial the box (es) you wish to agree to:

I understand that the information collected about me will be anonymised and may be used in	
future ethically approved research studies.	

Appendix M The UK Demographics Form

Demographics Form						
Participant ID		Date	/ /		Country	□UK
Setting	□ NHS Hospital	□ GP surgeries	□ Self- employment	□Community health centre	Method	□Interview
	□ Private Hospital			□ Clinic		□ Focus Group

The UK Demographics Form

INCLUSION CRITERIA

appropriate box

Please tick $\sqrt{}$ the

Pelvic Health Physiotherapists (PHPTs) who have been OR who are registered with the Health and Care Professional Council (HCPC).	□Yes	□No
Qualified Physiotherapists (PT) who have worked with Urinary Incontinence (UI) patients a minimum of one year and treated a minimum of 10 patients with UI in the past year.	□Yes	□No
Or qualified PHPTs with more than 2 years' experience who have treated less than 10 patients with UI in the past year.	□Yes	□No
Participants able to read, speak and write in English.	□Yes	□No
Able to give informed consent.	□Yes	□No
Academic PHPTs who are not necessarily treating patients with UI in the past year, but did research and/or teach students about UI.	□Yes	□No
Participant has read and understood the information sheet, has had the opportunity to ask questions and has completed the consent form.		

SOCIO-DEMOGRAPHIC INFORMATION

Sex:	□ Male	□ Female
Age: Please	tick as appropria	te
□ 20-25		□ 26-35
□ 36-45		□46-55
□ 56-65		

Qualifications *Please mark or write* the appropriate response

Diploma Physiothoropy	
Dipiona Physiomerapy	
Bachelor Physiotherapy	2
Masters of Physiotherapy	3
PhD in Physiotherapy	2
Other(specify)	
Full time	1
Part time	2
Other (specify)	

Number of years since graduation \Box

Employment status (paid or voluntary)

Where did you train?

How long did you work in that country?

Mumber of years since working with urinary incontinence □

Number of patients with urinary incontinence treated in the last year □

Please specify the common type of UI diagnosis...

Courses that you have attended in Urinary Incontinence (please specify the course title and duration):

Courses title

Duration

Who usually refers patients to the physiotherapy service in your setting?

Date	Signature
------	-----------

Appendix N KSA Demographics Form

Demographics	s Form					
Participant ID		Date	/ /	Date of Birth: / /	Country	□KSA
Setting	☐ Minestry of Health Hospital □ Private Hospital	 Tertiary Hospital 	□ Governmental Clinic	□Private Clinic	Method	□ Interview □ Focus Group

INCLUSION CRITERIA

F la	euse nek vine	αρριοριίατε σολ
PHPTs' have been OR are registered with SAUDI Commission for Health Specialities (SCHS) in KSA.	□Yes	□No
Qualified Physiotherapists (PT) who have worked with patients with urinary incontinence a minimum of one year and treated a minimum of	□Yes	□No
Or qualified PHPTs with more than 2 years' experience. Who were treating less than 10 patients with UI in the past year.	□Yes	□No
Participants able to read, speak and write in English.	□Yes	□No
Able to give inform consent.	□Yes	□No

Please tick $\sqrt{}$ the appropriate box

Or academic PHPTs, who are not necessarily, treating patients with UI	□Yes	□No
in the past year, but did research and/or teach students about UI.		

Participant has read and understood the information	oYes	□No
sheet, has had the opportunity to ask questions and has completed the consent form.		

SOCIO-DEMOGRAPHIC INFORMATION

Sex:	□ Male	□ Female
Age: Please t	ick as appropriate	
□ 20-25		□ 26-35
□ 36-45		□46-55

□ 56-65

Qualifications Please mark or write the appropriate response

Diploma Physiotherapy	1
Bachelor Physiotherapy	2
Masters of Physiotherapy	3
PhD in Physiotherapy	4
Other(specify)	5
Full time	1
Part time	2
Other (specify)	3

Employment status (paid or voluntary)

Number of years since graduation \Box	
Where did you train?	
How long did you work in that country	
Number of years since working with urinary incontinence \Box	
Number of patients with urinary incontinence treated in the last year \Box	
Please specify the common type of UI diagnosis: in Urinary Incontinence (please specify the course title and duration):	Courses that you have attended
Courses title	
Duration	
Who usually refers patients to the physiotherapy service in your sett	ing?

Date	Signature

Appendix O Interview Guide

Interview Guide

Adapted from Hennink (2007)

Introductory Statement: I would like to thank you for coming to this interview. My name is Jawahr Alagil from the University of Southampton, and I am conducting this interview as part of my PhD study to identify your views about your processes of decision making and clinical reasoning during the assessment and treatment of patients with urinary incontinence.

Decision making and clinical reasoning is a very important part of pelvic health physiotherapists in their clinical assessment and treatment for patients with urinary incontinence. I would like to focus on your processes of clinical reasoning during the assessment and treatment of your patients with urinary incontinence. This means the process of your thinking and how you reach your assessment and treatment decisions.

Your participation in this interview is completely voluntary. So, if you prefer to stop the interview, you are completely free to leave. However, I value your views on the topic and I hope that you will stay and share your opinions. The information that we will discuss today will remain confidential and will be used only for my research purposes. It will be securely stored so that it is not accessible to anyone outside the research team.

I would like to say that there are no right or wrong answers in this interview, there is a case scenario that I would like you to spend five minutes reading. Then we will simply discuss your views, opinions, and experiences on a range of topics related to this case, so please feel comfortable to say what you honestly feel.

During the discussion I will be taking notes and looking at my notes, which would remind me if I forgot to ask something. However, so that I do not have to worry about getting every word down on the paper, I would also like to digitally record the whole session. Please do not be concerned about this, the recording will remain completely confidential and will only be used for this research project. Are you comfortable with recording this discussion? I would like to spend about 60-120 minutes with you. Are there any questions before we start? Let's start.....

Think for a minute about one of your patients with urinary incontinence and while maintaining your patient confidentiality could you tell me....

How did you receive your patient referral?

Was the diagnosis clearly mentioned or did you have to make your own assessment?

If referral WAS NOT clearly and correctly diagnosed, ask:

What assessments did you do? How did you come to this decision? What treatments did you decide to use? How did you come to this decision? Prompts: What other kind of treatments would you usually prescribe? Why? If referral WAS clearly and correctly diagnosed, ask: How much did it inform your clinical reasoning about.....? What assessments to do? What treatments to use? *Prompts*: What other kind of treatments would you usually prescribe? Why? What helped you to decide what the clinical diagnosis or clinical problems were? How did you decide what treatments might be best? How did you reach the treatment decision? How did you solve any uncertainties that you had? How do you decide what treatments might be best? What role did the patient play in deciding what the treatment should be? *Prompts* – type of treatments, frequency of treatments,

Conclusion:

We are now reaching the end of the interview. Do you have any further comments to add before we conclude this session?

As this was a pilot interview, I wanted to ask you what you thought about the interview itself. Are there any other questions that I could have asked you that would have given me useful information?

I would like to thank you very much for your participation in this interview, your experience and opinions are very valuable to assist in improving our understanding of the clinical reasoning processes among pelvic health physiotherapists. I am interested to know more about the factors that might affect decision making and clinical reasoning, so I am planning on conducting a focus group discussion in the near future, I would be very interested to hear your views in discussion with other pelvic health physiotherapists. Please take your time to think about it and reply to me by e-mail if you would like to take part. You will receive a thank you e-mail from me, with an option to receive the findings of this study. Please reply to me if you are interested in reading the future study findings.

Appendix P Focus Group Guide

Focus Group Guide/ Framework

Adapted from (Lauri et al., 2001; Edwards et al., 2004b; Hennink, 2007; McGinnis et al., 2009a; Haas et al., 2012)

Introductory Statement

I would like to thank you all for coming to this group. My name is Jawahr Alagil from the University of Southampton, and I am conducting this discussion group as part of my PhD study to identify your views about the common factors that affect pelvic health physiotherapists' decision making and clinical reasoning while assessing and treating patients with urinary incontinence.

Decision making and clinical reasoning is a very important part of women's health physiotherapists in her clinical assessment and treatment for patients with urinary incontinence. We would like to focus on your view and experience about the factors that influence your clinical reasoning and decision making.

Your participation in this discussion group is completely voluntary. So, if you prefer not to be part of this discussion, you are completely free to leave. However, we value everyone's views on the topic, and I hope that you will stay and share your opinions. The information that we will discuss today will remain confidential and will be used only for research purposes. It will be securely stored so that it is not accessible to anyone outside the research team.

I would like to say that there are no right or wrong answers in this discussion, we will simply be discussing your views, opinions and experiences on a range of topics, so please feel comfortable to say what you honestly feel.

During the discussion a facilitator will be taking notes and reminding me if I forgot to ask something. However, so that she does not have to worry about getting every word down on the paper, we would also like to digitally record the whole session. Please do not be concerned about this, the recording will remain completely confidential and will only be used for this research project. Is everyone comfortable with recording this discussion? (Ensure that everyone consents to recording.)

We do not want to miss anything that is said so it is important that only one person talks at a time. Remember we want to hear as many different points of view as possible, so feel free to disagree with someone else and share your own opinions. We would like you to have the chance to express your opinions, so please let everyone have their say. We would like to spend about 90-120 minutes with you. Please also help yourselves to the refreshments we have provided before discussion. Are there any questions before we start? Can I just check that you all sign the consent form and read participants information sheet? Let's start

Introduction question

As an introduction, let us go around the group and perhaps each person can introduce themselves by saying their name and tell us your area of work. (I will edit this in the transcript using pseudonym name and remove all identifying places if mentioned).

Prompts:

In your opinion/experience*, what are the main different between the factors that can affect pelvic health physiotherapists' decision making and other physiotherapy specialities?

What are the main factors, external to treatment that affects your clinical practice?

Multidimensional knowledge base

In your opinion/experience*, what are the central skills in pelvic health physiotherapy practice? How did you gain your knowledge and become women's health physiotherapists? What type of knowledge is more important than the other in pelvic health physiotherapy? What kind of knowledge do you use when defining your client's health problems? **Physiotherapist's experience** How do you think that you have grown as pelvic health physiotherapists over the last few years? As you think back over your clinical experience, has your approach to assessing and treating patients with urinary

incontinence changed? Why or why not?

How do you regard the activity of teaching in your practice?

Physiotherapists understanding of patient's problems

- Is there anything that limited your choices of assessment and treatment approaches?
- How did you use the information you gathered from your assessment?
- How did you know to look at these things? Is there anything from your demographic data form that influenced your choices?

Patients factors (Age and other related diseases)

- How your assessment and treatment varied is based on patient's age and the availability of other disease?
- Do you involve family members or caregivers in the management process?
- If so, in what ways (or how) do you do this?

Patient goals, preferences and perceptions toward assessment and treatment

- What role do your clients have in their management process?
- In your opinion, what is the best way to interact and socialize with your patients with urinary incontinence? (interactive reasoning Edwards et al. (2004))
- How can you manage the social relationships with your patients and colleagues, the emotional aspects of your work, and your own emotions?

Prescribing exercises for patients with urinary incontinence:

In what circumstances do you think patients need to do exercises in this context (for example, on face-to-face individual basis) rather than the other contexts explored (for example, doing a home exercise program or attending a group exercise program)?

Adherence to exercise

- Do you have problem with patient's adherence and, why do you think this is generally the case?
- What do you do to help promote patient adherence?

- Do you think this is effective?
- Why or why not?
- Is there anything else could be done to promote adherence that you do not or cannot do?
- What stops you from doing this?

Cultural constraints

- How cultural competence is integrated in pelvic health physiotherapy practice? at all three levels identified: those of awareness, respect and acceptance, and mutual collaboration(Norris and Allotey, 2008).
- How women's health physiotherapists consider different culture in her assessment and treatment of patients with urinary incontinence?
- How the way of taking patient consent to assessment in urinary incontinence is varied among different culture?
- Are there any specific guidelines that you follow?

Relationship with inter-disciplinary team

- Do you consider referring the patient to another specialist and why?
- How do you do that?

Utilizing the experience of doing interview in the UK and KSA:

- Based on the participant's interview that has been done in the UK and KSA, they mentioned that there are different in the clinical practice between UK and KSA. In your opinion what are the factors that lead to this difference?
- Why there is different?

Conclusion:

We are now reaching the end of the discussion. Does anyone have any further comments to add before we conclude this session? I would like to thank you all very much for your participation in this discussion, your experiences and opinions are very valuable to assist in improving our understanding to decision making and clinical reasoning process among pelvic health physiotherapists. You will receive a thank you e-mail from the researcher, with an option to receive the findings of the study. Please reply to me if you are interested in reading the future study findings. *Experience/opinion will be used interchangeable

Appendix Q Thank you e-mail

Subject line: Interview and /or Focus group Thank you e-mail

Dear Mrs. /Mr. Last Name,

I enjoyed meeting you and learning more about your experience in clinical reasoning and decision making while assessing and treating patients with urinary incontinence.

Thank you for taking part in this research. It was very helpful learning about your knowledge and experience in the field of women's health.

I am grateful for the time you spent with me during the interview and/or focus group.

Please let me know if you are interested in reading the future study findings, which will be available in about two-year time.

Sincerely,

Jawahr Alagil

Glossary

Adapted from Bø (2020); Frawley et al. (2021)

Pelvic Organ Prolapse: refers to falling or downward displacement of the uterus, vaginal compartments their neighbouring organs such as bladder, rectum or bowel.

Dyspareunia: Complaint of persistent or recurrent pain or discomfort associated with attempted or complete vaginal penetration.

Palpation/ Digital palpation: The process of using fingers/hands as part of assessment, to gather information about the tissues.

Lifestyle modification: is the application of interventions in the management of lifestylerelated pelvic floor dysfunctions. For instance, fluid consumption/restriction, dietary modification, dietary elimination and physical activity.

Urgency suppression techniques are methods/manoeuvres that are used to decrease the feeling of urgency, which may include, but are not limited to: distraction, PFM contraction and perineal pressure such as sitting on a hard chair.

Facilitation technique: any method of increasing recruitment/ response of a nonresponding muscle. In the case of very weak PFMs, this may include a quick stretch of the PFM, with tapping or stretching the PFM digitally. An overflow effect from a strong contraction of adjacent synergistic muscle (e.g., external rotators) may also assist facilitation or recruitment of PFMs.

Functional PFM training: training and exercises that include a correct PFM contraction into activities of daily living such as lifting, transferring out of bed, or coughing. A PFM contraction before a rise in intra-abdominal pressure, e.g., a sneezing ("the Knack") is part of functional PFM training.

Posterior Tibial Nerve Stimulation: Percutaneous neuromuscular electrical stimulation (e.g., posterior TNS) is a peripheral neuromodulation technique, in which the posterior tibial nerve is electrically stimulated, via insertion of a percutaneous needle electrode. This is joined with an adhesive reference surface electrode placed near to the needle. This intervention is offered to patients with overactive bladder.

Nocturia: is waking to pass urine during the main sleep period.

Urgency: is the complaint of a sudden compelling desire to pass urine, which is difficult to defer.

Stress Urinary Incontinence: is the complaint of involuntary leakage on effort or exertion, or on sneezing or coughing.

Urgency Urinary Incontinence: is the complaint of involuntary loss of urine associated with urgency.

Mixed Urinary Incontinence: is the complaint of involuntary leakage associated with urgency

Abraham, C. and Michie, S. (2008) 'A taxonomy of behavior change techniques used in interventions', *Health psychology*, 27(3), p. 379.

Abrams, P. *et al.* (2018) '6th International Consultation on Incontinence. Recommendations of the International Scientific Committee: evaluation and treatment of urinary incontinence, pelvic organ prolapse and faecal incontinence', *Neurourology and urodynamics*, 37(7), pp. 2271-2272. Agner, J. (2020) 'Moving from cultural competence to cultural humility in occupational therapy: a paradigm shift', *American Journal of Occupational Therapy*, 74(4), pp. 7404347010p1-7404347010p7.

Ajjawi, R. and Higgs, J. (2012) 'Core components of communication of clinical reasoning: a qualitative study with experienced Australian physiotherapists', *Advances in health sciences education*, 17(1), pp. 107-119.

Alewijnse, D. et al. (2001) 'Predictors of intention to adhere to physiotherapy among women with urinary incontinence', *Health Education Research*, 16(2), pp. 173-186.

Alghadir, A. *et al.* (2015) 'Physical therapy education in Saudi Arabia', *Journal of physical therapy science*, 27(5), pp. 1621-1623.

AlHaqwi, A.I. *et al.* (2016) 'Patient's desire and preference for provision of information toward greater involvement in shared care', *Saudi journal of medicine & medical sciences*, 4(3), p. 172. Aljuraifani, R. *et al.* (2019) 'Activity of deep and superficial pelvic floor muscles in women in response to different verbal instructions: A preliminary investigation using a novel electromyography electrode', *The journal of sexual medicine*, 16(5), pp. 673-679.

Alkhoraif, A. and McLaughlin, P. (2016). The International Academic Forum.

Alkhoraif, A. and McLaughlin, P. (2018) 'Organisational culture aspects that facilitate lean implementation: a pilot study', *International Journal of Agile Systems and Management*, 11(2), pp. 103-125.

Alluhidan, M. *et al.* (2020) 'Challenges and policy opportunities in nursing in Saudi Arabia', *Human Resources for Health*, 18(1), pp. 1-10.

Almalki, M., FitzGerald, G. and Clark, M. (2011) 'Health care system in Saudi Arabia: an overview', *EMHJ-Eastern Mediterranean Health Journal, 17 (10), 784-793, 2011*.

Alsadaan, N. *et al.* (2021) 'Challenges facing the nursing profession in Saudi Arabia: An integrative review', *Nursing Reports*, 11(2), pp. 395-403.

Alshehri, M.A. (2017) 'Teaching clinical reasoning to physiotherapists in Saudi Arabia: a theoretical report', *International Journal of Physiotherapy*, 4(2), pp. 106-111.

Alshehri, M.A. *et al.* (2018) 'Factors affecting the extent of utilization of physiotherapy services by physicians in Saudi Arabia', *Journal of physical therapy science*, 30(2), pp. 216-222.

Altaweel, W. and Alharbi, M. (2012) 'Urinary incontinence: prevalence, risk factors, and impact on health related quality of life in Saudi women', *Neurourology and urodynamics*, 31(5), pp. 642-645. Amundadottir, O.R. *et al.* (2018) 'Physical Therapists' Clinical Reasoning and Decision-Making

Processes When Mobilizing Patients Who Are Critically III: A Qualitative Study', *Cardiopulmonary Physical Therapy Journal*, 29(1), pp. 13-25.

Andreatta, S. and Ferraro, G. (2012) *Elements of culture: An applied perspective*. Cengage Learning.

Araklitis, G. and Cardozo, L. (2017) 'Safety issues associated with using medication to treat overactive bladder', *Expert opinion on drug safety*, 16(11), pp. 1273-1280.

Araujo, C.C., Marques, A.d.A. and Juliato, C.R. (2020) 'The adherence of home pelvic floor muscles training using a mobile device application for women with urinary incontinence: a randomized controlled trial', *Female pelvic medicine & reconstructive surgery*, 26(11), pp. 697-703.

Arksey, H. and O'Malley, L. (2005) 'Scoping studies: towards a methodological framework', *International journal of social research methodology*, 8(1), pp. 19-32.

Bahrami, S. *et al.* (2021) 'Pelvic floor ultrasound: when, why, and how?', *Abdominal Radiology*, 46(4), pp. 1395-1413.

Bakker, E. *et al.* (2018) 'International Continence Society supported pelvic physiotherapy education guideline', *Neurourology and urodynamics*, 37(2), pp. 869-876.

Bar-On, R.E. and Parker, J.D. (2000) *The handbook of emotional intelligence: theory, development, assessment, and application at home, school, and in the workplace*. Jossey-Bass.

Barbour, R.S. (2001) 'Checklists for improving rigour in qualitative research: a case of the tail wagging the dog?', *BMJ: British Medical Journal*, 322(7294), p. 1115.

Bascur-Castillo, C. *et al.* (2022) 'Effect of conservative treatment of pelvic floor dysfunctions in women: An umbrella review', *International Journal of Gynecology & Obstetrics*.

Bassett, S. and Tango, S. (2002) 'Maori people's experiences of being physiotherapy patients: A phenomenological study', *New Zealand Journal of Physiotherapy*, 30(1), pp. 30-40.

Battles, J.B. *et al.* (2006) 'Sensemaking of patient safety risks and hazards', *Health services research*, 41(4p2), pp. 1555-1575.

Bayliss, E.A. *et al.* (2003) 'Descriptions of barriers to self-care by persons with comorbid chronic diseases', *The Annals of Family Medicine*, 1(1), pp. 15-21.

Beagan, B.L. (2015) 'Approaches to culture and diversity: a critical synthesis of occupational therapy literature: Des approches en matière de culture et de diversité: une synthèse critique de la littérature en ergothérapie', *Canadian Journal of Occupational Therapy*, 82(5), pp. 272-282. Berghmans, B., Seleme, M. and Bernards, A. (2020) 'Physiotherapy assessment for female urinary incontinence', *International urogynecology journal*, 31(5), pp. 917-931.

Bernards, A.T.M. *et al.* (2014) 'Dutch guidelines for physiotherapy in patients with stress urinary incontinence: an update', *International Urogynecology Journal*, 25(2), pp. 171-179.

Bettany-Saltikov, J. (2012) *How To Do A Systematic Literature Review In Nursing: A Step-By-Step Guide*. Open University Press.

Beuttenmüller, L. *et al.* (2010) 'Muscle contraction of the pelvic floor and quality of life of women with stress urinary incontinence who underwent kinesitherapy', *Physiotherapy Quarterly*, 18(1), p. 35.

Birks, M. and Mills, J. (2011) 'Essentials of grounded theory', *Grounded theory: a practical guide*, pp. 11-26.

Bithell, C. (2007) 'Entry-level physiotherapy education in the United Kingdom: governance and curriculum', *Physical therapy reviews*, 12(2), pp. 145-155.

Blackledge, J., Ciarrochi, J. and Bailey, A. (2006) *Personal values questionnaire*. Unpublished manuscript, University of Wollongong, Australia.

Bloomfield, H.E. *et al.* (2014) 'Screening pelvic examinations in asymptomatic, average-risk adult women: an evidence report for a clinical practice guideline from the American College of Physicians', *Annals of internal medicine*, 161(1), pp. 46-53.

Bø, K. (2020) 'Physiotherapy management of urinary incontinence in females'.

Bo, K. *et al.* (2014) *Evidence-Based physical therapy for the pelvic floor-E-book: bridging science and clinical practice*. Elsevier Health Sciences.

Bo, K. *et al.* (2017) 'An International Urogynecological Association (IUGA)/International Continence Society (ICS) joint report on the terminology for the conservative and

nonpharmacological management of female pelvic floor dysfunction', *International urogynecology journal*, 28(2), pp. 191-213.

Bø, K. and Herbert, R.D. (2013) 'There is not yet strong evidence that exercise regimens other than pelvic floor muscle training can reduce stress urinary incontinence in women: a systematic review', *Journal of physiotherapy*, 59(3), pp. 159-168.

Bø, K. *et al.* (2009) 'Evidence for benefit of transversus abdominis training alone or in combination with pelvic floor muscle training to treat female urinary incontinence: a systematic review', *Neurourology and Urodynamics: Official Journal of the International Continence Society*, 28(5), pp. 368-373.

Bø, K., Talseth, T. and Holme, I. (1999) 'Single blind, randomised controlled trial of pelvic floor exercises, electrical stimulation, vaginal cones, and no treatment in management of genuine stress incontinence in women', *Bmj*, 318(7182), pp. 487-493.

Bø, K., Talseth, T. and Vinsnes, A. (2000) 'Randomized controlled trial on the effect of pelvic floor muscle training on quality of life and sexual problems in genuine stress incontinent women', *Acta obstetricia et gynecologica Scandinavica*, 79(7), pp. 598-603.

Bo, K.a.S. (2005) ' Evaluation of female pelvic-floor muscle function and strength', *Pherhys T*, 85(3), pp. 269-282.

Bonder, B.R., Martin, L. and Miracle, A.W. (2004) 'Culture emergent in occupation', *American Journal of Occupational Therapy*, 58(2), pp. 159-168.

Bone, R.C. *et al.* (1990) 'Ethical and moral guidelines for the initiation, continuation, and withdrawal of intensive care', *Chest*, 97(4), pp. 949-959.

Bonello, M. and Meehan, B. (2019) 'Transparency and coherence in a doctoral study case analysis: reflecting on the use of NVivo within a'Framework'approach', *The Qualitative Report*, 24(3), pp. 483-498.

Bonner, A. and Tolhurst, G. (2002) 'Insider-outsider perspectives of participant observation', *Nurse Researcher (through 2013)*, 9(4), p. 7.

Booth, J. *et al.* (2021) 'Tibial nerve stimulation compared with sham to reduce incontinence in care home residents: ELECTRIC RCT', *Health technology assessment (Winchester, England)*, 25(41), p. 1.

Borg, S. (2001) 'The research journal: A tool for promoting and understanding researcher development', *Language teaching research*, 5(2), pp. 156-177.

Botlero, R. *et al.* (2008) 'Prevalence and incidence of urinary incontinence in women: review of the literature and investigation of methodological issues', *International journal of urology*, 15(3), pp. 230-234.

Boyle, S. (2011) European Observatory on Health Systems and Policies in the United Kingdom (England): health system review. Available at: <u>https://apps.who.int/iris/handle/10665/330328</u>. Brandt, C. (2021) 'Physiotherapy and pelvic floor health within a contemporary biopsychosocial model of care: From research to education and clinical practice', *The South African Journal of Physiotherapy*, 77(1).

Brinkmann, S. (2014) 'Interview', in *Encyclopedia of Critical Psychology*. Springer, pp. 1008-1010. Brinkmann, S. and Kvale, S. (2008) 'Ethics in qualitative psychological research', *The Sage* handbook of qualitative research in psychology, 24(2), pp. 263-279.

Brown, A.D., Stacey, P. and Nandhakumar, J. (2008) 'Making sense of sensemaking narratives', *Human relations*, 61(8), pp. 1035-1062.

Brown, H.W. *et al.* (2020) 'Better together: multidisciplinary approach improves adherence to pelvic floor physical therapy', *International urogynecology journal*, 31(5), pp. 887-893.

Buckley, B.S. and Lapitan, M.C.M. (2010) 'Prevalence of urinary incontinence in men, women, and children—current evidence: findings of the Fourth International Consultation on Incontinence', *Urology*, 76(2), pp. 265-270.

Buddhadev, N.P. (2012) 'Physiotherapists' perception of patient compliance to home exercises in chronic musculoskeletal physiotherapy', *Natl J Integr Res Med*, 3(2), pp. 162-165.

Bulley, C. and Donaghy, M. (2008) 'Processes in the development of international specialist competencies and standards: the sports physiotherapy for All project', *Journal of allied health*, 37(3), pp. 221E-243E.

Buschman, T.J. and Miller, E.K. (2007) 'Top-down versus bottom-up control of attention in the prefrontal and posterior parietal cortices', *science*, 315(5820), pp. 1860-1862.

Cancelli, I. *et al.* (2009) 'Drugs with anticholinergic properties: cognitive and neuropsychiatric sideeffects in elderly patients', *Neurological Sciences*, 30(2), pp. 87-92.

Carneiro, E. *et al.* (2010) 'The anatomical-functional characteristics of the pelvic floor and quality of life of women with stress urinary incontinence subjected to perineal exercises', *Actas Urológicas Españolas (English Edition)*, 34(9), pp. 788-793.

Case, K., Harrison, K. and Roskell, C. (2000) 'Differences in the clinical reasoning process of expert and novice cardiorespiratory physiotherapists', *PHYSIOTHERAPY (LONDON)*, 86(1), p. 14. Case, R. and Bereiter, C. (1984) 'From behaviourism to cognitive behaviourism to cognitive development: Steps in the evolution of instructional design', *Instructional Science*, 13(2), pp. 141-

158.

CASP (2022) *Critical appraisal tools checklist in Qualitative studies* Available at: <u>https://casp-uk.net/casp-tools-checklists/</u> (Accessed: 24/05).

Cerderbom, S., Bjerk, M. and Bergland, A. (2020) 'The tensions between micro-, meso-and macrolevels: physiotherapists' views of their role towards fall prevention in the community–a qualitative study', *BMC health services research*, 20(1), pp. 1-11.

Chaliha, C. and Stanton, S. (1999) 'The ethnic cultural and social aspects of incontinence–a pilot study', *International Urogynecology Journal*, 10(3), pp. 166-170.

Chan, M.Y., Hamamura, T. and Janschewitz, K. (2013) 'Ethnic differences in physical pain sensitivity: Role of acculturation', *PAIN®*, 154(1), pp. 119-123.

Chapparo, C. and Ranka, J. (2008) 'Clinical reasoning in occupational therapy', *Clinical reasoning in the health professions*, 3, pp. 265-277.

Chapra, M.U., Khan, S. and Al Shaikh-Ali, A. (2008) *The Islamic vision of development in the light of maqasid al-Shariah*. liit.

Charmaz, K. (2013) *Constructing grounded theory : a practical guide through qualitative analysis*. London : Sage Publications, 2013, c2006.

Charmaz, K. and Belgrave, L.L. (2007) 'Grounded theory', *The Blackwell encyclopedia of sociology*. Collins, C.K., Johnson, R. and Masaracchio, M.

Corbin, J. and Strauss, A. (2008) Basics of qualitative research London: Sage.

Costa, A.A.L.F.d. *et al.* (2018) 'What do Cochrane systematic reviews say about non-surgical interventions for urinary incontinence in women?', *Sao Paulo Medical Journal*, 136, pp. 73-83. Coulter, A. (2018) *National strategies for implementing shared decision making*. Bertelsmann Stiftung.

Cowell, I. *et al.* (2019) 'The perspectives of physiotherapists on managing nonspecific low back pain following a training programme in cognitive functional therapy: A qualitative study', *Musculoskeletal Care*, 17(1), pp. 79-90.

Cox, J.L. and Simpson, M.D. (2020) 'Cultural Humility: A Proposed Model for a Continuing Professional Development Program', *Pharmacy*, 8(4), p. 214.

Coyne, K. *et al.* (2003) 'The impact on health-related quality of life of stress, urge and mixed urinary incontinence', *BJU international*, 92(7), pp. 731-735.

Coyne, K.S. *et al.* (2009) 'The prevalence of lower urinary tract symptoms (LUTS) in the USA, the UK and Sweden: results from the Epidemiology of LUTS (EpiLUTS) study', *BJU International*, 104(3), pp. 352-360.

Creswell, J.W. and Creswell, J.D. (2017) *Research design: Qualitative, quantitative, and mixed methods approaches*. Sage publications.

Creswell, J.W. and Poth, C.N. (2017) *Qualitative inquiry and research design: Choosing among five approaches*. Sage publications.

Cristancho, S.M. *et al.* (2013) 'When surgeons face intraoperative challenges: a naturalistic model of surgical decision making', *The American Journal of Surgery*, 205(2), pp. 156-162.

Croskerry, P. (2009a) 'Clinical cognition and diagnostic error: applications of a dual process model of reasoning', *Advances in health sciences education*, 14(1), pp. 27-35.

Croskerry, P. (2009b) 'A universal model of diagnostic reasoning', *Academic medicine*, 84(8), pp. 1022-1028.

Cruz, E.B., Moore, A. and Cross, V. (2012a) 'Clinical reasoning and patient-centred care in musculoskeletal physiotherapy in Portugal – A qualitative study', *Manual Therapy*, 17(3), pp. 246-250.

Cruz, E.B., Moore, A.P. and Cross, V. (2012b) 'A qualitative study of physiotherapy final year undergraduate students' perceptions of clinical reasoning', *Manual Therapy*, 17(6), pp. 549-553. Currie, G. and Brown, A.D. (2003) 'A narratological approach to understanding processes of organizing in a UK hospital', *Human Relations*, 56(5), pp. 563-586.

Daluiso-King, G. and Hebron, C. (2020) 'Is the biopsychosocial model in musculoskeletal physiotherapy adequate? An evolutionary concept analysis', *Physiotherapy theory and practice*, pp. 1-17.

Dan, M. (2020) 'Culture as a multi-Level and multi-layer construct', *Revista de Management Comparat International*, 21(2), pp. 226-240.

Dawson, N.V. (1993) 'Physician judgment in clinical settings: methodological influences and cognitive performance', *Clinical Chemistry*, 39(7), pp. 1468-1478.

DeLancey, J.O. (1994) 'Structural support of the urethra as it relates to stress urinary incontinence: the hammock hypothesis', *American journal of obstetrics and gynecology*, 170(5), pp. 1713-1723.

DeLancey, J.O. *et al.* (2003) 'The appearance of levator ani muscle abnormalities in magnetic resonance images after vaginal delivery', *Obstetrics & Gynecology*, 101(1), pp. 46-53.

DeLancey, J.O. *et al.* (2007) 'Comparison of levator ani muscle defects and function in women with and without pelvic organ prolapse', *Obstetrics & Gynecology*, 109(2), pp. 295-302.

Delany, C., Edwards, I. and Fryer, C. (2019) 'How physiotherapists perceive, interpret, and respond to the ethical dimensions of practice: A qualitative study', *Physiotherapy theory and practice*, 35(7), pp. 663-676.

Delany, C. and Frawley, H. (2012) 'A process of informed consent for student learning through peer physical examination in pelvic floor physiotherapy practice', *Physiotherapy*, 98(1), pp. 33-39. Denzin, N.K. and Lincoln, Y.S. (2008) *Strategies of qualitative inquiry*. Sage.

Denzin, N.K. and Lincoln, Y.S. (2011) *The Sage handbook of qualitative research*. Sage. Dey, I. (2003) *Qualitative data analysis: A user friendly guide for social scientists*. Routledge. Dierckx, K. *et al.* (2013) 'Implementation of Shared Decision Making in Physical Therapy: Observed Level of Involvement and Patient Preference', *Physical Therapy*, 93(10), pp. 1321-1330. Dietz, H.P. (2011) 'Pelvic floor ultrasound in incontinence: what's in it for the surgeon?', *International urogynecology journal*, 22(9), pp. 1085-1097.

Dixon, N. (2003) 'Med QIC Medicare Quality Improvement. Baltimore: Centers for Medicare and Medicaid Services', in Services, B.C.f.M.a.M. (ed.). Available at: <u>http://www.medgic.org</u>. Doherty, R.F. and Purtilo, R.B. (2015) *Ethical Dimensions in the Health Professions-E-Book*. Elsevier Health Sciences.

Domenech, J. *et al.* (2011) 'Impact of biomedical and biopsychosocial training sessions on the attitudes, beliefs, and recommendations of health care providers about low back pain: a randomised clinical trial', *Pain*, 152(11), pp. 2557-2563.

Doody, C. and McAteer, M. (2002) 'Clinical reasoning of expert and novice physiotherapists in an outpatient orthopaedic setting', *PHYSIOTHERAPY (LONDON)*, 88(5), p. 258.

Dowding, D. *et al.* (2016) 'Using sense-making theory to aid understanding of the recognition, assessment and management of pain in patients with dementia in acute hospital settings', *International journal of nursing studies*, 53, pp. 152-162.

Dowling, M. (2006) 'Approaches to reflexivity in qualitative research', *Nurse Researcher (through 2013)*, 13(3), p. 7.

Dressler, D. and Pils, P. (2009) 'A qualitative study on cross-cultural communication in postaccident in-patient rehabilitation of migrant and ethnic minority patients in Austria', *Disability and Rehabilitation*, 31(14), pp. 1181-1190.

Dukhu, S., Purcell, C. and Bulley, C. (2018) 'Person-centred care in the physiotherapeutic management of long-term conditions: a critical review of components, barriers and facilitators', *International Practice Development Journal*.

Dumoulin, C. *et al.* (2015) 'Pelvic-Floor-Muscle Training Adherence: Tools, Measurements and Strategies—2011 ICS State-of-the-Science Seminar Research Paper II of IV', *Neurourology and urodynamics*, 34(7), pp. 615-621.

Dumoulin, C., Cacciari, L.P. and Hay-Smith, E.J.C. (2018) 'Pelvic floor muscle training versus no treatment, or inactive control treatments, for urinary incontinence in women', *Cochrane database of systematic reviews*, (10).

Dumoulin, C., Hay-Smith, E.J. and Mac Habee-Seguin, G. (2014) 'Pelvic floor muscle training versus no treatment, or inactive control treatments, for urinary incontinence in women', *The Cochrane database of systematic reviews*, 5, p. CD005654.

Dumoulin, C. and Hay-Smith, J. (2010) 'Pelvic floor muscle training versus no treatment, or inactive control treatments, for urinary incontinence in women', *Cochrane Database Syst Rev*, 1(1).

Dune, T. *et al.* (2021) 'Australian mental health care practitioners' construing of non-White and White people: implications for cultural competence and therapeutic alliance', *BMC psychology*, 9(1), pp. 1-17.

Dunford, R. and Jones, D. (2000) 'Narrative in stractegic change', *Human relations*, 53(9), pp. 1207-1226.

Duthie, J.B. *et al.* (2011) 'Botulinum toxin injections for adults with overactive bladder syndrome', *Cochrane Database of Systematic Reviews*, (12).

Dwyer, S.C. and Buckle, J.L. (2009) 'The space between: On being an insider-outsider in qualitative research', *International journal of qualitative methods*, 8(1), pp. 54-63.

Edwards, I., Braunack-Mayer, A. and Jones, M. (2005) 'Ethical reasoning as a clinical-reasoning strategy in physiotherapy', *Physiotherapy*, 91(4), pp. 229-236.

Edwards, I. *et al.* (2004a) 'Clinical reasoning strategies in physical therapy', *PHYSICAL THERAPY*, 84(4), p. 312.

Edwards, I. *et al.* (2004b) 'Clinical reasoning strategies in physical therapy: Author response', *PHYSICAL THERAPY*, 84(4), p. 334.

Edwards, I., Jones, M. and Hillier, S. (2006) 'The interpretation of experience and its relationship to body movement: a clinical reasoning perspective', *Manual therapy*, 11(1), pp. 2-10.

Ekman, I. *et al.* (2011) 'Person-centered care—ready for prime time', *European journal of cardiovascular nursing*, 10(4), pp. 248-251.

Elstein, A.S. (2001) 'Naturalistic decision making and clinical judgment', *Journal of behavioral decision making*, 14(5), p. 363.

Elstein, A.S., Shulman, L.S. and Sprafka, S.A. (1978) 'Medical problem solving an analysis of clinical reasoning'.

Elvén, M. and Dean, E. (2017) 'Factors influencing physical therapists' clinical reasoning:

qualitative systematic review and meta-synthesis', *Physical Therapy Reviews*, 22(1-2), pp. 60-75. Elwyn, G. *et al.* (2012) 'Shared decision making: a model for clinical practice', *Journal of general internal medicine*, 27(10), pp. 1361-1367.

Elzubier, A.G. (2002) 'Doctor-patient communication: a skill needed in Saudi Arabia', *Journal of Family and Community Medicine*, 9(1), p. 51.

Engin, M. (2011) 'Research diary: A tool for scaffolding', *International Journal of Qualitative Methods*, 10(3), pp. 296-306.

Evans, J.S.B. (2008) 'Dual-processing accounts of reasoning, judgment, and social cognition', *Annu. Rev. Psychol.*, 59, pp. 255-278.

Evans, M., Sharp, D. and Shaw, A. (2012) 'Developing a model of decision-making about complementary therapy use for patients with cancer: A qualitative study', *Patient Education and Counseling*, 89(3), pp. 374-380.

Fielden, J.M. (2012) 'Managing the transition of Saudi new graduate nurses into clinical practice in the Kingdom of Saudi Arabia', *Journal of nursing management*, 20(1), pp. 28-37.

Finch, E., Geddes, E.L. and Larin, H. (2005) 'Ethically-based clinical decision-making in physical therapy: process and issues', *Physiotherapy Theory and Practice*, 21(3), pp. 147-162.

Finch, H. and Lewis, J. (2003) 'Focus groups', *Qualitative research practice: A guide for social science students and researchers*, pp. 170-198.

Fischer, R. and Mansell, A. (2009) 'Commitment across cultures: A meta-analytical approach', *Journal of International Business Studies*, 40(8), pp. 1339-1358.

Fisher-Borne, M., Cain, J.M. and Martin, S.L. (2015) 'From mastery to accountability: Cultural humility as an alternative to cultural competence', *Social Work Education*, 34(2), pp. 165-181. Fisher, J.D. *et al.* (2002) 'Information-motivation-behavioral skills model-based HIV risk behavior change intervention for inner-city high school youth', *Health Psychology*, 21(2), p. 177.

Fitz, F.F. *et al.* (2020) 'Pelvic floor muscle training for female stress urinary incontinence: a randomised control trial comparing home and outpatient training', *International urogynecology journal*, 31(5), pp. 989-998.

Fitzgerald, C.M. *et al.* (2011) 'Pelvic floor muscle examination in female chronic pelvic pain', *The Journal of reproductive medicine*, 56(3-4), pp. 117-122.

Flavell, J.H. (1979) 'Metacognition and cognitive monitoring: A new area of cognitive– developmental inquiry', *American psychologist*, 34(10), p. 906.

Flick, U. (2004) 'Triangulation in qualitative research', *A companion to qualitative research*, 3, pp. 178-183.

Fontana, J.S. (2004) 'A methodology for critical science in nursing', *Advances in Nursing Science*, 27(2), pp. 93-101.

Foster, N.E. and Delitto, A. (2011) 'Embedding psychosocial perspectives within clinical management of low back pain: integration of psychosocially informed management principles into physical therapist practice—challenges and opportunities', *Physical therapy*, 91(5), pp. 790-803.

Frawley, H. *et al.* (2021) 'An International Continence Society (ICS) report on the terminology for pelvic floor muscle assessment', *Neurourology and Urodynamics*.

Frawley, H.C., Neumann, P. and Delany, C. (2019) 'An argument for competency-based training in pelvic floor physiotherapy practice', *Physiotherapy theory and practice*, 35(12), pp. 1117-1130. Friedland, R. (1991) 'Bringing society back in: Symbols, practices, and institutional contradictions', *The new institutionalism in organizational analysis*, pp. 232-263.

Furze, J. *et al.* (2015) 'Exploration of students' clinical reasoning development in professional physical therapy education', *Journal of Physical Therapy Education*, 29(3), pp. 22-33.

Fusch, P., Fusch, G.E. and Ness, L.R. (2018) 'Denzin's paradigm shift: Revisiting triangulation in qualitative research', *Journal of social change*, 10(1), p. 2.

Gale, N.K. *et al.* (2013) 'Using the framework method for the analysis of qualitative data in multidisciplinary health research', *BMC medical research methodology*, 13(1), pp. 1-8.

Garber, C.E. *et al.* (2011) 'American College of Sports Medicine position stand. Quantity and quality of exercise for developing and maintaining cardiorespiratory, musculoskeletal, and neuromotor fitness in apparently healthy adults: guidance for prescribing exercise', *Medicine and science in sports and exercise*, 43(7), pp. 1334-1359.

Gill, A.K. and Harrison, K. (2019) ''I am talking about it because I want to stop it': Child sexual abuse and sexual violence against women in British South Asian communities', *The British Journal of Criminology*, 59(3), pp. 511-529.

Gilliland, S.W. and Day, D.V. (1999) 'Business management'.

Gilpin, S. *et al.* (1989) 'The pathogenesis of genitourinary prolapse and stress incontinence of urine. A histological and histochemical study', *BJOG: An International Journal of Obstetrics & Gynaecology*, 96(1), pp. 15-23.

Goldenkoff, R. (2004) 'Using focus groups', *Handbook of practical program evaluation*, pp. 340-362.

Goodrick, E. and Reay, T. (2011) 'Constellations of institutional logics: Changes in the professional work of pharmacists', *Work and Occupations*, 38(3), pp. 372-416.

Goodwin, D. (2014) 'Decision-making and accountability: differences of distribution', *Sociology of health & Illness*, 36(1), pp. 44-59.

Gosling, J. *et al.* (1981) 'A comparative study of the human external sphincter and periurethral levator ani muscles', *British journal of urology*, 53(1), pp. 35-41.

Gov.UK (2020) Ethnicity facts and figures. Available at: <u>https://www.ethnicity-facts-</u>

figures.service.gov.uk/uk-population-by-ethnicity/national-and-regional-populations/regionalethnic-diversity/latest (Accessed: [07 Sept 2021]).

Grant, M.J. and Booth, A. (2009) 'A typology of reviews: an analysis of 14 review types and associated methodologies', *Health information & libraries journal*, 26(2), pp. 91-108.

Green, J. (2007) *The use of focus groups in research into health*. Sage, London, UK. Greenwood, R. *et al.* (2010) 'The multiplicity of institutional logics and the heterogeneity of organizational responses', *Organization science*, 21(2), pp. 521-539.

Griffthis, M. (1993) Motivational interviewing-preparing people to change addictive behaviormiller, wr, rollnick, s. ST Andrews house, 48 Princess rd east, Leicester British psychological soc guideline, N. (2019) Urinary incontinence and pelvic organ prolapse in women: management. Available at: <u>https://www.nice.org.uk/guidance/NG123</u> (Accessed: 31 January). guideline, N. (2021) *Pelvic floor dysfunction: prevention and non-surgical management*. Available at: <u>https://www.nice.org.uk/guidance/ng210</u> (Accessed: 31 December).

Gynaecological, C.S.o.P.a.T.P.O.a. and Physiotherapy (2019) *Think physio and act local*. Available at: <u>https://www.csp.org.uk/system/files/documents/2019-</u>

<u>04/001603</u> ERN%20LTP%20Briefing%20-%20Pelvic%20Health.pdf (Accessed: 11th of September). Haas, R. *et al.* (2012) 'Clinical decision making in exercise prescription for fall prevention', *Physical Therapy*, 92(5), pp. 666-679.

Hagen, S. *et al.* (2020) 'Effectiveness of pelvic floor muscle training with and without electromyographic biofeedback for urinary incontinence in women: multicentre randomised controlled trial', *bmj*, 371.

Hall, A.J. *et al.* (2018) 'Are physiotherapists employing person-centred care for people with dementia? An exploratory qualitative study examining the experiences of people with dementia and their carers', *BMC geriatrics*, 18(1), pp. 1-10.

Hall, A.M. *et al.* (2016) 'The influence of the therapist-patient relationship on treatment outcome in physical rehabilitation: a systematic review', *Physical therapy*, 90(8), pp. 1099-1110.

Ham, C. (2009) Health policy in Britain. Macmillan International Higher Education.

Hamdan, A. (2005) 'Women and education in Saudi Arabia: Challenges and achievements', *International Education Journal*, 6(1), pp. 42-64.

Hammersley, M. (1990) *Reading ethnographic research: a critical guide*. London, UK: Longman. Hammond, K.R. (2000) *Human judgment and social policy: Irreducible uncertainty, inevitable error, unavoidable injustice*. Oxford University Press on Demand.

Hannestad, Y.S. *et al.* (2000) 'A community-based epidemiological survey of female urinary incontinence:: The Norwegian EPINCONT Study', *Journal of clinical epidemiology*, 53(11), pp. 1150-1157.

Hassanein, M. *et al.* (2021) 'Ramadan fasting in people with type 1 diabetes during COVID-19 pandemic: The DaR Global survey', *diabetes research and clinical practice*, 172, p. 108626. Haugevik, K.M. and Bratberg, O. (2011) 'The cultural superpower: British cultural projection abroad', *Journal of the British Politics Society Norway*, 6(1), pp. 1-12.

Hay-Smith, J. *et al.* (2012) 'Comparisons of approaches to pelvic floor muscle training for urinary incontinence in women: an abridged Cochrane systematic review', *European journal of physical and rehabilitation medicine*, 48(4), pp. 689-705.

Haylen, B.T. *et al.* (2010) 'An International Urogynecological Association (IUGA)/International Continence Society (ICS) joint report on the terminology for female pelvic floor dysfunction', *International urogynecology journal*, 21(1), pp. 5-26.

Heath, G. *et al.* (2012) 'Paediatric 'care closer to home': stake-holder views and barriers to implementation', *Health & place*, 18(5), pp. 1068-1073.

Hendriks, H. *et al.* (2000) 'The diagnostic process and indication for physiotherapy: a prerequisite for treatment and outcome evaluation', *Physical Therapy Reviews*, 5(1), pp. 29-47.

Hennink, M., Hutter, I. and Bailey, A. (2010) *Qualitative research methods*. Sage.

Hennink, M.M. (2007) *International focus group research: A handbook for the health and social sciences*. Cambridge University Press.

Herderschee, R. *et al.* (2013) 'Feedback or biofeedback to augment pelvic floor muscle training for urinary incontinence in women: shortened version of a Cochrane systematic review', *Neurourology and urodynamics*, 32(4), pp. 325-329.

Heritage, J. (2013) *Garfinkel and ethnomethodology*. John Wiley & Sons.

Higgs, J. (2008) *Clinical reasoning in the health professions*. Elsevier Health Sciences.

Higgs, J. and Jones, M.A. (2008) 'Clinical decision making and multiple problem spaces', *Clinical reasoning in the health professions*, 3, pp. 3-17.

Higgs, J. and Trede, F. (2018) '4 CLINICAL REASONING AND MODELS OF PRACTICE', *CLINICAL REASONING*, p. 45.

Hinchliffe, S.J. et al. (1997) 'Software for qualitative research: 2. Some thoughts on

'aiding'analysis', *Environment and Planning A*, 29(6), pp. 1109-1124.

Hirsch, O. *et al.* (2012) 'Reliability and validity of the German version of the OPTION scale', *Health Expectations*, 15(4), pp. 379-388.

Hoffmann, T., Jansen, J. and Glasziou, P. (2018) 'The importance and challenges of shared decision making in older people with multimorbidity', *PLoS medicine*, 15(3), p. e1002530.

Hoffmann, T.C. *et al.* (2014) 'Shared decision making: what do clinicians need to know and why should they bother?', *Medical Journal of Australia*, 201(1), pp. 35-39.

Hoffmann, T.C., Lewis, J. and Maher, C.G. (2020) 'Shared decision making should be an integral part of physiotherapy practice', *Physiotherapy*, 107, pp. 43-49.

Hofstede, G. (1984) *Culture's consequences: International differences in work-related values.* sage. Hofstede, G. (2003) 'Cultural dimensions', *www. geert-hofstede. com*.

Holdar, U., Wallin, L. and Heiwe, S. (2013) 'Why do we do as we do? Factors influencing clinical reasoning and decision-making among physiotherapists in an acute setting', *Physiotherapy Research International*, 18(4), pp. 220-229.

Holloway, I. and Galvin, K. (2016) *Qualitative research in nursing and healthcare*. John Wiley & Sons.

Holmes, A.G.D. (2020) 'Researcher Positionality--A Consideration of Its Influence and Place in Qualitative Research--A New Researcher Guide', *Shanlax International Journal of Education*, 8(4), pp. 1-10.

Hopton, A.K. *et al.* (2012) 'Acupuncture in practice: mapping the providers, the patients and the settings in a national cross-sectional survey', *Bmj Open*, 2(1).

Huang, A.J. *et al.* (2006) 'Quality-of-life impact and treatment of urinary incontinence in ethnically diverse older women', *Archives of internal medicine*, 166(18), pp. 2000-2006.

Huczynski, A. and Buchanan, D. (2001) *Organizational behaviour: An introductory text (instructor's manual)*. Financial Times/Prentice Hall.

Huhn, K. *et al.* (2019) 'Clinical reasoning in physical therapy: a concept analysis', *Physical therapy*, 99(4), pp. 440-456.

Hunskaar, S. *et al.* (2000) 'Epidemiology and natural history of urinary incontinence', *International urogynecology journal*, 11(5), pp. 301-319.

Hutting, N. *et al.* (2020) 'Self-management support for people with non-specific low back pain: a qualitative survey among physiotherapists and exercise therapists', *Musculoskeletal Science and Practice*, 50, p. 102269.

Iacono, V.L., Symonds, P. and Brown, D.H. (2016) 'Skype as a Tool for Qualitative Research Interviews', *Sociological Research Online*, 21(2), p. 12.

Imamura, M. et al. (2015) 'Lifestyle interventions for the treatment of urinary incontinence in adults', *Cochrane Database of Systematic Reviews*, (12).

Jenks, A.C. (2011) 'From "lists of traits" to "open-mindedness": emerging issues in cultural competence education', *Culture, Medicine, and Psychiatry*, 35(2), pp. 209-235.

Jones, L.E. *et al.* (2014) 'Shared decision-making in back pain consultations: an illusion or reality?', *European Spine Journal: Official Publication Of The European Spine Society, The European Spinal Deformity Society, And The European Section Of The Cervical Spine Research Society*, 23 Suppl 1, pp. S13-S19.

Jones, M. (1995) 'Clinical reasoning and pain', Manual Therapy, 1(1), pp. 17-24.

Jootun, D., McGhee, G. and Marland, G.R. (2009) 'Reflexivity: promoting rigour in qualitative research', *Nursing Standard (through 2013)*, 23(23), p. 42.

Kant, I. (1908) 'Critique of pure reason. 1781', *Modern Classical Philosophers, Cambridge, MA: Houghton Mifflin*, pp. 370-456.

Kanuha, V.K. (2000) '"Being" native versus "going native": Conducting social work research as an insider', *Social work*, 45(5), pp. 439-447.

Karnad, P. and McLean, S.M. (2011) 'Physiotherapists' perceptions on adherence to home exercises in chronic musculoskeletal rehabilitation', *International Journal of Physiotherapy & Rehabilitation*, 1(2), pp. 14-29.

Karvonen, E. *et al.* (2017) 'Clinical reasoning and critical reflection in physiotherapists' examinations of patients with low back pain in its early phase: a qualitative study from physiotherapists' point of view', *European journal of physiotherapy*, 19(4), pp. 185-193.

Kautz, D.D. *et al.* (2005) 'Promoting clinical reasoning in undergraduate nursing students: application and evaluation of the Outcome Present State Test (OPT) model of clinical reasoning', *International Journal of Nursing Education Scholarship*, 2(1).

Kavvadias, T., Baessler, K. and Schuessler, B. (2012) 'Pelvic pain in urogynecology. Part II: treatment options in patients with lower urinary tract symptoms', *International urogynecology journal*, 23(5), pp. 553-561.

Kearney, R., Sawhney, R. and DeLancey, J.O. (2004) 'Levator ani muscle anatomy evaluated by origin-insertion pairs', *Obstetrics and gynecology*, 104(1), p. 168.

Keles, H. *et al.* (2007) 'Effect of chronic diseases and associated psychological distress on health-related quality of life', *Internal medicine journal*, 37(1), pp. 6-11.

Khattab, H. (2001) 'The Muslim woman's handbook'.

Kim, A., Kim, S. and Kim, H.G. (2020) 'Current overview of surgical options for female stress urinary incontinence', *International Neurourology Journal*, 24(3), p. 222.

King, L. *et al.* (2018) 'Clinical reasoning in osteopathy: Experiences of novice and experienced practitioners', *International Journal of Osteopathic Medicine*, 28, pp. 12-19.

Kitchener, M. (2002) 'Mobilizing the logic of managerialism in professional fields: The case of academic health centre mergers', *Organization studies*, 23(3), pp. 391-420.

Kittelson, A.J. *et al.* (2020) 'Person-centered care and physical therapy: a "people-like-me" approach', *Physical Therapy*, 100(1), pp. 99-106.

Kitzinger, J. (1994) 'The methodology of focus groups: the importance of interaction between research participants', *Sociology of health & illness*, 16(1), pp. 103-121.

Klein, G. (2008) 'Naturalistic decision making', *Human factors*, 50(3), pp. 456-460.

Klein, G.A., Calderwood, R. and Clinton-Cirocco, A. (1986) *Proceedings of the human factors society annual meeting*. Sage Publications Sage CA: Los Angeles, CA.

Kmietowicz, Z. (2018) 'Use mesh implants for stress urinary incontinence only as last resort, says NICE'. British Medical Journal Publishing Group.

Koch, T. and Harrington, A. (1998) 'Reconceptualizing rigour: the case for reflexivity', *Journal of advanced nursing*, 28(4), pp. 882-890.

Kosior, K., Wall, T. and Ferrero, S. (2019) 'The role of metacognition in teaching clinical reasoning: Theory to practice', *Education in the Health Professions*, 2(2), p. 108.

Kraemer, W.J. *et al.* (2002) 'American College of Sports Medicine position stand. Progression models in resistance training for healthy adults', *Medicine and science in sports and exercise*, 34(2), pp. 364-380.

Kraft, M.E. and Furlong, S.R. (2019) *Public policy: Politics, analysis, and alternatives*. Cq Press. Kristiansen, M., Obstfelder, A. and Lotherington, A.T. (2015) 'Nurses' sensemaking of

contradicting logics: An underexplored aspect of organisational work in nursing homes', *Scandinavian Journal of Management*, 31(3), pp. 330-337.

Krueger, R.A. (2014) *Focus groups: A practical guide for applied research*. Sage publications. Krueger, R.A. and Casey, M.A. (2000) *Focus groups*. Thousand Oaks, CA: Sage.

Kruger, J. *et al.* (2019) 'Can you train the pelvic floor muscles by contracting other related muscles?', *Neurourology and Urodynamics*, 38(2), pp. 677-683.

Kumagai, A.K. (2008) 'A conceptual framework for the use of illness narratives in medical education', *Academic medicine*, 83(7), pp. 653-658.

Kuzel, A.J. (1992) 'Sampling in qualitative inquiry'.

Kvale, S. (2008) Doing interviews. Sage.

Kvale, S. and Brinkmann, S. (2009) *Interviews: Learning the craft of qualitative interviewing*. London: Sage.

Lai, H.H. *et al.* (2016) 'The relationship between anxiety and overactive bladder or urinary incontinence symptoms in the clinical population', *Urology*, 98, pp. 50-57.

Langridge, N. (2019) 'The skills, knowledge and attributes needed as a first-contact

physiotherapist in musculoskeletal healthcare', *Musculoskeletal Care*, 17(2), pp. 253-260.

Langridge, N., Roberts, L. and Pope, C. (2015) 'The clinical reasoning processes of extended scope physiotherapists assessing patients with low back pain', *Manual therapy*, 20(6), pp. 745-750.

Langridge, N., Roberts, L. and Pope, C. (2016) 'The role of clinician emotion in clinical reasoning: balancing the analytical process', *Manual therapy*, 21, pp. 277-281.

Lauri, S. *et al.* (2001) 'An exploratory study of clinical decision-making in five countries', *Journal of nursing scholarship*, 33(1), pp. 83-90.

Lawrence, J.M. *et al.* (2008) 'Prevalence and co-occurrence of pelvic floor disorders in community-dwelling women', *Obstetrics & Gynecology*, 111(3), pp. 678-685.

Laycock, J. (1991) 'Incontinence. Pelvic floor re-education', Nursing, 4(39), pp. 15-17.

Laycock, J. and Haslam, J. (2013) *Therapeutic management of incontinence and pelvic pain: pelvic organ disorders*. Springer Science & Business Media.

Lee, J. (2005) 'The impact of urinary incontinence levels on the social lives of older Chinese in Hong Kong', *Hallym International Journal of Aging*, 7(1), pp. 63-80.

Lefringhausen, K., Spencer-Oatey, H. and Debray, C. (2019) 'Culture, norms, and the assessment of communication contexts: Multidisciplinary perspectives' 50. Sage Publications Sage CA: Los Angeles, CA, pp. 1098-1111 10.

Lester, S. and Religa, J. (2017) "Competence" and occupational standards: observations from six European countries', *Education+ Training*.

Levin, M.E. *et al.* (2014) 'Feasibility of a prototype web-based acceptance and commitment therapy prevention program for college students', *Journal of American College Health*, 62(1), pp. 20-30.

Lewis-Beck, M., Bryman, A.E. and Liao, T.F. (2003) *The Sage encyclopedia of social science research methods*. Sage Publications.

Lewis, J.S. *et al.* (2021) 'Reframing how we care for people with persistent non-traumatic musculoskeletal pain. Suggestions for the rehabilitation community', *Physiotherapy*, 112, pp. 143-149.

Lieberman, M.D., Jarcho, J.M. and Satpute, A.B. (2004) 'Evidence-based and intuition-based self-knowledge: an FMRI study', *Journal of personality and social psychology*, 87(4), p. 421.

Linderoth, H.C. (2017) 'From visions to practice–The role of sensemaking, institutional logic and pragmatic practice', *Construction management and economics*, 35(6), pp. 324-337.

Lowe, P., Griffiths, F. and Sidhu, R. (2007) 'I got pregnant, I was so like... crying inside...': experiences of women of Pakistani ancestry seeking contraception in the UK', *Diversity in Health and Social Care*, 4(1), pp. 69-76.

Luquis, R.R. and Pérez, M.A. (2021) *Cultural competence in health education and health promotion*. John Wiley & Sons.

MacArthur, C., Lewis, M. and Knox, E.G. (1991) 'Health after childbirth', *BJOG: An International Journal of Obstetrics & Gynaecology*, 98(12), pp. 1193-1195.

Magala, S.J. (1997) 'The making and unmaking of sense', *Organization Studies*, 18(2), pp. 317-338. Malik, S. *et al.* (2021) 'Effects of fasting on patients with chronic kidney disease during Ramadan and practical guidance for healthcare professionals', *Clinical Kidney Journal*, 14(6), pp. 1524-1534. Manhas, K.P. *et al.* (2020) 'Experiences of shared decision-making in community rehabilitation: a focused ethnography', *BMC health services research*, 20(1), pp. 1-12.

Marcelissen, T. *et al.* (2018) 'Management of idiopathic overactive bladder syndrome: what is the optimal strategy after failure of conservative treatment?', *European urology focus*, 4(5), pp. 760-767.

Marcum, J.A. (2013) 'The role of emotions in clinical reasoning and decision making', *Journal of Medicine and Philosophy*, 38(5), pp. 501-519.

Mason, J. (2002) Qualitative researching. Sage.

Mason, J. (2017) Qualitative researching. Sage.

Mattingly, C. (1991) 'The narrative nature of clinical reasoning', *American Journal of Occupational Therapy*, 45(11), pp. 998-1005.

Mattingly, C. and Fleming, M.H. (1994) *Clinical reasoning: Forms of inquiry in a therapeutic practice*. FA Davis Philadelphia.

May, C. and Finch, T. (2009) 'Implementing, embedding, and integrating practices: an outline of normalization process theory', *Sociology*, 43(3), pp. 535-554.

May, S. *et al.* (2008a) 'Expert therapists use specific clinical reasoning processes in the assessment and management of patients with shoulder pain: a qualitative study', *Australian Journal of Physiotherapy*, 54(4), pp. 261-266.

May, S. *et al.* (2008b) 'Expert therapists use specific clinical reasoning processes in the assessment and management of patients with shoulder pain: a qualitative study', *The Australian Journal Of Physiotherapy*, 54(4), pp. 261-266.

May, S. *et al.* (2010a) 'Limited clinical reasoning skills used by novice physiotherapists when involved in the assessment and management of patients with shoulder problems: a qualitative study', *Journal of Manual & Manipulative Therapy*, 18(2), pp. 84-88.

May, S. *et al.* (2010b) 'Limited clinical reasoning skills used by novice physiotherapists when involved in the assessment and management of patients with shoulder problems: a qualitative study', *Journal of Manual & Manipulative Therapy (Maney Publishing)*, 18(2), pp. 84-88.

Mays, N. and Pope, C. (2020) 'Quality in qualitative research', *Qualitative research in health care*, pp. 211-233.

Mazur-Bialy, A.I. *et al.* (2020) 'Urinary incontinence in women: modern methods of physiotherapy as a support for surgical treatment or independent therapy', *Journal of clinical medicine*, 9(4), p. 1211.

McClurg, D. *et al.* (2013) 'A multi-professional UK wide survey of undergraduate continence education', *Neurourology and urodynamics*, 32(3), pp. 224-229.

McClurg, D. *et al.* (2015) 'Scoping review of adherence promotion theories in pelvic floor muscle training–2011 ics state-of-the-science seminar research paper i of iv', *Neurourology and urodynamics*, 34(7), pp. 606-614.

McClurg, D. *et al.* (2017) *NEUROUROLOGY AND URODYNAMICS*. WILEY 111 RIVER ST, HOBOKEN 07030-5774, NJ USA.

McGinnis, P.Q. *et al.* (2009a) 'Factors that influence the clinical decision making of physical therapists in choosing a balance assessment approach', *PHYSICAL THERAPY*, 89(3), p. 233. McGinnis, P.Q. *et al.* (2009b) 'Factors that influence the clinical decision making of physical

therapists in choosing a balance assessment approach', *Physical therapy*, 89(3), pp. 233-247.

McNamara, C. (2009) 'General guidelines for conducting interviews. Retrieved January 11, 2010'. McPherson, K., Nahon, I. and Waddington, G. (2020) 'Entry level women's health physiotherapy curricula in Australia', *European Journal of Physiotherapy*, pp. 1-5.

Melville, J.L. *et al.* (2009) 'Major depression and urinary incontinence in women: temporal associations in an epidemiologic sample', *Am J Obstet Gynecol*, 201(5), pp. 490.e1-7.

Merriam, S.B. and Baumgartner, L.M. (2020) *Learning in adulthood: A comprehensive guide*. John Wiley & Sons.

Mescouto, K. *et al.* (2020) 'A critical review of the biopsychosocial model of low back pain care: time for a new approach?', *Disability and Rehabilitation*, pp. 1-15.

Messer, K.L. *et al.* (2007) 'Self-efficacy as a predictor to PFMT adherence in a prevention of urinary incontinence clinical trial', *Health education & behavior*, 34(6), pp. 942-952.

Miles, M.B., Huberman, A.M. and Saldana, J. (2013) *Qualitative data analysis*. Sage.

Miller, J.M. *et al.* (2008) 'Clarification and confirmation of the Knack maneuver: the effect of volitional pelvic floor muscle contraction to preempt expected stress incontinence', *International Urogynecology Journal*, 19(6), pp. 773-782.

Miller, P.A., Rivett, D. and Isles, R. (2009) *Pattern recognition is a clinical reasoning process in musculoskeletal physiotherapy*. University of Newcastle.

Miller, W.R. and Rollnick, S. (2012) *Motivational interviewing: Helping people change*. Guilford press.

Millward, L.J. (1995) 'Focus groups', *Research methods in psychology*, 2, pp. 274-292. Minkov, M. (2012) *Cross-cultural analysis: The science and art of comparing the world's modern societies and their cultures*. SAGE publications.

Mirzaei, S. *et al.* (2019) 'Administrative challenges of clinical governance in military and university hospitals of Kerman/Iran', *The International Journal of Health Planning and Management*, 34(2), pp. e1293-e1301.

Mobeireek, A. *et al.* (2008) 'Information disclosure and decision-making: the Middle East versus the Far East and the West', *Journal of Medical Ethics*, 34(4), pp. 225-229.

Mobeireek, A.F. *et al.* (1996) 'Communication with the seriously ill: physicians' attitudes in Saudi Arabia', *Journal of medical ethics*, 22(5), pp. 282-285.

Moher, D. *et al.* (2009) 'Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement', *PLoS medicine*, 6(7), p. e1000097.

Moisander, J. and Valtonen, A. (2006) *Qualitative marketing research: A cultural approach*. Sage. Morgan, D.L. (1996) *Focus groups as qualitative research*. Sage publications.

Morison, M. and Moir, J. (1998) 'The role of computer software in the analysis of qualitative data: efficient clerk, research assistant or Trojan horse?', *Journal of advanced nursing*, 28(1), pp. 106-116.

Mosier, K. *et al.* (2018) 'Expert professional judgments and" naturalistic decision making"'. Mousa, W. and Aldehayyat, J.S. (2018) 'Regional efficiency of healthcare services in Saudi Arabia', *Middle East Development Journal*, 10(1), pp. 152-174.

Murphy, J.I. (2004) 'Using focused reflection and articulation to promote clinical reasoning: An evidence-based teaching strategy', *Nursing Education Perspectives*, 25(5), pp. 226-231.

Myers, M.D. and Newman, M. (2007) 'The qualitative interview in IS research: Examining the craft', *Information and organization*, 17(1), pp. 2-26.

Nadan, Y. and Ben-Ari, A. (2013) 'What can we learn from rethinking 'multiculturalism'in social work education?', *Social Work Education*, 32(8), pp. 1089-1102.

Nambiar, A.K. *et al.* (2018) 'EAU guidelines on assessment and nonsurgical management of urinary incontinence', *European urology*.

Nicola-Richmond, K.M., Pépin, G. and Larkin, H. (2016) 'Transformation from student to occupational therapist: Using the D elphi technique to identify the threshold concepts of occupational therapy', *Australian occupational therapy journal*, 63(2), pp. 95-104.

Norris, M. and Allotey, P. (2008) 'Culture and physiotherapy', *Diversity in Health & Social Care*, 5(2).

Øberg, G.K., Normann, B. and Gallagher, S. (2015) 'Embodied-enactive clinical reasoning in physical therapy', *Physiotherapy theory and practice*, 31(4), pp. 244-252.

Okeahialam, N.A. *et al.* (2022) 'Prevention and non-surgical management of pelvic floor dysfunction: summary of NICE guidance', *bmj*, 376.

Oleribe, O.O. *et al.* (2019) 'Identifying key challenges facing healthcare systems in Africa and potential solutions', *International journal of general medicine*, 12, p. 395.

Oosman, S. *et al.* (2019) 'Essential elements for advancing cultural humility through a communitybased physical therapy practicum in a Métis community', *Physiotherapy Canada*, 71(2), pp. 146-157.

Oostendorp, R.A., Elvers, J.H. and Trijffel van, E. (2020) 'Concept Analysis of Clinical Reasoning in Physical Therapist Practice', *Physical therapy*, 100(8), pp. 1353-1356.

Organization, W.H. (2003) *Diet, nutrition, and the prevention of chronic diseases: report of a joint WHO/FAO expert consultation*. World Health Organization.

Ostaszkiewicz, J., Chestney, T. and Roe, B. (2004) 'Habit retraining for the management of urinary incontinence in adults', *Cochrane Database of Systematic Reviews*, (2).

Øvretveit, J. (1985) 'Medical dominance and the development of professional autonomy in physiotherapy', *Sociology of health & illness*, 7(1), pp. 76-93.

Pache, A.-C. and Santos, F. (2010) 'When worlds collide: The internal dynamics of organizational responses to conflicting institutional demands', *Academy of management review*, 35(3), pp. 455-476.

Parahoo, K. (2014) *Nursing research: principles, process and issues*. Palgrave Macmillan. Parmelli, E. *et al.* (2011) 'The effectiveness of strategies to change organisational culture to improve healthcare performance: a systematic review', *Implementation Science*, 6(1), pp. 1-8. Patton, M.Q. (2002) *Qualitative research and evaluation methods*. Thousand Oaks, Calif. : Sage Publications, c2002. Paul, S. (1996) 'Sensemaking in organizations'. Academy of Management Briarcliff Manor, NY 10510.

Payton, O.D. (1985) 'Clinical reasoning process in physical therapy', *Physical Therapy*, 65(6), pp. 924-928.

Peate, I. (2019) 'Urinary incontinence in women: treatment recommendations', *British Journal of Nursing*, 28(22), pp. 1486-1488.

Peek, K. *et al.* (2020) 'Characteristics associated with high levels of patient-reported adherence to self-management strategies prescribed by physiotherapists', *International Journal of Therapy And Rehabilitation*, 27(1), pp. 1-15.

Peek, K. *et al.* (2016) 'Interventions to aid patient adherence to physiotherapist prescribed selfmanagement strategies: a systematic review', *Physiotherapy*, 102(2), pp. 127-135.

Peng, L. *et al.* (2019) 'Magnetic stimulation for female patients with stress urinary incontinence, a meta-analysis of studies with short-term follow-up', *Medicine*, 98(19).

Pereira, V.S., Correia, G.N. and Driusso, P. (2011) 'Individual and group pelvic floor muscle training versus no treatment in female stress urinary incontinence: a randomized controlled pilot study', *European Journal of Obstetrics & Gynecology and Reproductive Biology*, 159(2), pp. 465-471.

Perry, S., McGrother, C.W. and Turner, K. (2006) 'An investigation of the relationship between anxiety and depression and urge incontinence in women: development of a psychological model', *British journal of health psychology*, 11(3), pp. 463-482.

Phillippi, J. and Lauderdale, J. (2018) 'A guide to field notes for qualitative research: Context and conversation', *Qualitative health research*, 28(3), pp. 381-388.

Pierce, H. *et al.* (2015) 'Pelvic floor health: a concept analysis', *Journal of advanced nursing*, 71(5), pp. 991-1004.

Pirotta, M. (1994) 'Cultural aspects of medical practice in Saudi Arabia', *The Medical journal of Australia*, 161(2), pp. 153, 156-9.

Pope, C., Van Royen, P. and Baker, R. (2002) 'Qualitative methods in research on healthcare quality', *BMJ Quality & Safety*, 11(2), pp. 148-152.

Porter, S. (2008) *Tidy's physiotherapy*. Elsevier Health Sciences.

Primeau, L.A. (2003) 'Reflections on self in qualitative research: Stories of family', *American Journal of Occupational Therapy*, 57(1), pp. 9-16.

Prokop, T.R. (2018) 'Use of the dual-processing theory to develop expert clinical reasoning in physical therapy students', *Journal of Physical Therapy Education*, 32(4), pp. 355-359.

Public Health England (2019) PHE Strategy 2020-25: Executive Summary. Available at:

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/ file/830105/PHE_Strategy_2020-25_Executive_Summary.pdf (Downloaded: 07 September 2021).

Quek, P. (2005) 'A critical review on magnetic stimulation: what is its role in the management of pelvic floor disorders?', *Current opinion in urology*, 15(4), pp. 231-235.

Rabiee, F. (2004) 'Focus-group interview and data analysis', *Proceedings of the nutrition society*, 63(4), pp. 655-660.

Rahman, R. and Al-Borie, H.M. (2020) 'Strengthening the Saudi Arabian healthcare system: role of vision 2030', *International Journal of Healthcare Management*, pp. 1-9.

Reay, T. and Hinings, C.R. (2005) 'The recomposition of an organizational field: Health care in Alberta', *Organization studies*, 26(3), pp. 351-384.

Reay, T. and Hinings, C.R. (2009) 'Managing the rivalry of competing institutional logics', *Organization studies*, 30(6), pp. 629-652.

Reed, P. *et al.* (2020) 'Impact of strength and nature of patient health values on compliance and outcomes for physiotherapy treatment for pelvic floor dysfunction', *Urology*, 136, pp. 95-99. Rehman, H. *et al.* (2017) 'Traditional suburethral sling operations for urinary incontinence in women', *Cochrane Database of Systematic Reviews*, (7).

Reiter-Palmon, R. *et al.* (2015) 'Naturalistic decision making in after-action review meetings: The implementation of and learning from post-fall huddles', *Journal of Occupational and Organizational Psychology*, 88(2), pp. 322-340.

Reza, I.F. (2020) 'Wudu as Islamic Psychotherapy to Improve Sleep Quality in Young Women', *Journal An-Nafs: Kajian Penelitian Psikologi*, 5(1), pp. 64-75.

Ritchie, J. et al. (2013) Qualitative research practice: A guide for social science students and researchers. sage.

Ritchie, J. et al. (1994) 'Analysing qualitative data'.

Rizvi, L.J. and Hussain, Z. (2021) 'Empowering woman through legal reforms-evidence from Saudi Arabian context', *International Journal of Law and Management*.

Robert, G. *et al.* (2011) 'The adoption, local implementation and assimilation into routine nursing practice of a national quality improvement programme: the Productive Ward in England', *Journal of clinical nursing*, 20(7-8), pp. 1196-1207.

Robinson, D. and Cardozo, L. (2021) 'Pharmacological Treatment of Urinary Incontinence and Overactive Bladder: The Evidence', in *Pelvic Floor Disorders*. Springer, pp. 351-363.

Robson, C. and McCartan, K. (2016) Real world research. John Wiley & Sons.

Rolls, L. and Relf, M. (2004) 'Bracketing Interviews: A method for increasing objectivity in bereavement and palliative care research. As cited in Dowling, M.(2007). From Husserl to van Manen. A Review of Different Phenomenological Approaches', *International Journal of Nursing Studies*, 44, pp. 131-142.

Rollston, R. *et al.* (2020) 'Comprehensive sexuality education to address gender-based violence', *The lancet*, 396(10245), pp. 148-150.

Rosness, R. *et al.* (2016) 'Prospective sensemaking in the operating theatre', *Cognition, Technology & Work*, 18(1), pp. 53-69.

Rubin, H.J. and Rubin, I.S. (2011) *Qualitative interviewing: The art of hearing data*. Sage. Sackett, D.L. *et al.* (1996) 'Evidence based medicine: what it is and what it isn't', 312(7023), pp. 71-72.

Sam, S. *et al.* (2020) 'Shared decision making in musculoskeletal pain consultations in low-and middle-income countries: a systematic review', *International Health*, 12(5), pp. 455-471. Sanders, T. *et al.* (2013) 'Biopsychosocial care and the physiotherapy encounter: physiotherapists'

accounts of back pain consultations', *BMC musculoskeletal disorders*, 14(1), pp. 1-10.

Sawrikar, P. and Katz, I. (2018) 'Preventing child sexual abuse (CSA) in ethnic minority communities: A literature review and suggestions for practice in Australia', *Children and youth services review*, 85, pp. 174-186.

Schatzman, L. and Strauss, A.L. (1973) *Field research: Strategies for a natural sociology*. Prentice Hall.

Schein, E.H. (1983) 'The role of the founder in creating organizational culture', *Organizational dynamics*, 12(1), pp. 13-28.

Schein, E.H. (1985) 'Defining organizational culture', *Classics of organization theory*, 3(1), pp. 490-502.

Schmidt, H., Norman, G. and Boshuizen, H. (1990) 'A cognitive perspective on medical expertise: theory and implications', *Academic medicine*, 65(10), pp. 611-621.

Schuwirth, L. (2002) 'Can clinical reasoning be taught or can it only be learned?', *Medical education*, 36(8), p. 695.

Scott, R.W., Scott, W.R. and Meyer, J.W. (1994) *Institutional environments and organizations: Structural complexity and individualism*. Sage.

Scott, W.R. (2013) *Institutions and organizations: Ideas, interests, and identities*. Sage publications.

Seale, C. (2012) 'Sampling. I Clive Seale (red.). Researching society and culture'. SAGE Publications Ltd, London, tredje upplagan.

Senra, C. and Pereira, M.G. (2015) 'Quality of life in women with urinary incontinence', *Revista da Associação Médica Brasileira*, 61(2), pp. 178-183.

Siddiqui, N.Y. *et al.* (2014) 'Perceptions about female urinary incontinence: a systematic review', *International urogynecology journal*, 25(7), pp. 863-871.

Siehl, C. and Martin, J. (1989) *Organizational culture: A key to financial performance?* : Graduate School of Business, Stanford University.

Silverman, D. (2015) Interpreting qualitative data. Sage.

Simmens, S.J. (2009) 'Cross-cultural comparison of the patient-centeredness of the hidden curriculum between a Saudi Arabian and 9 US medical schools Rasha Al-Bawardy, Benjamin Blatt, MD, Saad Al-Shohaib, MD \$'.

Simon, H.A. (1990) 'Alternative visions of rationality', *Rationality in action: Contemporary approaches*, pp. 189-204.

Sinclair, A.J. and Ramsay, I.N. (2011) 'The psychosocial impact of urinary incontinence in women', *The Obstetrician & Gynaecologist*, 13(3), pp. 143-148.

Singh, G. *et al.* (2018) 'Exploring the lived experience and chronic low back pain beliefs of Englishspeaking Punjabi and white British people: a qualitative study within the NHS', *BMJ open*, 8(2), p. e020108.

Singla, M. *et al.* (2015) 'Physiotherapists' assessment of patients' psychosocial status: are we standing on thin ice? A qualitative descriptive study', *Manual therapy*, 20(2), pp. 328-334. Slade, S.C. *et al.* (2020) 'Attributes of physiotherapy continence clinicians: a qualitative perspective', *Physiotherapy*, 106, pp. 119-127.

Smith, A. *et al.* (2013) 'Management of urinary incontinence in women: summary of updated NICE guidance', *Bmj*, 347.

Smith, M., Higgs, J. and Ellis, E. (2008) 'Characteristics and processes of physiotherapy clinical decision making: a study of acute care cardiorespiratory physiotherapy', *PHYSIOTHERAPY RESEARCH INTERNATIONAL*, 13(4), p. 209.

Solvang, P.K. and Fougner, M. (2016) 'Professional roles in physiotherapy practice: Educating for self-management, relational matching, and coaching for everyday life', *Physiotherapy theory and practice*, 32(8), pp. 591-602.

Song, J. (2019) '"She Needs to Be Shy!": Gender, Culture, and Nonparticipation Among Saudi Arabian Female Students', *Tesol Quarterly*, 53(2), pp. 405-429.

Southampton, U.o. (2021) Research Data Management Policy. Available at:

https://www.southampton.ac.uk/~assets/doc/calendar/Research%20Data%20Management%20P olicy.pdf.

Spencer-Oatey, H. (2008) *Culturally speaking second edition: Culture, communication and politeness theory*. Bloomsbury Publishing.

Speziale, H. and Carpenter, D. (2007) *Qualitative research in nursing* Lippincott Williams & Wilkins: Lippincott Williams & Wilkins.

Stewart, E.C. and Bennett, M.J. (2011) *American cultural patterns: A cross-cultural perspective*. Hachette UK.

Sun, S. (2008) 'Organizational culture and its themes', *International Journal of Business and Management*, 3(12), pp. 137-141.

Swallow, V. *et al.* (2011) 'Fathers and mothers developing skills in managing children's long-term medical conditions: how do their qualitative accounts compare?', *Child: care, health and development*, 37(4), pp. 512-523.

Tarcan, T. *et al.* (2020) 'Should we routinely assess psychological morbidities in idiopathic lower urinary tract dysfunction: ICI-RS 2019?', *Neurourology and Urodynamics*, 39, pp. S70-S79.

Tascon, S. and Gatwiri, K. (2020) 'Towards cultural humility: Theorising cultural competence as institutionalised whiteness', *Social Work & Policy Studies: Social Justice, Practice and Theory*, 3(1). Te, M. *et al.* (2020) 'New graduate physiotherapists' perceptions and experiences working with people from culturally and linguistically diverse communities in Australia: a qualitative study', *Physiotherapy Theory and Practice*, pp. 1-12.

Tervalon, M. and Murray-Garcia, J. (1998) 'Cultural humility versus cultural competence: A critical distinction in defining physician training outcomes in multicultural education', *Journal of health care for the poor and underserved*, 9(2), pp. 117-125.

Thackray, D. and Roberts, L. (2017) 'Exploring the clinical decision-making used by experienced cardiorespiratory physiotherapists: A mixed method qualitative design of simulation, video recording and think aloud techniques', *Nurse Education Today*, 49, pp. 96-105.

Thomas, L. *et al.* (1995) 'Obtaining patients' views of nursing care to inform the development of a patient satisfaction scale', *International Journal for Quality in Health Care*, 7(2), pp. 153-163.

Thompson, C. and Stapley, S. (2011) 'Do educational interventions improve nurses' clinical decision making and judgement? A systematic review', *International journal of nursing studies*, 48(7), pp. 881-893.

Thompson, C. and Yang, H. (2009) 'Nurses' decisions, irreducible uncertainty and maximizing nurses' contribution to patient safety', *Healthcare Quarterly*, 12(Sp).

Thornton, P.H. and Ocasio, W. (2008) 'Institutional logics', *The Sage handbook of organizational institutionalism*, 840(2008), pp. 99-128.

Thornton, P.H., Ocasio, W. and Lounsbury, M. (2015) 'The institutional logics perspective', *Emerging trends in the social and behavioral sciences: An interdisciplinary, searchable, and linkable resource*, pp. 1-22.

Tiffen, J., Corbridge, S.J. and Slimmer, L. (2014) 'Enhancing clinical decision making: development of a contiguous definition and conceptual framework', *Journal of professional nursing*, 30(5), pp. 399-405.

Tong, A. *et al.* (2012) 'Enhancing transparency in reporting the synthesis of qualitative research: ENTREQ', *BMC medical research methodology*, 12(1), pp. 1-8.

Toye, F. and Barker, K.L. (2020) 'A meta-ethnography to understand the experience of living with urinary incontinence:'is it just part and parcel of life?'', *BMC urology*, 20(1), pp. 1-25.

Turner III, D.W. (2010) 'Qualitative interview design: A practical guide for novice investigators', *The qualitative report*, 15(3), pp. 754-760.

Van Delft, K., Thakar, R. and Sultan, A. (2015) 'Pelvic floor muscle contractility: digital assessment vs transperineal ultrasound', *Ultrasound in Obstetrics & Gynecology*, 45(2), pp. 217-222.

Vandoninck, V. *et al.* (2004) 'Posterior tibial nerve stimulation in the treatment of voiding dysfunction: urodynamic data', *Neurourology and urodynamics*, 23(3), pp. 246-251.

Vedin, E. (2021) 'The clinical reasoning among master students specializing in Orthopedic Manual Therapy'.

Virkud, A. (2011) 'Management of stress urinary incontinence', *Best Practice & Research Clinical Obstetrics & Gynaecology*, 25(2), pp. 205-216.

Wadham, J. *et al.* (2010) *Blackstone's guide to the Equality Act 2010*. Oxford University Press, USA. Wainwright, S.F. *et al.* (2011) 'Factors that influence the clinical decision making of novice and experienced physical therapists', *Physical Therapy*, 91(1), pp. 87-101.

Walker, M.U. (2002) *Moral contexts*. Rowman & Littlefield Publishers.

Wang, Y. *et al.* (2013) 'Acupuncture for stress urinary incontinence in adults', *Cochrane Database of Systematic Reviews*, (7).

Warner, R.M. and Greenwell, T.J. (2018) 'A comparison of the National Institute for Health and Care Excellence (NICE) and European Association of Urology (EAU) guidelines for the assessment and management of urinary incontinence in women', *Journal of Clinical Urology*, 11(2), pp. 88-100.

Wei, M. *et al.* (2017) 'Fasting-mimicking diet and markers/risk factors for aging, diabetes, cancer, and cardiovascular disease', *Science translational medicine*, 9(377), p. eaai8700.

Weick, K.E. (1969) *The social psychology of organizing*. Reading (Mass.) ; London: Addison-Wesley. Weick, K.E. (1995) *Sensemaking in organizations*. Sage.

Weick, K.E. *et al.* (2004) 'Managing the unexpected in prescribed fire and fire use operations a workshop on the high reliability organization', *USDA Forest Service-General Technical Report RMRS-GTR*, (137 RMRS-GTR), pp. 1-79.

Weick, K.E., Sutcliffe, K.M. and Obstfeld, D. (2005) 'Organizing and the process of sensemaking', *Organization science*, 16(4), pp. 409-421.

Weitzman, E. and Miles, M.B. (1995) *Computer programs for qualitative data analysis*. Sage. Welsh, E. (2002) *Forum Qualitative Sozialforschung/Forum: Qualitative Social Research*. Wilkinson, S. and Silverman, D. (2004) '10 Focus Group Research', *Qualitative research: Theory, method and practice*, pp. 177-199.

Willcoxson, L. and Millett, B. (2000) 'The management of organisational culture', Australian Journal of Management and Organisational Behaviour, 3(2), pp. 91-99.

Williams, K.S. *et al.* (2005) 'Clinical and cost-effectiveness of a new nurse-led continence service: a randomised controlled trial', *British Journal of General Practice*, 55(518), pp. 696-703.

Williams, V., Boylan, A.-M. and Nunan, D. (2020) 'Critical appraisal of qualitative research: necessity, partialities and the issue of bias', *BMJ Evidence-Based Medicine*, 25(1), pp. 9-11.

Woods, M. *et al.* (2016) 'Advancing qualitative research using qualitative data analysis software (QDAS)? Reviewing potential versus practice in published studies using ATLAS. ti and NVivo, 1994–2013', *Social Science Computer Review*, 34(5), pp. 597-617.

Xianhui, G. and Shah, S.A.A. (2021) 'Non-oil economic transition for economic and environmental sustainability in Saudi Arabia: a multi-factor analysis under fuzzy environment', *Environmental Science and Pollution Research*, pp. 1-15.

Yoshikawa, K. *et al.* (2020) 'Sociocultural factors influencing physiotherapy management in culturally and linguistically diverse people with persistent pain: a scoping review', *Physiotherapy*, 107, pp. 292-305.

Zaini, R.G. *et al.* (2011) 'Saudi Meds: a competence specification for Saudi medical graduates', *Medical teacher*, 33(7), pp. 582-584.

Zamawe, F.C. (2015) 'The implication of using NVivo software in qualitative data analysis: Evidence-based reflections', *Malawi Medical Journal*, 27(1), pp. 13-15.

Zhao, Y. *et al.* (2018) 'Acupuncture for adults with overactive bladder: a systematic review and meta-analysis of randomized controlled trials', *Medicine*, 97(8).

Zhou, X. *et al.* (2011) 'Anxiety symptom presentations in Han Chinese and Euro-Canadian outpatients: Is distress always somatized in China?', *Journal of Affective Disorders*, 135(1-3), pp. 111-114.

Zsambok, C.E. and Klein, G. (2014) Naturalistic decision making. Psychology Press.